



Swansea University E-Theses

Analysis of adaptive finite element solutions in elastoplasticity with reference to transfer operation techniques.

Orlando, Antonio

How to cite:

Orlando, Antonio (2002) Analysis of adaptive finite element solutions in elastoplasticity with reference to transfer operation techniques.. thesis, Swansea University. http://cronfa.swan.ac.uk/Record/cronfa42476

Use policy:

This item is brought to you by Swansea University. Any person downloading material is agreeing to abide by the terms of the repository licence: copies of full text items may be used or reproduced in any format or medium, without prior permission for personal research or study, educational or non-commercial purposes only. The copyright for any work remains with the original author unless otherwise specified. The full-text must not be sold in any format or medium without the formal permission of the copyright holder. Permission for multiple reproductions should be obtained from the original author.

Authors are personally responsible for adhering to copyright and publisher restrictions when uploading content to the repository.

Please link to the metadata record in the Swansea University repository, Cronfa (link given in the citation reference above.)

http://www.swansea.ac.uk/library/researchsupport/ris-support/

University of Wales Swansea School of Engineering



ANALYSIS OF ADAPTIVE FINITE ELEMENT SOLUTIONS IN ELASTOPLASTICITY WITH REFERENCE TO TRANSFER OPERATION TECHNIQUES

Antonio Orlando

Laurea in Ingegneria Civile (Universitá degli Studi di Napoli Federico II) MSC, DIC (Imperial College, London)

Thesis submitted to the University of Wales in Candidature for the degree of Doctor of Philosophy

October 2002

С/Рн/258/02

ProQuest Number: 10801706

All rights reserved

INFORMATION TO ALL USERS The quality of this reproduction is dependent upon the quality of the copy submitted.

In the unlikely event that the author did not send a complete manuscript and there are missing pages, these will be noted. Also, if material had to be removed, a note will indicate the deletion.



ProQuest 10801706

Published by ProQuest LLC (2018). Copyright of the Dissertation is held by the Author.

All rights reserved. This work is protected against unauthorized copying under Title 17, United States Code Microform Edition © ProQuest LLC.

> ProQuest LLC. 789 East Eisenhower Parkway P.O. Box 1346 Ann Arbor, MI 48106 – 1346

To Aqueró

Acknowledgements

Finally, I have arrived to this point. Embarking on a Ph.D. programme abroad is a very special and privileged experience. There is always an enrichment: scientific and especially cultural.

Letting the memory go through the four last years spent in Swansea is a touching emotion. It gives joy to remind all people that have shared with me an important part of my life, bringing each of them something new. Thus, it is with great pleasure that I recall all of them.

I wish to thank especially Prof. D. Perić, my supervisor, for having proposed the research topic of error estimation; for his enduring patience; for the long time spent in his office to discuss not only issues on error estimation and especially for his human qualities of support in very difficult time during my work. Grazie Prof. Perić.

Many thanks are also due to Dr. Eduardo de Souza Neto for helpful and the passionate discussions on computational plasticity. Grazie Eduardo.

I would like to thank Prof. P. Ladevèze for his kindness and Dr. L. Gallimard, hosts at the LMT - laboratory in Cachan where I found a very welcoming environment. They made my stay in Paris an enjoyable experience. While I was in Paris, I could also go to Lourdes, a touching and moving place. Grazie Prof. Ladevèze and Dr. Gallimard.

A special thanks goes to Prof. Giuseppe Cocurullo for supporting my cause with the Rotary Club of Salerno, first, and then the Rotary Foundation of Rotary International which awarded me with an Ambassadorial Scholar Grant. This allowed me to come to Swansea four years ago. Also, I would like to acknowledge additional support from my supervisor Prof. Perić and especially from my parents. Grazie a voi tutti.

Then, there are all my friends in Swansea and here the list is very long. Without them, I do not know what it would have been like. I cannot imagine my PhD studies without them. Thus, it is with great joy that I recall Wulf Dettmer; Igor Dyson; Holger Ettinger; Paul Ledger; Anna Gimigliano; Nicola Massarotti and Caroline; Arnaud Malan; William Pao; Francisco Pires; Siva Kusalegaram; James Mac-Faddam and Africa; Miguel X. Rodriguez Paz and Claire; Andreas Rippl; Sony Saksono and Meily; Sava Slijepčević, his wife Dragana and Ognjen; the sisters of the Catholic Chaplaincy and in special way sister Noreen; the choir at Saint Benedict's church: Sheila, Eric, Peter, Cliff, Hugh, Kitty, Sister Elisabeth, Jane, Mike and Pauline; Father Francis and Father Dan; Dr. J. Seinz, Maria and the Latin-American community in Swansea. A deep and heartful thanks to all of them. Grazie amici miei.

I am deeply thankful to Marielita, a precious gift to me from Our Father. I am grateful to her for giving me peace and trust, for her patience, understanding and especially for her encouragement to finish: "But Antonio, why don't you finish?" In three months, was always my answer. My three months have come to the end, Mariela. The completion is also thanks to you. Grazie Mariela.

Last, but not least, I am very grateful to my parents for giving me a nurturing home and a good education and to my brother and sister for their unmissable support in these years far from them. Grazie papà e mamma, Renato e Marilina.

And finally I say thanks to Our Father for blessing me with good health and overall luck in life. Grazie Padre nostro.

This work is dedicated to Our Lady of Lourdes Grazie "Aqueró" "No numerical method can be considered satisfactory, unless something is known about the behaviour of the error e(p) of the method as a function of p." Peter Henrici in "Discrete Variable Methods in Ordinary Differential Equations"

Contents

.

Ι	\mathbf{Er}	ror E	stimati	on. General Theory	1
1	Intr 1.1	oducti The sc	on ope of th	e thesis	2 3
	1.2	Layou	t		4
2	Ove	rview	on a pos	steriori error estimates. Literature Review	6
	2.1	Introd	uction .		6
	2.2	Linear	problems	5	6
		2.2.1	Residua	type error estimates	10
			2.2.1.1	Explicit residual a posteriori error estimates	10
			2.2.1.2	Implicit residual a posteriori error estimates	13
		2.2.2	Recovery	y based error estimators	21
	2.3	Nonlin	lear probl	ems	24
		2.3.1	Nonlinea	ar incremental problem	25
		2.3.2	Analysis	of the time discretization error	29
		2.3.3	Error in	E - Indiante -	30 20
	<u>ე</u> /	2.3.4 Conch	Heuristie	c Error Indicators	39 42
	2.4	Conch	iding rem	arks	40
3	The	error	in the	constitutive equations for dissipative nonlinear	
	prol	olems			45
	3.1	Introd	uction .		45
	3.2	The re	ference co	ontinuum problem: The Initial Boundary Value Prob-	
		lem for	r a model	with internal variables	46
		3.2.1	Prelimin	aries	46
		3.2.2	Equilibr	ium Equation	46
		3.2.3	Compati	ibility Equations	46
		3.2.4	Constitu	tive Equations	47
			3.2.4.1	Thermodynamic Admissibility	47
			3.2.4.2	State variables	48
			3.2.4.3	Equations of State	48
			3.2.4.4	Complementary Equations. Associative Plasticity	51
			3.2.4.5	Examples of standard rate independent plasticity mod-	50
				els	56

3.3	Contra	activity of the elastoplastic flow	62	
3.4	Residu	al versus Error in Solution	64	
	3.4.1	A simple a posteriori error estimate via discrete energy dissi-		
		$pation \dots \dots \dots \dots \dots \dots \dots \dots \dots $	66	
3.5	Admis	ssible solution and measure of the error	69	
	3.5.1	Error in the constitutive equations for time continuous admis-		
		sible solution	71	
		3.5.1.1 Dissipation Error	72	
		3.5.1.2 Extended Dissipation Error	74	
	3.5.2	Error in the constitutive equations for admissible solution with		
		jump across time instant t_n	78	
		3.5.2.1 Augmented Dissipation Error	81	
		3.5.2.2 Augmented Extended Dissipation Error	90	
	3.5.3	Definition of error in solution		
		3.5.3.1 Extension of the Prager-Synge theorem to the Dissi-		
		pation Error	95	
3.6	Conclu	uding Remarks	96	

II Application to the Finite Element Solution of the IBVP in Elasto-Plasticity 98

4	The	Finite Element Solution of the IBVP in Elasto-Plasticity	99
	4.1	Introduction	. 99
	4.2	The displacement formulation of the IBVP	100
		4.2.1 Statement of the problem	102
	4.3	The time discrete problem	103
	4.4	The fully discrete problem: Constant finite element mesh	107
		4.4.1 Change of finite element mesh	110
	4.5	Overview on the different definitions of the initial state. Transfer	
		procedures	. 112
		4.5.1 Variationally consistent transfer	. 114
		4.5.1.1 Heat conduction	. 115
		4.5.1.2 Weak enforcement of the constitutive equations	. 117
		4.5.2 Weak enforcement of the continuity	. 122
		4.5.3 Smoothing transfer	. 123
	4.6	Numerical techniques. Newton-Raphson	
		method	. 124
		4.6.1 Line search method	. 128
	4.7	Concluding Remarks	. 129
5	The	error in the constitutive equations to assess the quality of the	е
9	finit	e element solution with mesh <i>constant</i> in time	
	5.1	Introduction	. 130

	5.2	Extend	led Dissipation Error	131
		5.2.1	Construction of the admissible solution	133
		5.2.2	Error Expressions	138
		5.2.3	Numerical example	141
		5.2.4	Analysis of the error	155
	5.3	Dissipa	ation Error	163
		5.3.1	Construction of the admissible solution	164
		5.3.2	Error Expressions	165
		5.3.3	Comparison between the two errors	167
	5.4	Conclu	ding Remarks	169
c	NL		-tli	
6	Nur	nerical	studies of transfer operations for adaptive finite element	1 197 1
6	Nur solu	nerical tions	studies of transfer operations for adaptive finite element	171
6	Nur solu 6.1	nerical i tions Introdu	studies of transfer operations for adaptive finite element	1 71 171
6	Nur solu 6.1 6.2	nerical tions Introdu Numer	studies of transfer operations for adaptive finite element uction	171 171
6	Nur solu 6.1 6.2	nerical tions Introdu Numer solution	studies of transfer operations for adaptive finite element uction	1 71 171 171
6	Nur solu 6.1 6.2	nerical tions Introdu Numer solution 6.2.1	studies of transfer operations for adaptive finite element uction	171 171 172 172
6	Nur solu 6.1 6.2	nerical tions Introdu Numer solution 6.2.1	studies of transfer operations for adaptive finite element uction	1 71 171 172 172 172 172
6	Nur solu 6.1 6.2	nerical tions Introdu Numer solution 6.2.1	studies of transfer operations for adaptive finite element uction	171 171 172 172 172 172 174
6	Nur solu 6.1 6.2	nerical tions Introdu Numer solution 6.2.1	studies of transfer operations for adaptive finite element uction	171 171 172 172 172 172 174 177
6	Nur solu 6.1 6.2	nerical tions Introdu Numer solution 6.2.1 Conclu	studies of transfer operations for adaptive finite element uction	171 171 172 172 172 172 174 177 206
6	Nur solu 6.1 6.2 6.3 Con	nerical tions Introdu Numer solution 6.2.1 Conclu	studies of transfer operations for adaptive finite element uction	171 171 172 172 172 172 174 177 206 207
6 7	Nur solu 6.1 6.2 6.3 Con 7 1	nerical tions Introdu Numer solution 6.2.1 Conclu	studies of transfer operations for adaptive finite element uction	171 171 172 172 172 172 174 177 206 207 208

Part I

Error Estimation. General Theory

.

Chapter 1

Introduction

In the scientific approach to the solution of engineering problems mathematical models are developed that describe the essential physical aspects of the problem. Mathematical models are mostly expressed in terms of complex differential and/or integral equations, and their solution in the majority of practical situations is impossible in the closed form. Therefore various approximation techniques have been used in obtaining some form of the approximate solution of the original problem.

It was not until the mid-fifties and advances in computer technology that made the approximate methods a powerful approach to the solution of practical engineering problems. Since then the approximate methods, and the finite element method in particular, have become the principal approach to solution of a large number of industrial applications in all areas of engineering including structural, civil, mechanical, aeronautical, chemical and since recently biomedical engineering.

One of the principal difficulties associated with the use of the finite element, and other approximate methods, is related to the accuracy, i.e., closeness of the approximation to the solution of the original problem. Since the closed form is not, in general, available the so-called error estimation techniques have been proposed.

The interest is here given to the discretization errors, which are caused by the numerical discretization of the continuous mathematical model. These involve approximations with the finite elements for the space variable and with the backward Euler method for the time variable. A first question is, therefore, to quantify the distance between the approximate solution and the exact solution. This is typified by the choice of a norm which is usually indicated by the functional setting in which the variational formulation is posed. These measures have usually global character, in the sense that the values of the function and/or its derivatives all over the space-time domain of interest are involved. In general, one can define measure of the error as a non-negative scalar function depending on the approximate and exact solution and describing the extent to which the approximate solution fails to coincide with the exact solution. This definition generally involves the knowledge of the exact solution which is unknown. Thus, the question of providing an estimate of this distance comes quite naturally. In particular, our interest is in *a posteriori* error estimates, that is, estimates of a given measure of the error that are constructed after the finite element solution has been computed, and they utilize the finite element solution and the input data of the concrete case of interest. These are different from *a priori* error estimates which are based on a knowledge of the characteristics of the exact solution, and provide qualitative information about the asymptotic rate of convergence of the approximation as the discretization parameters approach their corresponding limit values.

A posteriori error estimates play an important role in two related aspects of finite element calculations. First, such estimates provide the user of a finite element code with valuable information about the overall accuracy and reliability of the calculation. Second, since most *a posteriori* error estimates are computed locally, they also contain significant information about the distribution of error among individual elements, referred to as error indicators, which can form the basis of adaptive procedures. However, error indicators can also be developed on heuristics and may have no direct relation with the error.

Use of adaptive strategies in the finite element solution of history-dependent problems with incremental methods is of paramount importance. An adaptive strategy can be defined as a computational procedure which delivers the finite element solution for the problem at hand to the prescribed accuracy. Key ingredients are: (i) the availability of an error estimator which accounts for the sources of error associated with the approximation, (ii) error indicators for the choice of the optimal discretization parameters, and finally (iii) a data transfer procedure when the current finite element mesh is different from the one of the previous time step.

In the finite element analysis of these problems the quality of the simulation is generally assessed by physical or heuristic arguments based on the experience and judgement of the analyst. Frequently such arguments are later proved to be flawed, they are specific for the problem under consideration and often they fail to account for all the discretizations introduced, which therefore can produce a misleading trust in the accuracy of the approximate solution produced.

1.1 The scope of the thesis

The extended dissipation error developed in Ladevèze *et al.* (1999) applied to the assessment of the accuracy of the finite element solution obtained by a fully implicit displacement formulation of the elastoplastic problem is able to account for the effects of time and space discretization. The analysis, however, is carried out by assuming the finite element mesh constant throughout the loading process. A property of this error is its non-decreasing character in time due to the accumulation of the discretization errors. As a result, during the computation with incremental procedures, one may need to modify the parameters which define the fully discrete scheme, namely time step size and finite element mesh, in order to obtain the corresponding solution to the prescribed global accuracy.

When only variation of the time step is sufficient to improve the accuracy of the solution, the extended dissipation error can be used to assess the global quality of the finite element approximation because of the time continuity of the associated admissible solution. On contrary, when the finite element mesh is changed at time t_n , two finite element solutions are considered for the same load level: the one at t_n^- , which is associated with the old mesh \mathcal{T}_{h_n} , and the other at t_n^+ , which is associated with the new mesh $\mathcal{T}_{h_{n+1}}$. The solution at t_n^+ is employed to define a time linear interpolation function which we require to satisfy the following property

$$\lim_{\Delta t \downarrow 0} f_d(t_n + \Delta t) = \lim_{\Delta t \downarrow 0} f_i(t_n + \Delta t)$$

where $f_i = f_i(t_n + \Delta t)$ denotes the time linear interpolation over the time interval $[t_n^+, t_{n+1}^-]$ of the discrete values f_n^+ and f_{n+1}^- whereas $f_d = f_d(t_n + \Delta t)$ is the function which associates with any given Δt the solution of the discrete scheme corresponding to the given Δt and data f_n^+ . Consequently, a discontinuity jump appears in the time linear interpolation of the discrete values across the time node t_n as a result of the change of mesh and transfer procedure. In the development of reliable *a posteriori* error estimators, one needs, therefore, to account not only for the time step and finite element mesh size but also for the value of the jump.

In this work, attention is given only to the error estimation procedure itself. With this regard, an error estimator is proposed which allows the assessment of the effects of transfer procedures in displacement finite element solution of rate-independent plasticity discretized in time with the backward Euler method. The extended dissipation error developed in Ladevèze *et al.* (1999) will be augmented consistently by a term which accounts for the time discontinuity in the admissible solution. The new theory is formulated in tensorial notation and its applicability is illustrated on a one dimensional model problem where a detailed study of transfer procedures (Ortiz & Quigley, 1991; Perić *et al.*, 1996; Rashid, 2002) is carried out with numerical results providing confirmation of theoretical developments. With such a posteriori error estimator at hand, indications on how to change the finite element space and define the corresponding data can be given and the assessment of the several transfer operations can be finally framed in the context of the ensuing error.

1.2 Layout

This thesis is divided into two parts. The first one deals with the theory of the measure of the error in the constitutive equations. The theory is general and applies to admissible solutions for the problem under consideration. In the second part applications to the assessment of accuracy of finite element solutions of the initial boundary value problem in elastoplasticity are given.

The first part of the thesis, after this introductory chapter, is arranged as follows:

Chapter 2 gives a brief overview of some error estimators for linear and nonlinear problems. The objective is to illustrate the motivating ideas behind each of the proposed techniques and to provide motivation for the use of the error in the constitutive equations for the assessment of the accuracy of finite element solutions on evolving meshes.

Chapter 3 presents the general theory of the error in the constitutive equations to assess the quality of the so-called admissible solutions of dissipative nonlinear problems. We employ the theory of the extended dissipation error developed by Ladevèze *et al.* (1999) to accommodate admissible solutions with a discontinuity jump at the time instant t_n . This leads to a new error estimate which we call augmented extended dissipation error.

In the second part of the thesis, the arrangement is as follows:

Chapter 4 reports on the displacement finite element method for the solution of the initial boundary value problem of an elastoplastic model with internal variables and discusses the nature of the ensuing discretization errors. In particular, there is the fundamental observation that change of data and/or finite element mesh from one time interval to the other can be both related to a discontinuity jump of the approximate solution across the time instant t_n . Consequently, in the development of reliable *a posteriori* error estimates one needs to account also for the jump. A critical review of the current techniques to transfer data from one mesh to the other concludes the chapter.

Chapter 5 focuses mainly on how to use the extended dissipation error to assess the quality of the finite element solution with constant mesh in time. The main problem is, therefore, the definition of a corresponding admissible solution, which reflects the approximations associated with the finite element solution. After giving general guidelines, actual criteria to construct an admissible solution in the case of the Prandtl–Reuss model are given. The general theory is then applied to assess the quality of the finite element solution of one dimensional elastoplastic bar under axial load. The example shows that all trends on the error in the state laws and dissipation contribution are meaningful. Notable is also the comparison with classical measures of the exact error in solution. This shows that the extended dissipation error reflects quite well the evolution of the admissible solution with respect to the exact one as described by more classical measures of the error. Comparison with the classical dissipation error introduced in Ladevèze (1989) and developed in Ladevèze & Moës (1997) concludes the chapter.

Chapter 6 presents a general methodology for the assessment of the global quality of displacement finite element solutions of elastoplastic problems discretized in time with the backward Euler method on dynamically changing meshes. The methodology employs the extended dissipation error, augmented by the term which accounts for the time discontinuity in the admissible solutions. Its applicability is shown on a one dimensional model problem where a detailed study of the transfer operators is presented. The numerical results provide confirmation of the theoretical developments.

Chapter 7 presents a short summary and the conclusions of this work. Some suggestions for future research are finally given.

Chapter 2

Overview on a posteriori error estimates. Literature Review

2.1 Introduction

In this chapter we give a brief overview on some error estimators for linear and nonlinear problems. The objective is mainly to illustrate the motivating ideas behind each of the proposed techniques, rather than attempting to provide a (necessarily incomplete) list of error estimators. We start, therefore, with reviewing some a pos*teriori* error estimators for linear elliptic problems where it is possible to provide a theoretical unifying framework, which encompasses most of the existing procedures. Such analysis has been presented in Verfurth (1996), for instance. The advances obtained in the comprehension of the mechanism of error propagation corresponds to the maturity reached in the theory of linear elliptic partial differential equations (Evans, 1999) and their finite element approximation (Ciarlet, 1978). On contrary, the remaining class of problems, and in particular the mathematical models describing rate-independent and rate-dependent plasticity, present a far less unified approach, as the various types of nonlinearity are involved in quite different ways. However, for the class of problems which can be analysed with the methods of the convex analysis it is possible to identify some underlying threads. These derive from the duality theory which is a modern branch of the calculus of variation originated from the works of Fenchel, Moreau, Rockafellar and others. The key idea of the theory – simultaneous analysis of the primal and the so-called dual variational problem - is, for instance, exploited in the works of Ainsworth & Oden (1993); Ladevèze & Pelle (2001) and Paraschivoiu et al. (1997), thus representing an important tool in the *a posteriori* error analysis for those classes of problems.

2.2 Linear problems

The main concepts for the global control of the discretization error in energy norm for linear elliptic partial differential equations are next presented for the displacement formulation of the model of linear elasticity. At this end, some preliminaries are necessary.

Let Ω be a bounded open connected subset of the three dimensional Euclidean space with polyhedral boundary $\partial \Omega = \partial \Omega_d \cup \partial \Omega_t$ and $\partial \Omega_d \cap \partial \Omega_t = \emptyset$. Here, $\partial \Omega_d$ denotes the part of $\partial \Omega$ where a prescribed displacement u_d is fixed whereas the complementary part $\partial \Omega_t$ is where the boundary traction forces t are applied. The displacement field u = u(x) of the linear elastic model under the body force b is solution of the following variational problem

Find:
$$\boldsymbol{u} \in \boldsymbol{u}_d + \mathcal{V}_0$$

$$\int_{\Omega} \mathbf{C} \nabla_s \boldsymbol{u} \colon \nabla \boldsymbol{\eta} \, \mathrm{d}\Omega = \int_{\Omega} \boldsymbol{b} \cdot \boldsymbol{\eta} \, \mathrm{d}\Omega + \int_{\partial \Omega_t} \boldsymbol{t} \cdot \boldsymbol{\eta} \, \mathrm{d}s \quad \forall \, \boldsymbol{\eta} \in \mathcal{V}_0$$
(2.1)

where **C** is the definite positive Hooke's tensor, $\nabla_s \boldsymbol{u}$ is the symmetric part of the second order tensor $\nabla \boldsymbol{u}$, gradient of \boldsymbol{u}^{\S} , and \mathcal{V}_0 is the space of the test functions defined as $\mathcal{V}_0 = \{\boldsymbol{v} = \{\boldsymbol{v}_i\}_{i=1}^3 \in [\mathrm{H}^1(\Omega)]^3 | \boldsymbol{v} = \boldsymbol{0} \text{ on } \partial \Omega_d\}$ with $\mathrm{H}^1(\Omega)$ the standard Sobolev space of scalar functions v_i of $L^2(\Omega)$ with finite norm

$$\int_{\Omega} v_i^2 \,\mathrm{d}\Omega + \int_{\Omega} (\nabla v_i)^2 \,\mathrm{d}\Omega < \infty.$$

The well-possessedness of problem (2.1) follows from the Lax-Milgram theorem, for $\partial \Omega_d$ has positive measure (Ciarlet, 1978). Furthermore, the latter and the properties of **C** permit one to conclude that the bilinear form

$$(\boldsymbol{v}, \boldsymbol{w})_{\mathcal{V}_0} = \int_{\Omega} \mathbf{C} \nabla_s \boldsymbol{v} \colon \nabla \boldsymbol{w} \,\mathrm{d}\Omega$$
 (2.2)

defines an inner product in the space \mathcal{V}_0 . The associated norm is referred to as the energy norm; it is equivalent to the standard norm of \mathcal{V}_0 and it is given by

$$|||\boldsymbol{v}||| \stackrel{\text{def}}{=} \left(\int_{\Omega} \mathbf{C} \nabla_{\boldsymbol{s}} \boldsymbol{v} \colon \nabla \boldsymbol{v} \, \mathrm{d}\Omega \right)^{\frac{1}{2}} = \sup_{\boldsymbol{w} \in \mathcal{V}_0} \frac{\int_{\Omega} \mathbf{C} \nabla_{\boldsymbol{s}} \boldsymbol{v} \colon \nabla \boldsymbol{w} \, \mathrm{d}\Omega}{|||\boldsymbol{w}|||}, \quad (2.3)$$

where the second equality follows from the Cauchy-Scwartz's inequality. Hereafter, the space \mathcal{V}_0 is endowed with the energy norm.

We will consider conforming finite element approximations of the problem (2.1). With this regard, let $\mathcal{T}_h = \{\Omega_e\}$ be a finite element partition of Ω made up of polyhedrons Ω_e with faces γ . We denote with $\mathcal{E}^{h,\Omega}$ and $\mathcal{E}^{h,\partial\Omega_t}$ the sets of the faces which are contained in Ω (i.e., the interior faces) and in $\partial\Omega_t$, respectively. For each face $\gamma \in \mathcal{E}^{h,\Omega}$ we denote a fixed unit normal \boldsymbol{n} , chosen arbitrarily from the two possibilities. For the faces $\gamma \in \mathcal{E}^{h,\partial\Omega_t}$, \boldsymbol{n} is the outward normal to Ω . The definition of the elements $\Omega_{e,in}$ and $\Omega_{l,out}$ in relation to \boldsymbol{n} is depicted in Figure 2.1. Let $\mathcal{V}_0^h \subset \mathcal{V}$

[§]The symbol : in (2.1) denotes the double contraction operator. When it acts between second order tensors it delivers a scalar whereas when it acts between a second order tensor and a vector the outcome is a vector. The symbol \cdot is, on the other hand, the inner product between vectors. We also recall that the action of a fourth order tensor on a second order tensor is a second order tensor. For the definitions of operations on tensors we refer to Gurtin (1981).



Figure 2.1: Definition of the elements $\Omega_{e,in}$ and $\Omega_{l,out}$ in relation to n

be any conforming finite element space associated with \mathcal{T}_h and $\{\mathbb{N}_i^h\}$ the basis of the finite element shape functions. The finite element solution $u_h \in u_d + \mathcal{V}_0^h$ of the problem (2.1) is given by

Find:
$$\boldsymbol{u}_h \in \boldsymbol{u}_d + \mathcal{V}_0^h$$

$$\int_{\Omega} \mathbf{C} \nabla_s \boldsymbol{u}_h \colon \nabla \boldsymbol{\eta}_h \, \mathrm{d}\Omega = \int_{\Omega} \boldsymbol{b} \cdot \boldsymbol{\eta}_h \, \mathrm{d}\Omega + \int_{\partial \Omega_t} \boldsymbol{t} \cdot \boldsymbol{\eta}_h \, \mathrm{d}s \quad \forall \, \boldsymbol{\eta}_h \, \in \, \mathcal{V}_0^h$$
(2.4)

Thus, the discretization error $e = u - u_h$ is solution of the variational problem

Find:
$$\boldsymbol{e} \in \mathcal{V}_0$$

$$\int_{\Omega} \mathbf{C}(\nabla_s \boldsymbol{e}): \nabla \boldsymbol{\eta} \, \mathrm{d}\Omega = \int_{\Omega} \boldsymbol{b} \cdot \boldsymbol{\eta} \, \mathrm{d}\Omega + \int_{\partial \Omega_t} \boldsymbol{t} \cdot \boldsymbol{\eta} \, \mathrm{d}s - \int_{\Omega} \mathbf{C} \nabla_s \boldsymbol{u}_h: \nabla \boldsymbol{\eta} \, \mathrm{d}\Omega, \quad \forall \, \boldsymbol{\eta} \in \mathcal{V}_0$$
(2.5)

and, one has the error representation formula

$$|||\boldsymbol{e}|||^{2} = \int_{\Omega} \mathbf{C}(\nabla_{s}\boldsymbol{e}) \colon \nabla \boldsymbol{e} \,\mathrm{d}\Omega = \int_{\Omega} \boldsymbol{b} \cdot \boldsymbol{e} \,\mathrm{d}\Omega + \int_{\partial\Omega_{t}} \boldsymbol{t} \cdot \boldsymbol{e} \,\mathrm{d}s - \int_{\Omega} \mathbf{C}\nabla_{s}\boldsymbol{u}_{h} \colon \nabla \boldsymbol{e} \,\mathrm{d}\Omega.$$
(2.6)

The functional

$$\mathcal{R}_{\boldsymbol{u}_h}(\boldsymbol{\eta}) = \int_{\Omega} \boldsymbol{b} \cdot \boldsymbol{\eta} \, \mathrm{d}\Omega + \int_{\partial \Omega_t} \boldsymbol{t} \cdot \boldsymbol{\eta} \, \mathrm{d}s - \int_{\Omega} \mathbf{C} \nabla_s \boldsymbol{u}_h \colon \nabla \boldsymbol{\eta} \, \mathrm{d}\Omega, \qquad (2.7)$$

which appears at the right hand side of equation (2.5), is referred to as residual functional of u_h with respect to (2.1). It can be shown that it is an element of the dual topological space[‡] \mathcal{V}_0^* of \mathcal{V}_0 . From (2.4) and accounting for the definition (2.7) it follows

$$\mathcal{R}_{\boldsymbol{u}_h}(\boldsymbol{\eta}_h) = \int_{\Omega} \boldsymbol{b} \cdot \boldsymbol{\eta}_h \,\mathrm{d}\Omega + \int_{\partial\Omega_t} \boldsymbol{t} \cdot \boldsymbol{\eta}_h \,\mathrm{d}s - \int_{\Omega} \mathbf{C} \nabla_s \boldsymbol{u}_h \colon \nabla \boldsymbol{\eta}_h \,\mathrm{d}\Omega = 0, \quad \forall \boldsymbol{\eta}_h \in \mathcal{V}_0^h$$
(2.8)

[†] Let \mathcal{V} be a topological vector space. The dual topological space \mathcal{V}^* of \mathcal{V} is the vector space of the linear continuous functionals over \mathcal{V} . If \mathcal{V} is a normed space, a linear continuous functional $\mathcal{F}(v)$ over \mathcal{V} is a bounded functional, that is, $\sup_{v \in \mathcal{V}} \frac{|\mathcal{F}(v)|}{\|v\|_{\mathcal{V}}} < \infty$. In this case, the vector space \mathcal{V}^* is endowed with the norm $\|\mathcal{F}\|_{\mathcal{V}^*} = \sup_{v \in \mathcal{V}} \frac{|\mathcal{F}(v)|}{\|v\|_{\mathcal{V}}}$ (Brezis, 1986).

that is, the residual functional $\mathcal{R}_{\boldsymbol{u}_h}(\boldsymbol{\eta}_h)$ vanishes over $\mathcal{V}_0^h \subset \mathcal{V}_0$. Thus, it is also

$$(\boldsymbol{e},\boldsymbol{\eta}_h)_{\mathcal{V}_0} = \int_{\Omega} \mathbf{C} \nabla_s \boldsymbol{e} \colon \nabla \boldsymbol{\eta}_h \, \mathrm{d}\Omega = \int_{\Omega} \left(\boldsymbol{\sigma}_{ex} - \mathbf{C} \nabla_s \boldsymbol{u}_h \right) \colon \nabla \boldsymbol{\eta}_h \, \mathrm{d}\Omega = 0, \quad \forall \boldsymbol{\eta}_h \in \mathcal{V}_0^h$$

$$(2.9)$$

which is the orthogonality condition between the discretization error e and the finite element space \mathcal{V}_0^h with respect to the inner product defined by (2.2). Condition (2.9) means that the error $e \in \mathcal{V}_0$ solution of (2.5) presents zero component in the space \mathcal{V}_0^h .

The localization of the integrals in (2.7) over each finite element Ω_e and use of integration by parts gives

$$\mathcal{R}_{\boldsymbol{u}_{h}}(\boldsymbol{\eta}) = \sum_{\Omega_{e}\in\mathcal{T}^{h}} \int_{\Omega_{e}} \boldsymbol{r}_{\boldsymbol{u}_{h}} \cdot \boldsymbol{\eta} \, \mathrm{d}\Omega + \sum_{\gamma\in\mathcal{E}^{h,\partial\Omega_{t}}\cup\mathcal{E}^{h,\Omega}} \int_{\gamma} \boldsymbol{J}_{\boldsymbol{u}_{h}}^{\gamma} \cdot \boldsymbol{\eta} \, \mathrm{d}s =$$
$$= \sum_{\Omega_{e}\in\mathcal{T}^{h}} \left\{ \int_{\Omega_{e}} \boldsymbol{r}_{\boldsymbol{u}_{h}} \cdot \boldsymbol{\eta} \, \mathrm{d}\Omega + \sum_{\gamma\in\partial\Omega_{e}} \int_{\gamma} \boldsymbol{J}_{\boldsymbol{u}_{h}}^{\gamma} \cdot \boldsymbol{\eta} \, \mathrm{d}s \right\}.$$
(2.10)

In equation (2.10), $\mathbf{r}_{u_h} = \operatorname{div} \mathbf{C} \nabla_s u_h + \mathbf{b}$ is the regular part of the global residual associated with the lack of equilibrium of the finite element solution within the interior of the elements Ω_e , whereas $J_{u_h}^{\gamma}$ has the following definition

$$oldsymbol{J}_{oldsymbol{u}_h}^{\gamma} = \left\{egin{array}{cc} \left[{f C}
abla_s oldsymbol{u}_h \colon oldsymbol{n}
ight]_{\gamma} & ext{on } \gamma \in \mathcal{E}^{h,\Omega} \ t - {f C}
abla_s oldsymbol{u}_h \colon oldsymbol{n} & ext{on } \gamma \in \mathcal{E}^{h,\partial\Omega_t} \end{array}
ight.$$

where $[\mathbf{C}\nabla_s \boldsymbol{u}_h:\boldsymbol{n}]_{\gamma}$ denotes the jump of $\mathbf{C}\nabla_s \boldsymbol{u}_h:\boldsymbol{n}$ across the edge $\gamma \in \mathcal{E}^{h,\Omega}$; this value is independent on the choice of \boldsymbol{n} . Thus, $J_{\boldsymbol{u}_h}^{\gamma}$ represents the singular part of the global residual due to the lack of equilibrium in the normal tractions across the interelement boundaries and on the boundary $\partial\Omega_t$, that is, on $\gamma \in \mathcal{E}^{h,\partial\Omega_t} \cup \mathcal{E}^{h,\Omega}$.

Evaluating the discretization error means to solve the same linear elastic model as (2.1) but with different boundary conditions and external loads. Now, the boundary conditions are given by $\boldsymbol{e} = \boldsymbol{0}$ on $\partial\Omega_d$ and $\boldsymbol{C}\nabla_s \boldsymbol{e} \colon \boldsymbol{n} = \boldsymbol{t} - \boldsymbol{C}\nabla_s \boldsymbol{u}_h \colon \boldsymbol{n}$ on $\partial\Omega_t$, whereas the body forces are $-\operatorname{div} \boldsymbol{C}\nabla_s \boldsymbol{u}_h - \boldsymbol{b}$ over each element Ω_e and $[\boldsymbol{C}\nabla_s \boldsymbol{u}_h \colon \boldsymbol{n}]_{\gamma}$ are the surface loads applied on the faces $\gamma \in \mathcal{E}^{h,\Omega}$.

Nevertheless, problem (2.5) presents the same difficulty as the original problem (2.1), for it is posed in the infinite dimensional space \mathcal{V}_0 . One could, therefore, think of computing a finite element approximation e_h of e. The adoption of the same finite element space \mathcal{V}_0^h would, however, deliver $e_h = 0$, because of the orthogonality of e with respect to \mathcal{V}_0^h . If a more accurate approximation for the error e is sought, this would be equivalent to solve the original problem. Furthermore, this would also involve a computational effort that could be directed toward the evaluation of a better approximation for the solution of (2.1). In such a case, however, the error of the new more accurate finite element solution should presumably be estimated in any case, so that the same dilemma re-appears. Keeping at the minimum the

computation cost for the assessment of the accuracy of an approximate solution is, indeed, a fundamental feature of any error estimation technique.

The current schemes for accurate and quantitative estimates for the discretization error are usually classified according to how estimates of a given norm (or linear functional) of e are obtained. Our attention is here mainly directed to the control of the accuracy in energy norm. In particular, we will consider the residual type and the averaging type error estimates.

2.2.1 Residual type error estimates

The residual functional $\mathcal{R}_{\boldsymbol{u}_h}(\boldsymbol{v})$ is the forcing of the problem (2.5) that defines the finite element error \boldsymbol{e} . As a result, the solution of (2.5) will depend on $\mathcal{R}_{\boldsymbol{u}_h}(\boldsymbol{v})$. The starting point for this class of estimators can be assumed to be the equality between the energy norm of the error and the norm of the residual functional in \mathcal{V}_0^{\star} . This follows easily from the definition of norm of residual (see note \ddagger), equation (2.6) and (2.3),

$$\|\mathcal{R}_{\boldsymbol{u}_h}\|_{\boldsymbol{\mathcal{V}}_0^\star} = \sup_{\boldsymbol{v}\in\boldsymbol{\mathcal{V}}_0} \frac{|\mathcal{R}_{\boldsymbol{u}_h}(\boldsymbol{v})|}{|||\boldsymbol{v}|||} = \sup_{\boldsymbol{v}\in\boldsymbol{\mathcal{V}}_0} \frac{\int_{\Omega} \mathbf{C}(\nabla_s \boldsymbol{e}) \colon \nabla \boldsymbol{v} \, \mathrm{d}\Omega}{|||\boldsymbol{v}|||} = \frac{\int_{\Omega} \mathbf{C}(\nabla_s \boldsymbol{e}) \colon \nabla \boldsymbol{e} \, \mathrm{d}\Omega}{|||\boldsymbol{e}|||} = |||\boldsymbol{e}|||.$$

Estimates for |||e||| are, therefore, obtained by providing estimates of $||\mathcal{R}_{u_h}||_{\mathcal{V}_0^*}$. In turn, these can be obtained either through a direct computation using the finite element solution and the available data, or by solving local auxiliary problems, which give a representation of the functional $\mathcal{R}_{u_h} = \mathcal{R}_{u_h}(v)$. The first class of residual error estimates are referred to as *explicit* whereas the second one is called *implicit*.

2.2.1.1 Explicit residual a posteriori error estimates

These estimators were first introduced in Babuska & Rheinboldt (1978b) for the assessment of the accuracy of finite element approximations with higher order elements of 1D elliptic problems and then extended to 2D in Babuska & Rheinboldt (1979b). The bound can be expressed, in general, as follows

$$|||\boldsymbol{e}||| \leq \sum_{\Omega_{e} \in \mathcal{T}_{h}} \left\{ C_{I,1}^{e} h_{e} \|\boldsymbol{r}_{\boldsymbol{u}_{h}}\|_{[L^{2}(\Omega_{e})]^{3}} + \sum_{\gamma \in \partial \Omega_{e}} C_{I,2}^{e} h_{e}^{\frac{1}{2}} \|\boldsymbol{J}_{\boldsymbol{u}_{h}}^{\gamma}\|_{[L^{2}(\partial \Omega_{e})]^{3}} \right\}$$
(2.11)

where $C_{I,i}^{e}$, i = 1, 2 are interpolation constants (Ciarlet, 1978) which depend on the shape of the element and the local order of the polynomial approximation, whereas h_e is the diameter of the element Ω_e .

Apart from the constant $C_{I,i}^e$, all of the quantities on the right-hand side can be computed directly from the finite element approximation and the data for the problem of interest.

The relative importance of the two terms which appear in (2.11), the one associated with the interior residual r_{u_h} and the one associated with the jump $J_{u_h}^{\gamma}$,

were analysed for two-dimensional problems in Babuska & Miller (1987) and in Babuska & Yu (1987) for the case of irregular grids of bilinear quadrilaterals and biquadratic approximations, respectively. In the first case, the dominant term of the estimate was the residual jump whereas in the second case the error could be expressed only in terms of the residual in the interior of each element (see also the work of Carstensen & Verfurth (1999), where it is proven that for general meshes of linear triangles, the energy norm of the error may be estimated by employing only the jump $J_{u_h}^{\gamma}$).

Error estimates in norms other than energy norm were analysed in Babuska & Rheinboldt (1981), though for one dimensional problems. However, the theoretical analysis appears quite cumbersome and not providing for an immediate extension. A streamlining of the estimation technique was contributed noteworthily by Johnson and coworkers in Eriksson & Johnson (1991); Johnson & Hansbo (1992); Eriksson *et al.* (1995). Their works involve a number of basic ideas which represent also the basis of the estimates for the quantity of interest. The gist of the procedure is the duality argument used by Aubin and Nitsche for the derivation of a *priori* error estimates in norms other than the energy norm (Ciarlet, 1978). The duality argument is also used for the purpose of deriving a *posteriori* error estimates through the following points (Johnson, 1994):

- 1. Error representation formula by means of a dual problem
- 2. Orthogonality of the Galerkin approximation
- 3. Interpolation error estimates
- 4. Strong stability of the dual problem

As an illustration of this procedure, we next sketch the control of the error in L^2 norm. Also, for simplicity, we will assume that $\partial \Omega_d = \partial \Omega$. In this case, the error representation formula is given by

$$\|\boldsymbol{e}\|_{[L^2(\Omega)]^3}^2 = \int_{\Omega} \boldsymbol{e} \cdot \boldsymbol{e} \,\mathrm{d}\Omega = \int_{\Omega} \boldsymbol{C} \nabla_s \boldsymbol{e} \colon \nabla \boldsymbol{\varphi} \,\mathrm{d}\Omega$$

where $\varphi \in \mathcal{V}_0$ is solution of the dual problem

Find:
$$\boldsymbol{\varphi} \in \mathcal{V}_0$$

$$\int_{\Omega} \mathbf{C} \nabla_s \boldsymbol{\varphi} \colon \nabla \boldsymbol{\eta} \, \mathrm{d}\Omega = \int_{\Omega} \boldsymbol{e} \cdot \boldsymbol{\eta} \, \mathrm{d}\Omega \quad \forall \, \boldsymbol{\eta} \in \mathcal{V}_0$$
(2.12)

As a result of the Galerkin orthogonality (2.9), it is also

$$\|\boldsymbol{e}\|_{[L^{2}(\Omega)]^{3}}^{2} = \int_{\Omega} \mathbf{C} \nabla_{\boldsymbol{s}} \boldsymbol{e} \colon \left(\nabla \boldsymbol{\varphi} - \nabla \underbrace{\mathcal{I}_{\mathcal{V}_{0}^{h}} \boldsymbol{\varphi}}_{\boldsymbol{\varphi}_{h}} \right) \mathrm{d}\Omega$$
(2.13)

where $\mathcal{I}_{\mathcal{V}_0^h}$: $\mathcal{V}_0 \to \mathcal{V}_0^h$ is a suitable \mathcal{V}_0^h -interpolation operator (Clément, 1975; Bernardi & Girault, 1998).

From equation (2.5), the localization of the integral in (2.13) and the use of Cauchy-Schwartz's inequality, it follows

$$\begin{split} \|\boldsymbol{e}\|_{[L^{2}(\Omega)]^{3}}^{2} &\leq \sum_{\Omega_{\boldsymbol{e}}\in\mathcal{T}_{h}} \left\{ \|\boldsymbol{r}_{\boldsymbol{u}_{h}}\|_{[L^{2}(\Omega_{\boldsymbol{e}})]^{3}} \|\boldsymbol{\varphi}-\boldsymbol{\varphi}_{h}\|_{[L^{2}(\Omega_{\boldsymbol{e}})]^{3}} + \right. \\ &\left. + \sum_{\boldsymbol{\gamma}\in\partial\Omega_{\boldsymbol{e}}} \|\boldsymbol{J}_{\boldsymbol{u}_{h}}^{\boldsymbol{\gamma}}\|_{[L^{2}(\boldsymbol{\gamma})]^{3}} \|\boldsymbol{\varphi}-\boldsymbol{\varphi}_{h}\|_{[L^{2}(\boldsymbol{\gamma})]^{3}} \right\} \end{split}$$

The terms $\|\varphi - \varphi_h\|_{[L^2(\Omega_e)]^3}$ and $\|\varphi - \varphi_h\|_{[L^2(\partial\Omega_e)]^3}$ describe the weight of the terms $\|r_{\boldsymbol{u}_h}\|_{[L^2(\Omega_e)]^3}$ and $\|J_{\boldsymbol{u}_h}^{\gamma}\|_{[L^2(\partial\Omega_e)]^3}$ in the local contribution to the error in L^2 -norm, respectively.

The use of the interpolation error estimates (Ainsworth & Oden, 2000)

$$\|\varphi-\varphi_h\|_{[L^2(\Omega_e)]^3} \leq C_{Ie,1} \|\nabla\varphi\|_{[L^2(\tilde{\Omega}_e)]^{3\times 3}}, \quad \|\varphi-\varphi_h\|_{[L^2(\partial\Omega_e)]^3} \leq C_{Ie,1} \|\nabla\varphi\|_{[L^2(\tilde{\Omega}_e)]^{3\times 3}},$$

where $\tilde{\Omega}_e$ is the patch of elements associated with Ω_e^{\P} , along with the stability of the global dual problem (2.12),

$$\|
abla arphi\|_{[L^2(\Omega)]^{3 imes 3}} \leq C_s \|oldsymbol{e}\|_{[L^2(\Omega)]^3}$$

where C_s is the stability constant, it finally, gives

$$\|\boldsymbol{e}\|_{[L^{2}(\Omega)]^{3}} \leq C_{s} \sum_{\Omega_{e} \in \mathcal{T}_{h}} \left\{ C_{Ie,1}h_{e} \|\boldsymbol{r}_{\boldsymbol{u}_{h}}\|_{[L^{2}(\Omega_{e})]^{3}} + \sum_{\gamma \in \partial \Omega_{e}} C_{Ie,2}h_{e}^{\frac{1}{2}} \|\boldsymbol{J}_{\boldsymbol{u}_{h}}^{\gamma}\|_{[L^{2}(\gamma)]^{3}} \right\}.$$

The above error estimate does not admit cancellation between different elements Ω_e and on element level, as well. As a result, it is not very sharp. Furthermore, the mechanism of error propagation is accounted for only through the global stability constant C_s . In order to reflect better the local contribution of the local residual, in the context of control of quantity of interest in linear elasticity, Rannacher & Suttmeier (1997) implement only the steps 1, 2 and 3 of the above procedure. The term $\|\nabla \varphi\|_{[L^2(\tilde{\Omega}_e)]^{3\times 3}}$ is computed numerically by simply taking the first order difference quotient of an approximate solution $\varphi_h \in \mathcal{V}_0^h$ of the dual problem. This procedure was further improved by Suli & Houston (2001) in the case of error control of output of hyperbolic problems. Only the steps 1 and 2 were implemented and an approximation of the dual solution was retained in the bound as a local weight-function.

[¶]Given an element Ω_e of the mesh \mathcal{T}_h , the patch $\tilde{\Omega}_e$ of elements associated with Ω_e is the set of elements Ω_l which share an edge with Ω_e . Given a vertex *i* of the mesh \mathcal{T}_h , the patch ω_i of elements associated with *i* is the set of elements Ω_l which have *i* as one of its vertices (Verfurth, 1996).

2.2.1.2 Implicit residual a posteriori error estimates

Estimates of the norm of the residual can be also obtained by solving local problems which define approximations to the local representation of the residual as opposite to the *explicit* estimators described in the previous section, which are computed directly in terms of the norm of the residuals r_{u_h} and $J_{u_h}^{\gamma}$. Despite the simplicity of implementation, the main disadvantages of the *explicit* estimators are the presence of generally unknowns constants $C_{I,i}$ and the lack of sharpness of the bound. The latter is consequence of various applications of the Cauchy-Schwartz inequality which provokes the loss of cancellation between the various types of residual (Ainsworth & Oden, 2000).

With the *implicit* approach, on contrary, in the definition of the estimate, one tries to retain the structure of the equation (2.5), which defines the error, as far as possible.

These estimators have, generally, the following format (Babuska & Strouboulis, 2001)

$$\eta = \sqrt{\sum_{\omega_i \in \mathcal{R}} \|\boldsymbol{e}_{\omega_i}\|^2}$$

where $\mathcal{R} = \{\omega_i\}$ is a covering of Ω and e_{ω_i} is solution of a boundary value problem of the form

Find:
$$\boldsymbol{e}_{\omega_i} \in \mathcal{S}(\omega_i)$$

$$\int_{\Omega} \mathbf{C} \nabla_s \boldsymbol{e}_{\omega_i} \colon \nabla \boldsymbol{\eta} \, \mathrm{d}\Omega = \mathcal{F}_{\omega_i}(\boldsymbol{\eta}), \quad \forall \boldsymbol{\eta} \in \mathcal{S}(\omega_i)$$
(2.14)

with $\mathcal{S}(\omega_i)$ a suitable solution space and $\mathcal{F}_{\omega_i}(\eta)$ is defined in terms of the residuals r_{u_h} and $J_{u_h}^{\gamma}$.

According to the formulation of the auxiliary problems to solve, we distinguish

- subdomain residual error estimators;
- element residual error estimators;
- equilibrated element residual error estimators.

The subdomain residual error estimator was first introduced by Babuska & Rheinboldt (1978a). The main argument is a localization via a partition of unity of Ω which leads to problems posed on the patch of elements ω_i associated with each node *i*. The solution space $S(\omega_i)$ is a finite element space with its elements vanishing on $\partial \omega_i - \partial \Omega_t$, continuous and piecewise polynomials of a sufficiently high degree whereas $\mathcal{F}_{\omega_i}(\boldsymbol{\eta})$ is given by

$$\mathcal{F}_{\omega_i}(\boldsymbol{\eta}) = \int_{\omega_i} \boldsymbol{b} \cdot \boldsymbol{\eta} \, \mathrm{d}\Omega + \int_{\partial \omega_i \cap \partial \Omega_i} \boldsymbol{t} \cdot \boldsymbol{\eta} \, \mathrm{d}s - \int_{\omega_i} \mathbf{C} \nabla_s \boldsymbol{u}_h \colon \nabla \boldsymbol{\eta} \, \mathrm{d}\Omega.$$

The function e_{ω_i} is, therefore, solution of a Dirichlet problem with homogeneous essential boundary conditions on $\partial \omega_i$. Existence and uniqueness of this solution is

guaranteed by the Lax-Milgram's theorem. The local patches used in this technique are, however, rather expensive to approximate accurately. In effect, each element is treated several times according to the number of patches with which it is associated. Also, the error indicators $||e_{\omega_i}||$ are in this way associated with the patches ω_i and not with the single element Ω_e . This makes more difficult the definition of an optimal adaptive procedure based on $||e_{\omega_i}||$. The use of patch of elements ω_i is essentially a consequence of imposing Dirichlet boundary conditions on the auxiliary problems and of certain conditions required for the reliability of the error estimator (Verfurth, 1996).

On contrary, if only Neumann boundary conditions are imposed on the auxiliary problems, one can choose $\omega_i = \Omega_e$ (Verfurth, 1996). The resulting error estimators are referred to as *elemental* types and an immediate outcome of this approach is to have error indicators defined element by element. However, some care must be taken to insure the local Neumann problems are well posed.

The different estimation techniques differ in the way the well-posedness is achieved. With this regard, we distinguish the equilibrated elemental residual error estimators obtained by choosing the boundary data so that the underlying local problem is well posed and the elemental residual error estimators obtained by choosing the solution space $S(\Omega_e)$ so that the bilinear form $\int_{\Omega_e} \mathbf{C} \nabla_s \boldsymbol{v} : \nabla \boldsymbol{\eta} \, \mathrm{d}\Omega$ is coercive. For instance, the second and third version of the error estimates introduced by Bank & Weiser (1985) are of the latter type, whereas the first version is of the former type, which will be described later on.

In the elemental error estimates, the functional $\mathcal{F}_{\Omega_{e}}(\eta)$ is given by

$$\begin{aligned} \mathcal{F}_{\Omega_{e}}(\boldsymbol{\eta}) &= \int_{\Omega_{e}} \boldsymbol{r}_{\boldsymbol{u}_{h}}(\boldsymbol{\eta}) \, \mathrm{d}\Omega + \oint_{\partial\Omega_{e}} \boldsymbol{J}_{\boldsymbol{u}_{h},av}^{\gamma} \cdot \boldsymbol{\eta} \, \mathrm{d}s = \\ &= \int_{\Omega_{e}} \boldsymbol{b} \cdot \boldsymbol{\eta} \, \mathrm{d}\Omega + \int_{\partial\Omega_{e} \cap \partial\Omega_{t}} \boldsymbol{t} \cdot \boldsymbol{\eta} \, \mathrm{d}\Omega - \int_{\Omega_{e}} \mathbf{C} \nabla_{s} \boldsymbol{u}_{h} \colon \nabla \boldsymbol{\eta} \, \mathrm{d}\Omega + \oint_{\partial\Omega_{e}} \boldsymbol{J}_{\boldsymbol{u}_{h},av}^{\gamma} \cdot \boldsymbol{\eta} \, \mathrm{d}s, \end{aligned}$$

where $J_{\boldsymbol{u}_h,av}^{\gamma}$ is obtained by averaging the jump $J_{\boldsymbol{u}_h}^{\gamma}$ between the elements sharing the edge $\gamma \in \mathcal{E}_{h,\Omega}$ as follows

$$\boldsymbol{J}_{\boldsymbol{u}_h,av}^{\gamma} = \frac{1}{2} \Big[\boldsymbol{\mathsf{C}} \nabla_{\boldsymbol{s}} \boldsymbol{u}_h^{\Omega_{e,in}} \colon \boldsymbol{n} + \boldsymbol{\mathsf{C}} \nabla_{\boldsymbol{s}} \boldsymbol{u}_h^{\Omega_{l,out}} \colon \boldsymbol{n} \Big].$$

In this case, special consideration must be given to the choice of the finite element space $S(\Omega_e)$ in order to guarantee the solvability of the local problems (2.14) and to produce useful error estimators.

In the case of linear finite element approximations of scalar elliptic equations, the space $S(\Omega_e)$ used by Bank & Weiser (1985) to define the second and third version of their error estimates, is the space of the so-called bubble functions, that is, quadratic functions defined over Ω_e and vanishing at its vertices. This space has been then augmented by cubic bubble functions by Verfurth (1989) in the definition of an error estimator for linear finite element approximations of Stokes equations. Guidelines for choosing the space $S(\Omega_e)$ are well established in the case of first order finite element approximations and are discussed in Oden *et al.* (1989). In the case of higher order finite element approximations, the selection of $S(\Omega_e)$ is not an easy matter. In general, the criterion is the same as the one underlying the error estimates based on hierarchical bases (Bank & Smith, 1993): to increase the order of the space used to construct the original finite element approximation and then form the quotient space by subtracting the original finite element space. The influence of the choice of the different spaces on the solution of the local problem has been, however, investigated by Ainsworth (1996). A quite unsatisfactory state of affair has been shown, due to the sensitivity of the estimate to the choice of $S(\Omega_e)$. In some cases the estimator is a gross overestimate, yet in others the estimated error is zero, despite the true error being nonzero. An alternative possibility is given, therefore, by the equilibrated element residual error estimates.

Likewise the estimators described previously, the equilibrated element residual error estimates are obtained by solving local Neumann problems. In this case, the well-posedness of the local problems is achieved by imposing the consistency of the boundary data. The idea is to consider an equilibrated splitting of the interelement flux $J_{u_h}^{\gamma}$ such that

$$\begin{aligned} J_{\boldsymbol{u}_{h}}^{\gamma} &= J_{\boldsymbol{u}_{h}}^{\gamma,\Omega_{e,in}} + J_{\boldsymbol{u}_{h}}^{\gamma,\Omega_{l,out}} \\ \int_{\Omega_{e,in}} \boldsymbol{r}_{\boldsymbol{u}_{h}} \, \mathrm{d}\Omega + \oint_{\partial\Omega_{e,in}} J_{\boldsymbol{u}_{h}}^{\gamma,\Omega_{e,in}} \, \mathrm{d}s = \mathbf{0}, \end{aligned}$$

$$(2.15)$$

and the functional $\mathcal{F}_{\Omega_e}(\boldsymbol{\eta})$ given by

$$\mathcal{F}_{\Omega_{e,in}}(\boldsymbol{\eta}) = \int_{\Omega_{e,in}} \boldsymbol{r}_{\boldsymbol{u}_h} \cdot \boldsymbol{\eta} \, \mathrm{d}\Omega + + \oint_{\partial \Omega_{e,in}} \boldsymbol{J}_{\boldsymbol{u}_h}^{\boldsymbol{\gamma},\Omega_{e,in}} \cdot \boldsymbol{\eta} \, \mathrm{d}s.$$

The second condition in (2.15) is the consistency condition on the data of the following local Neumann problem

div
$$\mathbf{C} \nabla_s \boldsymbol{e}_{\Omega_e} = \boldsymbol{r}_{\boldsymbol{u}_h}$$
 in Ω_e
 $\mathbf{C} \nabla_s \boldsymbol{e}_{\Omega_e} : \boldsymbol{n} = \boldsymbol{J}_{\boldsymbol{u}_h}^{\gamma,\Omega_e}, \text{ on } \partial \Omega_e.$

which guarantees its well-posedness in $[H^1(\Omega_e)]^3$.

Different types of equilibrated element residual error estimates have been given in literature. These include the techniques proposed by Ladevèze & Leguillon (1983), Kelly (1984), the first version of the error estimate of Bank & Weiser (1985) and the error estimate given in Ainsworth & Oden (1992), among others. These are differentiated between each other, basically, by the assumption on the splitting of the residual jump across the element boundaries.

A unifying theoretical framework for the equilibrated element residual error estimators has been developed by Ainsworth & Oden (1993). The gist of their analysis is a localization of the primal-hybrid variational formulation (Raviart &

Thomas, 1977) of the problem (2.5) that characterizes the discretization error. The formulation is posed on the so-called broken Sobolev space $\mathcal{V}_0(\mathcal{T}_h)^{\dagger\dagger}$ associated with \mathcal{T}_h and it is obtained by relaxing the interelement continuity with the expense of introducing Lagrangian multipliers $\mu = \mu(v) \in \mathcal{M} \subset \mathcal{V}_0^*(\mathcal{T}_h)$. The latter are the linear and continuous functionals defined over $\mathcal{V}_0(\mathcal{T}_h)$ and vanishing over $\mathcal{V}_0 \subset \mathcal{V}_0(\mathcal{T}_h)$. The elements of \mathcal{M} permit the characterization of the interelement continuity of elements $v \in \mathcal{V}_0(\mathcal{T}_h)$. In this way, one can solve local problems which preserve the type of bound.

The main result is

$$-\frac{1}{2}|||\boldsymbol{e}|||^2 = \inf_{\boldsymbol{v}\in\mathcal{V}_0(\mathcal{T}_h)} \sup_{\mu\in\mathcal{M}} \mathcal{L}(\boldsymbol{v},\mu) = \sup_{\mu\in\mathcal{M}} \inf_{\boldsymbol{v}\in\mathcal{V}_0(\mathcal{T}_h)} \mathcal{L}(\boldsymbol{v},\mu) \geq \inf_{\boldsymbol{v}\in\mathcal{V}_0(\mathcal{T}_h)} \mathcal{L}(\boldsymbol{v},\mu) \quad \forall \mu\in\mathcal{M},$$

where $\mathcal{L}(\boldsymbol{v},\mu)$ is the Lagrangian functional defined as follows

$$\mathcal{L}(oldsymbol{v},oldsymbol{\mu}) = \sum_{\Omega_{oldsymbol{e}}\in\mathcal{T}_{oldsymbol{h}}} \mathcal{J}_{\Omega_{oldsymbol{e}}}(oldsymbol{v}) - \mu^{\star} + \mu$$

with

$$\mathcal{J}_{\Omega_{e}}(\boldsymbol{v}) = \frac{1}{2} \int_{\Omega_{e}} \mathbf{C} \nabla_{s} \boldsymbol{v} : \nabla \boldsymbol{v} \, \mathrm{d}\Omega - \int_{\Omega_{e}} \boldsymbol{b} \cdot \boldsymbol{v} \, \mathrm{d}\Omega - \int_{\partial\Omega_{e} \cap \Omega_{t}} \boldsymbol{t} \cdot \boldsymbol{v} \, \mathrm{d}s + \\ + \int_{\Omega_{e}} \mathbf{C} \nabla_{s} \boldsymbol{u}_{h} : \nabla \boldsymbol{v} \, \mathrm{d}\Omega + \oint_{\partial\Omega_{e}} \boldsymbol{g}_{\partial\Omega_{e}} \cdot \boldsymbol{v} \, \mathrm{d}s$$

and

$$\mu^{\star}(\boldsymbol{v}) = \sum_{\gamma \in \mathcal{E}_{h,\Omega}} \int_{\gamma} \boldsymbol{g}_{\gamma} \cdot [\boldsymbol{v}]_{\gamma} \, \mathrm{d}s.$$
(2.16)

In equation (2.16), g_{γ} is a smooth vector field associated with each $\gamma \in \mathcal{E}_{h,\Omega}$ and $[\boldsymbol{v}]_{\gamma}$ is the jump of $\boldsymbol{v} \in \mathcal{V}_0(\mathcal{T}_h)$ across γ (see note $\dagger \dagger$). The particular choice of \boldsymbol{g}_{γ} determines the error estimation method. Once \boldsymbol{g}_{γ} has been chosen, the vector field $\boldsymbol{g}_{\partial\Omega_e}$ is defined on $\partial\Omega_e$ for each element $\Omega_e \in \mathcal{T}_h$ such that

$$\sum_{\Omega_e \in \mathcal{T}_h} \oint_{\partial \Omega_e} \boldsymbol{g}_{\partial \Omega_e} \cdot \boldsymbol{v} \, \mathrm{d}s = \sum_{\gamma \in \mathcal{E}_{h,\Omega}} \int_{\gamma} \boldsymbol{g}_{\gamma} \cdot [\boldsymbol{v}]_{\gamma} \, \mathrm{d}s.$$

By choosing $\mu = \mu^*$, one obtains

$$|||\boldsymbol{e}|||^{2} \leq -2 \sum_{\Omega_{e} \in \mathcal{T}_{h}} \inf_{\boldsymbol{v} \in \mathcal{V}_{0}(\Omega_{e})} \mathcal{J}_{\Omega_{e}}(\boldsymbol{v})$$

$$(2.17)$$

^{††}The broken Sobolev space $\mathcal{V}_0(\mathcal{T}_h)$ is the space of the functions \boldsymbol{v} of class $[\mathrm{H}^1(\Omega_e)]^3$ over each element $\Omega_e \in \mathcal{T}_h$ which meet homogeneous essential boundary conditions on $\partial\Omega_e \cap \partial\Omega_d$. As a result, an element $\boldsymbol{v} \in \mathcal{V}_0(\mathcal{T}_h)$ may be discontinuous across $\gamma \in \mathcal{E}_{h,\Omega}$. We define $[\boldsymbol{v}]_{\gamma} = \lim_{\alpha \downarrow 0} \boldsymbol{v}(\boldsymbol{x} + \alpha)$

$$\alpha n) - \lim_{\alpha \downarrow 0} \boldsymbol{v}(\boldsymbol{x} - \alpha n). \text{ If we let } \mathcal{V}_0(\Omega_e) = \left\{ \boldsymbol{v} \in [\mathrm{H}^1(\Omega_e)]^3 \middle| \boldsymbol{v} = \boldsymbol{0} \text{ on } \partial\Omega_e \cap \partial\Omega_d \right\}, \text{ it is } \mathcal{V}_0(\mathcal{T}_h) = \prod_{\Omega_e \in \mathcal{T}_h} \mathcal{V}_0(\Omega_e).$$

where $\mathcal{V}_0(\Omega_e)$ is the restriction of $\mathcal{V}_0(\mathcal{T}_h)$ to Ω_e (see note $\dagger \dagger$).

The inequality (2.17) gives the link of the error estimate η to the solution of the following elemental primal problems

Find
$$\boldsymbol{w} \in \mathcal{V}_0(\Omega_e)$$

 $\mathcal{J}_{\Omega_e}(\boldsymbol{w}) = \inf_{\boldsymbol{v} \in \mathcal{V}_0(\Omega_e)} \mathcal{J}_{\Omega_e}(\boldsymbol{v}).$ (2.18)

The complementary principle applied to the solution of (2.2.1.2) delivers

Find
$$\boldsymbol{p} \in \mathcal{W}_{\Omega_{\boldsymbol{e}}}$$

$$\mathcal{G}_{\Omega_{\boldsymbol{e}}}(\boldsymbol{p}) = \sup_{\boldsymbol{q} \in \mathcal{W}_{\Omega_{\boldsymbol{e}}}} \underbrace{-\frac{1}{2} \int_{\Omega_{\boldsymbol{e}}} \boldsymbol{q} : \boldsymbol{q} \, \mathrm{d}\Omega}_{\mathcal{G}_{\Omega_{\boldsymbol{e}}}(\boldsymbol{q})},$$

where \mathcal{W}_{Ω_e} is the set of the stress tensors q solution of the following problem over Ω_e

$$\operatorname{div} \boldsymbol{q} = \boldsymbol{r}_{\boldsymbol{u}_h} \quad \text{in } \Omega_e$$
$$\boldsymbol{q} \colon \boldsymbol{n} = \boldsymbol{g}_{\Omega_e}, \quad \text{on } \partial \Omega_e$$

This set is not empty if the following condition is satisfied

$$\int_{\Omega_e} \boldsymbol{r}_{\boldsymbol{u}_h} \,\mathrm{d}\Omega + \oint_{\partial\Omega_e} \boldsymbol{g}_{\Omega_e} \,\mathrm{d}s = \boldsymbol{0}, \qquad (2.19)$$

which is the equilibration condition. Since it is

$$\inf_{\boldsymbol{v}\in\mathcal{V}_0(\Omega_e)}\mathcal{J}_{\Omega_e}(\boldsymbol{v})=\sup_{\boldsymbol{q}\in\mathcal{W}_{\Omega_e}}\mathcal{G}_{\Omega_e}(\boldsymbol{q})$$

then it follows

$$|||e|||^2 \leq -2\sum_{\Omega_e\in\mathcal{T}_h} \sup_{oldsymbol{q}\in\mathcal{W}_{\Omega_e}}\mathcal{G}_{\Omega_e}(oldsymbol{q}) \leq -2\sum_{ig||e|||\atop b_e\in\bigcup_n}\mathcal{G}_{\Omega_e}(oldsymbol{q}) \quad orall oldsymbol{q}\in\mathcal{W}_{\Omega_e},$$

that is, a concrete realization of an upper bound for |||e||| depends on the definition of $g_{\partial\Omega_e}$ and on the choice of $q \in \mathcal{W}_{\Omega_e}$.

The equilibration of the data (2.19), which is necessary for the model problem under consideration, is desirable to realize also when low order terms are present in the elliptic operator. In this case as the mesh size $h \rightarrow 0$, these terms can become preponderant and make the energy of the local solution blowing up. By imposing the equilibration of the data also in this case, the error estimator becomes finite (Ainsworth & Oden, 2000).

The error estimator introduced in Ladevèze & Leguillon (1983) can be casted into the previous framework by choosing $g_{\partial\Omega_e}$ to be $J^{\gamma}_{u_h,av}$ plus a suitable piecewise linear vector field on $\partial\Omega_e$ (Verfurth, 1999). However, this error estimate has been obtained by starting from other considerations which lead to the class of errors in $\inf_{\boldsymbol{v}\in\mathcal{V}_0(\Omega_e)}\mathcal{J}_{\Omega_e}(\boldsymbol{v}) = \sup_{\boldsymbol{q}\in\mathcal{W}_0}$

the constitutive equations and they will be considered next. The previous analysis provides also theoretical support to the heuristic error estimate introduced by Kelly (1984) consisting in the solution of local complementary problems.

Finally, it is worth mentioning that Paraschivoiu *et al.* (1997) have developed an extension of this theory to the estimates of output of interest. Besides the relaxation of the interelement continuity, an additional constraint is introduced represented by the equilibrium equations over the broken Sobolev space, so that the admissible set is constituted by the only solution of the problem (Patera & Peraire, 2001). The value of this generalization lies in the application to problems which can be expressed in terms of minimization of a convex functional. An instance of such extension to a hyperelastic model has been given in Bonet *et al.* (2002).

The error in the constitutive equation. The Prager-Synge theorem

The notion of error in the constitutive equations has been introduced for the first time by Ladevèze in 1975 (Ladevèze, 1975; Ladèveze, 1995) by exploiting the convex functional structure of the constitutive equations.

For linear elastic problems, this notion can be grasped quite easily. Let $\sigma_{ex} = \sigma_{ex}(x)$ and $u_{ex} = u_{ex}(x)$ be the exact stress field and the exact displacement field, respectively. This means that σ_{ex} does satisfy the equilibrium equations, u_{ex} is a kinematically admissible displacement field, that is, u_{ex} meets the internal and external compatibility conditions and finally, u_{ex} and σ_{ex} are related to each other by the constitutive equation

$$\boldsymbol{\sigma}_{ex} - \mathbf{C} \nabla_s \boldsymbol{u}_{ex} = \mathbf{0}. \tag{2.20}$$

Consider now a kinematically admissible displacement field $u_{ad} = u_{ad}(x)$. The energy norm of the error associated with $u_{ad} = u_{ad}(x)$ is defined as

$$|||\boldsymbol{u}_{ad} - \boldsymbol{u}_{ex}||| = \left(\int_{\Omega} [\boldsymbol{\sigma}_{ex} - \mathbf{C}\nabla_{s}\boldsymbol{u}_{ad}] : \mathbf{C}^{-1}[\boldsymbol{\sigma}_{ex} - \mathbf{C}\nabla_{s}\boldsymbol{u}_{ad}] \,\mathrm{d}\Omega\right)^{\frac{1}{2}}$$
(2.21)

Upper bound for (2.21) is obtained as

$$\eta(\boldsymbol{u}_{ad},\boldsymbol{\sigma}_{ad}) = \left(\int_{\Omega} [\boldsymbol{\sigma}_{ad} - \boldsymbol{\mathsf{C}} \nabla_{s} \boldsymbol{u}_{ad}] : \boldsymbol{\mathsf{C}}^{-1}[\boldsymbol{\sigma}_{ad} - \boldsymbol{\mathsf{C}} \nabla_{s} \boldsymbol{u}_{ad}] \,\mathrm{d}\Omega\right)^{\frac{1}{2}}$$
(2.22)

where $\sigma_{ad} = \sigma_{ad}(x)$ is any statically admissible stress field, that is, a stress tensor field that satisfies the equilibrium equation. As a result, σ_{ad} meets also the condition

$$\int_{\Omega} (\boldsymbol{\sigma}_{ex} - \boldsymbol{\sigma}_{ad}) \colon (\nabla_s \boldsymbol{u}_{ex} - \nabla_s \boldsymbol{u}_{ad}) \mathrm{d}\Omega = 0, \qquad (2.23)$$

which follows, by standard arguments, from the principle of the virtual work.

Equation (2.22) is a measure of the extent to which the pair (σ_{ad}, u_{ad}) fail to satisfy the constitutive equation (2.20) and is obtained by reformulating equation

(2.20) into an equivalent form which uses the convex free elastic potential and its Legendre transform, given by the complementary potential.

The validity of the bound

$$\begin{vmatrix} \text{Given } \boldsymbol{u}_{ad} \\ |||\boldsymbol{e}_{ex}||| \leq \eta(\boldsymbol{u}_{ad}, \boldsymbol{\sigma}_{ad}), \quad \forall \boldsymbol{\sigma}_{ad} \end{vmatrix}$$
(2.24)

is a simple consequence of the Prager–Synge's theorem (Prager & Synge, 1947) which states the orthogonality between the fields $\sigma_{ex} - \mathbf{C} \nabla_s u_{ad}$ and $\sigma_{ad} - \sigma_{ex}$. This reads as

$$\int_{\Omega} [\boldsymbol{\sigma}_{ad} - \mathbf{C} \nabla_{s} \boldsymbol{u}_{ad}] : \mathbf{C}^{-1} [\boldsymbol{\sigma}_{ad} - \mathbf{C} \nabla_{s} \boldsymbol{u}_{ad}] \, \mathrm{d}\Omega =$$

$$= \int_{\Omega} [\boldsymbol{\sigma}_{ex} - \mathbf{C} \nabla_{s} \boldsymbol{u}_{ad}] : \mathbf{C}^{-1} [\boldsymbol{\sigma}_{ex} - \mathbf{C} \nabla_{s} \boldsymbol{u}_{ad}] \, \mathrm{d}\Omega +$$

$$+ \int_{\Omega} [\boldsymbol{\sigma}_{ad} - \mathbf{C} \nabla_{s} \boldsymbol{u}_{ex}] : \mathbf{C}^{-1} [\boldsymbol{\sigma}_{ad} - \mathbf{C} \nabla_{s} \boldsymbol{u}_{ex}] \, \mathrm{d}\Omega.$$
(2.25)

Proof. In the case of linear elasticity, by accounting for the equivalence

$$\boldsymbol{\sigma}_{ex} - \mathbf{C}(\nabla_s \boldsymbol{u}_{ex}) = \mathbf{0} \Leftrightarrow \frac{1}{2} \boldsymbol{\sigma}_{ex} \colon \mathbf{C}^{-1} \boldsymbol{\sigma}_{ex} + \frac{1}{2} \nabla_s \boldsymbol{u}_{ex} \colon \mathbf{C} \nabla_s \boldsymbol{u}_{ex} - \boldsymbol{\sigma}_{ex} \colon \nabla_s \boldsymbol{u}_{ex} = 0$$

it is an easy matter to show the validity of the following equality

$$\frac{1}{2}\boldsymbol{\sigma}_{ad}: \mathbf{C}^{-1}\boldsymbol{\sigma}_{ad} + \frac{1}{2}\nabla_{s}\boldsymbol{u}_{ad}: \mathbf{C}\nabla_{s}\boldsymbol{u}_{ad} - \boldsymbol{\sigma}_{ad}: \nabla_{s}\boldsymbol{u}_{ad} = \\
= \frac{1}{2}\boldsymbol{\sigma}_{ad}: \mathbf{C}^{-1}\boldsymbol{\sigma}_{ad} + \frac{1}{2}\nabla_{s}\boldsymbol{u}_{ex}: \mathbf{C}\nabla_{s}\boldsymbol{u}_{ex} - \boldsymbol{\sigma}_{ad}: \nabla_{s}\boldsymbol{u}_{ex} + \\
+ \frac{1}{2}\boldsymbol{\sigma}_{ex}: \mathbf{C}^{-1}\boldsymbol{\sigma}_{ex} + \frac{1}{2}\nabla_{s}\boldsymbol{u}_{ad}: \mathbf{C}\nabla_{s}\boldsymbol{u}_{ad} - \boldsymbol{\sigma}_{ex}: \nabla_{s}\boldsymbol{u}_{ad} + \\
- (\boldsymbol{\sigma}_{ex} - \boldsymbol{\sigma}_{ad}): (\nabla_{s}\boldsymbol{u}_{ex} - \nabla_{s}\boldsymbol{u}_{ad}), \quad (2.26)$$

so that by integrating both sides of equation (2.26) and accounting for (2.23), equation (2.25) follows.

In the case of conforming finite element displacement approximations, one assumes $u_{ad} = u_h$ so that the actual realization of an upper bound η for the energy norm of the error $|||e_{ex}|||$ resolves in the definition of a statically admissible stress field σ_{ad} . However, for the efficiency of η the definition of σ_{ad} must be linked with the finite element solution $\sigma^h = \mathbb{C}\nabla_s u_h$. This is realized with the so-called prolongation condition introduced by Ladevèze & Leguillon (1983) which is a localization at each element of the Galerkin orthogonality (2.9) which holds for the global residual \mathcal{R}_{u_h} . This condition distinguishes the statically admissible stress fields σ_{ad} which satisfy the following equation for every shape function \mathbb{N}_i and for all the elements Ω_e^h ,

$$\int_{\Omega_{\epsilon}} \left(\boldsymbol{\sigma}_{ad} - \mathbf{C} \nabla_{s} \boldsymbol{u}_{h} \right) : \nabla \mathbb{N}_{i} \, \mathrm{d}\Omega = 0.$$

This condition, finally, corresponds to making an assumption on the splitting of the residual jump $J_{u_h}^{\gamma}$ across the interelement boundaries and σ_{ad} is obtained as solution of the following local problem stated for each element Ω_e

$$\operatorname{div} \boldsymbol{\sigma}_{ad} + \boldsymbol{b} = \boldsymbol{0} \ \boldsymbol{\sigma}_{ad} \colon \boldsymbol{n} = \boldsymbol{J}_{\boldsymbol{u}_h}^{\gamma,\Omega_e},$$

where $J_{\boldsymbol{u}_h}^{\gamma,\Omega_e}$ is the part of the jump $J_{\boldsymbol{u}_h}^{\gamma}$ across $\gamma \in \partial \Omega_e$ which is assigned to Ω_e .

Error indicators are obtained simply by the localization of the integral (2.22) as

$$\eta(\boldsymbol{u}_{ad}, \boldsymbol{\sigma}_{ad}) = \left(\sum_{\Omega_e \in \mathcal{T}_h} \eta_e^2\right)^{rac{1}{2}}$$

where

$$\eta_e^2 = \int_{\Omega_e} [\boldsymbol{\sigma}_{ad} - \mathbf{C} \nabla_s \boldsymbol{u}_{ad}] \colon \mathbf{C}^{-1} [\boldsymbol{\sigma}_{ad} - \mathbf{C} \nabla_s \boldsymbol{u}_{ad}] \,\mathrm{d}\Omega$$

The proof of (2.24) has large validity so that the error in the constitutive equations has also been applied to 2D and 3D elasticity by Ladevèze *et al.* (1991) and Coorevits *et al.* (1998), respectively; incompressible elasticity by Gastine *et al.* (1992) and to anisotropic meshes in Ladèveze (1994) and Ladevèze & Rougeot (1997). In each of these problems, the crunch of the estimation technique was always the definition of the equilibrating element tractions recovered by the finite element solution. A general procedure for such construction in the case of 2D finite element models has been developed in Ladevèze & Maunder (1996).

As we have mentioned earlier, local equilibrium problems with repartition of the residual jump have been proposed on heuristic basis also by Kelly (1984). It can be noted that the repartition that Kelly assumes in 1D corresponds to the prolongation condition introduced in Ladevèze & Leguillon (1983).

The concept of using two approximate solutions to build estimates of the error had been put forward also by Synge (1957) in establishing the hypercicle method. By using two approximate solutions located in spaces intersecting at the exact solution Synge builds estimates of solutions of the torsion problem.

We conclude this section by mentioning the work of Fraeijs de Veubeke (1965) that can be cast within the previous framework. Fraeijs de Veubeke provides estimates of the energy norm of the exact solution $|||\boldsymbol{u}_{ex}|||$ starting from the two-sided bounds for the exact free elastic energy in terms of the total elastic energy and the total complementary elastic energy. Let

$$\mathcal{J}(\boldsymbol{u}_{ad}) = \frac{1}{2} \int_{\Omega} \mathbf{C} \nabla_{s} \boldsymbol{u}_{ad} : \nabla \boldsymbol{u}_{ad} \, \mathrm{d}\Omega - \int_{\Omega} \boldsymbol{b} \cdot \boldsymbol{u}_{ad} \, \mathrm{d}\Omega - \int_{\partial \Omega_{t}} \boldsymbol{t} \cdot \boldsymbol{u}_{ad} \, \mathrm{d}s$$

be the total elastic energy defined over the affine space of the kinematically admissible displacement fields and

$$\mathcal{G}(\boldsymbol{\sigma}_{ad}) = -\frac{1}{2} \int_{\Omega} \boldsymbol{\sigma}_{ad} \colon \mathbf{C}^{-1} \boldsymbol{\sigma}_{ad} \,\mathrm{d}\Omega$$

the total complementary elastic energy defined over the affine space of the statically admissible stress fields. It is (Mikhlin, 1964)

$$orall oldsymbol{u}_{ad}, \quad \mathcal{J}(oldsymbol{u}_{ad}) \geq \underbrace{\mathcal{J}(oldsymbol{u}_{ex}) = \mathcal{G}(oldsymbol{\sigma}_{ex})}_{-rac{1}{2}|||oldsymbol{u}_{ex}||^2} \geq \mathcal{G}(oldsymbol{\sigma}_{ad}), \quad orall oldsymbol{\sigma}_{ad}.$$

Thus, estimates of the energy norm of the exact solution $|||u_{ex}|||$ were obtained as

$$\eta(\boldsymbol{u}_{ad}, \boldsymbol{\sigma}_{ad}) = \sqrt{-2\mathcal{G}(\boldsymbol{\sigma}_{ad}) + 2\mathcal{J}(\boldsymbol{u}_{ad})}$$

This procedure, however, failed to gain popularity being based on the global solution of the dual finite element method for the model under consideration, and also because the estimate cannot be expressed in terms of the contributions from each element, necessary for the optimization of the finite element meshes.

2.2.2 Recovery based error estimators

The recovery based error estimators represent certainly the class of error estimates that has met a big success in the engineering community for its relatively simple implementation. They were first introduced by Zienkiewicz & Zhu (1987) and since then many error estimators have been developed which employ the main idea. This relies on the following fact. The energy norm of the error, given by equation (2.3), can also be re-written as follows

$$|||\boldsymbol{e}||| = \left(\int_{\Omega} \left(\boldsymbol{\sigma}_{ex} - \mathbf{C}\nabla_{s}\boldsymbol{u}_{h}\right) : \mathbf{C}^{-1}\left(\boldsymbol{\sigma}_{ex} - \mathbf{C}\nabla_{s}\boldsymbol{u}_{h}\right) \mathrm{d}\Omega\right)^{\frac{1}{2}}, \qquad (2.27)$$

where the exact stress field σ_{ex} is unknown. Therefore, estimates to |||e||| are obtained by assuming in place of σ_{ex} in (2.27) approximations σ^* recovered by suitable postprocessing of the finite element solution $\sigma^h = \mathbb{C}\nabla_s u_h$, that is

$$|||\boldsymbol{e}||| \approx \eta = \left(\int_{\Omega} \left(\boldsymbol{\sigma}^{\star} - \mathbf{C}\nabla_{s}\boldsymbol{u}_{h}\right) : \mathbf{C}^{-1}\left(\boldsymbol{\sigma}^{\star} - \mathbf{C}\nabla_{s}\boldsymbol{u}_{h}\right) \mathrm{d}\Omega\right)^{\frac{1}{2}}.$$

The quality and reliability of this type of error estimator is however dependent on the accuracy of the recovered solution. In general, it can be said that if σ^* is such that

$$\int_{\Omega} \left(\boldsymbol{\sigma}_{ex} - \boldsymbol{\sigma}^{\star} \right) \colon \mathbf{C}^{-1} \left(\boldsymbol{\sigma}_{ex} - \boldsymbol{\sigma}^{\star} \right) \mathrm{d}\Omega \ll \int_{\Omega} \left(\boldsymbol{\sigma}^{\star} - \mathbf{C} \nabla_{s} \boldsymbol{u}_{h} \right) \colon \mathbf{C}^{-1} \left(\boldsymbol{\sigma}^{\star} - \mathbf{C} \nabla_{s} \boldsymbol{u}_{h} \right) \mathrm{d}\Omega$$

then

$$\int_{\Omega} \left(\boldsymbol{\sigma}^{\star} - \mathbf{C} \nabla_{s} \boldsymbol{u}_{h} \right) : \mathbf{C}^{-1} \left(\boldsymbol{\sigma}^{\star} - \mathbf{C} \nabla_{s} \boldsymbol{u}_{h} \right) \mathrm{d}\Omega \approx \int_{\Omega} \left(\boldsymbol{\sigma}_{ex} - \mathbf{C} \nabla_{s} \boldsymbol{u}_{h} \right) : \mathbf{C}^{-1} \left(\boldsymbol{\sigma}_{ex} - \mathbf{C} \nabla_{s} \boldsymbol{u}_{h} \right) \mathrm{d}\Omega.$$

and one can define

$$\eta = \left(\int_{\Omega} \left(\boldsymbol{\sigma}^{\star} - \mathbf{C} \nabla_{s} \boldsymbol{u}_{h}\right) : \mathbf{C}^{-1} \left(\boldsymbol{\sigma}^{\star} - \mathbf{C} \nabla_{s} \boldsymbol{u}_{h}\right) d\Omega\right)^{\frac{1}{2}}$$

as an *a posteriori* error estimator (with respect to the energy norm).

The procedures to build σ^* are, generally, referred to as the stress recovery or derivative recovery techniques. The definition of these methods finds their motivation in the observation that, under some conditions on the domain, mesh and regularity of the solution, there exist certain points of the domain where the derivatives of the finite element solution, $\mathbf{C}\nabla_s \boldsymbol{u}_h$, which are usually one order lower than that of the finite element solution itself \boldsymbol{u}_h , have superior accuracy (Barlow, 1976). This phenomenon is known as superconvergence. If superconvergent derivatives can be recovered by a particular post-processing method, an asymptotically exact error estimator is then obtained (Ainsworth & Oden, 2000).

The recovery technique given initially in Zienkiewicz & Zhu (1987) assumes σ^* interpolated by the same functions as the displacements, i.e.

$$\sigma^{\star} = \mathbb{N}\bar{\sigma}^{\star} \tag{2.28}$$

where $\bar{\sigma}^*$ are the nodal values of the continuous field σ^* . These are obtained by imposing that $\sigma^* - \mathbf{C} \nabla_s u_h$ is orthogonal to the space described by the shape functions \mathbb{N} , that is,

$$\int_{\Omega} \left(\boldsymbol{\sigma}^{\star} - \mathbf{C} \nabla_{s} \boldsymbol{u}_{h} \right) \colon \mathbb{N} \, \mathrm{d}\Omega = 0.$$

The Zienkiewicz–Zhu (Z^2) error estimator, whose corresponding error indicators are obtained simply by localization of the integral, was analysed in Ainsworth et al. (1989). It was found that while the estimator performs quite well for linear triangular and quadratic quadrilateral elements, it is not necessarily asymptotically exact. This property is shown to hold in the case of smooth solutions and parallel meshes by Babuska & Rodriguez (1993) and in Verfurth (1996) who refer to the analysis carried out by Rodriguez (1994). For other types of elements, the Z^{2-} method was often found to behave poorly with the effectivity index converging to zero in some cases. In others, the error indicators were misleading in steering the discretization process (Strouboulis & Haque, 1992). For this reason, Zhu & Zienkiewicz (1990) introduced first for one dimensional problems a new stress recovery procedure, termed as superconvergent patch recovery, by means of which superconvergent derivatives of the finite element solution are determined everywhere in the domain. The recovery procedure was then developed for 2D problems in Zienkiewicz & Zhu (1992a) and applied to error estimation in Zienkiewicz & Zhu (1992b). The continuous stress field σ^{\star} is, as usual, assumed to be given by equation (2.28). The nodal values $\bar{\sigma}^{\star}$ are obtained by considering a continuous polynomial expansion on an element patch surrounding the nodes where the recovery is desired. This expansion is made to fit locally the superconvergent points, called also sampling points, in a least-squares manner or simply be an L^2 -projection of the finite element derivatives. For the least-squares fitting, the superconvergent recovery is observed by the numerical test; for the local L^2 -projection fitting, a considerable improvement for the nodal values is achieved.

Improvements of the method were contributed by Wiberg & Abdulwahab (1993) that include the governing equilibrium equation on the recovered derivatives.

As in Zienkiewicz & Zhu (1992a), these are assumed to be interpolated by the same shape functions as the finite element displacement field, that is, σ^* is assumed of the form (2.28). The nodal values of the recovered stress field are also here obtained by assuming for the stresses a polynomial expansion over the patch of elements around the given node. The coefficients of this expansion are then computed by minimizing, in a least-square sense, the residual of the stresses at the superconvergent points and the weighted residual in the equilibrium equation over the local patch of elements. This recovery techniques was successively improved by Wiberg *et al.* (1994) for the recovery of derivatives near the boundaries where either tractions or displacements are prescribed. This was obtained by including seemingly a weighted residual error at the boundary points in the patch recovery and a pronounced improvement in the post processed gradients of the finite element solution was finally observed.

A complete analysis of the several recovery based error estimators in terms of the operator that defines the improved stress as function of the consistent derivatives of the finite element solution can be found in Ainsworth & Oden (2000). Carstensen & Funken (2000) analyze, on the other hand, their robustness with respect to violated (local) symmetry of meshes or superconvergence and with respect to incompressible locking.

These error estimators, however, are justified to varying extent by superconvergence properties which are known to hold only in special cases. In Babuska & Strouboulis (2001), therefore, a new definition of superconvergence - the η %-superconvergence - is considered which generalizes the classical idea of superconvergence to general meshes. By means of this property one can choose the best position of the sampling points when properties of superconvergence do not hold.

Remark 2.1. If the smoothed stress field σ^* is chosen as an equilibrated stress field σ_{ad} , for instance, with the criteria given in the previous section, then one retrieves the equilibrated element residual error estimates.

A numerical methodology which determines the quality of a posteriori error estimators has been set up by Babuska et al. (1994b) and Babuska et al. (1994a). The authors observe that the use of general benchmarks to validate a posteriori error estimates can lead to wrong conclusions if they are not properly chosen to isolate the basic factors which influence the performance of the estimator. As a result, an objective and standardized means to assess the robustness of an estimator that exercises all the feature of the particular estimator is given. However, this methodology presents its own limitations. The procedure allows the evaluation of the extreme bounds of the effectivity indices for the estimator when certain effects such as the influence of the singularities, the effect due to the boundary of the domain and mesh grading have been isolated. Moreover, the effectivity indices are those that would be obtained in the asymptotic limit when the mesh size approaches to zero. The preasymptotic behaviour of the estimators might well lead to rather different conclusions concerning the suitability of a particular estimator. Thus, this methodology must be seen not as a means to justify an estimator, but rather as a minimal criterion the estimator must meet. For the details of the procedure and its motivating ideas we refer to the above works and to Ainsworth & Oden (2000).

The main general conclusions of the studies carried out in Babuska *et al.* (1994b) and Babuska *et al.* (1994a) on the quality of estimators for piecewise affine finite approximations on triangular elements can be summarized as follows

- The performance of an estimator depends on the class of meshes, solutions and materials of interest.
- Among the residual estimators tested by the above authors, the implicit element residual estimator with equilibration was the most robust, namely it gives good results for several mesh types, for highly orthotropic materials and arbitrary grid material orientations. In particular, the equilibration proposed in Ladevèze & Leguillon (1983) was recommended.
- The Superconvergence patch recovery error estimator developed in Zienkiewicz & Zhu (1992b) gives good results for the class of smooth solutions approximated on patchwise uniform grids of linear and quadratic elements.
- Asymptotic exactness for an estimator can occur for special uniform grids only and cannot give a measure of quality for the estimator for the general meshes employed in engineering computations.
- The quality of the analysed error estimators tends to deteriorate on anisotropic meshes.

The value of the methodology lies also in the fact that it requires only the solution of small problems in the region of interest; it is inexpensive and it can be used to check the quality of any new estimator even if it is only available as a black-box computer subroutine.

2.3 Nonlinear problems

Unlike the linear problems analysed in the previous section, where a certain maturity has been reached in the comprehension of the mechanisms of propagation of the error, in the case of nonlinear problems, and in particular for those dependent also on the time, the theory of error estimation can be considered still in its infancy. This is reflected in the paucity of studies dedicated to the matter and of originality of the approaches, which usually try to adapt ideas developed for linear problems. Although it is difficult to make a classification of the techniques of estimation, for the nonlinearities are involved in a quite different ways and for the different nature in the approximation of the time and space variable, it may be useful to distinguish the several contributions as follows (Gallimard, 1994):

- Error estimators for problems where the time variable does not appear;
- Error estimators which attempt to estimate also the effects of the time discretization;

- Error in the constitutive equations;
- Methods based on heuristic considerations and direct to the development of error indicators.

For each of this class, the more meaningful works will be outlined, especially in relation to plasticity.

2.3.1 Nonlinear incremental problem

An approach to a theoretically justified *a posteriori* error estimate for the finite element approximation of plasticity problems was given in Johnson & Hansbo (1992). These authors analysed the regularized version of the Hencky problem in small strain perfect plasticity with Von Mises yield criterion given by

$$\begin{vmatrix} \operatorname{div} \boldsymbol{\sigma}_{\mu} = \boldsymbol{b} & \operatorname{in} \Omega \\ \boldsymbol{u}_{\mu} = \boldsymbol{0} & \operatorname{on} \partial\Omega \\ \boldsymbol{C}^{-1} \boldsymbol{\sigma}_{\mu} + \frac{1}{\mu} (\boldsymbol{\sigma}_{\mu} - \boldsymbol{P} \boldsymbol{\sigma}_{\mu}) = \nabla_{s} \boldsymbol{u}_{\mu} & \operatorname{in} \Omega, \end{aligned}$$
(2.29)

where u_{μ} and σ_{μ} denote the solutions of the regularized problem. In equation (2.29) μ represents the regularization parameter, whereas P is the projection operator on the yield surface defined by

$$\boldsymbol{P\tau} \stackrel{\text{def}}{=} \begin{cases} \boldsymbol{\tau} & \text{if } \|\boldsymbol{\tau}^{D}\| \leq \sigma_{y} \\ \frac{\boldsymbol{\tau}^{D}}{\|\boldsymbol{\tau}^{D}\|} \sigma_{y} + \frac{1}{3} \text{Tr}[\boldsymbol{\tau}] \boldsymbol{I} & \text{if } \|\boldsymbol{\tau}^{D}\| > \sigma_{y} \end{cases}$$
(2.30)

where σ_y is a material constant and $\tau^D = \tau - \frac{1}{3} \text{Tr}[\tau] I$ is the deviator tensor of τ . Equation (2.29) describes also the physical model of viscoplasticity which is well-posed in the usual Sobolev spaces (Duvaut & Lions, 1976). For the original model of perfect plasticity, on contrary, the displacements must be sought in the more technical space of the bounded deformations (Temam, 1985) if one requires to guarantee their existence.

Let the complementary energy norm of the error on the stresses be defined as

$$\|\boldsymbol{\sigma}_{\mu}-\boldsymbol{\sigma}_{\mu}^{h}\|_{E}^{2} \stackrel{\text{def}}{=} \int_{\Omega} [\boldsymbol{\sigma}_{\mu}-\boldsymbol{\sigma}_{\mu}^{h}] \colon \mathbf{C}^{-1}[\boldsymbol{\sigma}_{\mu}-\boldsymbol{\sigma}_{\mu}^{h}] \,\mathrm{d}\Omega,$$

where σ^{h}_{μ} denotes the consistent finite element stress tensor obtained from the displacement finite element approximation of (2.29). From the monotony of the operator $I - P^{\dagger}$ it follows

$$\| oldsymbol{\sigma}_{\mu} - oldsymbol{\sigma}_{\mu}^{h} \|_{E}^{2} \leq \mathcal{R}_{oldsymbol{\sigma}_{\mu}^{h}}(oldsymbol{e})$$

[†]Let \mathcal{V} be a Hilbert space. An operator $A: \mathcal{V} \to \mathcal{V}$ is called monotone if $\langle x, Ax \rangle \geq 0 \,\forall x \in \mathcal{V}$. A linear operator P is called a projection if $P^2 = P$. If P is a projection, then $||Px|| \leq ||x|| \,\forall x \in \mathcal{V}$, thus $\langle x, (I-P)x \rangle \geq 0$, i.e. I-P is monotone (Brezis, 1986).

where $e = u_{\mu} - u_{\mu,h}$ and the functional $\mathcal{R}_{\sigma_{\mu}^{h}}(\eta)$ is the residual produced by σ_{μ}^{h} in the equilibrium equations, which is given by

$$\mathcal{R}_{\boldsymbol{\sigma}^{h}_{\mu}}(\boldsymbol{\eta}) = \int_{\Omega} \boldsymbol{b} \cdot \boldsymbol{\eta} \, \mathrm{d}\Omega - \int_{\Omega} \boldsymbol{\sigma}^{h}_{\mu} \colon \nabla \boldsymbol{\eta} \, \mathrm{d}\Omega.$$

Further to a heuristic argument, Johnson & Hansbo (1992) distinguish two contributions into $\mathcal{R}_{\sigma_{\mu}^{h}}$. One contribution comes from the part Ω^{el} of the domain Ω which remains elastic both in the continuous and in the finite element model, whereas the other contribution comes from the complementary part $\Omega^{pl} = \Omega - \Omega^{el}$. As a result, they finally propose the following estimate

$$\|\boldsymbol{\sigma}_{\mu} - \boldsymbol{\sigma}_{\mu}^{h}\|_{E} \leq \left(\sum_{j=1}^{2} \|hC_{j}^{i}R_{j}(\boldsymbol{\sigma}_{\mu}^{h})\|_{L^{2}(\Omega^{el})}^{2} + C^{s}\sum_{j=1}^{2} \|hC_{j}^{i}R_{j}(\boldsymbol{\sigma}_{\mu}^{h})\|_{L^{\infty}(\Omega^{pl})}^{2}\right)^{\frac{1}{2}}$$
(2.31)

where

$$R_{1}(\boldsymbol{\sigma}_{\mu}^{h}) = \boldsymbol{r}_{\boldsymbol{\sigma}_{\mu}^{h}} = \operatorname{div} \boldsymbol{\sigma}_{\mu}^{h} + \boldsymbol{b} \qquad \text{on } \Omega_{e} \in \mathcal{T}_{h}$$

$$R_{2}(\boldsymbol{\sigma}_{\mu}^{h}) = \max_{\boldsymbol{\gamma} \subset \partial \Omega_{e}} \sup_{\boldsymbol{\gamma}} \frac{1}{2} \frac{\|\boldsymbol{J}_{\boldsymbol{\sigma}_{\mu}^{h}}^{\boldsymbol{\gamma}}\|_{L^{2}(\boldsymbol{\gamma})}}{h} \qquad \text{on } \Omega_{e} \in \mathcal{T}_{h}$$

$$\boldsymbol{J}_{\boldsymbol{\sigma}_{\mu}^{h}}^{\boldsymbol{\gamma}} = [\boldsymbol{\sigma}_{\mu}^{h} : \boldsymbol{n}]_{\boldsymbol{\gamma}} \qquad \text{on } \boldsymbol{\gamma} \in \mathcal{E}^{h,\Omega}$$

$$C^{s} = \|\nabla_{s} \boldsymbol{u}_{\mu}\|_{[L^{1}(\Omega)]^{3\times 3}} + \|\nabla_{s} \boldsymbol{u}_{\mu,h}\|_{[L^{1}(\Omega)]^{3\times 3}}$$

and C_j^i are the usual interpolation constants.

The two contributions to the error estimate in (2.31) reflect the type of dependence on the mesh size present in the *a priori* error estimate found by Johnson (1976b) for finite element approximation of this problem. Therein, the estimate has the following structure

$$\|\boldsymbol{\sigma}_{\mu} - \boldsymbol{\sigma}_{\mu}^{h}\|_{E} \leq O(h) + O(\sqrt{h})$$

with the O(h) and $O(\sqrt{h})$ -terms related to Ω^{el} and Ω^{pl} , respectively. However, since the *a priori* error estimate is sub-optimal, one expects the mesh will be more refined in the plastic part Ω^{pl} , where the stresses are suspected to be rather smooth (Fuchs & Seregin, 2000).

Furthermore, the estimate (2.31) is not a full *a posteriori* error estimate, since Ω^{el} and C^s depend on u_{μ} . Therefore, the authors suggest to replace u_{μ} with $u_{\mu,h}$ for the computation of C^s and assume for Ω^{el} only the part of the discrete model which remains elastic. This is quite arbitrary, for it implies that the plastification zone is already correctly captured on the current mesh.

The analysis of the Hencky problem in small strain perfect plasticity with Von Mises yield criterion has been considered also by Rannacher & Suttmeier (1998). As a special case of the control of output of interest, they obtain an *a posteriori* error estimate of the energy norm of the error on the stresses via duality argument applied
to a linearized dual problem. Starting from the non-linear variational equations which describe the regularized version of the Hencky problem and its corresponding finite element approximation, they obtain the non-linear Galerkin orthogonality relation

$$\int_{\Omega} \left[\boldsymbol{P}(\boldsymbol{C}\nabla_{s}\boldsymbol{u}_{\mu}) - \boldsymbol{P}(\boldsymbol{C}\nabla_{s}\boldsymbol{u}_{\mu,h}) \right] : \nabla\boldsymbol{\eta}_{h} \,\mathrm{d}\Omega =$$

$$= \int_{0}^{1} \int_{\Omega} \boldsymbol{P}'(\nabla(s\boldsymbol{u}_{\mu} + (1-s)\boldsymbol{u}_{\mu,h})) \nabla(\boldsymbol{u}_{\mu} - \boldsymbol{u}_{\mu,h}) : \nabla\boldsymbol{\eta}_{h} \,\mathrm{d}\Omega \,\mathrm{d}s = 0 \quad \forall \boldsymbol{\eta}_{h} \in \mathcal{V}_{h}$$
(2.32)

where the mean integral theorem has been invoked and P' is the tangent stiffness matrix sampled at $\nabla(s\boldsymbol{u}_{\mu}+(1-s)\boldsymbol{u}_{\mu,h})$ with $s \in]0,1[$. By computing the linearization (2.32) at $\boldsymbol{u}_{\mu,h}$, they consider the solution of the following linear dual problem

Find
$$\varphi \in \mathcal{V}$$

$$\int_{0}^{1} \int_{\Omega} \boldsymbol{P}'(\nabla \boldsymbol{u}_{\mu,h}) \nabla \boldsymbol{\eta} \colon \nabla \varphi \, \mathrm{d}\Omega \, \mathrm{d}s = \mathcal{J}(\boldsymbol{\eta}) \quad \forall \boldsymbol{\eta} \in \mathcal{V},$$

where $\mathcal{J}(\boldsymbol{\eta})$ is the output functional of interest which is taken equal to

$$\mathcal{J}_E(\boldsymbol{\eta}) = |||\boldsymbol{e}|||^{-1} \int_{\Omega} \mathbf{C} \nabla_s \boldsymbol{\eta} \colon \nabla \boldsymbol{e} \, \mathrm{d}\Omega$$

in the case of control of the error in energy norm. Following standard arguments, finally, they obtain

$$\|oldsymbol{u}_{\mu}-oldsymbol{u}_{\mu,h}\|_E \leq \sum_{\Omega_e\in\mathcal{T}_h} \, \omega_{\Omega_e}
ho_{\Omega_e}$$

with the local residuals defined by

$$\rho_{\Omega_e} = h_{\Omega_e} \| \boldsymbol{r}_{\boldsymbol{\sigma}_{\mu}^{h}} \|_{[L^2(\Omega_e)]^3} + h_{\Omega_e}^{\frac{1}{2}} \sum_{\boldsymbol{\gamma} \subset \partial \Omega_e} \| \boldsymbol{J}_{\boldsymbol{\sigma}_{\mu}^{h}}^{\boldsymbol{\gamma}} \|_{[L^2(\partial \Omega_e)]^3}$$

whereas the weights are approximated as in the linear case.

Since this estimation technique involves a linearization, the resulting estimate will be valid only asymptotically, that is for $u_{\mu,h}$ close to u_{μ} . Moreover, one is required to provide an approximation for e in order to define the forcing $\mathcal{J}_E(\eta)$ in the dual problem, which can provoke a deterioration of the quality of the bound.

Duality has also been employed to derive a posteriori error estimates for nonlinear variational problems. The work of Repin & Xanthis (1996) represents an important contribution in this sense. The authors, indeed, develop a rigorous mathematical analysis based on duality theory of the calculus of variations which leads to the concept of *duality error estimators* for approximations to nonlinear problems defined by a special class of convex functionals. In particular, the theory is presented for the Nadai deformation theory (Nadai, 1937) of hardening elasto-plastic material which gives rise to two variational problems: the primal for displacements and the dual for the stresses,

$$\forall \boldsymbol{\tau} \in \Sigma_{EQ}, \quad \mathcal{G}(\boldsymbol{\tau}) \leq \mathcal{G}(\boldsymbol{\sigma}_{ex}) = \sup_{\boldsymbol{\tau} \in \Sigma_{EQ}} \mathcal{G}(\boldsymbol{\tau}) = \mathcal{J}(\boldsymbol{u}) = \inf_{\boldsymbol{v} \in \mathcal{V}} \mathcal{J}(\boldsymbol{v}) \leq \mathcal{J}(\boldsymbol{v}), \quad \forall \boldsymbol{v} \in \mathcal{V}$$

where $\mathcal{J}(\boldsymbol{v})$ corresponds to the potential energy of the elasto-plastic body defined over the affine space \mathcal{V} of the kinematically admissible displacement fields and $\mathcal{G}(\boldsymbol{\tau})$ corresponds to the complementary energy of the elasto-plastic body defined over the affine space Σ_{EQ} of the statically admissible stress tensor fields. For the definition of the functionals $\mathcal{J}(\boldsymbol{v})$ and $\mathcal{G}(\boldsymbol{\tau})$ associated with the material model under consideration we refer to Repin & Xanthis (1996). Likewise Fraeijs de Veubeke (1965), Repin & Xanthis (1996) assume the difference $\mathcal{E}(\boldsymbol{v},\boldsymbol{\tau}) = \mathcal{J}(\boldsymbol{v}) - \mathcal{G}(\boldsymbol{\tau})$ as measure of the energy norm of the approximation error. However, due to the computational cost and difficulty for building equilibrated stress fields $\boldsymbol{\tau} \in \Sigma_{EQ}$, the functional $\mathcal{E}(\boldsymbol{v},\boldsymbol{\tau})$ is extended over the whole space Σ of the stress tensors. As a result, Repin & Xanthis (1996) obtain the following estimate

$$rac{1}{2}|||oldsymbol{v}-oldsymbol{u}|||^2\leq \mathcal{E}(oldsymbol{v},oldsymbol{ au})=\mathcal{E}_{\Lambda}(oldsymbol{v},oldsymbol{ au})+\mathcal{E}_{eq}(oldsymbol{ au}).$$

For the expressions of $\mathcal{E}_{\Lambda}(\boldsymbol{v},\boldsymbol{\tau})$ and \mathcal{E}_{eq} we refer again to Repin & Xanthis (1996). Here, in relation to further developments we note that the analysis of Repin & Xanthis (1996) shows that $\mathcal{E}_{\Lambda}(\boldsymbol{v},\boldsymbol{\tau}) = 0$ if and only if the constitutive equation is satisfied, whereas $\mathcal{E}_{eq}(\boldsymbol{\tau}) = 0$ if and only if $\boldsymbol{\tau}$ satisfies the equilibrium equations. For this reason, $\mathcal{E}_{\Lambda}(\boldsymbol{v},\boldsymbol{\tau})$ measures the error in the constitutive law and $\mathcal{E}_{eq}(\boldsymbol{\tau})$ measures the error in the equilibrium equations. The estimate, which is an extension of the one proposed by Ladevèze & Leguillon (1983) for linear problems, can, therefore, be applied by using the known approximate solution \boldsymbol{v} of the primal problem and the corresponding stress tensor obtained by the constitutive relation, even though this will not meet the equilibrium equations. However, the actual computation of the estimate, which is developed for any conforming approximation, not necessarily meeting an orthogonality condition, requires the solution of a quadratic minimization problem posed over an infinite dimensional which renders its practical application rather difficult. Furthermore, the estimate does not distinguish the elemental contributions, which form the basis of any adaptive process.

An a posteriori error estimate for the primal variational formulation of elastoplasticity with linear hardening given in Han & Reddy (1999) is provided by Alberty et al. (1999). The a posteriori error estimate refers to the finite element approximation of displacement and plastic strain field in the nonlinear incremental boundary value problem obtained within one time step of the backward Euler. With this formulation, the continuous formulation and finite element discretization of the nonlinear incremental boundary value problem can equivalently be expressed as minimization of a Lipschitz-continuous non-smooth convex functionals. By exploiting this property Alberty et al. (1999) obtain the following estimate

$$\|\boldsymbol{\sigma}-\boldsymbol{\sigma}^h\|_{[L^2(\Omega)]^{3\times 3}}^2+\|\boldsymbol{\epsilon}^p-\boldsymbol{\epsilon}_h^p\|_{[L^2(\Omega)]^{3\times 3}}^2+\|\nabla_s\boldsymbol{u}-\nabla_s\boldsymbol{u}_h\|_{[L^2(\Omega)]^{3\times 3}}^2\leq C\sum_{\Omega_e\in\mathcal{T}_h}\eta_{\Omega_e}^2,$$

where C is a constant depending on the hardening modulus, the discrete stress field σ^h is

$$\boldsymbol{\sigma}^h = \mathbf{C}(\nabla_s \boldsymbol{u}_h - \boldsymbol{\epsilon}_h^p),$$

whereas the elemental error indicators $\eta_{\Omega_e}^2$ are given by

$$\eta_{\Omega_e}^2 = h_{\Omega_e}^2 \|\boldsymbol{r}_{\sigma^h}\|_{[L^2(\Omega_e)]^3}^2 + \sum_{\gamma \subset \partial \Omega_e} h_{\gamma} \|\boldsymbol{J}_{\sigma^h}^{\gamma}\|_{[L^2(\gamma)]^3}^2$$

which are the same as in pure elasticity. This circumstance is motivated by the observation that the evolution law in the plastic material law (within one time step) is satisfied exactly on each element whence the material law has a vanishing residual. Thus, excluding the error accumulation for progressing time-steps, the only remaining residuals are those produced in the discrete equilibrium conditions.

2.3.2 Analysis of the time discretization error

The error estimates presented in the previous section, though based on solid theoretical background, by definition do not account for the effects of time discretization. These estimates, indeed, have been developed by looking at the error associated with the finite element approximation of the non-linear incremental boundary value problem obtained from a one-time step discretization of the initial boundary value problem. Therefore, the error estimate cannot take into account the error deriving from the replacement of the rate quantities appearing in the initial boundary value problem with the difference quotients. It is, indeed, this replacement that produces an error, called the time discretization error, which in certain cases may not be negligible.

An inherent difficulty in obtaining a complete *a posteriori* error estimation for the solution of the fully discrete scheme is due to the different nature of discretization: finite difference-type in time and finite element-type in space. With a finite element approximation-type, the error can be linked to the residual associated with the approximation. This is obtained simply by inserting the approximate solution into the exact equation. With a finite difference approximation-type, on the other hand, it seems impossible to insert the discrete solution into the exact equation and compute the residual. In this case, indeed, the error estimates developed for finite difference approximations of ordinary differential equations are traditionally based on predictor-corrector algorithms. In these algorithms the difference in solutions obtained by schemes with different orders of truncation error is used as rough estimates of the error. This estimate is in turn used to adjust the time step (Gear, 1971; Reiher, 1987; Abbo & Sloan, 1996). An alternative is the estimate proposed by Nochetto et al. (2000) that develop a posteriori error estimates for the backward Euler approximations of a special class of abstract evolution equations in Hilbert space. In their method it is the accuracy of time interpolant functions of the nodal values, such as continuous piecewise linear or discontinuous piecewise constant functions, to be assessed.

This unsatisfactory state of affair, however, does already become clear in the development of *a posteriori* error estimates for parabolic equations which represent the simplest class of evolution problems. For this reason, it seems appropriate to first recall briefly some works for parabolic problems. This is considered useful in relation to the evolution of elastoplastic systems.

Eriksson & Johnson (1991) develop an adaptive algorithm for the heat equation based on *a posteriori* error estimates of approximations obtained by the discontinuous Galerkin method which is based on space-time discretization. In the context of the error analysis, one of the advantages of using the discontinuous Galerkin method is the availability of a unique global variational structure which provides directly the approximation in space and in time. In this manner, one can represent the error in terms of the residual produced by the approximation in the global variational formulation of the problem which will, therefore, account for the discretization effects in space and time (Estep *et al.*, 2000). The use then of a space-time duality argument allows one to realize the control of the quantity of interest. The main idea of this technique, which extends to time dependent problems the procedure developed for elliptic problems, can be comprehended by considering the following abstract evolution equation in the Hilbert space \mathcal{H} (Nochetto *et al.*, 2000)

Find
$$u: t \in [0, T] \rightarrow \mathcal{H}$$

 $\dot{u} + \mathcal{F}(u) = 0$
 $u(t = 0) = u_0$ (2.33)

where $\mathcal{F}: \mathcal{H} \to \mathcal{H}$ is a given (nonlinear) operator of \mathcal{H} in \mathcal{H} . Let \mathcal{R} be the residual

$$\mathcal{R} = -\dot{U} - \mathcal{F}(U), \qquad (2.34)$$

where U is the approximate solution. By adding (2.33) and (2.34), we obtain the error equation for e = u - U

$$\dot{e} + \mathcal{U}e = \mathcal{R} \tag{2.35}$$

where

$$\mathcal{U} \stackrel{\text{def}}{=} \int_0^1 \nabla \mathcal{F}(su + (1-s)U) \mathrm{d}s.$$

Multiplying (2.35) by $\varphi \in C^1([0,T]; \mathcal{H})$ and integrating by parts over [0,T] delivers the error representation formula

$$\langle e(T), \varphi(T) \rangle = \langle e(0), \varphi(0) \rangle + \int_0^T \langle e, \dot{\varphi} - \mathcal{U}^* \varphi \rangle dt + \int_0^T \langle \mathcal{R}, \varphi \rangle dt$$
(2.36)

where \mathcal{U}^* is the transpose of \mathcal{U} . The *a posteriori* error estimate follows then by selecting φ in (2.36) as the solution of the backward dual problem

Find
$$\varphi(t) : t \in [0,T] \to \mathcal{H}$$

 $\dot{\varphi} - \mathcal{U}^* \varphi = 0$
 $\varphi(T) = e(T)$

and using strong stability properties of φ , such as bound for $\dot{\varphi}$, for evaluating the initial error and the residual terms.

The implementation of these ideas to the discontinuous Galerkin finite element method dG(0)cG(1), which combines continuous piecewise linear polynomial approximation in space with discontinuous piecewise constant polynomial approximation in time (which then reduces to the backward Euler with the rectangle rule applied to the integrals) delivers the estimate (Eriksson *et al.*, 1996)

$$\begin{aligned} \|u(t_N) - U_N\|_{L^2(\Omega)} &\leq L_N C_i \max_{1 \leq n \leq N} \left(\|h_n^2 R_2(U)\|_{I_n} + \|h_n^2 f\|_{I_n} + \|U_{n-1}\|_{L^2(\Omega)} + \|k_n f\|_{I_n} + \|\frac{h_n^2}{k_n} [U_{n-1}]\|_{L^2(\Omega)}^{\star} \right) \end{aligned}$$

$$(2.37)$$

where

$$\begin{aligned} \|\bullet\|_{I_n} &= \max_{t_{n-1} \leq t \leq t_n} \|\bullet\|_{L^2(\Omega)}, \\ u(t_N) &= u(\bullet, t_N), \end{aligned}$$

 C_i and L_N are constants depending on $k_n = t_n - t_{n-1}$ and t_n ,

$$R_2(U) = \max_{\gamma \subset \partial \Omega_e^n} \sup_{\gamma} \frac{1}{2} \frac{\|[\nabla U]\|_{L^2(\gamma)}}{h_n},$$

$$\Omega_e^n \in \mathcal{T}_{h_n}, \text{ with the finite element partition } \mathcal{T}_{h_n} \text{ associated with } I_n = [t_{n-1}, t_n],$$

 h_n meshsize of \mathcal{T}_{h_n} ,

and the starred term is different from zero only in presence of change between nonembedded finite element meshes. In equation (2.37) one can distinguish the terms that measure the residual error of the space discretization from those that measure the residual error of the time discretization. Among the latter, we consider further the starred term. This can be interpreted as error of the initial data of the single one step problem due to different interpolation assumption: in presence of nonembedded finite element spaces, $\mathcal{V}_{h_{n-1}} \not\subset \mathcal{V}_{h_n}$, the L^2 projection of $U_{n-1} \in \mathcal{V}_{h_{n-1}}$ onto \mathcal{V}_{h_n} is different from U_{n-1} .

Phase change phenomena represent a class of problems interesting to consider for they share some common features with the evolution of an elastoplastic medium, such as the presence of variational equations and differential inclusions in the governing equations. In the problem studied by Chen *et al.* (2000b), for example, we have

Find
$$\theta, \chi$$

 $\langle \frac{\partial \theta}{\partial t}, \eta \rangle + \langle \frac{\partial \chi}{\partial t}, \eta \rangle + \langle \nabla \theta, \nabla \eta \rangle = \langle f, \eta \rangle, \quad \forall \eta$
 $\theta \in \epsilon \frac{\partial \chi}{\partial t} + \Lambda(\chi)$

where ϵ is a small scalar parameter, θ stands for the temperature of a substance that occupies the domain Ω and undergoes solidification, χ is the phase variable and Λ

is a multivalued operator given by the inverse of the sign function. The primary variables θ and χ are approximated with continuous piecewise linear finite elements in space and the backward Euler method is employed for the time discretization.

In the error analysis carried out by the same authors, time interpolation functions of the nodal values of χ and θ , and also the constant interpolant function $\bar{\theta}(t) = \theta_n$, with $t_{n-1} < t \leq t_n$ are employed in the definition of the error. Starting from a representation of this error, the authors finally derive an estimate which is expressed as sum of several contributions. These include the term associated with the jump residual, the internal residual, the coarsening, the initial error, the time residual, the quadrature and the error on the data.

The extension of the previous procedures to the control of the error in the displacement formulation of an elastoplastic continuum is not straightforward. This is due mainly to two reasons. One is the high nonlinearity involved in the elastoplastic behaviour while the other is the use of primary variables, such as displacements, which do not appear in rate form in the formulation of the problem. Consequently, studies on the global control of the error are rare in the current literature. Rannacher & Suttmeier (1999) present a fairly complete analysis for the mixed-dual formulation of the quasi-static Prandtl-Reuss model defined by the following equations

Find:
$$\boldsymbol{\sigma} \in \Pi \mathcal{W}, \boldsymbol{v} = \dot{\boldsymbol{u}} \in \mathcal{V}$$

 $\langle \boldsymbol{\sigma}, \nabla \boldsymbol{\varphi} \rangle = \langle \boldsymbol{b}, \boldsymbol{\varphi} \rangle, \quad \forall \boldsymbol{\varphi} \in \mathcal{V}$
 $\langle \dot{\boldsymbol{\sigma}} - \mathbf{C} \nabla_s \boldsymbol{v}, \boldsymbol{\sigma} - \boldsymbol{\tau} \rangle \ge 0, \quad \forall \boldsymbol{\tau} \in \Pi \mathcal{W}$

$$(2.38)$$

where $\mathcal{V} = \{ \boldsymbol{v} \in [L^2(\Omega)]^3 : \boldsymbol{v} = \boldsymbol{0} \text{ on } \partial\Omega_d \}$ and $\Pi \mathcal{W} = \{ \boldsymbol{\tau} \in \mathcal{W} : \|\boldsymbol{\tau}^D\| - \sigma_y \leq 0, \text{ a.e. in } \Omega \}$, with $\mathcal{W} = [L^2(\Omega)]^{3 \times 3}$.

After performing time discretization with the backward Euler method, and the introduction of the nonlinear operator P defined in (2.30), problem (2.38) is transformed into the nonlinear variational equation

Data:
$$(\sigma_{n-1}, u_{n-1}) \in \mathcal{W} \times \mathcal{V}$$

Find: $(\sigma_n, u_n) \in \mathcal{W} \times \mathcal{V}$
 $\langle \sigma_n - P[\sigma_{n-1} + k_n \mathbb{C} \nabla_s (u_n - u_{n-1})], \tau \rangle + k_n \langle \sigma_n, \nabla \varphi \rangle = \langle b_n, \varphi \rangle,$
 $\forall (\tau, \varphi) \in \mathcal{W} \times \mathcal{V}.$
(2.39)

Equation (2.39) describes a material behaviour of Hencky-type in the case $k_n = 1$, $\sigma_{n-1} = 0$ and $u_{n-1} = 0$ which is discretized with standard finite elements in space as follows

Data:
$$(\boldsymbol{\sigma}_{n-1}^{h}, \boldsymbol{u}_{n-1}^{h}) \in \mathcal{W}_{h} \times \mathcal{V}_{h}$$

Find: $(\boldsymbol{\sigma}_{n}^{h}, \boldsymbol{u}_{n}^{h}) \in \mathcal{W}_{h} \times \mathcal{V}_{h}$
 $\langle \boldsymbol{\sigma}_{n}^{h} - \boldsymbol{P}[\boldsymbol{\sigma}_{n-1}^{h} + k_{n} \mathbf{C} \nabla_{s} (\boldsymbol{u}_{n}^{h} - \boldsymbol{u}_{n-1}^{h})], \boldsymbol{\tau}^{h} \rangle + k_{n} \langle \boldsymbol{\sigma}_{n}^{h}, \nabla \boldsymbol{\varphi}^{h} \rangle = \langle \boldsymbol{b}_{n}, \boldsymbol{\varphi}^{h} \rangle,$
 $\forall (\boldsymbol{\tau}^{h}, \boldsymbol{\varphi}^{h}) \in \mathcal{W}^{h} \times \mathcal{V}^{h}.$

$$(2.40)$$

For the control of the global stress error with respect to the energy norm, $\|\boldsymbol{e}_{\boldsymbol{\sigma}}\|_{E} = \langle \boldsymbol{e}_{\boldsymbol{\sigma}}, \mathbf{C}^{-1}\boldsymbol{e}_{\boldsymbol{\sigma}} \rangle$, in order to account for effects of time and space discretization Rannacher & Suttmeier (1999) consider the split of the total error into the three components

$$\boldsymbol{\sigma}(t_n) - \boldsymbol{\sigma}_n^h = \underbrace{\boldsymbol{\sigma}(t_n) - \boldsymbol{\sigma}_n}_{\boldsymbol{e}_{n,\boldsymbol{\sigma}}} + \underbrace{\boldsymbol{\sigma}_n - \tilde{\boldsymbol{\sigma}}_n}_{\tilde{\boldsymbol{e}}_{n,\boldsymbol{\sigma}}} + \underbrace{\tilde{\boldsymbol{\sigma}}_n - \boldsymbol{\sigma}_n^h}_{\boldsymbol{e}_{n,\boldsymbol{\sigma}}^h}$$
(2.41)

where

- $\sigma(t)$ is the exact solution of problem (2.40),
- σ_n is the solution of problem (2.39) with data σ_{n-1} and u_{n-1} ,
- $\tilde{\sigma}_n$ is again solution of problem (2.39) with data σ_{n-1}^h and u_{n-1}^h
- σ_n^h is solution of problem (2.40),

thus

- $e_{n,\sigma}$ is the error produced by time discretization only,
- $\tilde{e}_{n,\sigma}$ is the error due to change of data in the solution of (2.39),
- $e_{n,\sigma}^h$ is the error produced by space discretization only.

While for $e_{n,\sigma}^h$ Rannacher & Suttmeier (1999) provide an *a posteriori* estimate which is the same as for the discretization error of the Hencky problem, for the other two terms only a rough *a priori* estimate is given. In particular, for $e_{n,\sigma}$ the following bound is obtained

$$\max_{0 \le n \le N+1} \|\boldsymbol{e}_{n,\boldsymbol{\sigma}}\|_{E} \le \|\boldsymbol{e}_{\boldsymbol{\sigma}}^{0}\|_{E} + T \max_{1 \le n \le N+1} \left\{ k_{n} \max_{t \in [t_{n}, t_{n+1}]} \|\ddot{\boldsymbol{\sigma}}\|_{E} \right\}$$

which shows that the error due to the incremental loading process grows at most linearly with time provided that the exact solution stayed bounded. As for the error $\tilde{e}_{n,\sigma}$ related to the use of inexact starting values in each incremental loading step, this depends on the stability properties of problem (2.39). Rannacher & Suttmeier (1999) give the following estimate

$$\max_{1 \le n \le N+1} \|\tilde{\boldsymbol{e}}_{n,\boldsymbol{\sigma}}\|_E \le \|\tilde{\boldsymbol{e}}_{\boldsymbol{\sigma}}^0\|_E + \max_{0 \le n \le N} \left\{\|\boldsymbol{e}_{n,\boldsymbol{\sigma}}\|_E\right\}$$

which assumes an unchanged propagation of the full size of the error of the data.

The error estimate is finally obtained by combining all the previous results. Nevertheless, the adaptive algorithm given in Rannacher & Suttmeier (1999) is based only on the estimate of the term $e_{n,\sigma}^{h}$, for the effects of time discretization are neglected.

The same authors set up also a theoretical framework for *a posteriori* error analysis for the time discretization error based on duality. To this end, the solution of (2.40) is seen as arising from the use of a space-time approximation which uses

discontinuous Galerkin method dG(0) for the time discretization and standard finite element for the discretization in space. Further to a space-time duality argument on the linearized equation that defines the error the following estimate is obtained

$$\|\boldsymbol{\sigma}(T) - \boldsymbol{\sigma}_{N+1}^{h}\|_{E} \leq \sum_{n=1}^{N+1} k_{n} \sum_{\Omega_{e}^{n} \in \mathcal{T}_{h_{n}}} h_{\Omega_{e}}^{2} \left\{ (h_{\Omega_{e}}^{2} + k_{n}^{2} \sum_{i=1}^{4} \rho_{\Omega_{e}}^{n,i} \omega_{\Omega_{e}}^{n,i}) \right\}$$

where the local residuals are given by

$$\begin{split} \rho_{\Omega_{e}}^{n,1} &= h_{\Omega_{e}}^{-1} k_{n}^{-\frac{1}{2}} \int_{t_{n-1}}^{t_{n}} \|\boldsymbol{R}_{1}\|_{[L^{2}(\Omega_{e})]^{3}}, \mathrm{d}t, \quad \rho_{\Omega_{e}}^{n,2} = h_{\Omega_{e}}^{-\frac{3}{2}} k_{n}^{-\frac{1}{2}} \int_{t_{n-1}}^{t_{n}} \sum_{\gamma \subset \partial \Omega_{e}^{n}} \|\boldsymbol{J}_{\boldsymbol{\sigma}^{h}}^{\gamma}\|_{[L^{2}(\gamma)]^{3}} \mathrm{d}t, \\ \rho_{\Omega_{e}}^{n,3} &= h_{\Omega_{e}}^{-1} k_{n}^{-\frac{1}{2}} \int_{t_{n-1}}^{t_{n}} \|\boldsymbol{r}_{\boldsymbol{\sigma}^{h}}\|_{[L^{2}(\Omega_{e})]^{3}} \mathrm{d}t, \quad \rho_{\Omega_{e}}^{n,4} = h_{\Omega_{e}}^{-1} k_{n}^{-1} \|\boldsymbol{C}^{\frac{1}{2}}[\boldsymbol{\sigma}^{h}]_{n}\|_{[L^{2}(\Omega_{e})]^{3\times3}}, \end{split}$$

with

$$oldsymbol{R}_1 \stackrel{ ext{def}}{=} \dot{oldsymbol{\sigma}}^h + rac{1}{\mu} (oldsymbol{\sigma}^h - oldsymbol{P} oldsymbol{\sigma}^h) -
abla_s oldsymbol{v}_h,$$

whereas the respective weights are obtained by the solution of the linear continuous dual problem. Here, it is interesting to note that the term $\rho_{\Omega_e}^{n,4}$ reflects the low order regularity of the time constant piecewise function due to the presence of the discontinuity jump across the time instant t_n .

A similar approach has also been adopted by Larsson *et al.* (2001) for the control of quantity of interest in the space-time discretization of viscoplasticity. Also here, the latter is realized with space-time finite elements that are based on the use of the discontinuous Galerkin method for the time discretization and standard finite elements for the space discretization. This formulation gives quite naturally the variational setting for the analysis of the global error. Consequently, the general principles of duality given at the beginning of this Section can be applied.

Both the theoretical analysis of Rannacher & Suttmeier (1999) and the study of Larsson *et al.* (2001) share the same limits. These are inherent to the technique of error estimation based on duality when it is applied to nonlinear problems. One limit comes from the need of linearization of the problem that defines the error, so that the eventual estimate has only an asymptotic character. Another one comes from the need of solving exactly the continuous dual problem. Finally, the lack of a sufficiently developed theory of problems of elasto-plasticity does not allow a complete regularity study, which is very important in the analysis of the weights. All this hampers the practical use of the method.

Remark 2.2. The discontinuous Galerkin method dG(0) applied to time discretization is the backward Euler method with $(\bullet)_n^- = (\bullet)_n^+$ and the rectangle rule applied to the integrals. This method, however, does not accommodate the case of using different starting values, that is, $(\bullet)_n^- \neq (\bullet)_n^+$. On the other hand, the use of the discontinuous Galerkin method dG(1) in place of dG(0), though it has $(\bullet)_n^- \neq (\bullet)_n^+$, would produce a different time scheme different from the backward Euler.

Following an heuristic argument based on the comparison of the Prandtl-Reuss elastoplasticity with Hencky's plasticity, Barthold *et al.* (1997, 1998) split the total error into spatial and time discretization errors. The error associated with space discretization is captured by three terms. The first term is related to the lack of equilibrium and likewise in the work of Johnson & Hansbo (1992) this term is given by

$$\eta_{eq}^{2} = \sum_{\Omega_{e} \in \mathcal{T}_{h}} \left\{ C_{1}h^{2} \| \boldsymbol{r}_{\boldsymbol{\sigma}^{h}} \|_{[L^{2}(\Omega_{e})]^{3}}^{2} + C_{2}h^{2} \sum_{\gamma \subset \partial \Omega_{e}} \| \boldsymbol{J}_{\boldsymbol{\sigma}^{h}}^{\gamma} \|_{[L^{2}(\gamma)]^{3}}^{2} \right\}$$

with C_1 and C_2 interpolation constants. The second term is due to the violation of the consistency condition at points of the domain other than the Gauss points, as a result of the interpolation of the variables. This term has been introduced in Barthold *et al.* (1996) and is given by

$$\eta_{KT}^2 = \|\dot{\lambda}f(\boldsymbol{\sigma},\alpha) - \dot{\lambda}_h f(\boldsymbol{\sigma}_h,\alpha_h)\|_{L^2(\Omega)} = \|\dot{\lambda}_h f(\boldsymbol{\sigma}_h,\alpha_h)\|_{L^2(\Omega)}$$

where $f(\boldsymbol{\sigma}_h, \alpha_h)$ denotes the yield function sampled at $(\boldsymbol{\sigma}_h, \alpha_h)$ with α_h being the kinematic-internal variable and $\dot{\lambda}_h$ the plastic multiplier, which are both obtained from the finite element computation. Finally, the last term is the error in the plastic dissipation given by

$$\eta_{dis}^2 = \|\boldsymbol{\sigma} \colon (\dot{\boldsymbol{\epsilon}}^p - \dot{\boldsymbol{\epsilon}}_h^p)\|_{L^2(\Omega)}^2 + \|\mathsf{H}\alpha(\dot{\alpha} - \dot{\alpha}_h)\|_{L^2(\Omega)}^2$$

where H is the hardening modulus.

The time discretization error, on the other hand, is due to the numerical integration of the flow rule. The error produced within the time interval $[t_n, t_{n+1}]$ is defined as

$$\eta_{\Delta t}^{2} = \|\boldsymbol{\sigma}(t_{n+1})\|_{[L^{2}(\Omega)]^{3\times3}} \|\int_{t_{n}}^{t_{n+1}} \dot{\boldsymbol{\epsilon}}^{p} \,\mathrm{d}t - \Delta \boldsymbol{\epsilon}^{p}\|_{[L^{2}(\Omega)]^{3\times3}}$$
(2.42)

where the weight $\|\sigma(t_{n+1})\|_{[L^2(\Omega)]^{3\times 3}}$ has been added by Barthold *et al.* (1997) in order to be unit consistent with the space discretization error. Further to plausible assumptions, (2.42) is estimated by the maximal change of the normal (with respect to the yield function) between two time (load) steps. It is indeed this result to be interpreted physically as measure of the deviation of Prandtl–Reuss plasticity from Hencky plasticity in the current time step.

However, the aforementioned error measure appears to rely more on heuristic considerations motivated by the physics of the phenomenon which is not met, rather than arising from a theoretical argument typical of an error analysis. Furthermore, the computation of the several terms is realized by assuming recovered post-processed solutions in place of the unknown exact values. As a result, the estimation technique lends itself to the same type of criticism as for the recovery based error estimators.

A family of error measures with clear physical meaning and capable to account in a simple manner for effects of time and space discretization is represented on the other hand by the error in the constitutive equations, which is the objective of the next section.

2.3.3 Error in the Constitutive Equations

The error in the constitutive equations which has been applied successfully for the assessment of the accuracy of conforming approximations to linear problems is extended in a natural way to nonlinear dissipative problems of evolution.

Fundamental notion of this class of error measures is the definition of the socalled admissible solution. This is a particular approximate solution of the given initial boundary value problem. Its definition depends on the conditions and equations which are satisfied, whereas its quality is assessed in terms of the conditions and equations that are not. The application, then, of this theory to assess the quality of a finite element solution consists in building a corresponding admissible solution that reflects the approximations associated with the finite element solution.

The first error measure which exploits these concepts is the one based on the Drucker's inequality (Drucker, 1964). The definition of this error is due to Ladevèze (1985) in the context of the nonincremental method LATIN applied to the solution of the evolution of elastoviscoplastic materials which follow the conditions of Drucker's stability. This error measure was then further investigated in Coffignal (1987) and applied in Ladéveze *et al.* (1986) to the control of incremental finite element solutions of the same class of problems.

The error in the constitutive equations based on Drucker's inequality applies to material models defined by the following functional constitutive equations

$$\boldsymbol{\sigma}(t) = \mathcal{A}(\boldsymbol{\epsilon}(\tau), \, \tau \le t), \tag{2.43}$$

where \mathcal{A} denotes the constitutive operator which is function of the strain history[†]. In particular, the material model is supposed to obey to the Drucker's inequality. This property of the model can be formulated as follows (Bussy & Remond, 1985; Ladevèze & Pelle, 2001). Consider any pair of stress-strain history ($\boldsymbol{\sigma}(t); \boldsymbol{\epsilon}(\tau), \tau \leq t$) and ($\tilde{\boldsymbol{\sigma}}(t); \tilde{\boldsymbol{\epsilon}}(\tau), \tau \leq t$) which meet $\forall t \in [0, T]$ the constitutive equation (2.43) and the initial conditions, which are assumed equal to zero. The constitutive model defined by the operator \mathcal{A} is called stable according to Drucker if the following condition is satisfied

$$\left| \begin{array}{l} \forall (\boldsymbol{\sigma}, \boldsymbol{\epsilon}), \ (\tilde{\boldsymbol{\sigma}}, \tilde{\boldsymbol{\epsilon}}) \text{ meeting equation (2.43),} \\ \forall t \in [0, T], \quad \int_{0}^{t} [\boldsymbol{\sigma}(\tau) - \tilde{\boldsymbol{\sigma}}(\tau)] \colon [\dot{\boldsymbol{\epsilon}}(\tau) - \dot{\tilde{\boldsymbol{\epsilon}}}(\tau)] \, \mathrm{d}\tau \ge 0, \end{array} \right.$$

whereas it is strictly stable if also the following additional condition is met

$$\forall t \in [0,T], \quad \int_0^t [\boldsymbol{\sigma}(\tau) - \tilde{\boldsymbol{\sigma}}(\tau)] \colon [\dot{\boldsymbol{\epsilon}}(\tau) - \dot{\tilde{\boldsymbol{\epsilon}}}(\tau)] \, \mathrm{d}\tau = 0 \Rightarrow \begin{vmatrix} \boldsymbol{\sigma} = \tilde{\boldsymbol{\sigma}} \\ \boldsymbol{\epsilon} = \tilde{\boldsymbol{\epsilon}} \end{vmatrix}$$
(2.45)

The error in the constitutive equations based on Drucker's inequality assesses the quality of admissible solutions $S_{ad} = (\sigma_{ad}, u_{ad})$ which are defined by assuming

[†]Recall that the restriction of the function $f(\tau)$ to times τ not later than the current time t is called the history of f up to the time t.

at any time instant $t \in [0, T]$, a kinematically admissible displacement field u_{ad} and a statically admissible stress field σ_{ad} . Both u_{ad} and σ_{ad} are required to meet the initial conditions at t = 0. As a result, S_{ad} is an approximate solution to the initial boundary value problem, for the constitutive equations (2.43) are generally not satisfied. The quality of S_{ad} is characterized by defining first the following two processes meeting the constitutive equation (2.43)

$$\hat{S} = (\underbrace{\mathcal{A}(\nabla_{s}\boldsymbol{u}_{ad}(\tau,\tau \leq t))}_{\hat{\boldsymbol{\sigma}}(t)}; \boldsymbol{\epsilon}_{ad}(\tau,\tau \leq t)$$

$$\check{S} = (\boldsymbol{\sigma}_{ad}(t); \underbrace{\mathcal{A}^{-1}(\boldsymbol{\sigma}_{ad}(\tau,\tau \leq t))}_{\boldsymbol{\check{\epsilon}}(\tau,\tau \leq t)}$$

where \mathcal{A}^{-1} denotes the inverse operator of \mathcal{A} , and then computing the following quantity

$$e_{Drucker}^{2}(T) = \sup_{t \leq T} \int_{\Omega} \underbrace{\int_{0}^{t} [\hat{\boldsymbol{\sigma}}(\tau) - \boldsymbol{\sigma}_{ad}(\tau)] : [\dot{\boldsymbol{\epsilon}}_{ad}(\tau) - \dot{\boldsymbol{\dot{\epsilon}}}(\tau)] \,\mathrm{d}\tau}_{\eta^{2}(\boldsymbol{x},t)} \,\mathrm{d}\Omega \geq 0.$$
(2.46)

Indeed, by means of equations (2.44) and (2.45), one can easily check that $e_{Drucker}$ meets the following properties

$$\begin{vmatrix} e_{Drucker} \ge 0 \\ e_{Drucker} = 0 \Rightarrow \begin{vmatrix} \hat{\sigma} = \sigma_{ad} \\ & \forall x \in \Omega, \quad \forall t \in [0, T], \\ \check{\epsilon} = \epsilon_{ad} \end{vmatrix}$$

that is, $e_{Drucker}$ can be assumed as measure of the extent to which S_{ad} fails to satisfy the constitutive equations, and therefore of the extent to which S_{ad} fails to coincide with S_{ex} , the exact solution of the problem. Elaborating on the expression of $\eta^2(\boldsymbol{x},t)$ in the case of material models described by internal variables, Ladevèze & Pelle (2001) show that $\eta^2(\boldsymbol{x},t)$ can be expressed as sum of the error in the state law and the error in the evolution law,

$$\eta^{2}(\boldsymbol{x},t) = \frac{1}{2}[(\boldsymbol{\sigma}-\tilde{\boldsymbol{\sigma}}): \mathbf{C}^{-1}(\boldsymbol{\sigma}-\tilde{\boldsymbol{\sigma}}) + (\boldsymbol{X}-\tilde{\boldsymbol{X}}): (\boldsymbol{Y}-\tilde{\boldsymbol{Y}})]\Big|_{t} + \frac{1}{2}\int_{0}^{t} \Gamma \mathrm{d}\tau$$

where X denotes the set of kinematic-type internal variables, Y are the respective thermodynamic conjugate variable and $\Gamma \geq 0$ is related to the residual in the evolution law. Without further details related to the definition of these quantities which will be analysed in Section 3.5.1.2, here, we want just to observe that this decomposition is consistent with the fact that S_{ad} does not satisfy the complete constitutive equations. A similar structure of the error will be noted also in the extended dissipation error introduced in Ladevèze *et al.* (1999) for material models with internal variables which admit a standard formulation (Halphen & Nguyen, 1975). For this new measure of error, the admissibility conditions include the kinematic compatibility relations, the equilibrium equations and the initial conditions. As a result, the only equations left apart are the complete constitutive equations, i.e. the state laws and the evolution laws.

The error measure (2.46) has a global character in time and space. Furthermore, due to its definition, it is not possible to distinguish the contribution to the error arising from time and space discretization, thus it cannot be used to drive an adaptive process. To remedy this, Gallimard (1994) and Gallimard *et al.* (1996) have applied the concept of error in the constitutive equations to the problems obtained by using only time and space discretization, respectively. In this way, the authors have obtained error indicators that separate the effects of time and space discretization, respectively, which then have been used to control the discretization process.

The first use of the error in the constitutive equations to constitutive models described in terms of internal variables and having an associative flow rule is due to Ladevèze (1989). The admissibility conditions combine the kinematic compatibility relations, the equilibrium equations, the state law and the initial conditions. The only equation which is left out is the evolution law that governs the dissipative phenomenon. By exploiting the convexity structure of these equations, Ladevèze (1989) introduces, therefore, the concept of dissipation error given by the residual in the evolution law, which is appropriately reformulated in terms of the dissipation pseudo-potential and its Legendre-Fenchel transform. The theory of this error will be detailed in Section 3.5.1.1.

Likewise the error measure based on Drucker's inequality, the dissipation error has then applied in Moës (1996) and Ladevèze & Moës (1997) to assess the accuracy of incremental finite element solutions of associative problems. Furthermore, error indicators which separate the sources of the different discretizations have been defined and used to drive the adaptive process in time and space.

As we have already mentioned, another measure of the error in the constitutive equations has been introduced in Ladevèze *et al.* (1999) for material models described by internal variable and admitting standard formulation. The new measure is an extension of the dissipation error obtained by removing the state laws from the admissibility conditions. Applications of this error measure have been given for an elastic-damage coupled model in Ladevèze *et al.* (1999) solved with the nonincremental LATIN method and to the Prandtl–Reuss plasticity model in Orlando & Peric (2000) solved with the classical incremental finite element method. The extended dissipation error will be detailed in Section 3.5.1.2 and will represent the tool employed to analyse the effects of change of finite element mesh in classical finite element incremental solution of elasto-plastic problems.

The measures of the error in the constitutive equations given in this section apply to associative material models. Their key property is the characterization of the constitutive equations as solution of a scalar equation defined by convex scalar potentials. If the equation is not satisfied, it assumes positive value. This same characterization of the constitutive equations has also been given by de Saxcè (1992) to a much larger class of materials, such as the non-associated model of cyclic-plasticity of Chaboque-Marquis and the non-associated model of Drucker-Prager, which are termed implicit standard models. The characterization is based on the introduction of a unique potential, function of the rates and the associated force, which is used by Ladevèze (1999) to extend the notion of dissipation error also to this class of material models, whereas applications of the new measure of error have been given in Hjiaj (1994).

2.3.4 Heuristic Error Indicators

Lately, the emergence of new supercomputer architectures has allowed large-scale simulations of engineering problems and the incorporation of more detailed physics in the model. This has led to a substantial increase in the complexity of the simulation process that also incorporates non-linear mathematical models. As a result, adaptive strategies for the approximate solution of these problems are often based on heuristic error indicators or adaption indicators. A physically based argument is generally adopted as justification of their definition, which is typical for the problem under consideration. The motivation in using error indicators is that their primary quality is to be able to distinguish the elements and time steps which contribute mostly to the global accuracy of the approximation, even though they finally may fail to yield accurate global estimates. As a result, they represent a tool for the adaptive construction of the approximation, that is, they may be used as a guide for a sequence of discrete choices (to refine or not to refine a given element; to reduce or not to reduce a given time step) and not for the definition of the stopping criterion which should be based on an error estimate. It also appears clear that works in this area are much more abundant than in that of the development of error estimate. In the following we give only a brief overview of some error indicators that have been employed in practice.

The recovery based error estimate introduced by Zienkiewicz & Zhu (1987) has been applied in Zienkiewicz *et al.* (1988) to assess the finite element approximation of the flow formulation of incompressible plastic flow in which elastic effects are neglected. The above authors observe that in extrusion problems the energy norm has a definite meaning as it is simply the error in the rate of energy dissipation. Consequently, the total energy dissipation norm is defined as

$$|||\boldsymbol{u}||| = \left[\int_{\Omega} |\boldsymbol{\sigma}: \dot{\boldsymbol{\epsilon}}| \ \mathrm{d}\Omega\right]^{\frac{1}{2}}$$

where u is the displacement field, and the error $e = u - u_h$ as

$$|||\boldsymbol{e}||| = \left[\int_{\Omega} \left| (\boldsymbol{\sigma} - \boldsymbol{\sigma}^{h}) \colon (\dot{\boldsymbol{\epsilon}} - \dot{\boldsymbol{\epsilon}}_{h}) \right| \, \mathrm{d}\Omega \right]^{\frac{1}{2}}.$$

Since the basic formulation is mixed and the pressure variable is introduced as a means of ensuring incompressibility, Zienkiewicz *et al.* (1988) consider the energy

norm of the deviatoric part of σ , whose constitutive equation is

$$\boldsymbol{\sigma}^{D} = \mu \mathbf{D} \dot{\boldsymbol{\epsilon}}$$

where **D** is an appropriate material tensor and μ is the viscosity which in general is dependent on the strain rate.

As a result, the energy norm of the error is written as

$$|||\boldsymbol{e}||| = \left[\int_{\Omega} (\boldsymbol{\sigma}^{D} - \boldsymbol{\sigma}^{h,D}) \colon (\boldsymbol{\mu} \mathbf{D})^{-1} (\boldsymbol{\sigma}^{D} - \boldsymbol{\sigma}^{h,D}) \,\mathrm{d}\Omega\right]^{\frac{1}{2}}.$$
 (2.47)

The estimate of the measure of the error defined by equation (2.47) is obtained by replacing the exact solution σ^{D} with an approximation $\sigma^{D,\star}$ of higher order to that given by the finite element solution and obtained with the recovery technique described in Zienkiewicz & Zhu (1987).

This error indicator by its definition accounts for only effects of space discretization. It is thus used as basis for the refinement of the finite element mesh in the current time step. Furthermore, it does not account for the error in the incompressibility condition, which is enforced in a weak sense in the mixed formulation.

A posteriori error estimates based on the Zienkiewicz–Zhu adaptive strategy and the energy norm have been appropriately modified by Perić *et al.* (1994) to account for the elastoplastic deformation of the conventional and Cosserat continuum model. The corresponding error estimates are based on the rate of plastic work and on the plastic dissipation. After introducing the following definitions of error,

$$e_{\mathcal{W}^{p}} = \left[\sum_{\Omega_{e}\in\mathcal{T}_{h}}\int_{\Omega_{e}}\left|(\boldsymbol{\sigma}-\boldsymbol{\sigma}^{h}):\left(\dot{\boldsymbol{\epsilon}}^{p}-\dot{\boldsymbol{\epsilon}}_{h}^{p}\right)\right|\,\mathrm{d}\Omega\right]^{\frac{1}{2}},$$

$$e_{\mathcal{D}} = \left[\sum_{\Omega_{e}\in\mathcal{T}_{h}}\int_{\Omega_{e}}\left|(\boldsymbol{\sigma}-\boldsymbol{\sigma}^{h}):\left(\dot{\boldsymbol{\epsilon}}^{p}-\dot{\boldsymbol{\epsilon}}_{h}^{p}\right)+\left(\boldsymbol{X}-\boldsymbol{X}^{h}\right):\left(\dot{\boldsymbol{\alpha}}-\dot{\boldsymbol{\alpha}}^{h}\right)\right|\,\mathrm{d}\Omega\right]^{\frac{1}{2}},$$

$$(2.48)$$

where α denote the set of kinematic-type internal variables and X the respective conjugate thermodynamics forces, estimates for $(2.48)_1$ and $(2.48)_2$ are obtained by simply replacing the exact values with postprocessed solutions of the finite element approximation. The error based on the dissipation functional, in particular, has been then employed as basis for the finite element adaptive solution of a strain localization problem. The latter has been described by resorting to the Cosserat continuum in order to overcome serious limitations exhibited by classical continuous models in the post instability region. The expression of the estimate has been particularized for the classical model of J_2 -elastoplasticity which is generalized by introducing additional degrees of freedom within the Cosserat continua. In this case it is shown that the expression of the plastic dissipation involves only classical quantities. Assuming time discretization of the evolution problem is performed by the backward Euler algorithm, within a generic time step $I_n = [t_n, t_{n+1}]$, the *a posteriori* error estimate ϵ_D based on the dissipation functional is given by the following expression

$$\epsilon_{\mathcal{D}} = \sum_{\Omega_{e} \in \mathcal{T}_{h}} \int_{\Omega_{e}} \left[(\boldsymbol{\sigma}_{n+1}^{\star} - \boldsymbol{\sigma}_{n+1}^{h}) : \frac{3}{2} \mathbf{P} (\frac{\boldsymbol{\sigma}_{n+1}^{\star}}{R_{n+1}^{\star}} - \frac{\boldsymbol{\sigma}_{n+1}^{h}}{R_{n+1}^{h}}) + \right. \\ \left. + \left| R_{n+1}^{\star} - R_{n+1}^{h} \right| \left] \left(\Delta \alpha_{n+1}^{\star} - \Delta \alpha_{n+1}^{h} \right) \mathrm{d}\Omega \right]$$

where \mathbf{P} is a second order tensor used to represent the yield function in the sevendimensional stress space for plane strain J_2 -elastoplasticity within a Cosserat continuum, whereas α denotes the accumulated plastic strain and R the conjugate variable. The starred quantities denote the postprocessed solutions. Likewise the previous error estimate, error indicators are simply obtained by localization of the integrals at each element. These error indicators are, in turn, used only to drive mesh refinement, for they reflect only effects of space discretization.

A comparative evaluation of various error estimators for isotropic, elastoplastic and viscoplastic solids undergoing large deformations has been carried out in Tetambe *et al.* (1995). Five distinct error estimators are studied. These are the L^2 -norm of the stress error,

$$I_1 = \left[\int_{\Omega} (\boldsymbol{\sigma} - \boldsymbol{\sigma}^h) \colon (\boldsymbol{\sigma} - \boldsymbol{\sigma}^h) \,\mathrm{d}\Omega \right]^{\frac{1}{2}},$$

the L^2 -norm of the total strain error,

$$I_{2} = \left[\int_{\Omega} \left(\boldsymbol{\epsilon} - \boldsymbol{\epsilon}^{h} \right) \colon \left(\boldsymbol{\epsilon} - \boldsymbol{\epsilon}^{h} \right) \mathrm{d}\Omega \right]^{\frac{1}{2}},$$

the L^2 -norm of the equivalent total strain error,

$$I_3 = \left[\int_{\Omega} (\epsilon_{eq} - \epsilon_{eq}^h)^2 \,\mathrm{d}\Omega \right]^{\frac{1}{2}},$$

where ϵ_{eq} is an equivalent scalar strain given by an appropriate norm of the total strain ϵ ; the L^2 -norm of the incremental total strain given by

$$I_4 = \left[\int_{\Omega} \left(\Delta \boldsymbol{\epsilon} - \Delta \boldsymbol{\epsilon}^h \right) \colon \left(\Delta \boldsymbol{\epsilon} - \Delta \boldsymbol{\epsilon}^h \right) \mathrm{d}\Omega \right]^{\frac{1}{2}},$$

and finally the energy rate error norm introduced by Zienkiewicz *et al.* (1988). As usual, estimates for these measures of error are obtained by simply replacing the exact values with approximations of higher order, obtained by a suitable postprocessing of the finite element approximation. This study shows that the error computed using each of these estimators increases from its initial value as the deformation continues in the plastic zone in a large strain analysis. In particular, the error estimators based on the energy rate and the L^2 -norm of the incremental strain were able to predict the region with the maximum error.

Since these error estimators are defined in terms of incremental quantities, they mirror to different extent the effects of only space discretization. As a result, they are not able to predict consistently a monotonically increasing error with the deformation.

The adaptive procedures for large-deformation finite element analysis of elastic and elasto-plastic problems implemented by Lee & Bathe (1994) is based on a pointwise indicator for error in stresses and a pointwise indicator error in plastic strain increments. The pointwise error in stress indicator is a pointwise version of the stress smoothing type of indicators. This indicator estimates the error in the stresses by giving the differences between the unaveraged stress σ^h and a smoothed stress σ^* . In particular, only the pressure and the effective stress (which are related to the first and second stress invariant) are considered in the evaluation of the accuracy of the finite element solution. The error in the plastic strain increment tensor is given by

$$\Delta \boldsymbol{e}_{\boldsymbol{\epsilon}^p} = \Delta \boldsymbol{\epsilon}^p - \Delta \boldsymbol{\epsilon}_h^p \tag{2.49}$$

where

$$\Delta \boldsymbol{\epsilon}^p = \int_{t_n}^{t_{n+1}} \dot{\boldsymbol{\epsilon}}^p \mathrm{d}t.$$

An estimate of the error (2.49) is computed by considering the L^2 -norm of the difference between the plastic strain increments obtained using the trapezoidal rule, which is second order accurate, and the Euler backward method. Though this is an indicator on the time discretization error, it is used in Lee & Bathe (1994) as indicator of the space discretization error along with the indicator for the error on the stresses.

In the context of adaptive strategies for problems with localization of deformations Ortiz & Quigley (1991) argue heuristically that localized solutions may reasonably expect to be of bounded variation. The total variation of the solution over each element is thus assumed as a suitable adaptor indicator to drive the adaptive process. For 1D problems, the bounded variation of a scalar function v = v(x)is defined as follows

Given:
$$v : [a, b] \subset \mathbb{R} \to \mathbb{R}$$

 $|v|_{BV,[a,b]} = \sup \sum_{k=1}^{N} |v(x_k) - v(x_{k-1})|$ (2.50)

where the supremum is taken over all sequences $a = x_0 \leq x_1 \leq \ldots \leq x_N = b$, $\forall N \in \mathbb{N}$. Elements are targeted for refinement when the variation of the solution within each element is determined to be too high for the interpolation to adequately resolve it. Ortiz & Quigley (1991) also show that in 1D problems this error indicator can be derived from interpolation error bounds, that is, the equidistribution of the indicator (2.50) over the mesh is shown to minimize the L^{∞} interpolation error.

Use of error indicators based on minimization of interpolation error have also been employed by Demkowicz *et al.* (1985) and Radovitzky & Ortiz (1999), among others. In these works, the methodology set by Diaz *et al.* (1983) for grid optimization, that is, for relocation of the nodes within a fixed number of degree of freedom, is employed for mesh refinement. The estimation of the local errors is based on interpolation error bounds and extraction formulas for highly accurate estimates of the derivatives of the exact solution which appear in the bound. In the methodology proposed by Radovitzky & Ortiz (1999), starting with a very fine mesh, the discretization error can be bounded from above by the interpolation error, once a linearised analysis is carried out. In order to compute the size of the exact solution, which appears in the *a priori* error bound, Radovitzky & Ortiz (1999) consider the error of a lower order approximation $\tilde{\varphi}_h^{p-1}$ such that the computed finite element approximation φ_h^p represents the finite element nodal interpolation of $\tilde{\varphi}_h^{p-1}$.

These error indicators are based on *a priori* error estimates, therefore, usually, they deliver crude estimates. However, they present the advantage since they can be applied for those problems where energy estimates based on the residual break down due to the loss of ellipticity of the governing equation. Therefore, they must be used with judgement, for they imply a certain regularity of the exact solution and require accurate techniques to calculate derivatives of the exact solution.

2.4 Concluding remarks

The considerations of the previous sections show that the formal structure of the theory of *a posteriori* error estimation for linear problems is well understood, as opposite to the non linear problems. This mirrors the different maturity achieved in the theoretical analysis of the two classes of problems.

In the class of the linear elliptic problems, the recovery based error estimators represent perhaps the most widely used *a posteriori* estimates in solid mechanics applications for its relatively simple implementation. Also the explicit residual type error estimates are quite easy to use, for they simply require computing norms of the residual, once a certain weight between the jump term and the internal residual has been set. However, the way that the residual influences the error is accounted for by the stability properties of the dual problem, whose solution can pose further difficulties. The implicit residual type estimation of the error, on contrary, deliver much more effective estimates at the fraction of cost of solving additional local finite element problems posed on local higher order finite element spaces. In this class, the error in the constitutive equations developed by Ladevèze & Leguillon (1983) appears to be the most robust.

For nonlinear problems, very few theoretically based error estimates have been developed. In particular, for time dependent problems, the measures of the error in the constitutive equations have clear physical meaning and are able to account for all the sources of discretization. In the finite element solution of elastoplastic problems discretized in time with the backward Euler method, which will be the class of problems hereafter considered, apart from time and space discretization, another source of error needs to be considered. This arises as a result of the change of finite element mesh from one time step to the other, which introduces a discontinuity jump in the time linear interpolation of the discrete values of the solution. As a result, only measures of error that account for time discretization effects can reflect the low order regularity of the approximation across the time t_n when the change of mesh occurs. Thus, the extended dissipation error introduced by Ladevèze *et al.* (1999) naturally lends itself for this aim.

The theoretical analysis of this measure of error and the way it is able to accommodate discontinuity jump in the admissible solution is the main objective of the next Chapter.

Chapter 3

The error in the constitutive equations for dissipative nonlinear problems

3.1 Introduction

In this chapter the general theory to assess the quality of the so called admissible solutions of dissipative nonlinear problems is presented following Ladevèze & Pelle (2001).

After introducing the set of conditions and equations that govern the behaviour of a standard generalized model with internal variables (Halphen & Nguyen, 1975), and noting the dissipative nature of the elastoplastic problem under examination, a simple error analysis of the following first order ordinary differential equation

$$\begin{vmatrix} u'(t) + a(t) u(t) = f(t), \\ u(t = 0) = u_0 \end{vmatrix}$$

with $a \ge 0$, is performed. This is done in order to motivate the use of residual of approximate solutions of dissipative problems as indication of the error produced and also to show the influence on the error of the discontinuity jump which is present in the approximate solution at the time instant t_n . An *a posteriori* error estimation analysis is given, which delivers an upper bound expressed in terms of the discontinuity jump.

As introduction to the error in the constitutive equations, the fundamental notion of admissible solution is given. This is a particular approximate solution of the problem under exam. Its definition depends on the conditions and equations which are satisfied, whereas its quality is assessed in terms of the conditions and equations that are not. Consequently, the concepts of dissipation error and extended dissipation error for time continuous admissible solutions, and the one with discontinuity jump at the time instant t_n are given. A new measure of error in the constitutive equations of rate-independent plasticity models is defined which is called augmented extended dissipation error. Finally, a definition of error in solution concludes the chapter.

3.2 The reference continuum problem: The Initial Boundary Value Problem for a model with internal variables

3.2.1 Preliminaries

Consider a generic continuum body \mathcal{B} with boundary $\partial \mathcal{B}$ occupying a regular domain Ω of the three dimensional Euclidean space. Let $\partial \mathcal{B}_d$ and $\partial \mathcal{B}_t$ be disjoint parts of $\partial \mathcal{B}$, with $\partial \mathcal{B}_d \cup \partial \mathcal{B}_t = \partial \mathcal{B}$, where displacements and surface tractions are prescribed, respectively. We will assume displacements to be small in the quasi-static evolutive process of the body so that geometry changes and inertial effects may be neglected. As a result, the analysis will be performed in the reference configuration which is identified with the domain Ω while the boundary $\partial \mathcal{B}$ is identified with $\partial \Omega$. Likewise, the identifications $\partial \mathcal{B}_t$ with $\partial \Omega_t$ and $\partial \mathcal{B}_d$ with $\partial \Omega_d$ are assumed, as well.

3.2.2 Equilibrium Equation

Denote with S the linear space of the symmetric second order stress tensors. The statically admissible stress fields, $\sigma = \sigma(x, t) \in S$, are such that the weak form of the equilibrium, given by the virtual work, is satisfied, that is,

$$\int_{\Omega} \boldsymbol{\sigma} : \nabla \boldsymbol{\eta} \, \mathrm{d}\Omega = \int_{\Omega} \boldsymbol{b} \cdot \boldsymbol{\eta} \, \mathrm{d}\Omega + \int_{\partial \Omega_t} \boldsymbol{t} \cdot \boldsymbol{\eta} \, \mathrm{d}s \quad \forall \, \boldsymbol{\eta} \in \mathcal{V}_0, \quad \forall \, t \in \mathcal{I} = [0, T] \quad (3.1)$$

where $\mathbf{b} = \mathbf{b}(\mathbf{x}, t)$ and $\mathbf{t} = \mathbf{t}(\mathbf{x}, t)$ are respectively the body force and surface traction fields and \mathcal{V}_0 is the linear space of the virtual displacements which vanish on $\partial \Omega_d$, whereas \mathcal{I} is the time interval of interest (Gurtin, 1972; Ladevèze, 1999). Whenever necessary, the equilibrium condition will also be written as

$$\langle \boldsymbol{\sigma}(\boldsymbol{x},t),
abla \boldsymbol{\eta}(\boldsymbol{x})
angle = \langle \boldsymbol{b}(\boldsymbol{x},t), \boldsymbol{\eta}(\boldsymbol{x})
angle + \langle \boldsymbol{t}(\boldsymbol{x},t), \boldsymbol{\eta}(\boldsymbol{x})
angle_{\partial\Omega_t},$$

with $\langle \bullet, \bullet \rangle$ and $\langle \bullet, \bullet \rangle_{\partial \Omega_t}$ denoting integrals over the domain Ω and the boundary $\partial \Omega_t$ where traction conditions have been assigned, respectively.

3.2.3 Compatibility Equations

The compatibility equations refer to the displacements and the associated deformation. The conditions which define a kinematically admissible displacement field guarantee internal and external compatibility. In mathematical terms, this means that a displacement field $\boldsymbol{u} = \boldsymbol{u}(\boldsymbol{x},t)$ is kinematically admissible if it is time continuous, at least once differentiable in space and meets the boundary conditions on $\partial \Omega_d$. We refer to \mathcal{V} as the space of the kinematically admissible displacements.

Let \mathcal{E} be the linear space of the symmetric second order strain tensors. The kinematically admissible strain fields, $\boldsymbol{\epsilon} = \boldsymbol{\epsilon}(\boldsymbol{x},t) \in \mathcal{E}$, are continuous fields which are obtained from the kinematically admissible displacement fields $\boldsymbol{u} = \boldsymbol{u}(\boldsymbol{x},t) \in \mathcal{V}$ as follows

$$\boldsymbol{\epsilon} = \frac{1}{2} (\nabla \boldsymbol{u} + \nabla \boldsymbol{u}^{\mathrm{T}}) \stackrel{\text{def}}{=} \nabla_{\boldsymbol{s}} \boldsymbol{u}, \qquad (3.2)$$

i.e., $\boldsymbol{\epsilon}$ is the symmetric part of the second order tensor $\nabla \boldsymbol{u}$.

3.2.4 Constitutive Equations

3.2.4.1 Thermodynamic Admissibility

In the constitutive modelling with internal variables, which is hereafter adopted, the functional dependence of the present state of the material upon the history of the total strain and temperature (which are the observable variables) is replaced by equations that define the current state of the material in terms of only the current value of both the observable variables and additional variables, called generally hidden or internal variables. They are macroscopic measures of irreversible phenomena (Lemaitre & Chaboche, 1990; Besson *et al.*, 2001). The success of such formulation is due to the simplicity, in general, of the mathematical and numerical analysis of the problem which governs finally the evolution of the whole system.

Our interest, in particular, will focus on constitutive models of rate independent plasticity in isothermal conditions belonging to the class of the so-called standard generalised materials as introduced by Halphen & Nguyen (1975). By exploiting the convex structure of these models, emphasis will be placed especially on scalar equivalent formulations of the tensorial constitutive equations.

The physical validity of a constitutive model is usually expressed by its thermodynamic admissibility (Coleman & Gurtin, 1967; Maugin, 1992), which is obtained by imposing the validity of the Clausius-Duhem inequality. For isothermal processes, the aforementioned inequality represents the difference between the external total power and the rate of variation of the reversible stored energy, that is,

$$-\dot{\psi} + \boldsymbol{\sigma} \colon \dot{\boldsymbol{\epsilon}} \ge 0, \tag{3.3}$$

where ψ is the free Helmholtz energy per unit volume defined in terms of the state variables which describe the model, and $\sigma: \dot{\epsilon}$ is the total external power.

The inequality (3.3) is obtained by combining the local form of the first principle of thermodynamics, which expresses the conservation of energy, and the second principle, which refers notably to the direction of irreversible processes.

A thermodynamically consistent model is, therefore, obtained by specifying the state variables along with the functional form of the free Helmholtz energy and the complementary laws which describe the time evolution of the internal variables in respect of (3.3).

3.2.4.2 State variables

In the definition of the state variables, which characterise uniquely the state of the system, we refer only to the so called kinematic type variables whereas the corresponding static one are obtained by duality as introduced by the total energy and the dissipation. In particular, we assume the following: (i) temperature to be constant with time and uniform in space so that it will not be considered hereafter; (ii) that the total strain can be uniquely decomposed additively into its elastic and plastic part, that is,

$$\boldsymbol{\epsilon} = \boldsymbol{\epsilon}^e + \boldsymbol{\epsilon}^p, \tag{3.4}$$

and finally (iii) that the local state of the material is described by means of additional internal variables α which may only include scalar and/or second order tensors (Coleman & Gurtin, 1967; Reddy & Martin, 1994), characterizing the internal changes of the material.

3.2.4.3 Equations of State

The reversible energy density stored under any form in the material is represented by the free Helmholtz energy ψ which is assumed to depend on the variables (ϵ^e, α). In particular, we assume $\psi(\epsilon^e, \alpha)$ to be expressed as a sum of two proper strictly



Figure 3.1: Energy repartition for a linear elastic and linear isotropic hardening model

convex and lower semicontinuous functions of each of its arguments: $\psi_e(\epsilon^e)$ which is the stored energy due to elastic strain and $\psi_p(\alpha)$ which is the stored energy due to plastic and internal parameters related to hardening effects (Nguyen, 2000), cf. Figure 3.1, i.e.,

$$\psi(\epsilon^e, \alpha) = \psi_e(\epsilon^e) + \psi_p(\alpha).$$

By expanding the Clausius-Duhem inequality which is required to hold for any admissible thermodynamic process $(\epsilon, \epsilon^p, \alpha)$ (Coleman & Gurtin, 1967), we obtain the state equations

$$\boldsymbol{\sigma} = \frac{\partial \psi_e}{\partial \boldsymbol{\epsilon}^e}(\boldsymbol{\epsilon}^e), \quad \boldsymbol{A} = \frac{\partial \psi_p}{\partial \boldsymbol{\alpha}}(\boldsymbol{\alpha}), \tag{3.5}$$

and the associated intrinsic mechanical dissipation

$$\boldsymbol{\sigma} \colon \dot{\boldsymbol{\epsilon}}^p - \boldsymbol{A} \colon \dot{\boldsymbol{\alpha}} \ge 0. \tag{3.6}$$

The state equations (3.5) are also called state laws and in particular we refer to $(3.5)_1$ as the elastic law, whereas we refer to $(3.5)_2$ as the hardening law. The force-type variable A, defined by the hardening law $(3.5)_2$, is termed the thermodynamic force conjugate to α (Lemaitre & Chaboche, 1990; Besson *et al.*, 2001). If the functions $\psi_e = \psi_e(\epsilon^e)$ and $\psi_p = \psi_p(\alpha)$ are not differentiable, then the concept of subgradient[†] is introduced (Germain *et al.*, 1983; Rockafellar, 1970; Ekeland & Temam, 1976).

For our subsequent developments, it is useful to consider the following equivalent formulation of the state equations (3.5) (Germain *et al.*, 1983; Ladevèze & Pelle, 2001),

$$\underbrace{\psi_{e}(\epsilon^{e}) + \psi_{e}^{*}(\sigma) - \sigma : \epsilon^{e}}_{\substack{s^{l} \eta_{\sigma,t}^{2}(\sigma; \epsilon^{e})}} = 0 \Leftrightarrow \sigma - \frac{\partial \psi_{e}}{\partial \epsilon^{e}}(\epsilon^{e}) = 0,$$

$$\underbrace{\psi_{p}(\alpha) + \psi_{p}^{*}(A) - A : \alpha}_{\substack{s^{l} \eta_{\sigma,t}^{2}(A; \alpha)}} = 0 \Leftrightarrow A - \frac{\partial \psi_{p}}{\partial \alpha}(\alpha) = 0,$$
(3.7)

where $\psi_e^*(\boldsymbol{\sigma})$ and $\psi_p^*(\boldsymbol{A})$ are the conjugate functions or Legendre-Fenchel transforms of $\psi_e(\boldsymbol{\epsilon}^e)$ and $\psi_p(\boldsymbol{\alpha})$, respectively, cf. Figure 3.2. These are defined as





[†] Let \mathcal{V} and \mathcal{V}^* be two linear topological spaces placed in duality by the separating bilinear form $\langle \bullet, \bullet \rangle_{\mathcal{V}^*, \mathcal{V}}$. Given a real valued function $f : v \in \mathcal{V} \to \mathbb{R}$ and $v_0 \in \mathcal{V}$, if there exists a neighbourhood I_{v_0} of v_0 and $a \in \mathcal{V}^*$ such that the following inequality is satisfied $f(v) - f(v_0) \ge \langle a, v - v_0 \rangle$, $\forall v \in I_{v_0}$, we say that f is subdifferentiable at v_0 and a is referred to as a subdifferential of f at v_0 . The set of vectors $a \in \mathcal{V}^*$ which satisfy the previous property is referred to as the subgradient of f at v_0 and is denoted with $\partial f(v_0)$. If f is convex on \mathcal{V} and is subdifferentiable at v_0 the previous property holds for all $v \in \mathcal{V}$.

$$\psi_{e}^{*}: \boldsymbol{\sigma} \in \mathcal{S} \to \psi_{e}^{*}(\boldsymbol{\sigma}) = \sup_{\boldsymbol{\epsilon}^{e} \in \mathcal{E}} \{\boldsymbol{\sigma}: \boldsymbol{\epsilon}^{e} - \psi_{e}(\boldsymbol{\epsilon}^{e})\} \in \mathbb{R} \cup \{+\infty\},$$

$$\psi_{p}^{*}: \boldsymbol{A} \in \mathcal{A} \to \psi_{p}^{*}(\boldsymbol{A}) = \sup_{\boldsymbol{\alpha} \in \Lambda} \{\boldsymbol{A}: \boldsymbol{\alpha} - \psi_{p}(\boldsymbol{\alpha})\} \in \mathbb{R} \cup \{+\infty\},$$
(3.8)

with $\sigma: \epsilon$ denoting the duality pairing of the two spaces S and \mathcal{E} , whereas $A: \alpha$ is the duality pairing of the space Λ of the strain-type internal variables α and the space \mathcal{A} of the thermodynamic forces A.



Figure 3.3: Geometrical interpretation of the inequalities (3.9) for linear elastic and linear isotropic hardening model

The functions defined by (3.8) are convex lower semicontinuous functions, for they are pointwise supremum of a family of continuous affine functionals (Rockafellar, 1970; Ekeland & Temam, 1976). Furthermore, if ψ_e and ψ_p are differentiable, their conjugates (3.8) amount to their Legendre transforms, respectively.

Upon the definition of the Legendre-Fenchel transforms, it follows that for any pair $(\sigma, \epsilon^e) \in S \times \mathcal{E}$ and $(\alpha, A) \in \Lambda \times \mathcal{A}$,

$$\psi_{e}^{*}(\boldsymbol{\sigma}) + \psi_{e}(\boldsymbol{\epsilon}^{e}) - \boldsymbol{\sigma} : \boldsymbol{\epsilon}^{e} \ge 0$$

$$\psi_{p}^{*}(\boldsymbol{A}) + \psi_{p}(\boldsymbol{\alpha}) - \boldsymbol{A} : \boldsymbol{\alpha} \ge 0$$
(3.9)

where the respective equality applies if and only if (σ, ϵ^e) is the solution of $(3.5)_1$ and (\mathbf{A}, α) is the solution of $(3.5)_2$, respectively, cf. Figure 3.3.

In conclusion, given its importance and for completeness, we sketch the equivalence between $(3.5)_1$ and $(3.7)_1$.

Proof. Given $\sigma \in S$, let $\bar{\epsilon}^e \in \mathcal{E}$ be such that, (cf. Figure 3.2),

$$\psi_e^*(\boldsymbol{\sigma}) = \boldsymbol{\sigma} : \bar{\boldsymbol{\epsilon}}^e - \psi_e(\bar{\boldsymbol{\epsilon}}^e) = \sup_{\boldsymbol{\epsilon}^e \in \mathcal{E}} \{ \boldsymbol{\sigma} : \boldsymbol{\epsilon}^e - \psi_e(\boldsymbol{\epsilon}^e) \}.$$

It follows

$$\boldsymbol{\sigma} \colon \tilde{\boldsymbol{\epsilon}}^e - \psi_e(\bar{\boldsymbol{\epsilon}}^e) \geq \boldsymbol{\sigma} \colon \boldsymbol{\epsilon}^e - \psi_e(\boldsymbol{\epsilon}^e), \quad \forall \boldsymbol{\epsilon}^e \in \mathcal{E}$$



Figure 3.4: Definition of subdifferential at $\bar{\epsilon}^e$. (a) Differentiable function. (b) Non Differentiable function

that is,

$$\psi_e(\epsilon^e) - \psi_e(\bar{\epsilon}^e) \ge \sigma : (\epsilon^e - \bar{\epsilon}^e), \quad \forall \epsilon^e \in \mathcal{E}$$

This, from the definition of subdifferential (cf. note † and Figure 3.4) means

$$\sigma \in \partial \psi_e(\bar{\epsilon}^e).$$

If the function $\psi_e(\epsilon^e)$ is differentiable at $\bar{\epsilon}^e$, then $\partial \psi_e(\bar{\epsilon}^e) = \left\{ \frac{\partial \psi_e}{\partial \epsilon^e}(\bar{\epsilon}^e) \right\}$ which is $(3.5)_1$.

Viceversa, if $\sigma \in \partial \psi_e(\bar{\epsilon}^e)$, then it follows

$$\psi_e(\epsilon^e) - \psi_e(\bar{\epsilon}^e) \ge \sigma : (\epsilon^e - \bar{\epsilon}^e), \quad \forall \epsilon^e \in \mathcal{E}$$

that is,

$$\sigma : ar{\epsilon}^e - \psi_e(ar{\epsilon}^e) \geq \sigma : \epsilon^e - \psi_e(\epsilon^e), \quad orall \epsilon^e \in \mathcal{E}.$$

Thus,

$$\psi_e^*(\boldsymbol{\sigma}) = \boldsymbol{\sigma} \colon \bar{\boldsymbol{\epsilon}}^e - \psi_e(\bar{\boldsymbol{\epsilon}}^e)$$

which is $(3.7)_1$.

The equivalence between $(3.5)_2$ and $(3.7)_2$ can be proved using the same arguments.

3.2.4.4 Complementary Equations. Associative Plasticity

In isothermal processes where irreversible phenomena occur, that is, the dissipation $-\dot{\psi} + \sigma$: $\dot{\epsilon}$ does not vanish, the state equations alone are not sufficient to define the current state of the material. Supplementary equations are needed, which allow one to characterize the history of the observable variables in terms of the internal variables (Bataille & Kestin, 1979). These equations, therefore, must have differential

character, that is, they have to relate the rate of the internal variables to the associated thermodynamic forces. The conjugacy is here meant in the sense of power as defined by the intrinsic mechanical dissipation $\boldsymbol{\sigma}: \dot{\boldsymbol{\epsilon}}^p - \boldsymbol{A}: \boldsymbol{\alpha}$ which is used to relate by duality the space $\dot{\boldsymbol{\Sigma}} = \dot{\boldsymbol{\mathcal{E}}} \times \dot{\boldsymbol{\Lambda}}$ of the rate of internal variables $(\dot{\boldsymbol{\epsilon}}^p, -\dot{\boldsymbol{\alpha}})$ with the space $\boldsymbol{\Sigma} = \boldsymbol{\mathcal{S}} \times \boldsymbol{\mathcal{A}}$ of the thermodynamic forces $(\boldsymbol{\sigma}, \boldsymbol{\mathcal{A}})$. Here, we have denoted with $\dot{\boldsymbol{\mathcal{E}}}$ the linear space of the rate of the strain tensors $\dot{\boldsymbol{\epsilon}}^p$, and with $\dot{\boldsymbol{\Lambda}}$ the linear space of the tensorial quantities represented by $\dot{\boldsymbol{\alpha}}$.

The complementary equations are only restricted to meet the intrinsic mechanical inequality (3.6). Since $\sigma: \dot{\epsilon}^p - A: \dot{\alpha}$ represents the part of the plastic power which is actually dissipated in heat form, assuming the non negativity of the intrinsic mechanical dissipation means that we require the irreversible processes described by the constitutive model under consideration to produce heat (Besson *et al.*, 2001).

There are several approaches to the definition of the complementary laws in the respect of (3.6) (Chaboche, 1996). A simple way to ensure a priori the thermodynamic consistency of the model is given by the class of the standard generalised materials introduced in Halphen & Nguyen (1975). In this model, one assumes the existence of a potential of dissipation $\varphi(\dot{\epsilon}^p, -\dot{\alpha})$ in the space $\dot{\Sigma} = \dot{\mathcal{E}} \times \dot{\Lambda}$ of rate of dissipative variables, which is positive, convex in its variables and such that $\varphi(\mathbf{0}, \mathbf{0}) = 0$. The complementary laws are then given by

$$(\boldsymbol{\sigma}, \boldsymbol{A}) \in \partial \varphi(\dot{\boldsymbol{\epsilon}}^p, -\dot{\boldsymbol{\alpha}})$$
 (3.10)

where the symbol ∂ denotes the subdifferential operator (Rockafellar, 1970; Ekeland & Temam, 1976). As a result of the properties of $\varphi(\dot{\epsilon}^p, -\dot{\alpha})$ and the definition (3.10) of the complementary laws, the thermodynamic consistency of the material model follows quite easily:

Proof. From the meaning of subdifferential of a convex function, and the duality established by the mechanical intrinsic dissipation, it follows

$$\varphi(\dot{\epsilon}^{p'},-\dot{\alpha}')-\varphi(\dot{\epsilon}^{p},-\dot{\alpha})\geq\boldsymbol{\sigma}\colon(\dot{\epsilon}^{p'}-\dot{\epsilon}^{p})-\boldsymbol{A}\colon(\dot{\alpha}'-\dot{\alpha}),\quad\forall(\dot{\epsilon}^{p'},-\dot{\alpha}')\in\dot{\Sigma}.$$

Thus, for $(\dot{\boldsymbol{\epsilon}}^{p'},-\dot{\boldsymbol{lpha}}')=(\mathbf{0},\mathbf{0})$ one obtains

$$arphi(\mathbf{0},\mathbf{0}) - arphi(\dot{m{\epsilon}}^p,-\dot{m{lpha}}) \geq -m{\sigma}\colon \dot{m{\epsilon}}^p + m{A}\colon \dot{m{lpha}}$$

and for the properties of $\varphi(\dot{\epsilon}^p, -\dot{\alpha})$, it follows finally

$$oldsymbol{\sigma} \colon \dot{oldsymbol{\epsilon}}^p - oldsymbol{A} \colon oldsymbol{lpha} \geq arphi(\dot{oldsymbol{\epsilon}}^p, -\dot{oldsymbol{lpha}}) \geq 0$$

The definition of the complementary laws (3.10) in terms of the subdifferential allows the important class of non differentiable dissipation potentials associated with time independent dissipative processes, which is hereafter considered, such as dry friction, plasticity, brittle fracture and damage, to be included in the same theoretical framework (Panagiotopoulos, 1985; Nguyen, 1994). The rate independent phenomena, in fact, are characterised by assuming the dissipation potential as a lower semicontinuous convex positive homogeneous function of degree one (Panagiotopoulos, 1985; Nguyen, 1994). In geometrical terms, such function presents a convex closed set as epigraph, and for the property,

$$\varphi(m(\dot{\epsilon}^p, \dot{\alpha})) = m\varphi(\dot{\epsilon}^p, \dot{\alpha}) \quad \forall m > 0,$$

which defines the positivity homogeneity of degree one of the function φ , the epigraph is, in particular, a cone with vertex in the origin. Consequently, therein, the function is not differentiable, cf. Figure 3.5. A function with the above properties



Figure 3.5: Epigraph of a positive homogeneous function of degree one

is usually called a positive closed gauge (Rockafellar, 1970; Han & Reddy, 1999; Hiriart-Urruty & Lemaréchal, 2001). In this case, it is possible to show (Maugin, 1992; Han & Reddy, 1999; Hiriart-Urruty & Lemaréchal, 2001) that the dissipation potential $\varphi(\dot{\epsilon}^p, -\dot{\alpha})$ may be characterized as support function of a closed convex domain $\mathbb{E} \subseteq \Sigma$, containing the origin $(\sigma, A) = (0, 0)$, that is,

$$\varphi(\dot{\boldsymbol{\epsilon}}^{p},-\dot{\boldsymbol{\alpha}}) = \sup_{(\boldsymbol{\sigma},\boldsymbol{A})\in\mathbb{E}} \{\boldsymbol{\sigma}:\dot{\boldsymbol{\epsilon}}^{p}-\boldsymbol{A}:\dot{\boldsymbol{\alpha}}\}, \quad \forall(\dot{\boldsymbol{\epsilon}}^{p},-\dot{\boldsymbol{\alpha}})\in\dot{\boldsymbol{\Sigma}}$$
(3.11)

with \mathbb{E} , called the elastic domain, defined by

$$\mathbb{E} = \{ (\boldsymbol{\sigma}, \boldsymbol{A}) \in \Sigma \mid \boldsymbol{\sigma} : \dot{\boldsymbol{\epsilon}}^p - \boldsymbol{A} : \dot{\boldsymbol{\alpha}} \le \varphi(\dot{\boldsymbol{\epsilon}}^p, -\dot{\boldsymbol{\alpha}}), \quad \forall (\dot{\boldsymbol{\epsilon}}^p, -\dot{\boldsymbol{\alpha}}) \in \dot{\Sigma} \}.$$
(3.12)

In the special case of a regular domain, in the space $\Sigma = S \times A$ of the generalised stresses $q = (\sigma, A)$, the elastic domain may be expressed as

$$\mathbb{E} = \{ \boldsymbol{q} = (\boldsymbol{\sigma}, \boldsymbol{A}) \in \Sigma \mid f(\boldsymbol{q}) \leq 0 \}$$

where $f = f(\boldsymbol{\sigma}, \boldsymbol{A})$, named the yield criterion, is lower semicontinuous, convex and such that $f(\mathbf{0}, \mathbf{0}) \leq 0$.

The set \mathbb{E} defines the locus of the admissible generalised stresses. The name derives from the observation that the thermodynamic forces conjugate to $(\dot{\epsilon}^p, -\dot{\alpha}) =$

(0, 0), i.e., to an instantaneous elastic response, belong to the elastic domain \mathbb{E} , that is,

$$(\boldsymbol{\sigma}, \boldsymbol{A}) \in \partial \varphi(\boldsymbol{0}, \boldsymbol{0}) = \mathbb{E}$$
 (3.13)

and also, if $(\dot{\epsilon}^p, -\dot{\alpha}) \neq (0, 0)$, we have

$$(\boldsymbol{\sigma}, \boldsymbol{A}) \in \partial \varphi(\dot{\boldsymbol{\epsilon}}^p, -\dot{\boldsymbol{\alpha}}) \subseteq \partial \mathbb{E}.$$
 (3.14)

As a result, for any $(\dot{\epsilon}^p, -\dot{\alpha})$ the associated thermodynamic forces (σ, A) in this model are constrained to belong always to the elastic domain \mathbb{E} . Next, we prove only (3.13), whereas (3.14) can be easily obtained once the equivalent formulation of (3.10) in terms of the Legendre-Fenchel transform of φ is given.

Proof. By definition of subdifferential,

$$\partial \varphi(\mathbf{0},\mathbf{0}) = \{ (\boldsymbol{\sigma},\boldsymbol{A}) \in \Sigma^* | \varphi(\dot{\boldsymbol{\epsilon}}^p, -\dot{\boldsymbol{\alpha}}) - \varphi(\mathbf{0},\mathbf{0}) \ge \boldsymbol{\sigma} \colon \dot{\boldsymbol{\epsilon}}^p - \boldsymbol{A} \colon \dot{\boldsymbol{\alpha}}, \quad \forall (\dot{\boldsymbol{\epsilon}}^p, -\dot{\boldsymbol{\alpha}}) \in \dot{\Sigma} \}$$

and since $\varphi(\mathbf{0}, \mathbf{0}) = 0$ we obtain finally (3.12), thus $\partial \varphi(\mathbf{0}, \mathbf{0}) = \mathbb{E}$.

In place of equation (3.10), it is usually much more convenient to refer to the inverse relations obtained by introducing the Legendre-Fenchel transform of φ defined as

$$\varphi^*(\boldsymbol{\sigma},\boldsymbol{A}): (\boldsymbol{\sigma},\boldsymbol{A}) \in \mathcal{S} \times \mathcal{A} \to \varphi^*(\boldsymbol{\sigma},\boldsymbol{A}) = \sup_{(\dot{\boldsymbol{\epsilon}}^p,-\dot{\boldsymbol{\alpha}})\in \dot{\Sigma}} \{\boldsymbol{\sigma}: \dot{\boldsymbol{\epsilon}}^p - \boldsymbol{A}: \dot{\boldsymbol{\alpha}} - \varphi(\dot{\boldsymbol{\epsilon}}^p,-\dot{\boldsymbol{\alpha}})\}.$$

Since $\varphi(\dot{\epsilon}^p, -\dot{\alpha})$ is the support function of \mathbb{E} , the dual dissipation potential $\varphi^*(\sigma, A)$ is then the indicator function of \mathbb{E} defined as

$$\varphi^*(\boldsymbol{\sigma}, \boldsymbol{A}) = I_{\mathbb{E}} \stackrel{\text{def}}{=} \begin{cases} 0 & \text{if} (\boldsymbol{\sigma}, \boldsymbol{A}) \in \mathbb{E} \\ +\infty & \text{if} (\boldsymbol{\sigma}, \boldsymbol{A}) \notin \mathbb{E}, \end{cases}$$

The evolution equations for the internal variables are therefore given by (Maugin, 1992),

$$(\dot{\epsilon}^{p}, -\dot{\alpha}) \in \partial \varphi^{*}(\boldsymbol{\sigma}, \boldsymbol{A}) = \begin{cases} \emptyset & \text{if } (\boldsymbol{\sigma}, \boldsymbol{A}) \notin \mathbb{E} \\ N_{\mathbb{E}}(\boldsymbol{\sigma}, \boldsymbol{A}) & \text{if } (\boldsymbol{\sigma}, \boldsymbol{A}) \in \mathbb{E} \end{cases}$$
(3.15)

where \emptyset is the empty set whereas the symbol $N_{\mathbb{E}}(\sigma, A)$ denotes the normal cone at (σ, A) to \mathbb{E} . This is a subset of the space $\dot{\Sigma}$ dual to Σ where \mathbb{E} belongs to, and is defined as

$$N_{\mathbb{E}}(\boldsymbol{\sigma},\boldsymbol{A}) = \left\{ (\dot{\boldsymbol{\epsilon}}^{p},-\dot{\boldsymbol{\alpha}}) \in \dot{\boldsymbol{\Sigma}} \middle| (\boldsymbol{\sigma}-\boldsymbol{\tau}) \colon \dot{\boldsymbol{\epsilon}}^{p} - (\boldsymbol{A}-\boldsymbol{B}) \colon \dot{\boldsymbol{\alpha}} \leq 0, \quad \forall (\boldsymbol{\tau},\boldsymbol{B}) \in \mathbb{E} \right\}$$

which is a closed convex cone in Σ .

Since Σ and Σ are finite dimensional spaces, the normal cone can be identified with the cone of the outwards normals at $(\boldsymbol{\sigma}, \boldsymbol{A})$ to \mathbb{E} , cf. Figure 3.6, which is a subset of Σ . Therefore, it is easy to conclude that there is a non zero $(\dot{\boldsymbol{\epsilon}}^p, -\dot{\boldsymbol{\alpha}})$ at each $(\boldsymbol{\sigma}, \boldsymbol{A}) \in \partial \mathbb{E}$ whereas $N_{\mathbb{E}}(\boldsymbol{\sigma}, \boldsymbol{A}) = \{(0, 0)\}$ if $(\boldsymbol{\sigma}, \boldsymbol{A}) \in \mathring{\mathbb{E}}$. In particular, then,



Figure 3.6: Normal cone to the convex set \mathbb{E}

at regular points (σ, A) of the boundary $\partial \mathbb{E}$ the normal cone $N_{\mathbb{E}}(\sigma, A)$ reduces to the one dimensional set spanned by the outward normal at (σ, A) , whereas at nonsmooth boundary points $N_{\mathbb{E}}(\sigma, A)$ is a nontrivial cone, cf. Figure 3.6.

The evolution laws given as in (3.15) are usually described as hypothesis of normal dissipation or associativity of the plastic model.

Remark 3.1. By definition of subdifferential, equation (3.15) can be interpreted as the following time inequality

Given:
$$(\sigma, A)$$

Find: $(\dot{\epsilon}^p, -\dot{\alpha})$
Such That:
 $\varphi^*(\tau, B) - \varphi^*(\sigma, A) \ge (\tau - \sigma): \dot{\epsilon}^p - (B - A): \dot{\alpha} \quad \forall (\tau, B) \in \mathbb{E}.$

$$(3.16)$$

Since $\varphi^*(\boldsymbol{\tau}, \boldsymbol{B}) = I_{\mathbb{E}}$, then (3.16) reads as

$$\boldsymbol{\sigma} \colon \dot{\boldsymbol{\epsilon}}^p - \boldsymbol{A} \colon \dot{\boldsymbol{\alpha}} \ge \boldsymbol{\tau} \colon \dot{\boldsymbol{\epsilon}}^p - \boldsymbol{B} \colon \dot{\boldsymbol{\alpha}} \quad \forall (\boldsymbol{\tau}, \boldsymbol{B}) \in \mathbb{E},$$
(3.17)

which is an extension of the principle of maximum plastic dissipation, well known in perfect plasticity (Hill, 1950), for the class of materials with hardening. \Box

Remark 3.2. For implementation purposes, one usually resorts to an equivalent formulation of (3.15) given by (Simo & Hughes, 1998; de Souza Neto *et al.*, 2002)

$$-\dot{\boldsymbol{\epsilon}}^{p} + \lambda \frac{\partial f}{\partial \boldsymbol{\sigma}}(\boldsymbol{\sigma}, \boldsymbol{A}) = 0$$

$$-\dot{\boldsymbol{\alpha}} - \lambda \frac{\partial f}{\partial \boldsymbol{\sigma}}(\boldsymbol{\sigma}, \boldsymbol{A}) = 0 \qquad (3.18)$$

$$-\alpha - \lambda \frac{\partial}{\partial A}(\sigma, A) = 0 \qquad (3.18)$$

$$\lambda \ge 0, \quad \lambda f(\boldsymbol{\sigma}, \boldsymbol{A}) = 0, \quad f(\boldsymbol{\sigma}, \boldsymbol{A}) \le 0$$
 (3.19)

which is complemented by the consistency condition

$$\lambda f(\boldsymbol{\sigma}, \boldsymbol{A}) = 0,$$

where λ is called plastic multiplier.

The equations (3.18) are often referred to as the Kuhn-Tucker conditions (Luenberger, 1984) as they express the optimality conditions for the solution of the maximum plastic dissipation principle described by (3.17).

For our subsequent developments, likewise for the state equations, the evolution laws are recast into an equivalent formulation (Germain *et al.*, 1983; Ladevèze, 1999) which exploits the convexity of the model,

$$\underbrace{\varphi(\dot{\epsilon}^{p},-\dot{\alpha})+\varphi^{*}(\boldsymbol{\sigma},\boldsymbol{A})-\boldsymbol{\sigma}:\dot{\epsilon}^{p}+\boldsymbol{A}:\dot{\alpha}}_{{}^{d}\eta^{2}_{\boldsymbol{\sigma},t}(\boldsymbol{\sigma},\boldsymbol{A};\dot{\epsilon}^{p},\dot{\alpha})}=0 \Leftrightarrow (\dot{\epsilon}^{p},-\dot{\alpha})\in\partial\varphi^{*}(\boldsymbol{\sigma},\boldsymbol{A}) \quad (3.20)$$

From the properties of the Legendre-Fenchel transform, it follows that for any state $(\sigma, \mathbf{A}; \dot{\epsilon}^p, -\dot{\alpha}) \in \Sigma^* \times \dot{\Sigma}$

$$\varphi(\dot{\epsilon}^p, \dot{\alpha}) + \varphi^*(\sigma, A) - \sigma : \dot{\epsilon}^p + A : \dot{\alpha} \ge 0,$$
(3.21)

where the equality holds if and only if $(\sigma, A; \dot{\epsilon}^p, \dot{\alpha})$ is a solution of (3.15). The proof of this equivalence follows the same pattern as for the state equations.

In summary, the constitutive model of standard generalised material is obtained by specifying: (i) by means of the Helmholtz free energy the equations of state of the material, which comprise a relationship between the current value of the independent state variables and their conjugates, and (ii) by means of the dissipation potential the evolution laws, which describe the time evolution of the independent internal variables by means of differential inclusions and confer heredity property to the material behaviour (Germain *et al.*, 1983; Besson *et al.*, 2001).

Finally, for the quasi-static process where inertial effects are neglected, the initial conditions for the variables appearing in rate form have to be given to complete the initial boundary value problem. These are given by

$$\boldsymbol{\epsilon}^{p}(\boldsymbol{x},t=0) = \boldsymbol{\epsilon}_{0}^{p}(\boldsymbol{x}) \text{ and } \boldsymbol{\alpha}(\boldsymbol{x},t=0) = \boldsymbol{\alpha}_{0}(\boldsymbol{x})$$
 (3.22)

3.2.4.5 Examples of standard rate independent plasticity models

In this section, we present two examples of standard rate independent plasticity models. The first one is a plasticity model with isotropic hardening whereas the second one is a model with isotropic-kinematic hardening.

The state laws (3.7) and the evolution laws (3.20) for the internal variables are formulated within the framework of convex analysis presented in the previous section. For the evolution laws, in particular, given the associativity of the model, it will be sufficient to specify only the elastic domain $\mathbb{E} \subseteq \Sigma$, for the dual of the dissipation potential is then the indicator function of \mathbb{E} .

Plasticity Model P1: Prandtl Reuss Model with linear elasticity

The Prandtl-Reuss plasticity model is a standard model obtained by using the Von Mises yield criterion and an isotropic hardening law (Besson *et al.*, 2001). The

internal variables are the plastic strain tensor ϵ^p , and the accumulated plastic strain p, while the conjugate variables are σ and R, respectively.

State Laws

Linear Elasticity

The model of linear elasticity is obtained by introducing the free elastic potential

$$\psi_e(\boldsymbol{\epsilon}^e) = \frac{1}{2} \mathbf{C} \boldsymbol{\epsilon}^e: \ \boldsymbol{\epsilon}^e \tag{3.23}$$

with C being a second order positive definite tensor, namely the Hooke tensor. The complementary free energy then follows as

$$\psi_e^{\star}(\boldsymbol{\sigma}) = \frac{1}{2} \mathbf{C}^{-1} \boldsymbol{\sigma} \colon \boldsymbol{\sigma}, \qquad (3.24)$$

which is defined as the Legendre transform of (3.23).

The pair (σ, ϵ^{e}) is said to define a model of linear elasticity if (see equation $(3.7)_{1}$)

$$\psi_e(\boldsymbol{\epsilon}^e) + \psi_e^{\star}(\boldsymbol{\sigma}) - \boldsymbol{\sigma} \colon \boldsymbol{\epsilon}^e = 0,$$

which, by accounting for (3.23) and (3.24), can also be written as

$$\frac{1}{2}(\boldsymbol{\sigma} - \mathbf{C}\boldsymbol{\epsilon}^{e}) \colon \mathbf{C}^{-1}(\boldsymbol{\sigma} - \mathbf{C}\boldsymbol{\epsilon}^{e}) = 0.$$

Finally the familiar Hooke law is obtained as (see equation $(3.5)_1$)

$$\boldsymbol{\sigma}-\mathbf{C}\boldsymbol{\epsilon}^{e}=\mathbf{0}.$$

Hardening Law

The hardening law is the state equation that relates the internal variables to the thermodynamically conjugate ones as a result of imposing the requirements of the second law of thermodynamics. In the following, we give the details on how to compute the Legendre–Fenchel transform of the free plastic energy for the more general isotropic hardening law. In particular, it will be concluded that the Legendre–Fenchel transform reduces to the Legendre transform.

Let g(p) be a positive and increasing scalar function of the accumulated plastic strain p with g(p = 0) = 0. The free plastic energy is defined as

$$\psi_p(p) = \int_0^p g(arepsilon) \mathrm{d}arepsilon$$

and is a strictly convex function.

The Legendre-Fenchel transform of $\psi_p(p)$ is given by

$$\psi_p^{\star}(R) = \sup_{p \ge 0} \{ Rp - \psi_p(p) \}.$$

For given $R \ge 0$, solve for \bar{p} such that

$$R\bar{p} - \psi_p(\bar{p}) = \sup_{p \ge 0} \{Rp - \psi_p(p)\}.$$

Since $F(p) = Rp - \psi_p(p)$ is differentiable and concave, \bar{p} is obtained by solving the equation $\left. \frac{\mathrm{d}F}{\mathrm{d}p} \right|_{p=\bar{p}} = 0$, i.e.,

$$R - g(\bar{p}) = 0, \tag{3.25}$$

and because of the invertibility of g(p), it follows

$$\bar{p} = g^{-1}(R).$$

Thus, with $\bar{p} \ge 0$, the Legendre–Fenchel transform of $\psi_p(p)$ is obtained as follows

$$\psi_p^{\star}(R) = F(\bar{p}) = Rg^{-1}(R) - \psi_p(g^{-1}(R)),$$

which is the Legendre transform of $\psi_p(p)$. <u>Example 3.2.1</u>: Linear Hardening Let

$$g(p) = \mathsf{H}p; \quad \psi_p(p) = \frac{1}{2}\mathsf{H}p^2.$$

Then equation (3.25) writes as

$$R - \mathsf{H}\bar{p} = 0; \quad \bar{p} = \frac{R}{\mathsf{H}}.$$

Hence,

$$\psi_p^{\star}(R) = R\frac{R}{\mathsf{H}} - \frac{1}{2}\mathsf{H}\left(\frac{R}{\mathsf{H}}\right)^2 = \frac{1}{2\mathsf{H}}R^2.$$

Thus, in summary, the hardening law can be defined in one of the following equivalent expressions, $(3.7)_2$ and $(3.5)_2$,

Given:
$$\psi_p(p) = \frac{1}{2} Hp^2$$
; $\psi_p^*(R) = \frac{1}{2H}R^2$
 $\left| \begin{array}{l} \psi_p(p) + \psi_p^*(R) - Rp = 0 \Leftrightarrow \\ \frac{1}{2H}(R - Hp)^2 = 0 \Leftrightarrow R - Hp = 0. \end{array} \right|$

Evolution Laws

Given the closed convex elastic domain,

$$\mathbb{E} = \{(\boldsymbol{\sigma}, R) | \|\boldsymbol{\sigma}_D\| - (R + R_0) \le 0, \quad R \ge 0\},\$$

the dual of the dissipation potential for this standard model is the indicator function of \mathbb{E} ,

$$\varphi^{\star}(\boldsymbol{\sigma},R)=I_{\mathbb{E}}.$$

The Legendre-Fenchel transform of $\varphi^*(\sigma, R)$, which represents the dissipation potential, is therefore the support function of \mathbb{E} defined as

$$\varphi(\dot{\epsilon}^p, -\dot{p}) = \sup_{(\boldsymbol{\sigma}, R) \in \mathbb{E}} \{ \boldsymbol{\sigma} \colon \dot{\epsilon}^p - R\dot{p} \},\$$

which can be transformed in the following form (Ladevèze, 1999; Han & Reddy, 1999)

$$\varphi(\dot{\boldsymbol{\epsilon}}^p, -\dot{p}) = R_0 \| \dot{\boldsymbol{\epsilon}}^p \| + I_{\mathbb{C}}$$
(3.26)

with $I_{\mathbb{C}}$ being the indicator function of the following closed convex set

$$\mathbb{C} = \{ (\dot{\epsilon}^p, -\dot{p}) \middle| \| \dot{\epsilon}^p \| - \dot{p} \le 0 \text{ and } \operatorname{Tr}[\dot{\epsilon}^p] = 0 \}.$$

Proof. Consider

$$\sup_{(\boldsymbol{\sigma},R)\in\mathbb{E}}\left\{\boldsymbol{\sigma}:\dot{\boldsymbol{\epsilon}}^{p}-R\dot{p}\right\}=\sup_{\substack{\|\boldsymbol{\sigma}_{D}\|-(R_{0}+R)\leq0\\R\geq0}}\left\{\boldsymbol{\sigma}_{D}:\dot{\boldsymbol{\epsilon}}^{p}+\frac{1}{3}\mathrm{Tr}[\boldsymbol{\sigma}]\mathrm{Tr}[\dot{\boldsymbol{\epsilon}}^{p}]-R\dot{p}\right\}$$

Using the Schwarz's inequality and noting that the constraint involves only the norm of $\|\sigma_D\|$ and not its direction, in the search for the supremum we can equivalently consider only the stress tensors σ such that their deviatoric part maximizes σ_D : $\dot{\epsilon}^p$, that is,

$$\sup_{\substack{\|\boldsymbol{\sigma}_D\| - (R_0 + R) \leq 0\\R \geq 0}} \left\{ \boldsymbol{\sigma}_D \colon \dot{\boldsymbol{\epsilon}}^p + \frac{1}{3} \operatorname{Tr}[\boldsymbol{\sigma}] \operatorname{Tr}[\dot{\boldsymbol{\epsilon}}^p] - R\dot{p} \right\} =$$
$$= \sup_{\substack{\|\boldsymbol{\sigma}_D\| - (R_0 + R) \leq 0\\R > 0}} \left\{ \|\boldsymbol{\sigma}_D\| \| \dot{\boldsymbol{\epsilon}}^p\| + \frac{1}{3} \operatorname{Tr}[\boldsymbol{\sigma}] \operatorname{Tr}[\dot{\boldsymbol{\epsilon}}^p] - R\dot{p} \right\}$$

Let

$$f = \|\boldsymbol{\sigma}_D\| - (R_0 + R)$$

with the constraint that $f \leq 0$, it is also

$$\sup_{\substack{\|\boldsymbol{\sigma}_{D}\| - (R_{0} + R) \leq 0 \\ R \geq 0}} \left\{ \|\boldsymbol{\sigma}_{D}\| \| \dot{\boldsymbol{\epsilon}}^{p}\| + \frac{1}{3} \operatorname{Tr}[\boldsymbol{\sigma}] \operatorname{Tr}[\dot{\boldsymbol{\epsilon}}^{p}] - R\dot{p} \right\} =$$
$$= \sup_{\substack{\|\boldsymbol{\sigma}_{D}\| - (R_{0} + R) \leq 0 \\ R > 0}} \left\{ R(\|\dot{\boldsymbol{\epsilon}}^{p}\| - \dot{p}) + f\| \dot{\boldsymbol{\epsilon}}^{p}\| + R_{0} \| \dot{\boldsymbol{\epsilon}}^{p}\| + \frac{1}{3} \operatorname{Tr}[\boldsymbol{\sigma}] \operatorname{Tr}[\dot{\boldsymbol{\epsilon}}^{p}] \right\}$$

and by using simple properties of the supremum (Hiriart-Urruty & Lemaréchal, 2001) it follows also

$$\varphi(\dot{\epsilon}^p, -\dot{p}) = R_0 \|\dot{\epsilon}^p\| + \sup_{f \le 0} f \|\dot{\epsilon}^p\| + \sup_{R \ge 0} R(\|\dot{\epsilon}^p\| - \dot{p}) + \sup_{\operatorname{Tr}[\boldsymbol{\sigma}] \in \mathbb{R}} \frac{1}{3} \operatorname{Tr}[\boldsymbol{\sigma}] \operatorname{Tr}[\dot{\epsilon}^p],$$

that is,

$$\begin{split} \sup_{f \leq 0} f \|\dot{\epsilon}^p\| &= 0, \\ \sup_{R \geq 0} R(\|\dot{\epsilon}^p\| - \dot{p}) &= \begin{cases} 0 & \text{if } \|\dot{\epsilon}^p\| - \dot{p} \leq 0 \\ \infty & \text{if } \|\dot{\epsilon}^p\| - \dot{p} > 0 \end{cases}, \\ \sup_{\mathrm{Tr}[\sigma] \in \mathbb{R}} \frac{1}{3} \mathrm{Tr}[\sigma] \mathrm{Tr}[\dot{\epsilon}^p] &= \begin{cases} 0 & \text{if } \mathrm{Tr}[\dot{\epsilon}^p] = 0 \\ \infty & \text{if } \mathrm{Tr}[\dot{\epsilon}^p] \neq 0 \end{cases}, \end{split}$$

which finally delivers (3.26).

Thus, we say that the set $(\sigma(\mathbf{x}, t), R(\mathbf{x}, t); \epsilon^p(\mathbf{x}, t), p(\mathbf{x}, t))$ satisfies the evolution law at the point $\mathbf{x} \in \Omega$ if the following differential problem is satisfied

$$\forall t \leq T, \begin{cases} \|\boldsymbol{\sigma}^{D}(\boldsymbol{x}, t)\| - [R_{0} + R(\boldsymbol{x}, t)] \leq 0 \\ \|\dot{\boldsymbol{\epsilon}}^{p}(\boldsymbol{x}, t)\| - \dot{p}(\boldsymbol{x}, t) \leq 0 \\ \operatorname{Tr}[\dot{\boldsymbol{\epsilon}}^{p}(\boldsymbol{x}, t)] = 0 \\ R_{0}\|\dot{\boldsymbol{\epsilon}}^{p}(\boldsymbol{x}, t)\| - \boldsymbol{\sigma}(\boldsymbol{x}, t) \colon \dot{\boldsymbol{\epsilon}}^{p}(\boldsymbol{x}, t) + R(\boldsymbol{x}, t)\dot{p}(\boldsymbol{x}, t) = 0 \end{cases}$$

Plasticity Model P2: Standard Marquis-Chaboche modified model with Linear Elasticity

The non-associated Marquis-Chaboche plasticity model has been modified into an associated model by Ladèveze & Rougee (1984) and Coffignal (1987). The model accounts for both isotropic and kinematic hardening. The internal variables are, therefore, the plastic strain tensor $\boldsymbol{\epsilon}^p$, the accumulated plastic strain p and a second order symmetric tensor $\boldsymbol{\alpha}$ with zero trace. The conjugate variables are $\boldsymbol{\sigma}$, R, \boldsymbol{X} , respectively. The position of the elastic domain in the space of the generalised stresses is controlled by the internal variable \boldsymbol{X} whereas its amplitude by R.

State Laws

The free elastic and complementary potential are, as usual, given by

$$\psi_e(\epsilon^e) = rac{1}{2} \mathbf{C} \epsilon^e : \ \epsilon^e;$$

 $\psi_e^{\star}(\boldsymbol{\sigma}) = rac{1}{2} \mathbf{C}^{-1} \boldsymbol{\sigma} : \ \boldsymbol{\sigma},$

whereas the free plastic potential and its Legendre transform are

$$\psi_p(\boldsymbol{\alpha}, p) = \frac{1}{2} \boldsymbol{\Lambda} \boldsymbol{\alpha} \colon \boldsymbol{\alpha} + \int_0^p g(\varepsilon) d\varepsilon;$$

$$\psi_p^{\star}(\boldsymbol{X}, R) = \frac{1}{2} \boldsymbol{\Lambda}^{-1} \boldsymbol{X} \colon \boldsymbol{X} + [Rg^{-1}(R) - \int_0^{g^{-1}(R)} g(\varepsilon) d\varepsilon],$$

where Λ is a second order positive definite tensor, and g(p) is the same as defined for the previous plasticity model P1.

The state equations can therefore be equivalently expressed as follows

$$\psi_{e}(\epsilon^{e}) + \psi_{e}^{\star}(\sigma) - \sigma: \ \epsilon^{e} = 0 \Leftrightarrow \sigma - \mathbf{C}\epsilon^{e} = \mathbf{0}$$
$$\psi_{p}(\alpha, p) + \psi_{p}^{\star}(\mathbf{X}, R) - \{\mathbf{X}: \ \alpha + Rp\} = 0 \Leftrightarrow \begin{vmatrix} R - g(p) = 0 \\ \mathbf{X} - \mathbf{A}\alpha = \mathbf{0} \end{vmatrix}$$

Evolution Laws

The closed convex elastic domain of this model is defined as follows

$$\mathbb{E} = \{(\boldsymbol{\sigma}, \boldsymbol{X}, R) | \|\boldsymbol{\sigma}_D - \boldsymbol{X}\| + \frac{a}{2c} \|\boldsymbol{X}\|^2 - (R + R_0) \le 0, \quad R \ge 0\},\$$

where $a \ge 0, c > 0$ are material constants. For the standard formulation of the evolution laws, the dual of the dissipation potential is obtained by setting

$$\varphi^{\star}(\boldsymbol{\sigma}, \boldsymbol{X}, R) = I_{\mathbb{E}}.$$

The Legendre-Fenchel transform of $\varphi^{\star}(\sigma, X, R)$ is therefore the support function of the domain \mathbb{E} ,

$$\varphi(\dot{\epsilon}^{p}, -\dot{\alpha}, -\dot{p}) = \sup_{(\boldsymbol{\sigma}, R, \boldsymbol{X}) \in \mathbb{E}} \left\{ \boldsymbol{\sigma} : \dot{\epsilon}^{p} - R\dot{p} - \boldsymbol{X} : \dot{\boldsymbol{\alpha}} \right\}$$
(3.27)

,

which can be given by the following explicit expression

$$\varphi(\dot{\epsilon}^{p}, -\dot{\alpha}, -\dot{p}) = \begin{cases} R_{0} \|\dot{\epsilon}^{p}\| + \frac{c}{2a} \frac{\|\dot{\epsilon}^{p} - \dot{\alpha}\|^{2}}{\|\dot{\epsilon}^{p}\|} + \psi_{\mathbb{C}} \text{ if } \dot{\epsilon}^{p} \neq \mathbf{0}, \\ I_{\mathbb{C}} & \text{ if } \dot{\epsilon}^{p} = \mathbf{0}, \, \dot{\alpha} = \mathbf{0}, \\ \infty & \text{ if } \dot{\epsilon}^{p} = \mathbf{0}, \, \dot{\alpha} \neq \mathbf{0}, \end{cases}$$
(3.28)

with the domain \mathbb{C} given in the Prandtl-Reuss plasticity model P1.

Proof. Following arguments similar to the one shown in the Prandtl-Reuss model, we obtain

$$\varphi(\dot{\boldsymbol{\epsilon}}^{p},-\dot{\boldsymbol{\alpha}},-\dot{p}) = \sup_{(\boldsymbol{\sigma},R,\boldsymbol{X})\in\mathbb{E}} \left\{\boldsymbol{\sigma}:\dot{\boldsymbol{\epsilon}}^{p}-R\dot{p}-\boldsymbol{X}:\dot{\boldsymbol{\alpha}}\right\} =$$
$$= \sup_{(\boldsymbol{\sigma},R,\boldsymbol{X})\in\mathbb{E}} \left\{(\boldsymbol{\sigma}_{D}-\boldsymbol{X}+\boldsymbol{X}):\dot{\boldsymbol{\epsilon}}^{p}+\frac{1}{3}\mathrm{Tr}[\boldsymbol{\sigma}]\mathrm{Tr}[\dot{\boldsymbol{\epsilon}}^{p}]-R\dot{p}-\boldsymbol{X}:\dot{\boldsymbol{\alpha}}\right\} =$$

$$= \sup_{(\boldsymbol{\sigma}, R, \boldsymbol{X}) \in \mathbb{E}} \left\{ (\boldsymbol{\sigma}_D - \boldsymbol{X}) : \dot{\boldsymbol{\epsilon}}^p + \frac{1}{3} \operatorname{Tr}[\boldsymbol{\sigma}] \operatorname{Tr}[\dot{\boldsymbol{\epsilon}}^p] - R\dot{p} + \boldsymbol{X} : (\dot{\boldsymbol{\epsilon}}^p - \dot{\boldsymbol{\alpha}}) \right\} =$$

$$= \sup_{(\boldsymbol{\sigma}, R, \boldsymbol{X}) \in \mathbb{E}} \left\{ \|\boldsymbol{\sigma}_D - \boldsymbol{X}\| \| \dot{\boldsymbol{\epsilon}}^p \| + \frac{1}{3} \operatorname{Tr}[\boldsymbol{\sigma}] \operatorname{Tr}[\dot{\boldsymbol{\epsilon}}^p] - R\dot{p} + \|\boldsymbol{X}\| \| \dot{\boldsymbol{\epsilon}}^p - \dot{\boldsymbol{\alpha}}\| \right\} =$$

$$= \sup_{\substack{R \ge 0, \|\boldsymbol{X}\| \ge 0\\ \operatorname{Tr}[\boldsymbol{\sigma}] \in \mathbb{R}, f \le 0}} \left\{ R(\| \dot{\boldsymbol{\epsilon}}^p \| - \dot{p}) + f\| \dot{\boldsymbol{\epsilon}}^p \| + R_0 \| \dot{\boldsymbol{\epsilon}}^p \| + \frac{1}{3} \operatorname{Tr}[\boldsymbol{\sigma}] \operatorname{Tr}[\dot{\boldsymbol{\epsilon}}^p] + \|\boldsymbol{X}\| \| \dot{\boldsymbol{\epsilon}}^p - \dot{\boldsymbol{\alpha}}\| \right\}$$

where we have let

$$f = \|\boldsymbol{\sigma}_D - \boldsymbol{X}\| + \frac{a}{2c} \|\boldsymbol{X}\|^2 - (R + R_0)$$

and made use of the Schwarz's inequality.

Since the variables may vary independently on each other, it follows

$$\begin{aligned} \varphi(\dot{\boldsymbol{\epsilon}}^{p},-\dot{\boldsymbol{\alpha}},-\dot{p}) &= R_{0} \|\dot{\boldsymbol{\epsilon}}^{p}\| + \sup_{R\geq 0} \left\{ R(\|\dot{\boldsymbol{\epsilon}}^{p}\|-\dot{p}) \right\} + \sup_{\mathrm{Tr}[\boldsymbol{\sigma}]\in\mathbb{R}} \frac{1}{3} \mathrm{Tr}[\boldsymbol{\sigma}] \mathrm{Tr}[\dot{\boldsymbol{\epsilon}}^{p}] + \\ + \sup_{f\leq 0} f \|\dot{\boldsymbol{\epsilon}}^{p}\| + \sup_{\|\boldsymbol{X}\|\geq 0} \left\{ \|\boldsymbol{X}\|\|\dot{\boldsymbol{\epsilon}}^{p}-\dot{\boldsymbol{\alpha}}\| - \frac{a}{2c} \|\boldsymbol{X}\|^{2} \|\dot{\boldsymbol{\epsilon}}^{p}\| \right\}. \end{aligned}$$

As for the last term, we note

$$\sup_{\|\boldsymbol{X}\|\geq 0} \left\{ \|\boldsymbol{X}\| \|\dot{\boldsymbol{\epsilon}}^{p} - \dot{\boldsymbol{\alpha}}\| - \frac{a}{2c} \|\boldsymbol{X}\|^{2} \|\dot{\boldsymbol{\epsilon}}^{p}\| \right\} = \begin{cases} \frac{c}{2a} \frac{\|\dot{\boldsymbol{\epsilon}}^{p} - \dot{\boldsymbol{\alpha}}\|^{2}}{\|\dot{\boldsymbol{\epsilon}}^{p}\|} & \text{if } \dot{\boldsymbol{\epsilon}}^{p} \neq \boldsymbol{0}, \\ I_{\mathbb{C}} & \text{if } \dot{\boldsymbol{\epsilon}}^{p} = \boldsymbol{0}, \, \dot{\boldsymbol{\alpha}} = \boldsymbol{0}, \\ \infty & \text{if } \dot{\boldsymbol{\epsilon}}^{p} = \boldsymbol{0}, \, \dot{\boldsymbol{\alpha}} \neq \boldsymbol{0}, \end{cases}$$

whereas for the other terms the same considerations as expressed for the Prandtl-Reuss model hold, so that finally, the expression (3.28) is obtained.

3.3 Contractivity of the elastoplastic flow

Object of this section is to show the contractivity of the elastoplastic flow for standard generalised materials described by a quadratic Helmholtz energy. For the more general case of strongly convex energy, we refer to Laborde & Nguyen (1990) where, however, the Lipschitz property of the elastoplastic flow with respect to the initial data is shown to hold.
Denote by $(\sigma, A; \epsilon, \epsilon^p, \alpha)$ the solution of the following problem

$$egin{aligned} &\langle \pmb{\sigma},
abla \pmb{\eta}
angle = \langle \pmb{b}, \pmb{\eta}
angle + \langle \pmb{t}, \pmb{\eta}
angle_{\partial \Omega_t} & orall \pmb{\eta} \in \mathcal{V}_0 \ &m{\epsilon} = \pmb{\epsilon}^e + \pmb{\epsilon}^p \ &m{\sigma} = \pmb{\mathsf{C}} \pmb{\epsilon}^e \ &m{A} = \pmb{\mathsf{H}} \pmb{lpha} \ &(\dot{\pmb{\epsilon}}^p, - \dot{\pmb{lpha}}) \in \partial arphi^\star(\pmb{\sigma}, \pmb{A}) \ &m{\epsilon}^p(t=0) = \pmb{\epsilon}_0^p \ &m{lpha}(t=0) = \pmb{lpha}_0 \end{aligned}$$

and by $(\tilde{\boldsymbol{\sigma}}, \tilde{\boldsymbol{A}}; \tilde{\boldsymbol{\epsilon}}, \tilde{\boldsymbol{\epsilon}}^p, \tilde{\boldsymbol{\alpha}})$ the solution of the same problem but with initial state given by $\tilde{\boldsymbol{\epsilon}}_0^p, \tilde{\boldsymbol{\alpha}}_0$. The monotony of the operator $\partial \varphi^*$, which is the subdifferential of a convex function (Rockafellar, 1970; Ekeland & Temam, 1976; Hiriart-Urruty & Lemaréchal, 2001), gives

$$(\boldsymbol{\sigma} - \tilde{\boldsymbol{\sigma}}) : (\dot{\boldsymbol{\epsilon}}^p - \dot{\tilde{\boldsymbol{\epsilon}}}^p) - (\boldsymbol{A} - \tilde{\boldsymbol{A}}) : (\dot{\boldsymbol{\alpha}} - \dot{\tilde{\boldsymbol{\alpha}}}) \ge 0,$$

that is,

$$(\boldsymbol{\sigma} - \tilde{\boldsymbol{\sigma}}) : (\dot{\boldsymbol{\epsilon}}^e - \tilde{\boldsymbol{\epsilon}}^e) + (\boldsymbol{A} - \boldsymbol{A}) : (\dot{\boldsymbol{\alpha}} - \tilde{\boldsymbol{\alpha}}) \le 0,$$
(3.29)

where we have accounted for the additivity of the total strain and the equality

$$(\boldsymbol{\sigma}-\tilde{\boldsymbol{\sigma}}):(\dot{\boldsymbol{\epsilon}}-\dot{\tilde{\boldsymbol{\epsilon}}})=0,$$

since $\sigma - \tilde{\sigma}$ is self equilibrated and $\epsilon - \tilde{\epsilon}$ corresponds to the difference of two kinematically admissible displacement fields for the same problem.

By accounting for the state equations in (3.29), it follows

$$\frac{\mathrm{d}}{\mathrm{d}t} \Big[\mathbf{C} (\boldsymbol{\epsilon}^{\boldsymbol{e}} - \tilde{\boldsymbol{\epsilon}}^{\boldsymbol{e}}) \colon (\boldsymbol{\epsilon}^{\boldsymbol{e}} - \tilde{\boldsymbol{\epsilon}}^{\boldsymbol{e}}) + \mathbf{H} (\boldsymbol{\alpha} - \tilde{\boldsymbol{\alpha}}) \colon (\boldsymbol{\alpha} - \tilde{\boldsymbol{\alpha}}) \Big] \leq 0,$$

which shows that the time dependent function

$$\mathbf{C}(\boldsymbol{\epsilon}^{e}(t) - \tilde{\boldsymbol{\epsilon}}^{e}(t)) \colon (\boldsymbol{\epsilon}^{e}(t) - \tilde{\boldsymbol{\epsilon}}^{e}(t)) + \mathbf{H}(\boldsymbol{\alpha}(t) - \tilde{\boldsymbol{\alpha}}(t)) \colon (\boldsymbol{\alpha}(t) - \tilde{\boldsymbol{\alpha}}(t))$$

is nonincreasing with time, thus

$$\begin{aligned} \mathbf{C}(\boldsymbol{\epsilon}^{e}(t) - \tilde{\boldsymbol{\epsilon}}^{e}(t)) &: (\boldsymbol{\epsilon}^{e}(t) - \tilde{\boldsymbol{\epsilon}}^{e}(t)) + \mathbf{H}(\boldsymbol{\alpha}(t) - \tilde{\boldsymbol{\alpha}}(t)) : (\boldsymbol{\alpha}(t) - \tilde{\boldsymbol{\alpha}}(t)) \leq \\ &\leq \mathbf{C}(\boldsymbol{\epsilon}_{0}^{e} - \tilde{\boldsymbol{\epsilon}}_{0}^{e}) : (\boldsymbol{\epsilon}_{0}^{e} - \tilde{\boldsymbol{\epsilon}}_{0}^{e}) + \mathbf{H}(\boldsymbol{\alpha}_{0} - \tilde{\boldsymbol{\alpha}}_{0}) : (\boldsymbol{\alpha}_{0} - \tilde{\boldsymbol{\alpha}}_{0}) \quad \forall t \geq 0, \end{aligned}$$

which expresses the contractivity of the elastoplastic flow with respect to the norm associated with the Helmholtz energy. In the following section, we show how this property allows the use of the time accumulated residual as indication of the error in solution.

3.4 Residual versus Error in Solution

For the considerations in this section, we basically follow Eriksson et al. (1996), which we refer to for further details.

Consider the scalar initial value problem,

$$\begin{vmatrix} \dot{u}(t) + a(t)u(t) = f(t) & t \in [0, T] \\ u(0) = u_0 \end{aligned}$$
(3.30)

with $a(t) \geq 0$.

The solution of (3.30) is given by

$$u(t) = \exp[-A(t)]u_0 + \int_0^t \exp[-(A(t) - A(\tau))]f(\tau) \,\mathrm{d}\tau$$
(3.31)

where $A(t) = \int_0^t a(\tau) d\tau$, so that the following *a priori* estimate can be easily obtained

$$|u(t)| \le |u_0| + \int_0^t |f(\tau)| \, \mathrm{d}\tau \quad \forall t \le T,$$
(3.32)

for the nondecreasing character of A = A(t).

Let $0 = t_1 < \ldots < t_n < \ldots < t_{N+1} = T$ be a partition of the time interval [0, T] of interest and consider a function U = U(t) to be approximation of the problem (3.30), which is differentiable over the intervals $[t_n, t_{n+1}]$. The function U = U(t) may have jump discontinuities at the time instants t_n , thus we let $U(t_n^+) - U(t_n^-) = \Delta_n$. For n = 1, we assume $U(t_1^-) = u_0$, thus $\Delta_1 = U_0 - u_0$. This means that U = U(t) is solution of the following problem

For
$$n = 1, ..., N$$

 $\dot{U}(t) + a(t)U(t) = f(t) + R(t)$ $t \in [t_n, t_{n+1}]$
 $U(t_n^+) = U(t_n^-) + \Delta_n$

where R = R(t) is the residual produced by U = U(t) within each time interval $[t_n, t_{n+1}]$ where U = U(t) is differentiable.

The error e(t) = u(t) - U(t) associated with the approximation U = U(t) is, therefore, solution of the following problem

For
$$n = 1, ..., N$$

 $\dot{e}(t) + a(t)e(t) = R(t)$ $t \in [t_n, t_{n+1}]$ (3.33)
 $e(t_n) = \Delta_n.$

Using for each subinterval $[t_n, t_{n+1}]$ the result given in (3.31), we obtain

$$e(t) = \sum_{n=1}^{N} \exp[-A(t-t_n)] \Delta_n \beta_n + \int_0^t \exp[-(A(t) - A(\tau))] R(\tau) \, \mathrm{d}\tau \tag{3.34}$$

where

$$\beta_n = \begin{cases} 0 & \text{if } t \le t_n \\ 1 & \text{if } t > t_n. \end{cases}$$

In equation (3.34) the term

$$\exp[-A(t-t_n)]\Delta_n\beta_n$$

gives the propagation at $t(\geq t_n)$ of the discontinuity jump Δ_n in the approximate solution U = U(t), whereas

$$\int_0^t \exp[-(A(t) - A(\tau))]R(\tau) \mathrm{d}\tau$$

can be interpreted as the sum of the time-elemental contributions to the total error at the time t. The time-elemental contributions are obtained by the propagation at time t of the residual error $R(\tau)d\tau$ produced within the time-elemental interval $[\tau, \tau + d\tau]$ at time $\tau \leq t$.

Remark 3.3. Equation (3.34) shows the influence of the jump discontinuities on the error. Also, note that for a continuous approximation solution U = U(t), that is, $\Delta_n = 0$, for $n = 1, \ldots, N$, the error depends only on the residual produced within the time intervals where the approximation is differentiable.

Applying (3.32) and the triangular inequality, we obtain the following a priori estimate of the solution (3.34),

$$|e(t)| \le \sum_{n=1}^{N} |\Delta_n| \beta_n + \int_0^t |R(\tau)| \, \mathrm{d}\tau \quad \forall t \le T,$$
(3.35)

which shows the accumulation in time of the jump discontinuities and of the residual as indication of the pointwise error. From (3.35), it is immediate to obtain also the following global estimate in time,

$$\sup_{t \le T} |e(t)| \le \sum_{n=1}^{N} |\Delta_n| + \int_0^T |R(\tau)| \, \mathrm{d}\tau.$$
(3.36)

Remark 3.4. If R(t) = 0, $\forall t \in [t_n, t_{n+1}]$, $\forall n$, that is, the approximate solution U(t) does satisfy exactly equation (3.30) over each time interval $[t_n, t_{n+1}]$, the second term on the r.h.s. of equation (3.34) disappears and the error is due to the occurrence of the jumps in U = U(t) across the time nodes t_n . Finally, this means that the error of the approximate solution U = U(t) is related to the error in the initial data over each time interval.

The property of problem (3.30), which has allowed development of an estimate in terms of the jumps and of the residual such as (3.36), is its dissipativity resulting from $a(t) \ge 0$. Dissipativity is represented by the contraction of the solution with respect to the initial state, i.e.,

$$\left|u(t)-\tilde{U}(t)\right|\leq\left|u_{0}-\tilde{U}_{0}\right|$$

where $\tilde{U} = \tilde{U}(t)$ denotes now a solution of problem (3.30) with the same f = f(t) but different initial state \tilde{U}_0 .

The contractivity of the elastoplastic flow is enjoyed by the standard generalised materials described by a quadratic Helmholtz energy as shown in the previous section. This, therefore, suggests the use of an error estimate (3.36) for the aforementioned class of problems.

Remark 3.5. The link between residual and error is well known in linear algebra. The residual of an approximate solution of the linear system, Ax = b, influences the error in solution not only by the residual size but also by the condition number of the system matrix. For those problems which have a condition number not greater than one the residual of the approximate solution can be assumed as an estimate of its error (Stewart, 1973; Estep *et al.*, 2000). In fact, we have the following result (Stewart, 1973)

Theorem. Let A be nonsingular, $Ax = b \neq 0$, and $A\tilde{x} = b + r$. Then

$$\frac{\|\boldsymbol{x} - \tilde{\boldsymbol{x}}\|}{\|\boldsymbol{x}\|} \leq \kappa(\boldsymbol{A}) \frac{\|\boldsymbol{r}\|}{\|\boldsymbol{b}\|}$$

where $\kappa(A) = ||A|| ||A^{-1}||$ is the condition number of A and r is the residual associated with the approximation \tilde{x} of the linear system Ax = b.

3.4.1 A simple *a posteriori* error estimate via discrete energy dissipation

In this Section we develop a simple *a posteriori* error estimate of the discretization error obtained by solving an ordinary differential equation of the first order with the backward Euler method. This technique extends the one developed by Nochetto *et al.* (2000) for *a posteriori* error estimation of the backward Euler approximations of abstract evolution equations in Hilbert space. In Nochetto *et al.* (2000) it is assumed that $U_n^- = U_n^+$, whereas in what follows, this hypothesis is removed. The meaning of the notation is kept the same as in the previous Section.

The theory is presented for the model problem

Find:
$$u = u(t)$$

 $\dot{u}(t) + \mathcal{F}(u) = 0$ (3.37)
 $u(t = 0) = u_0$

where $\mathcal{F}(u)$ is assumed to admit a convex potential $\phi = \phi(u)$, that is,

$$\mathcal{F}(u) = \frac{\mathrm{d}}{\mathrm{d}u}\phi(u). \tag{3.38}$$

For the problem (3.30), for example, $\phi(u) = \frac{1}{2}au^2$.

The use of the backward Euler method as time discretization scheme of (3.37) gives

For
$$n = 1, ..., N$$

Data: U_n^+
Find: U_{n+1}^-
 $\frac{U_{n+1}^- - U_n^+}{k_n} + \mathcal{F}(U_{n+1}^-) = 0$
(3.39)

where $k_{n+1} = t_{n+1} - t_n$ and the data U_n^+ of the algebraic equation relative to the time step $[t_n^+, t_{n+1}^-]$ is assumed to be different from the solution U_n^- of the previous time step.

From the definition of differential given in note (\dagger) of Section 3.2.4.3, problem (3.37) is equivalent to the following evolution variational inequality,

Find:
$$u = u(t)$$

 $\langle \dot{u}(t), u(t) - v(t) \rangle + \phi(u(t)) - \phi(v(t)) \le 0, \quad \forall v(t) \in \mathcal{C}^0(0, T)$

$$(3.40)$$

where $C^{0}(0,T)$ is the space of the continuous functions over [0,T]. Hence, the problem (3.39) is equivalent to the inequality

For
$$n = 1, ..., N$$

Data: U_n^+
Find: U_{n+1}^-
 $\frac{1}{k_{n+1}} \langle U_{n+1}^- - U_n^+, U_{n+1}^- - v \rangle + \phi(U_{n+1}^-) - \phi(v) \le 0, \quad \forall v \in \mathbb{R}.$
(3.41)

Hereafter, U = U(t) is the discontinuous function obtained as piecewise linear interpolant of the values U_n^+ and U_{n+1}^- over each time step $[t_n, t_{n+1}]$ for n = 1, ..., N.

The error estimate given in Nochetto *et al.* (2000) is related to the amount of energy dissipation associated with U = U(t) and it is obtained by exploiting the Lyapunov properties of $\phi = \phi(t)$, which decreases along solution paths of both (3.37) and (3.39). In fact, it is known that ϕ satisfies the energy identity (Eriksson *et al.*, 1996)

$$|\dot{u}(t)|^2 + \frac{\mathrm{d}}{\mathrm{d}t}\phi(u(t)) = 0, \quad \text{a.e. } t \in [0, T]$$
 (3.42)

and the discrete energy inequality

$$-\mathcal{E}_{n+1} \stackrel{\text{def}}{=} \left| \frac{U_{n+1}^{-} - U_{n}^{+}}{k_{n+1}} \right|^{2} + \frac{\phi(U_{n+1}^{-}) - \phi(U_{n}^{+})}{k_{n+1}} \le 0, \quad \forall 0 \le n \le N,$$
(3.43)

which follows directly from (3.41) upon choosing $v = U_n^+$. The discrete quantity \mathcal{E}_{n+1} in (3.43) is thus a measure of the residual produced by U = U(t) in the energy equation (3.42).

An a posteriori error bound for the solution (3.41) can be computed as follows

$$\max_{t \in [0, t_{N+1}]} |u(t) - U(t)| \le \sum_{n=0}^{N} k_{n+1} \sqrt{\mathcal{E}_{n+1}} + \sum_{n=1}^{N} |U_n^- - U_n^+| + |U_0 - u_0|, \quad (3.44)$$

where the estimate depends on both \mathcal{E}_{n+1} and $|U_n^- - U_n^+|$, with the latter vanishing in the classical use of the backward Euler.

Proof. The linear interpolant U = U(t) over $[t_n, t_{n+1}]$ is given by

$$U(t) = \frac{t - t_n}{k_{n+1}} U_{n+1}^- + \frac{t_{n+1} - t}{k_{n+1}} U_n^+.$$

Thus,

$$\dot{U}(t) = \frac{U_{n+1}^- - U_n^+}{k_{n+1}}, \quad \forall t \in [t_n, t_{n+1}].$$

As a result, it follows

$$\frac{1}{k_{n+1}} \langle U_{n+1}^{-} - U_{n}^{+}, U_{n+1}^{-} - v \rangle = \langle \dot{U}, U - v \rangle - \langle \dot{U}, U - U_{n+1}^{-} \rangle.$$

Hence, equation (3.41) can be re-written as

$$\langle \dot{U}, U - v \rangle + \phi(U) - \phi(v) \leq \underbrace{\langle \dot{U}, U - U_{n+1}^- \rangle + \phi(U) - \phi(U_{n+1}^-)}_{\mathcal{R}}$$

which resembles equation (3.40). The next step is to estimate \mathcal{R} . With this regard, note that

$$U - U_{n+1}^{-} = (t - t_{n+1})U$$

and for the convexity of ϕ , it follows

$$\phi(U) \leq \frac{t - t_n}{k_{n+1}} \phi(U_{n+1}^-) + \frac{t_{n+1} - t}{k_{n+1}} \phi(U_n^+).$$

Hence,

$$\phi(U) - \phi(U_{n+1}^{-}) \le (t - t_{n+1}) \frac{\phi(U_{n+1}^{-}) - \phi(U_{n}^{+})}{k_{n+1}}.$$

Thus,

$$\mathcal{R} \le (t - t_{n+1}) \langle \dot{U}, \dot{U} \rangle + (t - t_{n+1}) \frac{\phi(U_{n+1}^-) - \phi(U_n^+)}{k_{n+1}} = (t - t_{n+1}) \mathcal{E}_{n+1},$$

where \mathcal{E}_{n+1} has been defined in (3.43). Therefore, equation (3.41) can be written as follows

$$\langle U, U - v \rangle + \phi(U) - \phi(v) \le (t - t_{n+1})\mathcal{E}_{n+1}, \quad \forall v \in \mathbb{R}.$$

For v = U, one obtains

$$\langle U, U - u \rangle + \phi(U) - \phi(u) \le (t - t_{n+1})\mathcal{E}_{n+1},$$

٦

whereas equation (3.40) for v = U gives

$$\langle \dot{u}, u - U \rangle + \phi(u) - \phi(U) \leq 0.$$

Summing up the last two inequalities delivers

$$\langle \dot{u} - \dot{U}, u - U \rangle \le (t - t_{n+1}) \mathcal{E}_{n+1}, \text{ a.e. } t \in [t_n, t_{n+1}],$$

which can also be written as

$$\frac{1}{2}\frac{d}{dt}|u(t) - U(t)|^2 \le (t - t_{n+1})\mathcal{E}_{n+1}, \quad \text{a.e. } t \in [t_n, t_{n+1}].$$

Integration over the time step under consideration delivers

$$|u(t_{n+1}) - U_{n+1}^{-}|^2 \le k_{n+1}^2 \mathcal{E}_{n+1} + |U_n^+ - u(t_n)|^2$$

that is,§

$$\begin{aligned} \left| u(t_{n+1}) - U_{n+1}^{-} \right| &\leq k_{n+1} \sqrt{\mathcal{E}_{n+1}} + \left| U_n^+ - u(t_n) \right| \leq \\ &\leq k_{n+1} \sqrt{\mathcal{E}_{n+1}} + \left| U_n^+ - U_n^- \right| + \left| U_n^- - u(t_n) \right| \end{aligned}$$

and by induction one finally obtains (3.44).

3.5 Admissible solution and measure of the error

In section 3.2 we have described the properties and equations that define the behaviour of the standard generalised material model, which for the reader's convenience are summarized in Box 3.1.

We assume that the problem of computing the response of such model to given external actions is posed in the set of functions, $(\boldsymbol{\sigma}(\boldsymbol{x},t), \boldsymbol{A}(\boldsymbol{x},t); \boldsymbol{\epsilon}(\boldsymbol{x},t), \boldsymbol{\epsilon}^{p}(\boldsymbol{x},t), \boldsymbol{\alpha}(\boldsymbol{x},t))$, which gives a finite value to the global energy

$$\int_{\Omega} \int_{\Omega} \int_{e}^{sl} \eta_{\boldsymbol{x},t}^{2}(\boldsymbol{\sigma}; \boldsymbol{\epsilon}^{e}) \, \mathrm{d}\Omega + \int_{\Omega} \int_{p}^{sl} \eta_{\boldsymbol{x},t}^{2}(\boldsymbol{A}; \boldsymbol{\alpha}) \, \mathrm{d}\Omega + \int_{\Omega} \int_{0}^{T} \eta_{\boldsymbol{x},t}^{2}(\boldsymbol{\sigma}, \boldsymbol{A}; \boldsymbol{\epsilon}^{p}, \boldsymbol{\alpha}) \, \mathrm{d}t \, \mathrm{d}\Omega < \infty.$$

Also, we assume that the formulation is such that the problem has a solution which is unique.

In this class of functions, we distinguish a subset given by those functions which satisfy *only some* properties and equations given in Box 3.1. Any element of this set is referred to, in general, as an **admissible solution**. It is, therefore, clear that an admissible solution is the exact solution if and only if also *the remaining* equations are satisfied.

[§]If $a^2 \leq b^2 + c^2$, with $b, c \geq 0$, then it is $b^2 + c^2 \leq (b+c)^2$, thus it follows $a \leq b + c$.

Box 3.1. Initial Boundary Value Problem for Standard Generalised Models with Internal Variables

Find $\sigma(\mathbf{x}, t)$, $\mathbf{A}(\mathbf{x}, t)$; $\mathbf{u}(\mathbf{x}, t)$, $\epsilon(\mathbf{x}, t)$, $\epsilon^{p}(\mathbf{x}, t)$, $\alpha(\mathbf{x}, t)$ such that the following conditions are satisfied:

Kinematic Compatibility:

Continuity of the Displacement Field, $\boldsymbol{u}(\boldsymbol{x},t)$.

Time continuity of the Total Strain, $\boldsymbol{\epsilon}(\boldsymbol{x},t) = \nabla_{\boldsymbol{s}} \boldsymbol{u}(\boldsymbol{x},t)$.

Time continuity of the Plastic Strain, $\boldsymbol{\epsilon}^p(\boldsymbol{x},t)$.

Time continuity of the Internal Variables, $\alpha(x, t)$.

Displacement Boundary Conditions.

Additivity of the Strain Tensor:

 $\boldsymbol{\epsilon}(\boldsymbol{x},t) = \boldsymbol{\epsilon}^{\boldsymbol{e}}(\boldsymbol{x},t) + \boldsymbol{\epsilon}^{\boldsymbol{p}}(\boldsymbol{x},t),$ $\forall \, \boldsymbol{x} \in \Omega, \quad \forall \, t \in [0,T].$

Equilibrium:

 $egin{aligned} &\langle m{\sigma}(m{x},t),
abla m{\eta}(m{x})
angle = \langle m{b}(m{x},t), m{\eta}(m{x})
angle + \langle m{t}(m{x},t), m{\eta}(m{x})
angle_{\partial\Omega_t} \ & \forall m{\eta} \in \mathcal{V}_0, \quad \forall \, t \in [0,T]. \end{aligned}$

Initial Conditions:

$$\epsilon^{p}(\boldsymbol{x}, t = 0) = \boldsymbol{0},$$

 $\forall \boldsymbol{x} \in \Omega$
 $\alpha(\boldsymbol{x}, t = 0) = \boldsymbol{0},$

State Laws:

$$\begin{split} \psi_e(\boldsymbol{\epsilon}^e(\boldsymbol{x},t)) + \psi_e^*(\boldsymbol{\sigma}(\boldsymbol{x},t)) - \boldsymbol{\sigma}(\boldsymbol{x},t) &: \boldsymbol{\epsilon}^e(\boldsymbol{x},t) = 0, \\ \psi_p(\boldsymbol{\alpha}(\boldsymbol{x},t)) + \psi_p^*(\boldsymbol{A}(\boldsymbol{x},t)) - \boldsymbol{A}(\boldsymbol{x},t) &: \boldsymbol{\alpha}(\boldsymbol{x},t) = 0, \\ \forall \, \boldsymbol{x} \in \Omega, \quad \forall \, t \in [0,T]. \end{split}$$

Evolution Laws:

$$\begin{split} \varphi(\dot{\epsilon}^{p}(\boldsymbol{x},t),-\dot{\boldsymbol{\alpha}}(\boldsymbol{x},t)) + \varphi^{*}(\boldsymbol{\sigma}(\boldsymbol{x},t),\boldsymbol{A}(\boldsymbol{x},t)) + \\ -\boldsymbol{\sigma}(\boldsymbol{x},t) \colon \dot{\boldsymbol{\epsilon}}^{p}(\boldsymbol{x},t) + \boldsymbol{A}(\boldsymbol{x},t) \colon \dot{\boldsymbol{\alpha}}(\boldsymbol{x},t) = 0, \\ \forall \, \boldsymbol{x} \in \Omega, \quad \forall \, t \in [0,T]. \end{split}$$

If in the conditions defining the admissible solutions all the qualitative properties are included, the approximation quality of an admissible solution is described by the residual, which is produced in the equations defining the admissible subset.

Given the dissipative character of the problem under consideration, as shown in Section 3.3, and following from arguments given in Section 3.4, a direct measure of the residual can be used as an indication of the error associated with the problem.

The notion of error in the constitutive equations for non linear problems as introduced by Ladéveze et al. (1986) implements the above ideas by splitting the equations that govern the behaviour of the continuum in two groups: One group is used to define an admissible solution and combines the kinematic compatibility conditions, the additivity of the strain tensor, the equilibrium equations and the initial conditions. The second group comprises, on the other hand, the constitutive equations and is used to quantify the approximation of the admissible solution by means of the residual produced therein. As far as the measure of this residual is concerned, this depends on the type of constitutive formulation which is adopted. In the constitutive formulation with internal variables we exploit the convexity structure of the state laws and evolution laws, as described in Section 3.2.4, by referring to equivalent scalar formulations of the tensorial constitutive equations. These equivalent formulations, in turn, can be interpreted as offset of the energetic balance which is not met by the admissible solution. The notion of dissipation error and extended dissipation error are in this way introduced according to whether or not we include the state laws in the conditions that define the admissibility of a solution, respectively.

If a functional formalism is adopted for the constitutive modelling, for materials that strictly follow the conditions of Drucker stability (Drucker, 1964) the notion of Drucker's error is introduced (Ladéveze *et al.*, 1986; Coffignal, 1987; Gallimard, 1994). Here, the Drucker's inequality is used to quantify the quality of the admissible solution, which is not required to satisfy the constitutive equations.

The violation of a qualitative property, on the other hand, such as time continuity, for instance, requires a more specific treatment and this will be object of a further subsection.

3.5.1 Error in the constitutive equations for time continuous admissible solution

Throughout this section, in the definition of the admissible conditions we will always consider functions which are time continuous over the time interval of interest. In order to be more specific in the treatment, in the following the internal variables are denoted as p and α to indicate scalar and tensorial quantities, respectively. The associated thermodynamic forces are thus referred to as R and X, respectively.

3.5.1.1 Dissipation Error

This measure of the error has been introduced for the first time by Ladevèze (1989) within the context of the LATIN method applied to the solution of the initial boundary value problem of a material model formulated with internal variables. Its numerical performance has been assessed in Moës (1996); Ladevèze & Moës (1997) and Ladevèze & Moës (1999).

Definition of the Admissibility Conditions

The dissipation error is the error in the constitutive equations for a formulation with internal variables obtained by including the state laws in the definition of the admissibility conditions. More precisely, the field $(\sigma_{ad}(\boldsymbol{x},t), \boldsymbol{X}_{ad}(\boldsymbol{x},t), R_{ad}(\boldsymbol{x},t);$ $\boldsymbol{u}_{ad}(\boldsymbol{x},t), \boldsymbol{\epsilon}_{ad}(\boldsymbol{x},t), \boldsymbol{\epsilon}_{ad}^{p}(\boldsymbol{x},t), \boldsymbol{\alpha}_{ad}(\boldsymbol{x},t), p_{ad}(\boldsymbol{x},t))$ is an admissible solution with respect to the computation of the dissipation error if the following conditions are met:

Kinematic Compatibility:

Continuity of the Displacement Field, $u_{ad}(x, t)$.

Time continuity of the Total Strain, $\epsilon_{ad}(\boldsymbol{x},t) = \nabla_s \boldsymbol{u}_{ad}(\boldsymbol{x},t)$.

Time continuity of the Plastic Strain, $\epsilon_{ad}^{p}(\boldsymbol{x},t)$.

Time continuity of the Internal Variables, $\alpha_{ad}(x, t)$, $p_{ad}(x, t)$.

Displacement Boundary Conditions.

Additivity of the Strain Tensor:

 $\begin{aligned} \boldsymbol{\epsilon}_{ad}(\boldsymbol{x},t) &= \boldsymbol{\epsilon}_{ad}^{e}(\boldsymbol{x},t) + \boldsymbol{\epsilon}_{ad}^{p}(\boldsymbol{x},t), \\ \forall \, \boldsymbol{x} \in \Omega, \quad \forall \, t \in [0,T]. \end{aligned}$

Equilibrium:

 $\langle \sigma_{ad}(\boldsymbol{x},t), \nabla \boldsymbol{\eta}(\boldsymbol{x}) \rangle = \langle \boldsymbol{b}(\boldsymbol{x},t), \boldsymbol{\eta}(\boldsymbol{x}) \rangle + \langle \boldsymbol{t}(\boldsymbol{x},t), \boldsymbol{\eta}(\boldsymbol{x}) \rangle_{\partial \Omega_t}$ $\forall \boldsymbol{\eta} \in \mathcal{V}_0, \quad \forall t \in [0,T].$

Initial Conditions:

$$\begin{split} \boldsymbol{\epsilon}_{ad}^{p}(\boldsymbol{x},t=0) &= \boldsymbol{0}, \\ \boldsymbol{\alpha}_{ad}(\boldsymbol{x},t=0) &= \boldsymbol{0}, \quad p_{ad}(\boldsymbol{x},t=0) = \boldsymbol{0} \end{split} \quad \forall \, \boldsymbol{x} \in \boldsymbol{\Omega} \end{split}$$

State Laws:

$$\begin{split} \psi_e(\boldsymbol{\epsilon}_{ad}^e(\boldsymbol{x},t)) + \psi_e^*(\boldsymbol{\sigma}_{ad}(\boldsymbol{x},t)) &- \boldsymbol{\sigma}_{ad}(\boldsymbol{x},t) \colon \boldsymbol{\epsilon}_{ad}^e(\boldsymbol{x},t) = 0, \\ \psi_p(\boldsymbol{\alpha}_{ad}(\boldsymbol{x},t), p_{ad}(\boldsymbol{x},t)) + \psi_p^*(\boldsymbol{X}_{ad}(\boldsymbol{x},t), R_{ad}(\boldsymbol{x},t)) + \\ &- \left[\boldsymbol{X}_{ad}(\boldsymbol{x},t) \colon \boldsymbol{\alpha}_{ad}(\boldsymbol{x},t) + R_{ad}(\boldsymbol{x},t) p_{ad}(\boldsymbol{x},t) \right] = 0, \\ \forall \, \boldsymbol{x} \in \Omega, \quad \forall \, t \in [0,T]. \end{split}$$

Definition of Error

The only equation which in general is not satisfied by an admissible solution is therefore the evolution law. The quality of its approximation does then depend upon the residual produced therein. A natural way to measure this residual is obtained by resorting to a scalar equivalent formulation of the evolution law due to the convexity nature of the law. This formulation is discussed in Section 3.2.4.4 and especially notable are its properties (3.20) and (3.21). As a result, it is quite natural to assume the following definition of an error

$$e_{dis}^{2}(T) = 2 \int_{\Omega} \int_{0}^{T} d\eta_{\boldsymbol{x},t}^{2}(\boldsymbol{\sigma}_{ad}, \boldsymbol{X}_{ad}, \boldsymbol{R}_{ad}; \dot{\boldsymbol{\epsilon}}_{ad}^{p}, \dot{\boldsymbol{\alpha}}_{ad}, \dot{\boldsymbol{p}}_{ad}) dt d\Omega$$
(3.45)

where we have let

$$\begin{split} {}^{d}\eta_{\boldsymbol{x},t}^{2}(\boldsymbol{\sigma}_{ad},\,\boldsymbol{X}_{ad},\,R_{ad};\,\dot{\boldsymbol{\epsilon}}_{ad}^{p},\,\dot{\boldsymbol{\alpha}}_{ad},\,\dot{p}_{ad}) = \\ &= \varphi^{\star}(\boldsymbol{\sigma}_{ad}(\boldsymbol{x},t),\,\boldsymbol{X}_{ad}(\boldsymbol{x},t),\,R_{ad}(\boldsymbol{x},t)) + \varphi(\dot{\boldsymbol{\epsilon}}_{ad}^{p}(\boldsymbol{x},t),\,-\dot{\boldsymbol{\alpha}}_{ad}(\boldsymbol{x},t),\,-\dot{p}_{ad}(\boldsymbol{x},t)) + \\ &-\boldsymbol{\sigma}_{ad}(\boldsymbol{x},t):\,\,\dot{\boldsymbol{\epsilon}}_{ad}^{p}(\boldsymbol{x},t) + \boldsymbol{X}_{ad}(\boldsymbol{x},t):\,\,\dot{\boldsymbol{\alpha}}_{ad}(\boldsymbol{x},t) + R_{ad}(\boldsymbol{x},t)\dot{p}_{ad}(\boldsymbol{x},t) \end{split}$$

and we recall, once again, that the state laws are satisfied by the admissible solution.

Remark 3.6. Equation (3.45) defines the global error at time T as a sum of elemental contributions arising from the unaltered propagation at t of the error produced by the residual ${}^{d}\eta^{2}_{\boldsymbol{x},t}d\Omega d\tau$ within $d\Omega$ and $[\tau, \tau + d\tau]$ at time $\tau \leq t$. This is in agreement with the dissipative nature of the problem under consideration as discussed in Section 3.3 and Section 3.4.

If we denote by

$$s_{ad}(\boldsymbol{x},t) = \Big(\boldsymbol{\sigma}_{ad}(\boldsymbol{x},t), \boldsymbol{X}_{ad}(\boldsymbol{x},t), \boldsymbol{R}_{ad}(\boldsymbol{x},t); \, \boldsymbol{\epsilon}_{ad}(\boldsymbol{x},t), \boldsymbol{\epsilon}_{ad}^{p}(\boldsymbol{x},t), \boldsymbol{\alpha}_{ad}(\boldsymbol{x},t), \boldsymbol{p}_{ad}(\boldsymbol{x},t)\Big),$$

and

$$s_{ex}(\boldsymbol{x},t) = \Big(\boldsymbol{\sigma}_{ex}(\boldsymbol{x},t), \boldsymbol{X}_{ex}(\boldsymbol{x},t), \boldsymbol{R}_{ex}(\boldsymbol{x},t); \boldsymbol{\epsilon}_{ex}(\boldsymbol{x},t), \boldsymbol{\epsilon}_{ex}^{p}(\boldsymbol{x},t), \boldsymbol{\alpha}_{ex}(\boldsymbol{x},t), p_{ex}(\boldsymbol{x},t)\Big),$$

an admissible and the exact solution of the initial boundary value problem, respectively, the definition (3.45) can be assumed as a global measure of the error of the (kinematic) admissible solution in the following sense:

Theorem 3.1.

Given an admissible solution $s_{ad} = s_{ad}(x, t)$ with respect to the computation of the dissipation error, it follows

$$\begin{aligned} e_{dis}^2(T) &\geq 0 \\ e_{dis}^2(T) &= 0 \iff s_{ad}(\boldsymbol{x},t) = s_{ex}(\boldsymbol{x},t) \quad \forall \, \boldsymbol{x} \in \Omega, \quad \forall \, t \leq T. \end{aligned}$$

Proof. Because of the properties of the Legendre-Fenchel inequality, it follows that ${}^{d}\eta^{2}_{\boldsymbol{x},t} \geq 0$ and ${}^{d}\eta^{2}_{\boldsymbol{x},t} = 0$ if and only if the admissible solution does satisfy the evolution law at time t. As a result, $e^{2}_{dis}(t)$ is a non negative increasing scalar function. Therefore if $e^{2}_{dis}(T) = 0$, it is also $e^{2}_{dis}(t) = 0 \quad \forall t \leq T$, which then means that $s_{ad}(\boldsymbol{x},t)$ does satisfy the evolution law $\forall \boldsymbol{x} \in \Omega, \quad \forall t \leq T$, i.e., $s_{ad}(\boldsymbol{x},t) = s_{ex}(\boldsymbol{x},t) \quad \forall \boldsymbol{x} \in \Omega, \quad \forall t \leq T$.

Remark 3.7. The dissipation error $e_{dis}^2(T)$ is finite if and only if

$$(\boldsymbol{\sigma}_{ad}, \boldsymbol{X}_{ad}, \boldsymbol{R}_{ad}) \in dom \, \varphi^* \quad \forall \boldsymbol{x} \in \Omega, \, \forall t \leq T$$
$$(\dot{\epsilon}^p_{ad}, \dot{\boldsymbol{\alpha}}_{ad}, \dot{p}_{ad}) \in dom \, \varphi \quad \forall \boldsymbol{x} \in \Omega, \, \forall t \leq T$$

where dom stands for the effective domain of the function. These conditions are easy to impose for a model with linear hardening and linear elasticity in case of the convex elastic domain. For more general hardening laws, however, the model is required to be expressed first in normal form as introduced in Ladevèze (1989), that is, the state laws must be transformed in a linear form. For further details on the meaning of transformation of internal variables and conditions under which the above transformation is feasible, we refer to Ladevèze (1999) and Nguyen (2000). \Box

Finally, note that if we let $t_n \in]0, T[$, for the additivity of the integral, equation (3.45) can be also written as

$$e_{dis}^{2}(T) = e_{dis}^{2}(t_{n}) + 2 \int_{\Omega} \int_{t_{n}}^{T} d\eta_{\boldsymbol{x},t}^{2}(\boldsymbol{\sigma}_{ad}, \boldsymbol{X}_{ad}, R_{ad}; \dot{\boldsymbol{\epsilon}}_{ad}^{p}, \dot{\boldsymbol{\alpha}}_{ad}, \dot{p}_{ad}) dt d\Omega \qquad (3.46)$$

which presents the global error at the time T as a sum of the error at the time $t_n < T$ and the error associated with the admissible solution over $[t_n, T]$.

3.5.1.2 Extended Dissipation Error

This error measure has been introduced for the first time in Ladevèze *et al.* (1999) and Ladevèze (2001), where an application is given for an elastic-damage coupled model.

Definition of the Admissibility Conditions

An immediate extension of the dissipation error introduced in the previous section is obtained by removing the state laws from the definition of the admissibility conditions. This allows the recovery of the error in the constitutive equations for an admissible solution which is elastic, as introduced in Section ??. More precisely, the field $(\sigma_{ad}(\boldsymbol{x},t), \boldsymbol{X}_{ad}(\boldsymbol{x},t), R_{ad}(\boldsymbol{x},t); \boldsymbol{u}_{ad}(\boldsymbol{x},t), \epsilon_{ad}(\boldsymbol{x},t), \epsilon_{ad}^{p}(\boldsymbol{x},t), \alpha_{ad}(\boldsymbol{x},t), p_{ad}(\boldsymbol{x},t))$ is an admissible solution with respect to the computation of the extended dissipation error if the following conditions are met

Kinematic Compatibility:

Continuity of the Displacement Field, $u_{ad}(x, t)$. Time continuity of the Total Strain, $\epsilon_{ad}(x, t) = \nabla_s u_{ad}(x, t)$. Time continuity of the Plastic Strain, $\epsilon^p_{ad}(x, t)$. Time continuity of the Internal Variables, $\alpha_{ad}(x, t)$, $p_{ad}(x, t)$. Displacement Boundary Conditions.

Additivity of the Strain Tensor:

 $\epsilon_{ad}(\boldsymbol{x},t) = \epsilon_{ad}^{\boldsymbol{e}}(\boldsymbol{x},t) + \epsilon_{ad}^{\boldsymbol{p}}(\boldsymbol{x},t),$ $\forall \boldsymbol{x} \in \Omega, \quad \forall t \in [0,T].$

Equilibrium:

 $egin{aligned} &\langle \pmb{\sigma}_{ad}(\pmb{x},t), \nabla \pmb{\eta}(\pmb{x})
angle = \langle \pmb{b}(\pmb{x},t), \pmb{\eta}(\pmb{x})
angle + \langle \pmb{t}(\pmb{x},t), \pmb{\eta}(\pmb{x})
angle_{\partial \Omega_t} \\ &\forall \pmb{\eta} \in \mathcal{V}_0, \quad \forall t \in [0,T]. \end{aligned}$

Initial Conditions:

$$\begin{split} \boldsymbol{\epsilon}_{ad}^p(\boldsymbol{x},t=0) &= \boldsymbol{0}, \\ \boldsymbol{\alpha}_{ad}(\boldsymbol{x},t=0) &= \boldsymbol{0}, \quad p_{ad}(\boldsymbol{x},t=0) = \boldsymbol{0} \end{split} \quad \forall \, \boldsymbol{x} \in \boldsymbol{\Omega} \end{split}$$

Definition of Error

In this case, the equations that are not satisfied by an admissible solution are the state laws and the evolution laws. Therefore, besides the residual in the evolution laws for which the same considerations as in the previous section apply, now also the residual in the state laws must be considered to assess the quality of the approximation associated with the given admissible solution. A natural measure of this residual is provided by the equivalent formulation of the state equations which exploits the convexity properties of the law, as expressed notably by the equations (3.7) and (3.9). Furthermore, given the nature of the state laws that relate the current value of the kinematic variables to the corresponding static one, a global measure of the error is obtained by assuming an L^{∞} accumulation in time of the current value of the error in the state laws. Therefore, it is quite natural to assume

the following definition of error

$$e_{ext}^{2}(T) = \sup_{t \leq T} \left\{ \underbrace{2 \int_{\Omega} {}^{sl} \eta_{\boldsymbol{x},t}^{2}(\boldsymbol{\sigma}_{ad}, \boldsymbol{X}_{ad}, R_{ad}; \boldsymbol{\epsilon}_{ad}^{e}, \boldsymbol{\alpha}_{ad}, p_{ad}) \mathrm{d}\Omega}_{\boldsymbol{\theta}_{sl}^{2}(t)} + \underbrace{2 \int_{\Omega} \int_{0}^{t} {}^{d} \eta_{\boldsymbol{x},\tau}^{2}(\boldsymbol{\sigma}_{ad}, \boldsymbol{X}_{ad}, R_{ad}; \boldsymbol{\epsilon}_{ad}^{p}, \dot{\boldsymbol{\alpha}}_{ad}, \dot{p}_{ad}) \mathrm{d}\tau \mathrm{d}\Omega}_{\boldsymbol{\theta}_{d}^{2}(t)} \right\}$$

$$(3.47)$$

where, in general, the quantity

$$s^{l}\eta^{2}_{\boldsymbol{x},t}(\boldsymbol{\sigma}_{ad}, \boldsymbol{X}_{ad}, \boldsymbol{R}_{ad}; \boldsymbol{\epsilon}^{e}_{ad}, \boldsymbol{\alpha}_{ad}, \boldsymbol{p}_{ad}) =$$

$$= \psi^{*}(\boldsymbol{\sigma}_{ad}(\boldsymbol{x}, t), \boldsymbol{X}_{ad}(\boldsymbol{x}, t), \boldsymbol{R}_{ad}(\boldsymbol{x}, t)) + \psi(\boldsymbol{\epsilon}^{e}_{ad}(\boldsymbol{x}, t), \boldsymbol{\alpha}_{ad}(\boldsymbol{x}, t), \boldsymbol{p}_{ad}(\boldsymbol{x}, t)) +$$

$$-\left\{\boldsymbol{\sigma}_{ad}(\boldsymbol{x}, t): \boldsymbol{\epsilon}^{e}_{ad}(\boldsymbol{x}, t) + \boldsymbol{X}_{ad}(\boldsymbol{x}, t): \boldsymbol{\alpha}_{ad}(\boldsymbol{x}, t) + \boldsymbol{R}_{ad}(\boldsymbol{x}, t)\boldsymbol{p}_{ad}(\boldsymbol{x}, t)\right\}$$

is the residual in the state laws, and, likewise to the dissipation error, the term

$$\begin{split} {}^{d}\eta^{2}_{\boldsymbol{x},t}(\boldsymbol{\sigma}_{ad},\,\boldsymbol{X}_{ad},\,R_{ad};\,\dot{\boldsymbol{\epsilon}}^{p}_{ad},\,\dot{\boldsymbol{\alpha}}_{ad},\,\dot{\boldsymbol{p}}_{ad}) = \\ &= \varphi^{\star}(\boldsymbol{\sigma}_{ad}(\boldsymbol{x},t),\,\boldsymbol{X}_{ad}(\boldsymbol{x},t),\,R_{ad}(\boldsymbol{x},t)) + \varphi(\dot{\boldsymbol{\epsilon}}^{p}_{ad}(\boldsymbol{x},t),\,-\dot{\boldsymbol{\alpha}}_{ad}(\boldsymbol{x},t),\,-\dot{\boldsymbol{p}}_{ad}(\boldsymbol{x},t)) + \\ &-\boldsymbol{\sigma}_{ad}(\boldsymbol{x},t):\,\,\dot{\boldsymbol{\epsilon}}^{p}_{ad}(\boldsymbol{x},t) + \boldsymbol{X}_{ad}(\boldsymbol{x},t):\,\,\dot{\boldsymbol{\alpha}}_{ad}(\boldsymbol{x},t) + R_{ad}(\boldsymbol{x},t)\dot{\boldsymbol{p}}_{ad}(\boldsymbol{x},t). \end{split}$$

describes the residual produced in the evolution laws.

Remark 3.8. If we recall the result

Given
$$f(t) \ge 0 \quad \forall t \le T$$

 $(\sup_{t \le T} f(t))^{\frac{1}{2}} = \sup_{t \le T} f^{\frac{1}{2}}(t),$

we can also write

$$e_{ext}(T) = \sup_{t \leq T} \left\{ \underbrace{2 \int_{\Omega} {}^{sl} \eta_{\boldsymbol{x},t}^{2}(\boldsymbol{\sigma}_{ad}, \boldsymbol{X}_{ad}, \boldsymbol{R}_{ad}; \boldsymbol{\epsilon}_{ad}^{e}, \boldsymbol{\alpha}_{ad}, \boldsymbol{p}_{ad}) \mathrm{d}\Omega}_{\boldsymbol{\theta}_{sl}^{2}(t)} + \underbrace{2 \int_{\Omega} \int_{0}^{t} {}^{d} \eta_{\boldsymbol{x},\tau}^{2}(\boldsymbol{\sigma}_{ad}, \boldsymbol{X}_{ad}, \boldsymbol{R}_{ad}; \boldsymbol{\epsilon}_{ad}^{p}, \dot{\boldsymbol{\alpha}}_{ad}, \dot{\boldsymbol{p}}_{ad}) \mathrm{d}\tau \mathrm{d}\Omega}_{\boldsymbol{\theta}_{d}^{2}(t)} \right\}^{\frac{1}{2}}.$$

Thus, the extended dissipation error has the form of an L^{∞} norm in time of an energy type norm of the error in the state variables.

Definition (3.47) can be assumed as a global measure of the error of the (kinematic) admissible solution in the following sense:

Theorem 3.2.

Given an admissible solution $s_{ad} = s_{ad}(x, t)$ with respect to the computation of the extended dissipation error, i.e., s_{ad} is not required to meet the state laws, it follows

$$\begin{split} e^2_{ext}(T) &\geq 0 \\ e^2_{ext}(T) &= 0 \iff s_{ad}(x,t) = s_{ex}(x,t) \quad \forall \, x \in \Omega, \quad \forall \, t \leq T. \end{split}$$

Proof. Because of the properties of the Legendre-Fenchel inequality, we recall that ${}^{sl}\eta^2_{\boldsymbol{x},t} \geq 0$ and ${}^{d}\eta^2_{\boldsymbol{x},t} \geq 0$. Also, ${}^{sl}\eta^2_{\boldsymbol{x},t} = 0$, ${}^{d}\eta^2_{\boldsymbol{x},t} = 0$ if and only if the given time continuous admissible solution $s_{ad}(\boldsymbol{x},t)$ satisfies the state laws and the evolution laws, respectively. As a result, $e^2_{ext}(T)$ appears as supremum of $\theta^2_{sl}(t)$ and $\theta^2_d(t)$ which are non negative increasing scalar functions. Therefore, if $e^2_{ext}(T) = 0$, it follows $\theta^2_{sl}(t) = 0 \quad \forall t \leq T \text{ and } \theta^2_d(t) = 0 \quad \forall t \leq T$. Finally, due to the definition of $\theta^2_{sl}(t)$, it follows

$${}^{sl}\eta^2_{\boldsymbol{x},t} = 0, \quad \forall \, \boldsymbol{x} \in \Omega, \quad \forall \, t \leq T$$

 ${}^{d}\eta^2_{\boldsymbol{x},t} = 0, \quad \forall \, \boldsymbol{x} \in \Omega, \quad \forall \, t \leq T.$

Thus, the given time continuous admissible solution, $s_{ad}(x, t)$ satisfies the state laws and the evolution laws, respectively, that is, it coincides with the exact solution, $s_{ex}(x, t)$.

The viceversa is trivial.

Remark 3.9. The extended dissipation error $e_{ext}^2(T)$ is finite if and only if

$$(\boldsymbol{\sigma}_{ad}, \boldsymbol{X}_{ad}, \boldsymbol{R}_{ad}) \in dom \, \varphi^*$$

 $(\dot{\boldsymbol{\epsilon}}^p_{ad}, \dot{\boldsymbol{\alpha}}_{ad}, \dot{p}_{ad}) \in dom \, \varphi.$

Unlike the dissipation error, where the constraint of the state laws was imposed between kinematic and the conjugate static admissible variables, in this case the model is not required to be a priori transformed into normal form in order to compute the error, for the meeting of the above conditions is quite easy to realize due to the convexity of $dom \varphi^*$ and $dom \varphi$.

Likewise equation (3.46), for any $t_n \in]0, T[$, the extended dissipation error

(3.47) can also be expressed as follows

$$e_{ext}^{2}(T) = \operatorname{Max}\left\{e_{ext}^{2}(t_{n}), \sup_{t_{n} \leq t \leq T}\left\{2\int_{\Omega}{}^{sl}\eta_{x,t}^{2}(\boldsymbol{\sigma}_{ad}, \boldsymbol{X}_{ad}, \boldsymbol{R}_{ad}; \boldsymbol{\epsilon}_{ad}^{e}, \boldsymbol{\alpha}_{ad}, \boldsymbol{p}_{ad})\mathrm{d}\Omega + \theta_{d}^{2}(t_{n}) + \right.$$

$$\left. + 2\int_{\Omega}\int_{t_{n}}^{t}{}^{d}\eta_{x,\tau}^{2}(\boldsymbol{\sigma}_{ad}, \boldsymbol{X}_{ad}, \boldsymbol{R}_{ad}; \boldsymbol{\epsilon}_{ad}^{p}, \boldsymbol{\alpha}_{ad}, \boldsymbol{p}_{ad})\mathrm{d}\tau \mathrm{d}\Omega\right\}\right\}.$$

$$\left. (3.48)$$

This presents the global error at the time T in terms of the error at the time $t_n < T$ and the admissible solution over $[t_n, T]$.

3.5.2 Error in the constitutive equations for admissible solution with jump across time instant t_n

In the admissible conditions listed in the preceding section, time continuity had been always assumed. Object of this section is to show how the previous definitions of error in the constitutive equations can be extended to the case in which the hypothesis of time continuity is removed so that admissible solutions may include a discontinuity jump at a given time instant t_n , i.e., $s_{ad}(x, t_n^-) \neq s_{ad}(x, t_n^+)$.

The need for relaxing continuity may arise, for example, in presence of a finite element solution having discontinuity jump at the time instant t_n because of change of mesh, as will be seen in the following chapters.

In rate independent plasticity, the solution of the initial boundary value problem which governs the evolution of the continuum depends only on the sequence of load levels whereas time has just the function of ordering this sequence. This means



Figure 3.7: Fictitious Load Time Variations

that the response of the system under the loading paths depicted in Figure 3.7, for

instance, is the same with respect to any value of the fictitious time step Δt during which the load level is kept constant. In agreement with this behaviour, it can be assumed that the value of the admissible solution at t_n^+ is also the value at $t_n + \Delta t$ and is independent on Δt . In this way, a fictitious time continuous process over the time interval $[t_n, t_n + \Delta t]$ along which the discontinuity is assumed to be taking place can be defined, and one can analyse the error in the evolution law as the time step Δt shrinks to zero.

An example of this process is illustrated in Figure 3.8 which refers to the time



Fictitious Continuous $\epsilon^{p}_{ad,\Delta t}(\tau)$ Variation



Plastic Strain Rate $\dot{\epsilon}^p_{ad,\Delta t}(\tau)$ Variation



Figure 3.8: Definition of Fictitious State Variables Variation

variation of an admissible plastic strain with discontinuity jump across the time instant t_n . Figure 3.8 also hints to the procedure used for the above extension which shows the formation of the δ -Dirac at t_n for the plastic strain rate because of the discontinuity jump in $\epsilon_{ad}^p(t)$ at t_n .

Under constant load level equal to $b(x, t_n)$, we consider a family of fictitious time continuous admissible solutions over $[t_n, t_n + \Delta t]$ and parameterized by Δt

having as limit the given admissible solution, that is, we consider

$$\sigma_{ad,\Delta t}(\boldsymbol{x},\tau), \boldsymbol{X}_{ad,\Delta t}(\boldsymbol{x},\tau), \boldsymbol{R}_{ad,\Delta t}(\boldsymbol{x},\tau);$$

$$\epsilon_{ad,\Delta t}(\boldsymbol{x},\tau), \epsilon_{ad,\Delta t}^{p}(\boldsymbol{x},\tau), \boldsymbol{\alpha}_{ad,\Delta t}(\boldsymbol{x},\tau), \boldsymbol{p}_{ad,\Delta t}(\boldsymbol{x},\tau),$$

such that, $\forall x \in \Omega$,

$$\begin{split} &\lim_{\Delta t \to 0+} \sigma_{ad,\Delta t}(\boldsymbol{x},\tau) = \sigma_{ad}(\boldsymbol{x},\tau); \quad \lim_{\Delta t \to 0+} X_{ad,\Delta t}(\boldsymbol{x},\tau) = X_{ad}(\boldsymbol{x},\tau); \\ &\lim_{\Delta t \to 0+} R_{ad,\Delta t}(\boldsymbol{x},\tau) = R_{ad}(\boldsymbol{x},\tau); \\ &\lim_{\Delta t \to 0+} \epsilon_{ad,\Delta t}(\boldsymbol{x},\tau) = \epsilon_{ad}(\boldsymbol{x},\tau); \quad \lim_{\Delta t \to 0+} \epsilon_{ad,\Delta t}^{p}(\boldsymbol{x},\tau) = \epsilon_{ad}^{p}(\boldsymbol{x},\tau); \\ &\lim_{\Delta t \to 0+} \alpha_{ad,\Delta t}(\boldsymbol{x},\tau) = \alpha_{ad}(\boldsymbol{x},\tau); \quad \lim_{\Delta t \to 0+} p_{ad,\Delta t}(\boldsymbol{x},\tau) = p_{ad}(\boldsymbol{x},\tau). \end{split}$$

where $(\bullet)_{ad}(x,\tau)$ denote the functions with the time discontinuity jump.

With regard to each member of this family, the error in the evolution law can now be computed. Thus, if $\forall x \in \Omega$ the following limit exists and is finite,

$$^{\Delta}\zeta_{d}^{2}(\boldsymbol{x},t_{n}) \equiv \lim_{\Delta t \to 0+} \int_{t_{n}}^{t_{n}+\Delta t} \left\{ \varphi^{*}(\boldsymbol{\sigma}_{ad,\Delta t}(\boldsymbol{x},\tau),\boldsymbol{X}_{ad,\Delta t}(\boldsymbol{x},\tau),R_{ad,\Delta t}(\boldsymbol{x},\tau)) + \right.$$

$$+ \varphi(\dot{\boldsymbol{e}}_{ad,\Delta t}^{p}(\boldsymbol{x},\tau),-\dot{\boldsymbol{\alpha}}_{ad,\Delta t}(\boldsymbol{x},\tau),-\dot{\boldsymbol{p}}_{ad,\Delta t}(\boldsymbol{x},\tau)) + \\ \left. - \boldsymbol{\sigma}_{ad,\Delta t}(\boldsymbol{x},\tau) : \dot{\boldsymbol{e}}_{ad,\Delta t}^{p}(\boldsymbol{x},\tau) + \right.$$

$$+ \left. \left. \boldsymbol{X}_{ad,\Delta t}(\boldsymbol{x},\tau) : \dot{\boldsymbol{\alpha}}_{ad,\Delta t}(\boldsymbol{x},\tau) + R_{ad,\Delta t}(\boldsymbol{x},\tau) \dot{\boldsymbol{p}}_{ad,\Delta t}(\boldsymbol{x},\tau) \right\} d\tau$$

$$(3.49)$$

it seems natural to assume the limit to be an error in the constitutive equations at \cdot the point \boldsymbol{x} in presence of discontinuity.

Remark 3.10. The jump in the state variables modifies only the error component that is associated with the evolution law, since these are the equations that involve the rate of the internal variables.

Furthermore, the additional term, ${}^{\Delta}\zeta_d^2(\boldsymbol{x}, t_n)$, is always non negative as a result of limit of non negative functions due to the Legendre-Fenchel inequality. However, for the material models taken into account, the non negativity of the term ${}^{\Delta}\zeta_d^2(\boldsymbol{x}, t_n)$ will be checked directly. Since ${}^{\Delta}\zeta_d^2(\boldsymbol{x}, t_n) \ge 0$, the jump in the admissible solution will always produce an increase of the error component associated with the dissipation.

3.5.2.1 Augmented Dissipation Error

The dissipation error at the time T of the admissible solution with jump across the time $t_n \in [0, T[$ is given by

$${}^{\Delta}e_{dis}^{2}(T) = 2\int_{\Omega}\int_{0}^{T}{}^{d}\eta_{\boldsymbol{x},t}^{2}(\boldsymbol{\sigma}_{ad}, \boldsymbol{X}_{ad}, \boldsymbol{R}_{ad}; \dot{\boldsymbol{\epsilon}}_{ad}^{p}, \dot{\boldsymbol{\alpha}}_{ad}, \dot{p}_{ad})\,\mathrm{d}t\,\mathrm{d}\Omega + 2\int_{\Omega}{}^{\Delta}\zeta_{d}^{2}(\boldsymbol{x}, t_{n})\mathrm{d}\Omega$$

In order to compute ${}^{\Delta}\zeta_d^2(\boldsymbol{x}, t_n)$, the admissible solution is required to satisfy the state equations, which here we present in the following form

$$\sigma_{ad,\Delta t}(\boldsymbol{x},\tau) = \mathbf{C}[\epsilon_{ad,\Delta t}(\boldsymbol{x},\tau) - \epsilon_{ad,\Delta t}^{p}(\boldsymbol{x},\tau)];$$

$$X_{ad,\Delta t}(\boldsymbol{x},\tau) = \mathbf{\Lambda}\alpha_{ad,\Delta t}(\boldsymbol{x},\tau); \ R_{ad,\Delta t}(\boldsymbol{x},\tau) = g(p_{ad,\Delta t}(\boldsymbol{x},\tau))$$
(3.50)

where Λ and g(p) are the same as for the plasticity model P2.

In the following, we provide the expression for ${}^{\Delta}\zeta_d^2(\boldsymbol{x}, t_n)$ for the rate independent plasticity models introduced in section 3.2.4.5 by assuming linear hardening. This condition is here invoked in order to guarantee that by assuming a time linear interpolation of the kinematic admissible variables, the corresponding conjugate forces, given by (3.50), will also appear as time linear interpolation of the values at the ends of the time step. This results in the static admissibility of the conjugate forces because of the admissibility of the interpolant values and the convexity of the elastic domain.

Also, for each model, it will be shown that ${}^{\Delta}\zeta_d^2(\boldsymbol{x}, t_n)$ is non negative and that by setting ${}^{\Delta}\zeta_d^2(\boldsymbol{x}, t_n)$ equal to zero, we infer the time continuity at t_n of the admissible solution. This, finally, means that ${}^{\Delta}\zeta_d^2(\boldsymbol{x}, t_n)$ characterizes effectively the discontinuity jump across t_n .

Plasticity Model P1: Prandtl Reuss Model

Continuous Admissible Fictitious Solutions.

We assume, $\forall x \in \Omega$, continuous admissible fictitious solutions defined as follows

- □ Kinematic Admissible Solution: $\epsilon_{ad,\Delta t}(\boldsymbol{x},\tau), \epsilon_{ad,\Delta t}^{p}(\boldsymbol{x},\tau), p_{ad,\Delta t}(\boldsymbol{x},\tau)$ are obtained as linear interpolation over $[t_n, t_n + \Delta t]$ of the values at t_n and $t_n + \Delta t$.
- □ Static Admissible Solution:

 $\sigma_{ad,\Delta t}(\boldsymbol{x},\tau) = \mathbf{C}[\epsilon_{ad,\Delta t}(\boldsymbol{x},\tau) - \epsilon^{p}_{ad,\Delta t}(\boldsymbol{x},\tau)]$ and $R_{ad,\Delta t}(\boldsymbol{x},\tau) = \mathbf{H}p_{ad,\Delta t}(\boldsymbol{x},\tau)$ are obtained by imposing the state laws. Because of the linearity of these laws, $\sigma_{ad,\Delta t}(\boldsymbol{x},\tau)$ and $R_{ad,\Delta t}(\boldsymbol{x},\tau)$ are given by linear interpolation over $[t_n, t_n + \Delta t]$ of the values at t_n and $t_n + \Delta t$.

 $\Box \sigma_{ad,\Delta t}(x,\tau)$ is in equilibrium with $b(x,t_n)$ because of the convexity of the equilibrium condition.

- $\Box \ (\boldsymbol{\sigma}_{ad,\Delta t}(\boldsymbol{x},\tau), R_{ad,\Delta t}(\boldsymbol{x},\tau)) \in \mathbb{E} \quad \forall \tau \in [t_n, t_n + \Delta t] \text{ because of the convexity of}$ the elastic domain.
- $\square \ \epsilon_{\mathit{ad}, \Delta t}(\pmb{x}, \tau)$ is kinematically admissible because of the convexity of the compatibility conditions.

Finite Value Error Requirements

We also impose that $\forall x \in \Omega$ the admissible solutions satisfy the following conditions

$$\begin{vmatrix}
(\boldsymbol{\sigma}_{ad}(\boldsymbol{x}, t_{n}^{-}), R_{ad}(\boldsymbol{x}, t_{n}^{-})) \in \mathbb{E} \\
(\boldsymbol{\sigma}_{ad}(\boldsymbol{x}, t_{n}^{+}), R_{ad}(\boldsymbol{x}, t_{n}^{+})) \in \mathbb{E}
\end{vmatrix} \Rightarrow$$

$$\Rightarrow (\boldsymbol{\sigma}_{ad}(\boldsymbol{x}, \tau), R_{ad}(\boldsymbol{x}, \tau)) \in \mathbb{E}, \forall \tau \in [t_{n}, t_{n} + \Delta t], \qquad (3.51)$$

$$p_{ad}(\boldsymbol{x}, t_{n}^{+}) - p_{ad}(\boldsymbol{x}, t_{n}^{-}) \geq \|\boldsymbol{\epsilon}_{ad}^{p}(\boldsymbol{x}, t_{n}^{+}) - \boldsymbol{\epsilon}_{ad}^{p}(\boldsymbol{x}, t_{n}^{-})\|, \qquad (3.51)$$

$$\operatorname{Tr}(\boldsymbol{\epsilon}_{ad}^{p}(\boldsymbol{x}, t_{n}^{+}) - \boldsymbol{\epsilon}_{ad}^{p}(\boldsymbol{x}, t_{n}^{-})) = 0.$$

Condition (3.51) occurs because of the linearity of the state laws and convexity of the elastic domain.

Dissipation Error across the time discontinuity

Consider the expression for the dissipation error for this model given by equation (3.45), we have,

$$\forall \boldsymbol{x} \in \Omega$$

$$^{\Delta} \zeta_{d}^{2}(\boldsymbol{x}, t_{n}) = \lim_{\Delta t \to 0+} \int_{t_{n}}^{t_{n}+\Delta t} \left\{ R_{0} \| \dot{\boldsymbol{\epsilon}}_{ad,\Delta t}^{p}(\boldsymbol{x}, \tau) \| + \right.$$

$$- \boldsymbol{\sigma}_{ad,\Delta t}(\boldsymbol{x}, \tau) : \dot{\boldsymbol{\epsilon}}_{ad,\Delta t}^{p}(\boldsymbol{x}, \tau) + R_{ad,\Delta t}(\boldsymbol{x}, \tau) \dot{p}_{ad,\Delta t}(\boldsymbol{x}, \tau) \right\} d\tau =$$

$$= R_{0} \| \boldsymbol{\epsilon}_{ad}^{p}(\boldsymbol{x}, t_{n}^{+}) - \boldsymbol{\epsilon}_{ad}^{p}(\boldsymbol{x}, t_{n}^{-}) \| +$$

$$- \frac{\boldsymbol{\sigma}_{ad}(\boldsymbol{x}, t_{n}^{+}) + \boldsymbol{\sigma}_{ad}(\boldsymbol{x}, t_{n}^{-})}{2} : \left(\boldsymbol{\epsilon}_{ad}^{p}(\boldsymbol{x}, t_{n}^{+}) - \boldsymbol{\epsilon}_{ad}^{p}(\boldsymbol{x}, t_{n}^{-}) \right) +$$

$$+ \frac{R_{ad}(\boldsymbol{x}, t_{n}^{+}) + R_{ad}(\boldsymbol{x}, t_{n}^{-})}{2} (p_{ad}(\boldsymbol{x}, t_{n}^{+}) - p_{ad}(\boldsymbol{x}, t_{n}^{-}))$$

$$(3.52)$$

with the State Equations being satisfied at t_n^- and t_n^+ .

Theorem 3.3.

Given the admissible solutions $s_{ad}(t_n^-)$ and $s_{ad}(t_n^+)$, which satisfy the state equations at t_n^- and t_n^+ respectively, and the finite value error requirements, it follows that Δ

$$\Delta \zeta_d^2(\boldsymbol{x}, t_n) \ge 0$$

Proof. To keep the notation simple, let

.

$$\sigma^{+} = \sigma_{ad}(\boldsymbol{x}, t_{n}^{+}), \ \sigma^{-} = \sigma_{ad}(\boldsymbol{x}, t_{n}^{-});$$
$$\Delta \sigma = \sigma^{+} - \sigma^{-};$$
$$R^{+} = R_{ad}(\boldsymbol{x}, t_{n}^{+}), \ R^{-} = R_{ad}(\boldsymbol{x}, t_{n}^{-});$$
$$\Delta R = R^{+} - R^{-};$$
$$\Delta \epsilon = \epsilon_{ad}(\boldsymbol{x}, t_{n}^{+}) - \epsilon_{ad}(\boldsymbol{x}, t_{n}^{-});$$
$$\Delta \epsilon^{p} = \epsilon_{ad}^{p}(\boldsymbol{x}, t_{n}^{+}) - \epsilon_{ad}^{p}(\boldsymbol{x}, t_{n}^{-});$$
$$\Delta p = p_{ad}(\boldsymbol{x}, t_{n}^{+}) - p_{ad}(\boldsymbol{x}, t_{n}^{-}).$$

From the conditions giving a finite value to the error, it follows

$$\Delta p \geq \|\Delta \epsilon^p\|$$
 and $R \geq 0$.

Thus we get

$$R_0 \|\Delta \epsilon^p\| - \sigma^+ \colon \Delta \epsilon^p + R^+ \Delta p \ge R_0 \|\Delta \epsilon^p\| - \sigma^+ \colon \Delta \epsilon^p + R^+ \Delta \epsilon^p.$$

Since $\operatorname{Tr}[\Delta \epsilon^p] = 0$, we can also write,

$$\boldsymbol{\sigma}^+ \colon \Delta \boldsymbol{\epsilon}^p = \boldsymbol{\sigma}_D^+ \colon \Delta \boldsymbol{\epsilon}^p \le \|\boldsymbol{\sigma}_D^+\| \| \Delta \boldsymbol{\epsilon}^p \|.$$

Hence,

$$R_{0} \|\Delta \epsilon^{p}\| - \sigma^{+} \colon \Delta \epsilon^{p} + R^{+} \Delta \epsilon^{p} \ge$$
$$\geq R_{0} \|\Delta \epsilon^{p}\| - \|\sigma_{D}^{+}\| \|\Delta \epsilon^{p}\| + R^{+} \Delta \epsilon^{p} =$$
$$= (R_{0} + R^{+} - \|\sigma_{D}^{+}\|) \|\Delta \epsilon^{p}\| \ge 0,$$

that is,

$$R_0 \|\Delta \epsilon^p\| - \sigma^+ \colon \Delta \epsilon^p + R^+ \Delta p \ge 0.$$
(3.53)

Likewise, it can be proven that

$$R_0 \|\Delta \epsilon^p\| - \boldsymbol{\sigma}^- \colon \Delta \epsilon^p + R^- \Delta p \ge 0.$$
(3.54)

Thus, summing up term by term and for the linearity of the double contraction operator, it follows

$$^{\Delta}\zeta_d^2(\boldsymbol{x},t_n) = R_0 \|\Delta \boldsymbol{\epsilon}^p\| - \frac{\boldsymbol{\sigma}^+ + \boldsymbol{\sigma}^-}{2} \colon \Delta \boldsymbol{\epsilon}^p + \frac{R^+ + R^-}{2} \Delta p \ge 0 \tag{3.55}$$

Theorem 3.4.

Given the admissible solutions $s_{ad}(t_n^-)$ and $s_{ad}(t_n^+)$, satisfying the state equations at t_n^- and t_n^+ respectively, and the finite value error requirements, the following expressions are valid

$$^{\Delta}\zeta_d^2(\boldsymbol{x}, t_n) = 0 \quad \Rightarrow \quad \begin{cases} \boldsymbol{\sigma}_{ad}(\boldsymbol{x}, t_n^+) = \boldsymbol{\sigma}_{ad}(\boldsymbol{x}, t_n^-) \\ R_{ad}(\boldsymbol{x}, t_n^+) = R_{ad}(\boldsymbol{x}, t_n^-) \\ \boldsymbol{\epsilon}_{ad}(\boldsymbol{x}, t_n^+) = \boldsymbol{\epsilon}_{ad}(\boldsymbol{x}, t_n^-) \\ \boldsymbol{\epsilon}_{ad}^p(\boldsymbol{x}, t_n^+) = \boldsymbol{\epsilon}_{ad}^p(\boldsymbol{x}, t_n^-) \\ p_{ad}(\boldsymbol{x}, t_n^+) = p_{ad}(\boldsymbol{x}, t_n^-) \end{cases}$$

Proof. Because of (3.55), by accounting for (3.53) and (3.54), it follows

If
$${}^{\Delta}\zeta_d^2(\boldsymbol{x}, t_n) = 0 \Rightarrow \begin{cases} R_0 \|\Delta \boldsymbol{\epsilon}^p\| - \boldsymbol{\sigma}^+ \colon \Delta \boldsymbol{\epsilon}^p + R^+ \Delta p = 0 \\ R_0 \|\Delta \boldsymbol{\epsilon}^p\| - \boldsymbol{\sigma}^- \colon \Delta \boldsymbol{\epsilon}^p + R^- \Delta p = 0 \end{cases}$$

Thus,

$$\Delta \boldsymbol{\sigma} \colon \Delta \boldsymbol{\epsilon}^p = \Delta R \Delta p \tag{3.56}$$

From equilibrium at t_n^- and t_n^+ it follows

$$egin{aligned} &\int_{\Omega} \pmb{\sigma}^- \colon
abla \eta \, \mathrm{d}\Omega = \int_{\Omega} \pmb{b}(\pmb{x},t_n) \eta \, \mathrm{d}\Omega + \int_{\partial \Omega_t} \pmb{t}(\pmb{x},t_n) \eta \, \mathrm{d}\Omega & orall \eta \in \mathcal{V}_0 \ \end{aligned}$$
 $\int_{\Omega} \pmb{\sigma}^+ \colon
abla \eta \, \mathrm{d}\Omega = \int_{\Omega} \pmb{b}(\pmb{x},t_n) \eta \, \mathrm{d}\Omega + \int_{\partial \Omega_t} \pmb{t}(\pmb{x},t_n) \eta \, \mathrm{d}\Omega & orall \eta \in \mathcal{V}_0. \end{aligned}$

Hence, since $\Delta \epsilon \in \mathcal{V}_0$, subtracting term by term, we get

$$\int_{\Omega} \Delta \boldsymbol{\sigma} \colon \Delta \boldsymbol{\epsilon} \mathrm{d}\Omega = 0 \Leftrightarrow \int_{\Omega} \Delta \boldsymbol{\sigma} \colon (\Delta \boldsymbol{\epsilon}^{e} + \Delta \boldsymbol{\epsilon}^{p}) \mathrm{d}\Omega = 0$$
$$\Leftrightarrow \int_{\Omega} \Delta \boldsymbol{\sigma} \colon \Delta \boldsymbol{\epsilon}^{e} \mathrm{d}\Omega + \int_{\Omega} \Delta \boldsymbol{\sigma} \colon \Delta \boldsymbol{\epsilon}^{p} \mathrm{d}\Omega = 0$$
(3.57)

In the dissipation error the state laws are satisfied, i.e.

$$\Delta \boldsymbol{\sigma} = \mathbf{C} \Delta \boldsymbol{\epsilon}^{e}$$

$$\Delta R = \mathsf{H} \Delta p.$$

Hence, it follows

$$\Delta \boldsymbol{\sigma} : \ \Delta \boldsymbol{\epsilon}^{e} = \mathbf{C} \Delta \boldsymbol{\epsilon}^{e} : \ \Delta \boldsymbol{\epsilon}^{e} \ge 0$$

$$\Delta \boldsymbol{\sigma} : \ \Delta \boldsymbol{\epsilon}^{p} = \Delta R \Delta p = \mathsf{H} \Delta p^{2} \ge 0.$$
(3.58)

By accounting for (3.56), from (3.57) the following results are obtained

$$\Delta \boldsymbol{\sigma} \colon \Delta \boldsymbol{\epsilon}^{e} = 0 \Rightarrow \Delta \boldsymbol{\sigma} = \mathbf{0} \Rightarrow \Delta \boldsymbol{\epsilon}^{e} = \mathbf{0}$$
$$\Delta \boldsymbol{\sigma} \colon \Delta \boldsymbol{\epsilon}^{p} = \Delta R \Delta p = 0 \Rightarrow \mathsf{H} \Delta p^{2} = 0 \Rightarrow \Delta p = 0 \stackrel{\text{(a)}}{\Rightarrow} \Delta \boldsymbol{\epsilon}^{p} = \mathbf{0}$$

where implication (a) follows from the condition imposed on the admissible solution to deliver a finite error, given by (3.51).

Plasticity Model P2: Standard variant Marquis-Chaboche model

Continuous Admissible Fictitious Solutions.

We assume, $\forall x \in \Omega$, continuous admissible fictitious solutions defined as follows

- □ Kinematic Admissible Solution: $\epsilon_{ad,\Delta t}(\boldsymbol{x},\tau), \epsilon_{ad,\Delta t}^{p}(\boldsymbol{x},\tau), \alpha_{ad,\Delta t}(\boldsymbol{x},\tau), p_{ad,\Delta t}(\boldsymbol{x},\tau)$ are obtained as linear interpolation over $[t_n, t_n + \Delta t]$ of the values at t_n and $t_n + \Delta t$.
- □ Static Admissible Solution: $\sigma_{ad,\Delta t}(\boldsymbol{x},\tau) = \mathbf{C}[\epsilon_{ad,\Delta t}(\boldsymbol{x},\tau) - \epsilon_{ad,\Delta t}^{p}(\boldsymbol{x},\tau)], \ \boldsymbol{X}_{ad,\Delta t}(\boldsymbol{x},\tau) = \Lambda \alpha_{ad,\Delta t}(\boldsymbol{x},\tau) \text{ and } R_{ad,\Delta t}(\boldsymbol{x},\tau) = \mathsf{H}p_{ad,\Delta t}(\boldsymbol{x},\tau).$
- $\Box \sigma_{ad,\Delta t}(\boldsymbol{x},\tau)$ is in equilibrium with $\boldsymbol{b}(\boldsymbol{x},t_n)$ because of the convexity of the equilibrium condition.
- $\Box \ (\boldsymbol{\sigma}_{ad,\Delta t}(\boldsymbol{x},\tau), \boldsymbol{X}_{ad,\Delta t}(\boldsymbol{x},\tau), R_{ad,\Delta t}(\boldsymbol{x},\tau)) \in \mathbb{E} \quad \forall \tau \in [t_n, t_n + \Delta t] \text{ because of the convexity of the elastic domain.}$
- $\Box \epsilon_{ad,\Delta t}(\boldsymbol{x},\tau)$ is kinematically admissible because of the convexity of the compatibility conditions.

Finite Value Error Requirements

We also impose that $\forall x \in \Omega$ the admissible solutions satisfy the following conditions

$$\begin{vmatrix}
(\sigma_{ad}(\boldsymbol{x}, t_{n}^{-}), \boldsymbol{X}_{ad}(\boldsymbol{x}, t_{n}^{-}), R_{ad}(\boldsymbol{x}, t_{n}^{-})) \in \mathbb{E} \\
(\sigma_{ad}(\boldsymbol{x}, \tau), \boldsymbol{X}_{ad}(\boldsymbol{x}, \tau), R_{ad}(\boldsymbol{x}, \tau)) \in \mathbb{E}
\end{vmatrix} \Rightarrow \begin{cases}
(\sigma_{ad}(\boldsymbol{x}, t_{n}^{+}), \boldsymbol{X}_{ad}(\boldsymbol{x}, t_{n}^{+}), R_{ad}(\boldsymbol{x}, t_{n}^{+})) \in \mathbb{E} \\
\forall \tau \in [t_{n}, t_{n} + \Delta t].
\end{aligned}$$

$$p_{ad}(\boldsymbol{x}, t_{n}^{+}) - p_{ad}(\boldsymbol{x}, t_{n}^{-}) \geq \|\boldsymbol{\epsilon}_{ad}^{p}(\boldsymbol{x}, t_{n}^{+}) - \boldsymbol{\epsilon}_{ad}^{p}(\boldsymbol{x}, t_{n}^{-})\|,$$

$$\operatorname{Tr}[\boldsymbol{\epsilon}_{ad}^{p}(\boldsymbol{x}, t_{n}^{+}) - \boldsymbol{\epsilon}_{ad}^{p}(\boldsymbol{x}, t_{n}^{-})] = 0,$$

$$\boldsymbol{\epsilon}_{ad}^{p}(\boldsymbol{x}, t_{n}^{+}) - \boldsymbol{\epsilon}_{ad}^{p}(\boldsymbol{x}, t_{n}^{-}) = \mathbf{0} \Rightarrow \boldsymbol{\alpha}_{ad}(\boldsymbol{x}, t_{n}^{+}) - \boldsymbol{\alpha}_{ad}(\boldsymbol{x}, t_{n}^{-}) = \mathbf{0}.$$
(3.60)

Condition (3.59) occurs because of the linearity of the state laws and convexity of the elastic domain, whereas condition (3.60) derives from the definition $(3.28)_2$ of

 $\varphi(\dot{\boldsymbol{\epsilon}}^p,-\dot{\boldsymbol{\alpha}},-\dot{p}).$

Dissipation Error across the time discontinuity

For this model we have to distinguish the following two cases in relation to the definition $(3.28)_2$ of the Fenchel-Legendre conjugate of the dissipation pseudo-potential

$\forall x \in \Omega$

□ If $\epsilon_{ad}^p(\boldsymbol{x}, t_n^+) - \epsilon_{ad}^p(\boldsymbol{x}, t_n^-) = 0$ then it can also be written $\alpha_{ad}^p(\boldsymbol{x}, t_n^+) - \alpha_{ad}^p(\boldsymbol{x}, t_n^-) = 0$ because of (3.60). By accounting for (3.28), the dissipation error takes the following expression

$${}^{\Delta}\zeta_{d}^{2}(\boldsymbol{x},t_{n}) = \lim_{\Delta t \to 0+} \int_{t_{n}}^{t_{n}+\Delta t} R_{ad,\Delta t}(\boldsymbol{x},\tau)\dot{p}_{ad,\Delta t}(\boldsymbol{x},\tau)d\tau =$$
$$= \frac{R_{ad}(\boldsymbol{x},t_{n}^{+}) + R_{ad}(\boldsymbol{x},t_{n}^{-})}{2}(p_{ad}(\boldsymbol{x},t_{n}^{+}) - p_{ad}(\boldsymbol{x},t_{n}^{-})), \qquad (3.61)$$

with the State Equations being satisfied at t_n^- and t_n^+ .

□ If $\epsilon_{ad}^p(\boldsymbol{x}, t_n^+) - \epsilon_{ad}^p(\boldsymbol{x}, t_n^-) \neq 0$, and recalling equation (3.28)₁, the dissipation error takes the following expression

with the State Equations being satisfied at t_n^- and t_n^+ .

Theorem 3.5.

Given the admissible solutions $s_{ad}(t_n^-)$ and $s_{ad}(t_n^+)$, satisfying the state equations at t_n^- and t_n^+ respectively, and the finite value error requirements, it follows that

$$^{\Delta}\zeta_d^2(\boldsymbol{x},t_n) \ge 0$$

Proof. If $\Delta \epsilon^p = 0$, $\Delta \alpha^p = 0$,

$$^{\Delta}\zeta_d^2(\boldsymbol{x},t_n) = \frac{R^+ + R^-}{2}\Delta p \ge 0$$

because of the finite value error requirements.

If $\Delta \epsilon^p \neq 0$,

$$^{\Delta}\zeta_{d}^{2}(\boldsymbol{x},t_{n}) = R_{0} \|\Delta\boldsymbol{\epsilon}^{p}\| + \frac{c}{2a} \frac{\|\Delta\boldsymbol{\epsilon}^{p} - \Delta\boldsymbol{\alpha}\|^{2}}{\|\Delta\boldsymbol{\epsilon}^{p}\|} - \frac{\boldsymbol{\sigma}^{+} + \boldsymbol{\sigma}^{-}}{2} : \Delta\boldsymbol{\epsilon}^{p} + \frac{\boldsymbol{X}^{+} + \boldsymbol{X}^{-}}{2} : \Delta\boldsymbol{\alpha} + \frac{R^{+} + R^{-}}{2} \Delta p.$$

$$(3.63)$$

Let us consider the term

$$R_0 \|\Delta \epsilon^p\| + \frac{c}{2a} \frac{\|\Delta \epsilon^p - \Delta \alpha\|^2}{\|\Delta \epsilon^p\|} - \boldsymbol{\sigma} \colon \Delta \epsilon^p + \boldsymbol{X} \colon \Delta \boldsymbol{\alpha} + R\Delta p, \qquad (3.64)$$

where σ , X, R can all refer either to the time instant t_n^+ or t_n^- . In the following, we will show that (3.64) is non negative, which then determines the non negativity of (3.63).

Since,

$$\Delta p \geq \|\Delta \boldsymbol{\epsilon}^p\|,$$

it follows

$$R\Delta p \ge R \|\Delta \epsilon^p\|.$$

Also,

$$\sigma: \Delta \epsilon^{p} = \sigma_{D}: \Delta \epsilon^{p} = (\sigma_{D} - X + X): \Delta \epsilon^{p} =$$
$$= (\sigma_{D} - X): \Delta \epsilon^{p} + X: \Delta \epsilon^{p} \le ||\sigma_{D} - X|| ||\Delta \epsilon^{p}|| + X:$$

 $\Delta \epsilon^p$

Hence,

$$-\boldsymbol{\sigma}: \ \Delta \boldsymbol{\epsilon}^p \geq -\|\boldsymbol{\sigma}_D - \boldsymbol{X}\| \|\Delta \boldsymbol{\epsilon}^p\| - \boldsymbol{X}: \ \Delta \boldsymbol{\epsilon}^p.$$

By accounting for the above inequalities, it results

$$R_{0}\|\Delta\epsilon^{p}\| + \frac{c}{2a} \frac{\|\Delta\epsilon^{p} - \Delta\alpha\|^{2}}{\|\Delta\epsilon^{p}\|} - \boldsymbol{\sigma} \colon \Delta\epsilon^{p} + \boldsymbol{X} \colon \Delta\boldsymbol{\alpha} + R\Delta p \geq \frac{c}{2a} \frac{\|\Delta\epsilon^{p} - \Delta\alpha\|^{2}}{\|\Delta\epsilon^{p}\|} + (R_{0} + R)\|\Delta\epsilon^{p}\| - \|\boldsymbol{\sigma}_{D} - \boldsymbol{X}\|\|\Delta\epsilon^{p}\| + -\boldsymbol{X} \colon (\Delta\epsilon^{p} - \Delta\alpha).$$

From the yield condition, it follows

$$(R_0+R) - \|\boldsymbol{\sigma}_D - \boldsymbol{X}\| \geq \frac{a}{2c} \|\boldsymbol{X}\|^2,$$

hence,

$$\left[(R_0 + R) - \|\boldsymbol{\sigma}_D - \boldsymbol{X}\| \right] \|\Delta \boldsymbol{\epsilon}^p\| \ge \frac{a}{2c} \|\boldsymbol{X}\|^2 \|\Delta \boldsymbol{\epsilon}^p\|$$

Also, it follows

$$-\boldsymbol{X}: \ (\Delta \boldsymbol{\epsilon}^p - \Delta \boldsymbol{\alpha}) \geq - \|\boldsymbol{X}\| \|\Delta \boldsymbol{\epsilon}^p - \Delta \boldsymbol{\alpha}\|$$

so that finally, we have

$$R_{0}\|\Delta\epsilon^{p}\| + \frac{c}{2a} \frac{\|\Delta\epsilon^{p} - \Delta\alpha\|^{2}}{\|\Delta\epsilon^{p}\|} - \boldsymbol{\sigma}: \ \Delta\epsilon^{p} + \boldsymbol{X}: \ \Delta\alpha + R\Delta p \geq \\ \geq \frac{c}{2a} \frac{\|\Delta\epsilon^{p} - \Delta\alpha\|^{2}}{\|\Delta\epsilon^{p}\|} + \frac{a}{2c} \|\boldsymbol{X}\|^{2} \|\Delta\epsilon^{p}\| - \|\boldsymbol{X}\| \|\Delta\epsilon^{p} - \Delta\alpha\| = \\ = \left(\sqrt{\frac{c}{2a}} \frac{\|\Delta\epsilon^{p} - \Delta\alpha\|}{\sqrt{\|\Delta\epsilon^{p}\|}} - \sqrt{\frac{a}{2c}} \sqrt{\|\Delta\epsilon^{p}\|} \|\boldsymbol{X}\|\right)^{2} \geq 0 \quad \Box \qquad (3.65)$$

Theorem 3.6.

Given the admissible solutions $s_{ad}(t_n^-)$ and $s_{ad}(t_n^+)$, satisfying the state equations at t_n^- and t_n^+ respectively, and the finite value error requirements, it follows that

$$\begin{tabular}{l} & \Delta \zeta_d^2({\boldsymbol{x}},t_n) = 0 \end{tabular} \Rightarrow \end{tabular} \left\{ \begin{array}{l} \boldsymbol{\sigma}_{ad}({\boldsymbol{x}},t_n^+) = \boldsymbol{\sigma}_{ad}({\boldsymbol{x}},t_n^-) \\ & \boldsymbol{X}_{ad}({\boldsymbol{x}},t_n^+) = \boldsymbol{X}_{ad}({\boldsymbol{x}},t_n^-) \\ & \boldsymbol{R}_{ad}({\boldsymbol{x}},t_n^+) = \boldsymbol{R}_{ad}({\boldsymbol{x}},t_n^-) \\ & \boldsymbol{\epsilon}_{ad}({\boldsymbol{x}},t_n^+) = \boldsymbol{\epsilon}_{ad}({\boldsymbol{x}},t_n^-) \\ & \boldsymbol{\epsilon}_{ad}^p({\boldsymbol{x}},t_n^+) = \boldsymbol{\epsilon}_{ad}^p({\boldsymbol{x}},t_n^-) \\ & \boldsymbol{\alpha}_{ad}({\boldsymbol{x}},t_n^+) = \boldsymbol{\alpha}_{ad}({\boldsymbol{x}},t_n^-) \\ & \boldsymbol{p}_{ad}({\boldsymbol{x}},t_n^+) = p_{ad}({\boldsymbol{x}},t_n^-) \end{array} \right.$$

Proof. We again have to distinguish the two cases in relation to the expression of the dissipation error as given by equation (3.61) and (3.62). We first consider the case of $\Delta \epsilon^p \neq 0$ and then $\Delta \epsilon^p = 0$, $\Delta \alpha = 0$.

If $\Delta \epsilon^p \neq 0$, given the expression (3.62), and accounting for (3.65), it follows

$$If \ ^{\Delta}\zeta_{d}^{2}(\boldsymbol{x}, t_{n}) = 0 \Rightarrow \begin{cases} \boldsymbol{\sigma}^{+} : \ \Delta \boldsymbol{\epsilon}^{p} = R_{0} \| \Delta \boldsymbol{\epsilon}^{p} \| + \frac{c}{2a} \frac{\| \Delta \boldsymbol{\epsilon}^{p} - \Delta \boldsymbol{\alpha} \|}{\| \Delta \boldsymbol{\epsilon}^{p} \|} + \\ + R^{+} \Delta p + \boldsymbol{X}^{+} : \ \Delta \boldsymbol{\alpha} \end{cases}$$
$$\boldsymbol{\sigma}^{-} : \ \Delta \boldsymbol{\epsilon}^{p} = R_{0} \| \Delta \boldsymbol{\epsilon}^{p} \| + \frac{c}{2a} \frac{\| \Delta \boldsymbol{\epsilon}^{p} - \Delta \boldsymbol{\alpha} \|}{\| \Delta \boldsymbol{\epsilon}^{p} \|} + \\ + R^{-} \Delta p + \boldsymbol{X}^{-} : \ \Delta \boldsymbol{\alpha} \end{cases}$$

thus,

$$\Delta \boldsymbol{\sigma} \colon \Delta \boldsymbol{\epsilon}^p = \Delta R \Delta p + \Delta \boldsymbol{X} \colon \Delta \boldsymbol{\alpha}$$
(3.66)

From equilibrium at t_n^- and t_n^+ it follows (see equation (3.57))

$$\int_{\Omega} \Delta \boldsymbol{\sigma} \colon \Delta \boldsymbol{\epsilon} \mathrm{d}\Omega = 0 \Leftrightarrow \int_{\Omega} \Delta \boldsymbol{\sigma} \colon \Delta \boldsymbol{\epsilon}^{e} \mathrm{d}\Omega + \int_{\Omega} \Delta \boldsymbol{\sigma} \colon \Delta \boldsymbol{\epsilon}^{p} \mathrm{d}\Omega = 0$$
(3.67)

In the dissipation error the state laws are satisfied, i.e.

$$\Delta \boldsymbol{\sigma} = \mathbf{C} \Delta \boldsymbol{\epsilon}^{e}$$
$$\Delta R = \mathbf{H} \Delta p$$
$$\Delta \boldsymbol{X} = \mathbf{\Lambda} \Delta \boldsymbol{\alpha}$$

hence, it follows

$$\Delta \boldsymbol{\sigma} \colon \Delta \boldsymbol{\epsilon}^{e} = \mathbf{C} \Delta \boldsymbol{\epsilon}^{e} \colon \Delta \boldsymbol{\epsilon}^{e} \ge 0$$

$$\Delta \boldsymbol{\sigma} \colon \Delta \boldsymbol{\epsilon}^{p} = \Delta R \Delta p + \Delta \boldsymbol{X} \colon \Delta \boldsymbol{\alpha} = \mathsf{H} \Delta p^{2} + \mathbf{\Lambda} \Delta \boldsymbol{\alpha} \colon \Delta \boldsymbol{\alpha} \ge 0$$
(3.68)

for Λ is a second order positive definite tensor. By accounting for (3.68), from (3.67) it results

$$\Delta \boldsymbol{\sigma} \colon \Delta \boldsymbol{\epsilon}^{e} = 0 \Rightarrow \Delta \boldsymbol{\sigma} = \mathbf{0} \Rightarrow \Delta \boldsymbol{\epsilon}^{e} = \mathbf{0}$$
$$\Delta \boldsymbol{\sigma} \colon \Delta \boldsymbol{\epsilon}^{p} = \Delta R \Delta p + \Delta \boldsymbol{X} \colon \Delta \boldsymbol{\alpha} = 0 \Rightarrow$$
$$\Rightarrow \begin{vmatrix} \mathsf{H} \Delta p^{2} = 0 \Rightarrow \Delta p = 0 \stackrel{(a)}{\Rightarrow} \Delta \boldsymbol{\epsilon}^{p} = \mathbf{0} \\ \Lambda \Delta \boldsymbol{\alpha} \colon \Delta \boldsymbol{\alpha} = 0 \Rightarrow \Delta \boldsymbol{\alpha} = \mathbf{0} \end{vmatrix}$$

where implication (a) follows from the condition imposed on the admissible solution to deliver a finite error.

If $\Delta \epsilon^p = 0$, $\Delta \alpha = 0$, the dissipation error is given by

$${}^{\Delta}\zeta_d^2(\boldsymbol{x}, t_n) = \frac{R^- + R^+}{2}\Delta p = \frac{\mathsf{H}}{2}(p^{+2} - p^{-2})$$

where we have taken into account the meeting of the state equations at t_n^- and t_n^+ . Thus, it follows

$${}^{\Delta}\zeta_d^2(\boldsymbol{x},t_n) = 0 \Rightarrow p^{+^2} = p^{-^2} \Rightarrow p^+ = p^-, \text{ for } p \ge 0$$

which delivers

$$\Delta p = 0 \Rightarrow \Delta R = 0$$

From equations (3.67) and (3.68) we derive also

$$\Delta \epsilon^e = \mathbf{0} \Rightarrow \Delta \sigma = \mathbf{0}.$$

3.5.2.2 Augmented Extended Dissipation Error

The extended dissipation error at the time T of the admissible solution with jump across the time $t_n \in [0, T[$ is given by

$$^{\Delta}e_{ext}^{2}(T) = \sup_{t \leq T} \left\{ \underbrace{2 \int_{\Omega} {}^{sl} \eta_{x,t}^{2}(\boldsymbol{\sigma}_{ad}, \boldsymbol{X}_{ad}, \boldsymbol{R}_{ad}; \boldsymbol{\epsilon}_{ad}^{e}, \boldsymbol{\alpha}_{ad}, \boldsymbol{p}_{ad}) \mathrm{d}\Omega}_{\boldsymbol{\theta}_{sl}^{2}(t)} + \underbrace{2 \int_{\Omega} \int_{0}^{t} {}^{d} \eta_{x,\tau}^{2}(\boldsymbol{\sigma}_{ad}, \boldsymbol{X}_{ad}, \boldsymbol{R}_{ad}; \boldsymbol{\epsilon}_{ad}^{p}, \boldsymbol{\alpha}_{ad}, \boldsymbol{p}_{ad}) \mathrm{d}\tau \, \mathrm{d}\Omega + 2 \int_{\Omega} {}^{\Delta} \zeta_{d}^{2}(x, t_{n}) \mathrm{d}\Omega}_{\boldsymbol{\theta}_{sl}^{2}(t)} \right\}$$

$$(3.69)$$

where for ${}^{\Delta}\zeta_d^2(\boldsymbol{x}, t_n)$ we consider the same expressions as given in Section 3.5.2.1. For the applications, it is convenient to rewrite equation (3.69) as follows,

$$^{\Delta}e_{ext}^{2}(T) = \mathrm{MAX} \left\{ \underbrace{\sup_{t \leq t_{n}^{-}} \left[\underbrace{2 \int_{\Omega} {}^{sl} \eta_{x,t}^{2} \, \mathrm{d}\Omega}_{\theta_{sl}^{2}(t)} + \underbrace{2 \int_{\Omega} \int_{0}^{t} {}^{d} \eta_{x,\tau}^{2} \, \mathrm{d}\tau \, \mathrm{d}\Omega}_{\theta_{d}^{2}(t)} \right], }_{\theta_{sl}^{2}(t)} \right.$$

$$\sup_{t_{n}^{+} \leq t \leq T} \left[\underbrace{2 \int_{\Omega} {}^{sl} \eta_{x,t}^{2} \, \mathrm{d}\Omega}_{\theta_{d}^{2}(t_{n})} + \underbrace{2 \int_{\Omega} {}^{\Delta} \zeta_{d}^{2}(x,t_{n}) \, \mathrm{d}\Omega}_{\Delta \theta_{d}^{2}(t_{n})} + \underbrace{2 \int_{\Omega} {}^{\int_{t_{n}^{+}} {}^{t} \eta_{x,\tau}^{2} \, \mathrm{d}\tau \, \mathrm{d}\Omega}_{\theta_{d}^{2}(t_{n})} \right] \right\}$$

$$(3.70)$$

which highlights the different contributions to the error from the parts of the admissible solution which are continuous in time.

The next result guarantees that definition (3.69) can be assumed as measure of the error in the following sense

Theorem 3.7.

Given the admissible solutions $s_{ad}(t_n^-)$ and $s_{ad}(t_n^+)$, meeting the finite value error requirements and not necessarily the state equations at t_n^- and t_n^+ , respectively, it follows that

$$\begin{split} ^{\Delta} e^2_{ext}(T) &\geq 0 \\ ^{\Delta} e^2_{ext}(T) &= 0 \Longleftrightarrow s_{ad}(\boldsymbol{x},t) = s_{ex}(\boldsymbol{x},t) \quad \forall \, \boldsymbol{x} \in \Omega, \quad \forall \, t \leq T. \end{split}$$

Proof. The extended dissipation error ${}^{\Delta}e_{ext}^2(T)$ is defined as the supremum of a function which is sum of non negative terms, thus

$$^{\Delta}e_{ext}^{2}(T) = 0 \Rightarrow \begin{cases} \theta_{sl}^{2}(t) = 0 \forall t \leq T \quad \Leftrightarrow \quad \left| \begin{array}{c} {}^{sl}\eta_{x,t}^{2} = 0 \\ \forall x \in \Omega, \quad \forall t \leq T. \end{array} \right. (a) \\ \\ \left| \begin{array}{c} \Delta e_{ext}^{2}(T) = 0 \Rightarrow t \leq T \quad \Leftrightarrow \\ \Delta \theta_{d}^{2}(T) = 0 \forall t \leq T \quad \Leftrightarrow \\ \end{array} \right. \\ \left| \begin{array}{c} \Delta \zeta_{d}^{2}(x,t_{n}) = 0 \\ \forall x \in \Omega, \end{array} \right. (b) \\ \\ \left| \begin{array}{c} \partial \eta_{x,t}^{2} = 0 \\ \forall x \in \Omega, \quad \forall t \leq T. \end{array} \right. \\ \left| \begin{array}{c} \partial \eta_{x,t}^{2} = 0 \\ \forall x \in \Omega, \quad \forall t \leq T. \end{array} \right. \end{cases} \end{cases}$$

Condition (3.71a) means that the admissible solution $s_{ad}(x,t)$ satisfies the state laws, $\forall x \in \Omega$, $\forall t \leq T$. This condition along with (3.71b) allows one to conclude also that the jump in all the variables is zero as a result of the argument given in Section 3.5.2.1 for the rate independent plasticity models taken into account. Finally, condition (3.71c) allows to conclude that the time continuous admissible solution satisfies also the evolution law.

Even for the extended dissipation error a characterization of only the discontinuity can be given. Hereafter, the condition is proved only for the Prandtl Reuss model with linear hardening. The argument applies likewise to the standard variant Marquis-Chaboche model with linear hardening. For both the models with more general hardening laws, we believe, however, that similar conclusions can be obtained.

The condition characterizing the discontinuity is next given in a more general format which applies to admissible solutions with jump across time instant t_n in the case of rate-independent plasticity.

Denote by

$$s_{ad}(\boldsymbol{x}, t_n) = (\boldsymbol{\sigma}_{ad}(\boldsymbol{x}, t_n), R_{ad}(\boldsymbol{x}, t_n); \boldsymbol{\epsilon}_{ad}(\boldsymbol{x}, t_n), \boldsymbol{\epsilon}_{ad}^{\boldsymbol{p}}(\boldsymbol{x}, t_n), p_{ad}(\boldsymbol{x}, t_n))$$

and

$$s_{ad}(\boldsymbol{x}, t_n + \Delta t) = (\boldsymbol{\sigma}_{ad}(\boldsymbol{x}, t_n + \Delta t), R_{ad}(\boldsymbol{x}, t_n + \Delta t);$$

$$\boldsymbol{\epsilon}_{ad}(\boldsymbol{x}, t_n + \Delta t), \boldsymbol{\epsilon}_{ad}^p(\boldsymbol{x}, t_n + \Delta t), p_{ad}(\boldsymbol{x}, t_n + \Delta t))$$

any admissible solution at t_n and $t_n + \Delta t$, respectively, corresponding to the same load level and with $s_{ad}(t)$ the admissible solution obtained as time linear interpolation over $[t_n, t_n + \Delta t]$ of $s_{ad}(t_n)$ and $s_{ad}(t_n + \Delta t)$. The following theorem, then, can be stated

Theorem 3.8.

Given the admissible solutions $s_{ad}(t_n)$ and $s_{ad}(t_n + \Delta t)$, corresponding to the same load level, meeting the finite value error requirements and not necessarily the state equations at t_n and $t_n + \Delta t$, respectively, it follows $\forall x \in \Omega$,

$$\operatorname{IF} \begin{cases} {}^{sl}\eta_{\boldsymbol{x},t_n}^2 = {}^{sl}\eta_{\boldsymbol{x},t_n+\Delta t}^2 = {}^{sl}\eta_{\boldsymbol{x},t}^2 \\ \forall t \in [t_n, t_n + \Delta t]. \\ \Delta \zeta_{\boldsymbol{x},d}^2(\boldsymbol{x},t_n) = 0 \end{cases} \Rightarrow \begin{cases} \boldsymbol{\sigma}_{ad}(\boldsymbol{x},t_n) = \boldsymbol{\sigma}_{ad}(\boldsymbol{x},t_n + \Delta t) \\ R_{ad}(\boldsymbol{x},t_n) = R_{ad}(\boldsymbol{x},t_n + \Delta t) \\ \boldsymbol{\epsilon}_{ad}(\boldsymbol{x},t_n) = \boldsymbol{\epsilon}_{ad}(\boldsymbol{x},t_n + \Delta t) \end{cases}$$

Proof. For the Prandtl-Reuss model with linear hardening, the hypothesis of the theorem concerning the error in the state law writes as

$$\int_{\Omega} (\boldsymbol{\sigma}_{ad}(t) - \mathbf{C}\boldsymbol{\epsilon}_{ad}^{e}(t)): \ \mathbf{C}^{-1}(\boldsymbol{\sigma}_{ad}(t) - \mathbf{C}\boldsymbol{\epsilon}_{ad}^{e}(t)) \mathrm{d}\boldsymbol{x} + \frac{1}{\mathsf{H}} \int_{\Omega} (R_{ad}(t) - \mathsf{H}p_{ad}(t))^{2} \mathrm{d}\boldsymbol{x} = \mathrm{const} \quad \forall t \in [t_{n}, t_{n} + \Delta t].$$

Thus, differentiation with respect to time delivers

$$\begin{split} &\int_{\Omega} (\boldsymbol{\sigma}_{ad}(t) - \mathbf{C}\boldsymbol{\epsilon}_{ad}^{e}(t)) \colon \, \mathbf{C}^{-1}(\dot{\boldsymbol{\sigma}}_{ad}(t) - \mathbf{C}\dot{\boldsymbol{\epsilon}}_{ad}^{e}(t)) \mathrm{d}\boldsymbol{x} + \\ &+ \frac{1}{\mathsf{H}} \int_{\Omega} (R_{ad}(t) - \mathsf{H}p_{ad}(t))(\dot{R}_{ad}(t) - \mathsf{H}\dot{p}_{ad}(t)) \mathrm{d}\boldsymbol{x} = 0 \quad \forall t \in [t_n, \, t_n + \Delta t] \end{split}$$

which, because of the definition of $s_{ad}(t)$, can be written as

$$\int_{\Omega} (\boldsymbol{\sigma}_{ad}(t) - \mathbf{C}\boldsymbol{\epsilon}_{ad}^{e}(t)): \ \mathbf{C}^{-1}(\Delta\boldsymbol{\sigma}_{ad} - \mathbf{C}\Delta\boldsymbol{\epsilon}_{ad}^{e}) \mathrm{d}\boldsymbol{x} +$$

$$+ \frac{1}{\mathsf{H}} \int_{\Omega} (R_{ad}(t) - \mathsf{H}p_{ad}(t))(\Delta R_{ad} - \mathsf{H}\Delta p_{ad}) \mathrm{d}\boldsymbol{x} = 0 \quad \forall t \in [t_{n}, t_{n} + \Delta t]$$
(3.72)

where the same notation as in the proof of Theorem 3.3 has been employed provided that t_n^- is meant as t_n and t_n^+ as $t_n + \Delta t$.

After computing equation (3.72) for $t = t_n$ and $t = t_n + \Delta t$ and subtracting side to side one obtains

$$\int_{\Omega} (\Delta \boldsymbol{\sigma}_{ad} - \mathbf{C} \Delta \boldsymbol{\epsilon}_{ad}^{e}) \colon \mathbf{C}^{-1} (\Delta \boldsymbol{\sigma}_{ad} - \mathbf{C} \Delta \boldsymbol{\epsilon}_{ad}^{e}) \mathrm{d}\boldsymbol{x} + \frac{1}{\mathsf{H}} \int_{\Omega} (\Delta R_{ad} - \mathsf{H} \Delta p_{ad}) (\Delta R_{ad} - \mathsf{H} \Delta p_{ad}) \mathrm{d}\boldsymbol{x} = 0,$$

that is,

$$\int_{\Omega} (\Delta \boldsymbol{\sigma}_{ad} - \mathbf{C} \Delta \boldsymbol{\epsilon}_{ad}^{e}) : \mathbf{C}^{-1} (\Delta \boldsymbol{\sigma}_{ad} - \mathbf{C} \Delta \boldsymbol{\epsilon}_{ad}^{e}) d\boldsymbol{x} = 0$$
$$\frac{1}{\mathsf{H}} \int_{\Omega} (\Delta R_{ad} - \mathsf{H} \Delta p_{ad})^{2} d\boldsymbol{x} = 0,$$

so that the following relations hold

$$\Delta \boldsymbol{\sigma}_{ad} - \mathbf{C} \Delta \boldsymbol{\epsilon}_{ad}^{e} = \mathbf{0}$$
$$\Delta R_{ad} - \mathbf{H} \Delta p_{ad} = 0,$$

and the same arguments as in the proof of Theorem 3.3 can now be adopted. $\hfill \Box$

3.5.3 Definition of error in solution

Let

$$s_{ex}(x,t) = \left(\sigma_{ex}(x,t), X_{ex}(x,t), R_{ex}(x,t); \epsilon_{ex}(x,t), \epsilon_{ex}^{p}(x,t), \alpha_{ex}(x,t), p_{ex}(x,t)\right)$$

denote the exact solution of the initial boundary value problem defined in section 3.2, that is, $s_{ex}(x,t)$ is the time continuos function that meets all the equations given in Box 3.1. Let

$$s_{ad}^{kin}(\boldsymbol{x},t) = \left(\boldsymbol{u}_{ad}(\boldsymbol{x},t), \boldsymbol{\epsilon}_{ad}^{p}(\boldsymbol{x},t), \boldsymbol{\alpha}_{ad}(\boldsymbol{x},t), p_{ad}(\boldsymbol{x},t)\right),$$
(3.73)

be a kinematically admissible solution with \boldsymbol{u} meeting the compatibility conditions, and $\boldsymbol{\epsilon}_{ad}^{p}(\boldsymbol{x},t), \boldsymbol{\alpha}_{ad}(\boldsymbol{x},t), p_{ad}(\boldsymbol{x},t)$ meeting the initial conditions. The kinematically admissible solution may also present discontinuity jump across time instants t_n . We assume the error in the constitutive equations produced by

$$s_{ex,ad} = \left(\boldsymbol{\sigma}_{ex}, \boldsymbol{X}_{ex}, R_{ex}; \boldsymbol{u}_{ad}, \boldsymbol{\epsilon}_{ad}^{p}, \boldsymbol{\alpha}_{ad}, p_{ad}\right),$$

as global measure of the exact error in solution associated with s_{ad}^{kin} . This is defined as

$$e_{ex}^2(T) = \sup_{t \le T} \theta_{ex}^2(t), \qquad (3.74)$$

where

$$\theta_{ex}^{2}(t) = 2 \int_{\Omega}^{sl} \eta_{x,t}^{2}(\boldsymbol{\sigma}_{ex}, \boldsymbol{X}_{ex}, R_{ex}; \boldsymbol{\epsilon}_{ad}^{e}, \boldsymbol{\alpha}_{ad}, p_{ad}) \mathrm{d}\Omega \Big|_{t} + 2 \int_{\Omega} \int_{0}^{t} d\eta_{x,\tau}^{2}(\boldsymbol{\sigma}_{ex}, \boldsymbol{X}_{ex}, R_{ex}; \dot{\boldsymbol{\epsilon}}_{ad}^{p}, \dot{\boldsymbol{\alpha}}_{ad}, \dot{p}_{ad}) \mathrm{d}\tau \mathrm{d}\Omega.$$

The following heuristic argument show that this definition of error is meaningful. First, note that

$$e_{ex}^{2}(T) \text{ FINITE} \Rightarrow \begin{cases} (\boldsymbol{\sigma}_{ex}, \boldsymbol{X}_{ex}, R_{ex}) \in \mathbb{E} & \forall \boldsymbol{x} \in \Omega, \quad \forall t \leq T \quad (a) \\ (\dot{\boldsymbol{\epsilon}}_{ad}^{p}, \dot{\boldsymbol{\alpha}}_{ad}, \dot{p}_{ad}) \in \mathbb{C} & \forall \boldsymbol{x} \in \Omega, \quad \forall t \leq T \quad (b) \end{cases}$$
(3.75)

where $\mathbb{E} \times \mathbb{C}$ is the effective domain of the function ${}^{d}\eta^{2}_{\boldsymbol{x},t}(\boldsymbol{\sigma}, \boldsymbol{X}, R; \dot{\boldsymbol{\epsilon}}^{p}, \dot{\boldsymbol{\alpha}}, \dot{p})$. For the plasticity models under consideration, if the admissible solution is discontinuous across the time instant t_{n} , conditions (3.75b) imply that

$$\Delta p_{ad} \ge \|\Delta \boldsymbol{\epsilon}_{ad}^p\|. \tag{3.76}$$

Now, we can give the following result:

Theorem 3.9.

Given a kinematic admissible solution $s_{ad}^{kin} = (u_{ad}, \epsilon_{ad}^{p}, \alpha_{ad}, p_{ad})$, it follows that

$$e_{ex}^{2}(T) = 0$$
 IF and ONLY IF $\begin{cases} s_{ex,ad} \text{ is time continuous} \\ s_{ex,ad}(x,t) = s_{ex}(x,t) \quad \forall x \in \Omega, \quad \forall t \leq T \end{cases}$

Proof. Denote with

$$egin{aligned} & ilde{\pmb{\sigma}} = \pmb{\mathsf{C}}[\pmb{\epsilon}_{ad} - \pmb{\epsilon}^p_{ad}], \ & ilde{\pmb{X}} = \pmb{\Lambda}\pmb{lpha}_{ad}, \ & ilde{\pmb{R}} = g(p_{ad}) \end{aligned}$$

the forces conjugate to the admissible kinematic variables $(\epsilon_{ad}, \epsilon_{ad}^p, p_{ad}, \alpha_{ad})$. Since $e_{ex}^2(T)$ is defined as the supremum of a function which is sum of non negative terms, it follows

$$e_{ex}^{2}(T) = 0 \Rightarrow \begin{cases} {}^{sl}\eta_{x,t}^{2} = 0 & \forall x \in \Omega, \quad \forall t \leq T, \quad (a) \\ {}^{d}\eta_{x,t}^{2} = 0 & \forall x \in \Omega, \quad \forall t \leq T. \quad (b) \end{cases}$$

Condition (a) means that

$$\begin{split} \tilde{\boldsymbol{\sigma}}(\boldsymbol{x},t) &= \boldsymbol{\sigma}_{ex}(\boldsymbol{x},t) \quad \forall \, \boldsymbol{x} \in \Omega, \quad \forall \, t \leq T, \\ \tilde{\boldsymbol{X}}(\boldsymbol{x},t) &= \boldsymbol{X}_{ex}(\boldsymbol{x},t) \quad \forall \, \boldsymbol{x} \in \Omega, \quad \forall \, t \leq T, \\ \tilde{R}(\boldsymbol{x},t) &= R_{ex}(\boldsymbol{x},t) \quad \forall \, \boldsymbol{x} \in \Omega, \quad \forall \, t \leq T. \end{split}$$

Because of the time continuity of $(\sigma_{ex}(x, t), X_{ex}(x, t), R_{ex}(x, t))$ and of the continuity of the functional relations that define the state equations, the time continuity of $(\epsilon^{e}_{ad}, p_{ad}, \alpha_{ad})$ also follows. In turn, the time continuity of p_{ad} , along with the condition (3.76) implies also the continuity of ϵ^{p}_{ad} , hence the continuity of $\epsilon_{ad} = \epsilon^{e}_{ad} + \epsilon^{p}_{ad}$, as well. Then, it follows that

$$\left(\tilde{\boldsymbol{\sigma}}, \ \tilde{\boldsymbol{X}}, \ \tilde{R}; \ \boldsymbol{u}_{ad}, \boldsymbol{\epsilon}^p_{ad}, \boldsymbol{\alpha}_{ad}, p_{ad}\right) = s_{ex,ad}$$

is time continuous, statically admissible, meets the state laws and the evolution laws. Thus it coincides with the exact solution, s_{ex} .

3.5.3.1 Extension of the Prager-Synge theorem to the Dissipation Error

In the case of the dissipation error, for a time continuous admissible solution s_{ad} , Ladevèze (1999) shows that the following equation holds for a normal formulation of the model

$$\int_{\Omega} \int_{0}^{t} d\eta_{\boldsymbol{x},\tau}^{2} (\boldsymbol{\sigma}_{ad}, \boldsymbol{X}_{ad}, \boldsymbol{R}_{ad}; \dot{\boldsymbol{\epsilon}}_{ad}^{p}, \dot{\boldsymbol{\alpha}}_{ad}, \dot{\boldsymbol{p}}_{ad}) d\tau d\Omega =
= \frac{1}{2} \int_{\Omega} (\boldsymbol{\sigma}_{ex} - \boldsymbol{\sigma}_{ad}) : \mathbf{C}^{-1} (\boldsymbol{\sigma}_{ex} - \boldsymbol{\sigma}_{ad}) d\Omega \Big|_{t} +
+ \frac{1}{2} \int_{\Omega} (\boldsymbol{X}_{ex} - \boldsymbol{X}_{ad}) : \mathbf{\Lambda}^{-1} (\boldsymbol{X}_{ex} - \boldsymbol{X}_{ad})] d\Omega \Big|_{t} +
+ \frac{1}{2} \int_{\Omega} \left[(\boldsymbol{R}_{ex} - \boldsymbol{R}_{ad}) \mathbf{H}^{-1} (\boldsymbol{R}_{ex} - \boldsymbol{R}_{ad}) \right] d\Omega \Big|_{t} +
+ \int_{\Omega} \int_{0}^{t} d\eta_{\boldsymbol{x},\tau}^{2} (s_{ad}, s_{ex}) d\tau d\Omega$$
(3.77)

where

$${}^{d}\eta^{2}_{\boldsymbol{x},t}(s_{ad}, s_{ex}) = {}^{d}\eta^{2}_{\boldsymbol{x},t}(s_{ex,ad}) + {}^{d}\eta^{2}_{\boldsymbol{x},t}(s_{ad,ex})$$

with

$$s_{ex,ad} = \left(\boldsymbol{\sigma}_{ex}, \boldsymbol{X}_{ex}, R_{ex}; \boldsymbol{u}_{ad}, \boldsymbol{\epsilon}_{ad}^{p}, \boldsymbol{\alpha}_{ad}, p_{ad}\right),$$

$$s_{ad,ex} = \left(\boldsymbol{\sigma}_{ad}, \boldsymbol{X}_{ad}, R_{ad}; \boldsymbol{u}_{ex}, \boldsymbol{\epsilon}_{ex}^{p}, \boldsymbol{\alpha}_{ex}, p_{ex}\right),$$

which can be easily obtained by elaborating on the expression of ${}^d\eta^2_{\boldsymbol{x},t}(s_{ad})$ and accounting of the properties of the exact solution s_{ex} .

Equation (3.77) can be considered as an extension of the Prager-Synge theorem to the Dissipation Error and, likewise for the linear elasticity, allows one to show easily that the dissipation error is an upper bound for the error in the solution as defined by equation (3.74). Indeed, we have the following result:

Theorem 3.10.

Given an admissible solution, s_{ad} , with respect to the computation of the dissipation error, it follows

$$e_{dis}^2(T) \ge e_{ex}^2(T)$$
 (3.78)

Proof. Let

$$\begin{aligned} \theta_{ex}^{2}(t) &= \int_{\Omega} \left(\boldsymbol{\sigma}_{ex} - \boldsymbol{\sigma}_{ad} \right) \colon \mathbf{C}^{-1} \left(\boldsymbol{\sigma}_{ex} - \boldsymbol{\sigma}_{ad} \right) \mathrm{d}\Omega \Big|_{t} + \\ &+ \int_{\Omega} \left(\boldsymbol{X}_{ex} - \boldsymbol{X}_{ad} \right) \colon \boldsymbol{\Lambda}^{-1} \left(\boldsymbol{X}_{ex} - \boldsymbol{X}_{ad} \right) \mathrm{d}\Omega \Big|_{t} + \\ &+ \int_{\Omega} \left[\left(R_{ex} - R_{ad} \right) \mathbf{H}^{-1} \left(R_{ex} - R_{ad} \right) \right] \mathrm{d}\Omega \Big|_{t} + \\ &+ 2 \int_{\Omega} \int_{0}^{t} d\eta_{x,\tau}^{2}(s_{ex,ad}) \, \mathrm{d}\tau \, \mathrm{d}\Omega \end{aligned}$$

 and

$$\theta_d^2(t) = 2 \int_{\Omega} \int_0^t {}^d \eta_{\boldsymbol{x},\tau}^2(s_{ad}) \,\mathrm{d}\tau \,\mathrm{d}\Omega.$$

Since ${}^{d}\eta^{2}_{\boldsymbol{x},\tau}(s_{ex,ad}) \geq 0$ and ${}^{d}\eta^{2}_{\boldsymbol{x},\tau}(s_{ad,ex}) \geq 0$, it follows from equation (3.77),

$$\theta_d^2(t) \ge \theta_{ex}^2(t) \quad \forall t \le T.$$

Thus, it is

$$\sup_{t \le T} \theta_d^2(t) \ge \sup_{t \le T} \theta_{ex}^2(t) = e_{ex}^2(T)$$

where

$$\sup_{t < T} \theta_d^2(t) = \theta_d^2(T) \equiv e_{dis}^2(T),$$

since $\theta_d^2(t)$ is an increasing function of time.

3.6 Concluding Remarks

This Chapter represents the theoretical core of the thesis. Here, the theory of the error in the constitutive equations for material models with internal variables and associative flow rule has been presented following the works of Ladevèze (1989) and Ladevèze *et al.* (1999).

The fundamental notion of admissible solution has been given. The error in the constitutive equations developed by Ladevèze *et al.* (1999) has been extended to admissible solutions with discontinuity jump at the time instant t_n in the case of rate-independent plasticity models. Theorem 3.8 represents the main proposition of the Chapter. The theorem provides a characterization of the discontinuity jump in the admissible solution as a function of the augmented term $^{\Delta}\zeta_{x,d}^{2}(x,t_{n})$ and of the behaviour of the error component associated with the residual in the state law.

As a result, the augmented extended dissipation error can be employed as a basis of methodology for the assessment of the global accuracy in time of finite element solutions on evolving meshes.

With this regard, in the next Chapter we first recall the governing finite element equations and subsequently, we continue discussing the several sources of discretization errors which are introduced, in particular those arising from the change of finite element mesh from one time increment to the other.

Part II

Application to the Finite Element Solution of the IBVP in Elasto-Plasticity
Chapter 4

The Finite Element Solution of the IBVP in Elasto-Plasticity

4.1 Introduction

In this chapter we report on the displacement finite element method for the solution of the initial boundary value problem of an elastoplastic model with internal variables. After reformulating the general problem in a way which presents the displacement field $\boldsymbol{u} = \boldsymbol{u}(\boldsymbol{x},t) \in \mathcal{V}$ as the sole principal unknown, we continue by discussing the corresponding discrete schemes so that the nature of the ensuing discretization errors can be understood.

With this regard and the aim to set a general framework for handling evolving finite element meshes, we need to invert the usual sequence of first spatial and then temporal discretization by considering first the semidiscrete scheme in time, which is here obtained by a backward Euler integration in time. The initial boundary value problem, continuous in time and space, is transformed into the recursive solution of nonlinear problems, continuous with respect to the space, which are referred to as incremental boundary value problems, henceforth abbreviated as InBVP. In these problems, the state of the system at the time instant t_n is a data of the problem, whereas the principal unknown is the displacement field at the time instant t_{n+1} .

The fully discrete scheme of the initial boundary value problem is therefore obtained by a finite element discretization of these incremental boundary value problems. This is obtained by replacing the general infinite dimensional affine spaces where the principal unknown and test functions belong to, with finite dimensional affine spaces. Here, we observe that change of data and/or of finite element mesh from one time interval to the other can be both related to a discontinuity jump of the approximate solution across the time instant t_n . As a result of the observations expressed in Section 3.4, in the developments of reliable *a posteriori* error estimators, one needs, therefore, to account also for the jump. With such *a posteriori* error estimator at hand, indication on how to change the finite element space and define the corresponding data can be given and the assessment of the several transfer operations proposed in literature can be framed in the context of the ensuing error. We, therefore, will give a brief review of some of the current techniques to transfer data from one mesh to the other, and recall the numerical techniques adopted for the solution of the nonlinear algebraic system of equations before concluding the chapter.

4.2 The displacement formulation of the IBVP

In section 3.2 the properties and equations that define the behaviour of a standard generalised material model have been given, which for the reader's convenience have been summarized in Box 3.1. There, the constitutive equations were presented in the alternative scalar equivalent formulations for the purpose of the theory of the error in the constitutive equations. The latter are now given in a more general format, not necessarily restricted to the class of standard models, which is used as basis of the numerical discretization. The general formulation of the problem, therefore, reads as in Box 4.1 where \mathfrak{F} and \mathfrak{G} denote two multivalued tensorial functions which assure the thermodynamic admissibility of the model and enjoy the necessary regularity properties to guarantee a solution of the constitutive initial value problem for given total strain. Furthermore, as for the specific choice of the functional spaces, that is, of the functional setting in which the initial boundary value problem is posed, in the following we assume, if not stated otherwise, that the spaces are endowed with those minimum regularity properties that make the operations involved meaningful and guarantee at least the existence of a solution, (Brezis, 1986; Fiorenza, 1988). However, given the generality of the formulation in Box 4.1, in the current literature, there are no results on its well-posedness. These are available only for some special classes of material models and for some formulations of the initial boundary value problem, as it can be found, among others, in the works of Moreau (1974); Duvaut & Lions (1976); Johnson (1976a, 1978); Suquet (1981); Temam (1985, 1986); Han & Reddy (1999); Alberty & Carstensen (2000) and Fuchs & Seregin (2000).

Remark 4.1. We would like to point out the two main techniques which have been adopted for the proof of existence of solutions and which differentiate mainly in the first part of the proof. One makes a systematic use of Rothe's method (Kacur, 1985) in which first approximate solutions are constructed by semidiscretization in time, and then one passes to the limit using compactness arguments. The other technique applies the methods of the constructive theory of partial differential equations in which a family of regularized problems is examined before passing to the limit using likewise compactness arguments (Evans, 1999). For both the techniques it is fundamental that uniform *a priori* estimates of the approximate solutions are available.

Box 4.1. General formulation of the Initial Boundary Value Problem for a model with internal variables

Given $\boldsymbol{b}(\boldsymbol{x},t)$ on Ω , $\boldsymbol{t}(\boldsymbol{x},t)$ on $\partial\Omega_t$ and the initial state (here assumed as free) Find $\forall t \in [0, T]$ $\begin{vmatrix} u = u(\bullet, t) \in \mathcal{V} \\ \epsilon^p = \epsilon^p(\bullet, t) \in \mathcal{E} \\ \alpha = \alpha(\bullet, t) \in \Lambda \\ \sigma = \sigma(\bullet, t) \in \mathcal{S} \\ A = A(\bullet, t) \in \mathcal{A} \end{vmatrix}$ such that the following equations are satisfied: $\langle \boldsymbol{\sigma}(\boldsymbol{x},t), \nabla \boldsymbol{\eta}(\boldsymbol{x}) \rangle = \langle \boldsymbol{b}(\boldsymbol{x},t), \boldsymbol{\eta}(\boldsymbol{x}) \rangle + \langle \boldsymbol{t}(\boldsymbol{x},t), \boldsymbol{\eta}(\boldsymbol{x}) \rangle_{\partial \Omega_t}$ $\forall \boldsymbol{\eta} \in \mathcal{V}_0, \quad \forall t \in [0, T],$ and, $\forall \boldsymbol{x} \in \Omega, \quad \forall t \in [0, T],$ $\boldsymbol{\epsilon}(\boldsymbol{x},t) = \nabla_{\boldsymbol{s}} \boldsymbol{u}(\boldsymbol{x},t)$ Constitutive Initial Value Problem (CIVP) $\boldsymbol{\epsilon}(\boldsymbol{x},t) = \boldsymbol{\epsilon}^{\boldsymbol{e}}(\boldsymbol{x},t) + \boldsymbol{\epsilon}^{\boldsymbol{p}}(\boldsymbol{x},t)$ $\boldsymbol{\sigma}(\boldsymbol{x},t) = \frac{\partial \psi_{\boldsymbol{e}}}{\partial \boldsymbol{\epsilon}^{\boldsymbol{e}}}(\boldsymbol{\epsilon}^{\boldsymbol{e}}(\boldsymbol{x},t))$ $\boldsymbol{A}(\boldsymbol{x},t) = rac{\partial \psi_p}{\partial \boldsymbol{lpha}}(\boldsymbol{lpha}(\boldsymbol{x},t))$ $\partial_t \epsilon^p(oldsymbol{x},t) \in \mathfrak{F}(oldsymbol{\sigma}(oldsymbol{x},t),oldsymbol{A}(oldsymbol{x},t))$ $\partial_t \boldsymbol{\alpha}(\boldsymbol{x},t) \in \mathfrak{G}(\boldsymbol{\sigma}(\boldsymbol{x},t),\boldsymbol{A}(\boldsymbol{x},t))$ $\boldsymbol{\epsilon}^p(\boldsymbol{x},t=0) = \boldsymbol{0}$ $\boldsymbol{\alpha}(\boldsymbol{x},t=0) = \boldsymbol{0}$



4.2.1 Statement of the problem

The displacement formulation of the initial boundary value problem defined in Box 4.1 is obtained further the observation that the constitutive initial value problem can be solved at any point $\boldsymbol{x} \in \Omega$ with respect to the stress tensor once the displacement field is given.

Box 4.2. Displacement formulation of the initial boundary value problem for

a model with internal variables Given $\boldsymbol{b}(\boldsymbol{x},t)$ on Ω , $\boldsymbol{t}(\boldsymbol{x},t)$ on $\partial\Omega_t$ and the initial state (here assumed as free) Find $\forall t \in [0, T]$ $\boldsymbol{u} = \boldsymbol{u}(\bullet, t) \in \mathcal{V}$ such that the following equation is satisfied: $\langle \boldsymbol{\sigma}(\boldsymbol{x},t), \nabla \boldsymbol{\eta}(\boldsymbol{x}) \rangle = \langle \boldsymbol{b}(\boldsymbol{x},t), \boldsymbol{\eta}(\boldsymbol{x}) \rangle + \langle \boldsymbol{t}(\boldsymbol{x},t), \boldsymbol{\eta}(\boldsymbol{x}) \rangle_{\partial \Omega_{t}}$ $\forall \boldsymbol{\eta} \in \mathcal{V}_0, \quad \forall t \in [0, T],$ where $\boldsymbol{\sigma} = \boldsymbol{\sigma}(\boldsymbol{x}, t)$ is the stress tensor field obtained by solving at any point $x \in \Omega$ the following constitutive initial value problem with prescribed strain $\epsilon(x, t) =$ $\nabla_{\boldsymbol{s}} \boldsymbol{u}(\boldsymbol{x},t), \, \forall \, t \in [0,\,T]$ Constitutive Initial Value Problem (CIVP) $\boldsymbol{\epsilon}(\boldsymbol{x},t) = \boldsymbol{\epsilon}^{\boldsymbol{e}}(\boldsymbol{x},t) + \boldsymbol{\epsilon}^{\boldsymbol{p}}(\boldsymbol{x},t)$ $\boldsymbol{\sigma}(\boldsymbol{x},t) = rac{\partial \psi_{e}}{\partial \boldsymbol{\epsilon}^{e}}(\boldsymbol{\epsilon}^{e}(\boldsymbol{x},t))$ $\boldsymbol{A}(\boldsymbol{x},t) = rac{\partial \psi_p}{\partial \boldsymbol{lpha}}(\boldsymbol{lpha}(\boldsymbol{x},t))$ $\partial_t \epsilon^p(\boldsymbol{x},t) \in \widetilde{\mathfrak{F}}(\boldsymbol{\sigma}(\boldsymbol{x},t),\boldsymbol{A}(\boldsymbol{x},t))$ $\partial_t \boldsymbol{\alpha}(\boldsymbol{x},t) \in \mathfrak{G}(\boldsymbol{\sigma}(\boldsymbol{x},t),\boldsymbol{A}(\boldsymbol{x},t))$ $\boldsymbol{\epsilon}^p(\boldsymbol{x},t=0) = \boldsymbol{0}$ $\boldsymbol{\alpha}(\boldsymbol{x},t=0)=\boldsymbol{0}$

We assume that given the displacement field, $\boldsymbol{u} = \boldsymbol{u}(\boldsymbol{x}, t)$, the constitutive initial value problem given in Box 4.1 has solution.

By replacing then the stress tensor into the equilibrium equation, we obtain an equation with respect to only the displacement field. The displacement formulation of the initial boundary value problem reads, therefore, as in Box 4.2 (de Souza Neto *et al.*, 2002).

Remark 4.2. The solution of the constitutive model with respect to the stress tensor in terms of the displacement field is implied, on a conceptual level and under general smoothness assumptions, by the equivalence of the two type of formulations of a constitutive model, the one in terms of the history of the mechanical variables and the other in terms of the internal variables (Bataille & Kestin, 1979; Ladevèze, 1999). Also, it is worth mentioning that the solution of the constitutive initial value problem (henceforth abbreviated as CIVP) with prescribed strain field $\epsilon(x,t) = \nabla_s u(x,t)$ delivers the field of all the variables which describe the state of the system (Laborde & Nguyen, 1990).

Remark 4.3. A formal expression of the problem given in Box 4.2 in the case of perfect plasticity with the equation expressed only in terms of the displacement field can be found, for instance, in Rannacher & Suttmeier (1998). This formulation is, in fact, not different from the one given in Box 4.2 since the authors let $\boldsymbol{\sigma} = \Pi(\nabla \boldsymbol{u})$ where Π denotes a very general operator obtained by solving the CIVP with respect to \boldsymbol{u} .

4.3 The time discrete problem

The displacement formulation of the initial boundary value problem given in Box 4.2 is discretized in time by the backward Euler method. This is the time discrete scheme which will be the focus of our considerations in the following because of its stability and accuracy properties for finite time step (Ortiz & Popov, 1985).

Let $0 = t_1 < \ldots < t_n < \ldots < t_{N+1} = T$ be a partition of the time interval of interest [0, T] and set $k = \max_{1 \le n \le N} \{k_n = t_{n+1} - t_n\}$. A family of fully implicit approximations of the problem in Box 4.2 is obtained as recursive solution of the non linear spatially continuous variational problems defined in Box 4.4. These are obtained by replacing the rate quantities with backward difference quotients and by sampling all the other functions at t_{n+1} . Thus, the constitutive initial value problem is transformed into the constitutive incremental nonlinear problem (henceforth, abbreviated as CInNP), and the resulting global problem appears, therefore, in the form of a system of variational equations, which expresses the equilibrium, and unilateral constraints in the presence of the inclusions which define the stress tensor field. This format is much clearer if, for instance, we refer to the model of linear elasticity and associative plasticity with linear hardening (Simo & Hughes, 1998; Rannacher & Suttmeier, 1998; Han & Reddy, 1999), which is given in Box 4.3. Here, the CInNP is obtained by backward Euler discretization of the principle of maximum plastic dissipation (3.17).

The effect of replacing the time derivative with backward difference quotients produces an error which is referred to as the time discretization error. The main effect of this error is visualized in Figure 4.1 for a model of perfect plasticity with regard to the error on the direction of the plastic flow under the assumption that the point \boldsymbol{x} experiences plastic loading passing from t_n to t_{n+1} . However, also an error in the intensity of the plastic flow must be noted, in general. The magnitude

Box 4.3. Fully implicit scheme of the IBVP for the model of linear elasticity and associative plasticity with linear hardening

For:	$n = 1, 2, \ldots, N$	
Given:	External Loading	$egin{aligned} & m{b}_{n+1}(m{x}) = m{b}(ullet, t_{n+1}), \ ext{on } \Omega \ & m{t}_{n+1}(m{x}) = m{t}(ullet, t_{n+1}), \ ext{on } \partial \Omega_t \end{aligned}$
	State of the system at t_n	$egin{array}{ccc} oldsymbol{\epsilon}_n^p(oldsymbol{x}) &= oldsymbol{\epsilon}^p(ullsymbol{\bullet},t_n) \in \mathcal{E} \ oldsymbol{lpha}_n(oldsymbol{x}) &= oldsymbol{lpha}(ullsymbol{\bullet},t_n) \in \Lambda \end{array}$
Find:	$\boldsymbol{u}_{n+1}(\boldsymbol{x}) = \boldsymbol{u}(ullet, t_{n+1}) \in \mathcal{V}$	
Such That the following equation is satisfied:		
$egin{aligned} &\langle m{\sigma}_{n+1}(m{x}), abla m{\eta}(m{x}) angle = \langle m{b}_{n+1}(m{x}), m{\eta}(m{x}) angle + \langle m{t}_{n+1}(m{x}), m{\eta}(m{x}) angle_{\partial \Omega_t} \ &orall m{\eta} \in \mathcal{V}_0, \end{aligned}$		
with $\sigma_{n+1}(x) = \sigma(x, t_{n+1}) \in \Sigma_{n+1}(x)$, where $\Sigma_{n+1}(x)$ is obtained by solving at any point $x \in \Omega$ the following variational inequality defined by the prescribed strain $\epsilon(x, t_{n+1}) = \nabla_s u_{n+1}(x)$,		
$ig (\sigma^{ ext{trial}}_{n+1}(oldsymbol{x}) - oldsymbol{\sigma}_{n+1}(oldsymbol{x})) \colon C^{-1}(oldsymbol{ au}(oldsymbol{x}) - oldsymbol{\sigma}_{n+1}(oldsymbol{x})) +$		
$+(Holdsymbollpha_n(oldsymbol x)-oldsymbol A_{n+1}(oldsymbol x))\colonH^{-1}(oldsymbol B(oldsymbol x)-oldsymbol A_{n+1}(oldsymbol x))\leq 0$		
	$orall \left(oldsymbol{ au},oldsymbol{B} ight) \in\mathbb{E}$	
,		

where

$$oldsymbol{\sigma}^{ ext{trial}}_{n+1}(oldsymbol{x}) = {f C}(
abla_soldsymbol{u}_{n+1}(oldsymbol{x}) - oldsymbol{\epsilon}^p_n(oldsymbol{x}))$$

of this error depends on the accuracy for finite time step of the time integration scheme. This accuracy is, usually, assessed numerically with the so-called isoerror maps (Krieg & Krieg, 1977; Schreyer *et al.*, 1979; Ortiz & Popov, 1985).

In proposing the above time discrete scheme, we are faced with two questions: one refers to the existence and uniqueness of the solution for the single nonlinear variational problem while the second regards the convergence of the family of the approximations with their respective rate. Unfortunately, for the formulation given in Box 4.4, likewise the continuous formulation, there are no general results of well posedness and, *a fortiori*, of convergence of the approximation. However, some considerations can be done in merit, though of heuristic character, as extension of those holding for formulations for which a complete analysis has been provided by Han & Reddy (1999).

In stating the single one step variational problem relative to $[t_n, t_{n+1}]$, the state of the system at the time instant t_n defined by $\epsilon_n^p(x)$, $\alpha_n(x)$ is a data of the



Figure 4.1: Effect of the time discretization error in the direction of the plastic flow, $\dot{\epsilon}^p(t) \approx \frac{\epsilon_{n+1}^p - \epsilon_n^p}{k_n}$ with $t \in [t_n, t_{n+1}]$ for a perfect plasticity model with yield surface given by equation $f(\sigma) = 0$.

problem whereas the unknown is given by the function $u_{n+1}(x)$. Therefore, we assume that the single nonlinear variational problem has solution, as long as very general regularity properties are met by the data. This can be easily shown, for example, for the time discrete scheme of the dual variational formulation for the model of linear elasticity and associative plasticity with linear hardening, given in Han & Reddy (1999), where (u, σ, A) are assumed as primary variables and the change of data corresponds to the projection onto a closed convex set at a different point.

Also, we require that $\epsilon_n^p(\boldsymbol{x})$, $\alpha_n(\boldsymbol{x})$ are obtained from the pointwise solution of the CInNP relative to the previous time interval $[t_{n-1}, t_n]$ and to the solution $\boldsymbol{u}_n(\boldsymbol{x})$. This condition, which can be expressed as continuity of the piecewise linear interpolant of the discrete solutions $\{\epsilon_n^p(\boldsymbol{x})\}_{n=1}^{N+1}, \{\alpha_n(\boldsymbol{x})\}_{n=1}^{N+1}$, is invoked in order to obtain an *a priori* estimate for the family of solutions which is independent on k. This is a basic result which is used for a compactness argument to prove finally that as $k \to 0$, the limit of the interpolants is in fact a solution of the continuous problem. Part of the analysis of the semidiscrete scheme would be also an *a priori* error estimate of the approximate solutions which would have in general the following format

$$\max_{1 \le n \le N} \| \boldsymbol{u}_{ex}(\boldsymbol{x}, t_{n+1}) - \boldsymbol{u}_{n+1}(\boldsymbol{x}) \| \le O(k^q), \quad q > 0,$$
(4.1)

which describes the time discretization error with its dependence on the discretization parameter k along with its rate of convergence and $\|\bullet\|$ is an appropriate norm.

Remark 4.4. In relation to the particular constitutive model, the solution of CInNP can result in more or less complex algorithm. The procedures for its solution are generally called *constitutive update algorithms*. In rate independent plasticity models, in particular, they are also referred to as the *stress return algorithms*, because the stresses must be returned to the yield surface (Ortiz & Stainier, 1999). For an overview of these procedures, we refer to Simo (1998); Armero & Pérez-Foguet

Box 4.4. Fully implicit scheme of the problem given in Box 4.2

For: $n = 1, 2, \dots, N$ Given:External Loading $b_{n+1}(x) = b(\bullet, t_{n+1}), \text{ on } \Omega$ $t_{n+1}(x) = t(\bullet, t_{n+1}), \text{ on } \partial \Omega_t$ State of the system at t_n $\epsilon_n^p(x) = \epsilon^p(\bullet, t_n) \in \mathcal{E}$ $\alpha_n(x) = \alpha(\bullet, t_n) \in \Lambda$

Find: $u_{n+1}(x) = u(\bullet, t_{n+1}) \in \mathcal{V}$

Such That the following equation is satisfied:

$$egin{aligned} &\langle \sigma_{n+1}(x),
abla \eta(x)
angle = \langle m{b}_{n+1}(x), m{\eta}(x)
angle + \langle m{t}_{n+1}(x), m{\eta}(x)
angle_{\partial \Omega_t} \ & orall m{\eta} \in \mathcal{V}_0, \end{aligned}$$

where $\sigma_{n+1}(x) = \sigma(x, t_{n+1})$ is the stress tensor field obtained by solving, at any point $x \in \Omega$, the following constitutive nonlinear problem with prescribed strain $\epsilon(x, t_{n+1}) = \nabla_s u_{n+1}(x)$,

Constitutive Incremental Nonlinear Problem (CInNP)

$$\begin{split} \boldsymbol{\epsilon}(\boldsymbol{x},t_{n+1}) &= \boldsymbol{\epsilon}^{\boldsymbol{e}}(\boldsymbol{x},t_{n+1}) + \boldsymbol{\epsilon}^{\boldsymbol{p}}(\boldsymbol{x},t_{n+1}) \\ \boldsymbol{\sigma}(\boldsymbol{x},t_{n+1}) &= \frac{\partial \psi_{\boldsymbol{e}}}{\partial \boldsymbol{\epsilon}^{\boldsymbol{e}}} (\boldsymbol{\epsilon}^{\boldsymbol{e}}(\boldsymbol{x},t_{n+1})) \\ \boldsymbol{A}(\boldsymbol{x},t_{n+1}) &= \frac{\partial \psi_{\boldsymbol{p}}}{\partial \boldsymbol{\alpha}} (\boldsymbol{\alpha}(\boldsymbol{x},t_{n+1})) \\ \frac{\boldsymbol{\epsilon}^{\boldsymbol{p}}(\boldsymbol{x},t_{n+1}) - \boldsymbol{\epsilon}^{\boldsymbol{p}}(\boldsymbol{x},t_{n})}{k_{n}} \in \mathfrak{F}(\boldsymbol{\sigma}(\boldsymbol{x},t_{n+1}),\boldsymbol{A}(\boldsymbol{x},t_{n+1})) \\ \frac{\boldsymbol{\alpha}(\boldsymbol{x},t_{n+1}) - \boldsymbol{\alpha}(\boldsymbol{x},t_{n})}{k_{n}} \in \mathfrak{G}(\boldsymbol{\sigma}(\boldsymbol{x},t_{n+1}),\boldsymbol{A}(\boldsymbol{x},t_{n+1})) \end{split}$$

(2002) and de Souza Neto et al. (2002).

We conclude this section by observing first that the nonlinear variational problem of Box 4.4 can be given the following compact form of a nonlinear variational

equation (Rannacher & Suttmeier, 1998),

Find:
$$u_{n+1}(x) \in \mathcal{V}$$

Such that:
 $\mathcal{L}(\nabla_s u_{n+1}, \eta) \stackrel{\text{def}}{=}$
 $\stackrel{\text{def}}{=} \langle \sigma_{n+1} (\nabla_s u_{n+1}(x)), \nabla \eta(x) \rangle - \langle b_{n+1}(x), \eta(x) \rangle - \langle t_{n+1}(x), \eta(x) \rangle_{\partial \Omega_t} = 0$
 $\forall \eta \in \mathcal{V}_0,$
(4.2)

where the stress tensor

$$\sigma_{n+1}\Big(\epsilon_n^p(x), \, \alpha_n(x); \, \epsilon_{n+1}(x) = \nabla_s u_{n+1}(x)\Big) \tag{4.3}$$

is obtained as part of the solution of the CInNP with data $\epsilon_n^p(x)$ and $\alpha_n(x)$. It is understood that, in (4.3), $\epsilon_{n+1}(x)$ must be considered as variable. Secondly, the function $\mathcal{L} = \mathcal{L}(\nabla_s u_{n+1}, \eta)$ is defined over $\mathcal{V} \times \mathcal{V}_0$, where we recall \mathcal{V} to be the space of the kinematically admissible displacement fields $u_{n+1} = u_{n+1}(x)$, principal unknown of the problem. In the time discrete problem, the displacement field $u_{n+1} = u_{n+1}(x)$ may be any element of \mathcal{V} .

4.4 The fully discrete problem: Constant finite element mesh

The single one step problem (4.2) arising from the time discretization of the initial boundary value problem has the same structure as an elliptic problem apart from substitution into the equilibrium equation of the stress tensor with a nonlinear function of the displacement, $u_{n+1} = u_{n+1}(x)$, primary unknown of the problem. The latter is therefore amenable to the discretization methods for this class of problems (Glowinski *et al.*, 1981) in particular the finite element method. In this way the complete discretization of the initial boundary value problem is achieved, which is referred to as the fully discrete scheme.

More precisely, approximations to the single nonlinear incremental boundary value problem (4.2) with standard displacement finite element procedure are obtained by simply replacing the infinite dimensional affine spaces \mathcal{V} and \mathcal{V}_0 of the trial functions, $\boldsymbol{u}_{n+1} = \boldsymbol{u}_{n+1}(\boldsymbol{x})$, and test functions, $\boldsymbol{\eta} = \boldsymbol{\eta}(\boldsymbol{x})$, with finite dimensional affine subspaces, \mathcal{V}^h and \mathcal{V}_0^h , respectively, which are intended to be finite element spaces. For the construction of such spaces with respective terminology, we refer to standard textbooks on finite element methods (Ciarlet, 1978; Zienkiewicz & Taylor, 2000).

The weak enforcement of the equilibrium with respect to only some test functions produces an error which is referred to as the space discretization error. As it has been already noted for linear problems in Section 2.2, the main effect of this error is the lack of the pointwise equilibrium over the domain Ω , along the boundary traction $\partial \Omega_t$ and the element boundaries. The latter, in turn, means lack of continuity of the stress tensor field therein.

The discrete problem relative to the time interval $[t_n, t_{n+1}]$ reads as follows

Given:External Loading $b_{n+1}(x)$, on Ω
 $t_{n+1}(x)$, on $\partial\Omega_t$ State of the system at t_n $\epsilon_n^p(x) \in \mathcal{E}$
 $\alpha_n(x) \in \Lambda$ Find: $u_{n+1}^h(x) \in \mathcal{V}^h$ Such That $\langle^h \sigma_{n+1}(x), \nabla \eta^h(x) \rangle = \langle b_{n+1}(x), \eta^h(x) \rangle + \langle t_{n+1}(x), \eta^h(x) \rangle$

$$\langle {}^{\prime \sigma} \sigma_{n+1}(\boldsymbol{x}), \vee \eta^{\prime \prime}(\boldsymbol{x}) \rangle = \langle \boldsymbol{b}_{n+1}(\boldsymbol{x}), \eta^{\prime \prime}(\boldsymbol{x}) \rangle + \langle \boldsymbol{t}_{n+1}(\boldsymbol{x}), \eta^{\prime \prime}(\boldsymbol{x}) \rangle_{\partial \Omega_{t}}$$

$$\forall \eta^{h} \in \mathcal{V}_{0}^{h},$$

$$(4.4)$$

where the stress tensor field ${}^{h}\sigma_{n+1}(x)^{\P}$ is the function defined by solving at any point $x \in \Omega$ the CInNP for any given strain $\epsilon_{n+1}^{h}(x) = \nabla_{s} u_{n+1}^{h}(x)$ and fixed data $\epsilon_{n}^{p}(x), \alpha_{n}(x)$.

Since a complete theory for the fully discrete approximations to the displacement formulation of the initial boundary value problem is still to be developed, the considerations that follow will have, therefore, heuristic character and will assume minimum regularity requirements. In rate independent plasticity with positive hardening, this leads, in our case, to the assumption that $\epsilon^{p}(\boldsymbol{x},t), \boldsymbol{\alpha}(\boldsymbol{x},t)$ are continuous in time and in space, whereas $\dot{\epsilon}^{p}(\boldsymbol{x},t), \dot{\boldsymbol{\alpha}}(\boldsymbol{x},t)$ can experience discontinuity in time.

Remark 4.5. The following observations take their motivation primarily from the analysis of fully discrete approximations to other formulations of plasticity, basically the dual variational formulation of the model of linear elasticity and associative plasticity with linear hardening (Johnson, 1976b, 1977; Hlaváček, 1980; Han & Reddy, 1999) and of other evolutive processes, in general, (Kacur, 1985; Evans, 1999) and in particular, parabolic equations (Raviart & Thomas, 1983), and degenerate parabolic equations (Nochetto *et al.*, 1997; Chen *et al.*, 2000a). Instrumental is also the work by Dorfler & Wilderotter (2000), though for elliptic problems, on the development of *a posteriori* error estimates which account for data error.

Likewise the Galerkin finite element approximation of the primal formulation of an elliptic problem, we can, therefore, assume that the results of well posedness

[¶] Hereafter, the symbol ${}^{h}(\bullet)$ will refer to secondary variables which are obtained from the solution of the CInNP for the prescribed strain ϵ_{n+1}^{h} . They do not have to be confused with the respective finite dimensional discretizations, which are here not considered. Only the interpolation of the primary variable, that is, the displacement field, has been assumed which is remarked with the superscript h on the right.

for the continuous problem (4.2) carry over also to its finite element counterparts represented by problem (4.4); in particular, we emphasize the existence of solution as long as the data $\epsilon_n^p(x)$, $\alpha_n(x)$ satisfy general regularity properties.

The result of the approximation which involves only the unknown function $u_{n+1}(x)$ is an error whose value depends on the approximation properties of the finite element subspaces. This is usually mirrored by an *a priori* error estimate of the discretization error which is of the following type

$$\|u_{n+1}(x) - u_{n+1}^h(x)\| \le O(h^p)$$

where h denotes the mesh size of the triangulation \mathcal{T}_h of the domain Ω associated with the finite element space \mathcal{V}^h , and $u_{n+1}(x)$ is the solution of the continuous problem (4.2). If we combine the above result with (4.1) and use the triangular inequality, finally, we can obtain

$$\max_{1 \le n \le N} \| \boldsymbol{u}_{ex}(\boldsymbol{x}, t_{n+1}) - \boldsymbol{u}_{n+1}^{h}(\boldsymbol{x}) \| \le O(k^{q}) + O(h^{p})$$
(4.5)

which shows the effects of time and space discretization on the full discrete approximation along with their respective order of convergence.

In posing problem (4.4), however, the data $\epsilon_n^p(x), \alpha_n(x)$ are unknowns, for being solution at t_n of the time discrete scheme. Therefore, a fully discrete approximation to the problem (4.2), actually, calls also for an approximation to the data of each nonlinear incremental boundary value problem.

If the finite element space \mathcal{V}^h does not change from one time interval to the other, a fully discrete scheme is formulated by assuming the data ${}^h \epsilon_n^p(\boldsymbol{x}), {}^h \alpha_n(\boldsymbol{x}),$ which denote the pointwise solution at t_n of the CInNP with ${}^h \epsilon_{n-1}^p(\boldsymbol{x}), {}^h \alpha_{n-1}(\boldsymbol{x})$ and prescribed strain $\epsilon_n^h(\boldsymbol{x}) = \nabla_s \boldsymbol{u}_n^h(\boldsymbol{x})$ corresponding to the finite element solution at t_n . As a result of this choice, the piecewise linear interpolant of the discrete solutions $\{{}^h \epsilon_n^p(\boldsymbol{x})\}_{n=1}^{N+1}, \{{}^h \alpha_n(\boldsymbol{x})\}_{n=1}^{N+1}$, is continuous. By exploiting then properties of the equations, one can, therefore, envisage to get a uniform a priori estimate for the family of solutions which proves to be crucial for the error analysis of the fully discrete scheme as it results from the convergence studies carried out in Johnson (1976b, 1977); Hlaváček (1980) and Han & Reddy (1999) which finally deliver an error estimate in the form of relation (4.5).

The above considerations suggest that also other approximations to the data of (4.2) can be imagined, provided that then it is possible to prove the convergence of the resulting fully discrete problems. However, the effects of different approximations to the data from the one proposed above and of change of finite element space from one time step to the other, necessary in the adaptive finite element solution of the given initial boundary value problem, are similar and related somehow to each other. Both can be related, in fact, to the introduction of a discontinuity in the solution as it will be shown in the next Section.

4.4.1 Change of finite element mesh

In this Section we consider change of finite element mesh from one time interval to the other and assume that we are able to solve the CInNP at any point $\boldsymbol{x} \in \Omega$. More precisely, let \mathcal{V}^{h_n} and $\mathcal{V}^{h_{n+1}}$ be the finite element spaces adopted for the discretization of the problem (4.2) relative to the time interval $[t_{n-1}, t_n]$ and $[t_n, t_{n+1}]$, respectively. Set $h = \max_{1 \le n \le N} h_n$ where h_n is the meshsize of the triangulation \mathcal{T}_{h_n} of the domain Ω associated with the finite element space \mathcal{V}^{h_n} and denote with ${}^{h_n}\boldsymbol{\epsilon}_n^p(\boldsymbol{x})$, ${}^{h_n}\boldsymbol{\alpha}_n(\boldsymbol{x})$, ${}^{h_n}\boldsymbol{\sigma}_n(\boldsymbol{x})$, the pointwise solution of the CInNP corresponding to the finite element solution $\boldsymbol{u}_n^{h_n}(\boldsymbol{x})$ at t_n .

Assume $h_n^{\mu} \epsilon_n^{\nu}(\mathbf{x})$, $h_n \alpha_n(\mathbf{x})$ as data of the problem (4.2) relative to $[t_n, t_{n+1}]$ and the



Figure 4.2: External loads and Finite element solutions at t_n and t_{n+1} with equilibration of the initial state for change of finite element mesh $\mathcal{T}_{h_n} \to \mathcal{T}_{h_{n+1}}$ at the time instant t_n .

finite element space $\mathcal{V}^{h_{n+1}}$ for its discretization.

If we envision to solve first for $\Delta t \to 0$, that is, at $t_n + \Delta t$ for Δt very small (henceforth, this time instant will be denoted as t_n^+), one expects in general

$${}^{h_n}\epsilon^p_n(x)
eq {}^{h_{n+1}}\epsilon^p_n(x), \quad {}^{h_n}lpha_n(x)
eq {}^{h_{n+1}}lpha_n(x), \quad {}^{h_n}\sigma_n(x)
eq {}^{h_{n+1}}\sigma_n(x),$$

where

$${}^{h_{n+1}}\epsilon_n^p(x),\,{}^{h_{n+1}}lpha_n(x),\,{}^{h_{n+1}}\sigma_n(x)$$

are obtained as usual from the solution of the CINNP with data ${}^{h_n}\epsilon_n^p(\boldsymbol{x})$, ${}^{h_n}\alpha_n(\boldsymbol{x})$, and prescribed strain $\epsilon_n^{h_{n+1}}(\boldsymbol{x}) = \nabla_s \boldsymbol{u}_n^{h_{n+1}}(\boldsymbol{x})$, where $\boldsymbol{u}_n^{h_{n+1}}(\boldsymbol{x}) \in \mathcal{V}^{h_{n+1}}$ is the finite element solution at t_n^+ . Indeed, ${}^{h_{n+1}}\sigma_n(\boldsymbol{x})$ is in equilibrium with respect to $\mathcal{V}^{h_{n+1}}$, whereas ${}^{h_n}\sigma_n(\boldsymbol{x})$ is not if $\mathcal{V}^{h_{n+1}} \neq \mathcal{V}^{h_n}$, even though they correspond to the same external load level at t_n . Figure 4.2 sketches the different quantities in a finite element solution with change of finite element mesh. The solution at t_n^+ , sometimes, is said to be obtained further to the equilibration of the initial state (Cirak & Ramm, 2000). **Remark 4.6.** If the finite element space does not change, i.e. $\mathcal{V}^{h_n} = \mathcal{V}^{h_{n+1}}$, then it is

$${}^{h_n} \epsilon_n^p(x) = {}^{h_{n+1}} \epsilon_n^p(x), \quad {}^{h_n} \alpha_n(x) = {}^{h_{n+1}} \alpha_n(x), \quad {}^{h_n} \sigma_n(x) = {}^{h_{n+1}} \sigma_n(x),$$
, the interpolant will be continuous at t_n .

that is, the interpolant will be continuous at t_n .

Consider now also the solution at t_{n+1} obtained with data ${}^{h_{n+1}}\epsilon_n^p(x)$, ${}^{h_{n+1}}\alpha_n(x)$ and corresponding to the finite element solution $u_{n+1}^{h_{n+1}}(x) \in \mathcal{V}^{h_{n+1}}$ of (4.4). It is easy to realize that the piecewise linear interpolant of the discrete solutions

$$\dots, \underbrace{\overset{h_n}{\underset{\text{sol. at } t_n^-}{}}}_{\text{sol. at } t_n^-} \underbrace{\overset{h_{n+1}}{\underset{\text{sol. at } t_n^+}{}}}_{\text{sol. at } t_n^+} \underbrace{\overset{h_{n+1}}{\underset{\text{sol. at } t_{n+1}^-}{}}}_{\text{sol. at } t_{n+1}^-} \underbrace{\overset{h_{n+1}}{\underset{\text{sol. at } t_{n+1}^-}{}}}_{\text{sol. at } t_{n+1}^-}} \underbrace{\overset{h_{n+1}}{\underset{\text{sol. at } t_{n+1}^-}{}}}_{\text{sol. at } t_{n+1}^-}}$$

will be discontinuous across the time node t_n as is depicted in Figure 4.3 at a generic point $\boldsymbol{x} \in \Omega$. As a result, the considerations of Section 4.4 holding for static finite



Figure 4.3: The time interpolant of the plastic strain at a generic point $x \in \Omega$ is discontinuous at t_n further to the change of finite element mesh $\mathcal{T}_{h_n} \to \mathcal{T}_{h_{n+1}}$.

element meshes can no more be extended and, consequently, nothing can be said, in general, on the convergence of the interpolants as $k, h \rightarrow 0$. In this case, in fact, by accounting for the observations in Section 3.4, the residual error, which is obtained by substituting the approximation into the general equations given in Box 4.2 and representing the forcing in the problem that defines the global error, has two components: One component is regular, which is present also with static finite element mesh and depends essentially on the time step and mesh size. The other component is singular for the presence of the rate quantities $\dot{\epsilon}^p(x,t)$, $\dot{\alpha}(x,t)$ and the discontinuity jump in the time interpolant of ϵ^p and α . The singular components, therefore, depend on the value of the discontinuities which, for the way the fully discrete schemes have been formulated in this section, can be arbitrary. Consequently, in principle, they can have an important influence on the global error. It is, therefore, clear that reliable a *posteriori* estimates of the error of such approximations will have to depend also on these singular components of the residual.

Remark 4.7. Convergence studies of fully discrete approximations in presence of changing mesh are completely missing from the current literature, even for those variational formulations of plasticity models which have been investigated in detail in the case of finite element mesh constant in time. Furthermore, it must be noted that the issue of convergence of fully discrete approximations in presence of evolving finite element meshes is shared also by other evolutive processes, in general, as it has been pointed out since by Dupont (1982). This problem, actually, calls for constraints on the type of change of mesh as shown in Nochetto *et al.* (1997) and Dawson & Kirby (1999) in order to deliver family of discrete schemes that are stable and convergent. Also, it is interesting to note in the *a posteriori* error estimates of these approximations the presence of terms that are due to change of mesh (Eriksson & Johnson, 1991; Estep *et al.*, 2000; Chen *et al.*, 2000a).

In conclusion, if no change of mesh occurs, k and h are the only parameters that control the accuracy of the approximation, thus, both *a priori* and *a posteriori* error estimates will depend somehow on them. On the contrary, if change of mesh occurs, we introduce a discontinuity in the solution which can be arbitrary. The residual will have singular components depending on the jump. Since the error depends on the residual, its estimates, both *a priori* and *a posteriori*, will have to account for the discontinuity.

Also, the considerations expressed in this section point out an important issue in presence of change of mesh. This is the development of *a posteriori* error estimates which by including terms that depend on the jump allow one to assess the effects of change of mesh and data for the solution of (4.4). We will see in Chapter 6 how the error in the constitutive equations will permit to account for the discontinuity in a natural and consistent way.

Remark 4.8. A similar outcome, though starting from a different point of view, can be retrieved from the analysis carried out in Rannacher & Suttmeier (1999), which has been highlighted in section 2.3.2. The authors refer to a dual variational formulation of plasticity and propose a splitting of the error which distinguishes the component due to time discretization, to the space discretization and to the effect, through the stability of the nonlinear incremental boundary value problem, of the error for using different data in posing this problem. \Box

4.5 Overview on the different definitions of the initial state. Transfer procedures

In the previous section, we have assumed the data ${}^{h_n} \epsilon_n^p(\boldsymbol{x})$, ${}^{h_n} \alpha_n(\boldsymbol{x})$ to be known at any point $\boldsymbol{x} \in \Omega$ and equal to the solution at t_n relative to the previous time interval $[t_{n-1}, t_n]$. However, in the actual computation, we do not need in general to know such fields but only their value at discrete points, namely the integration points of each element. In fact, the integrals that appear in the variational formulation (4.4) are seldom computed exactly. Instead, they are approximated through a process of numerical integration, such as, for instance, Gaussian quadrature formula. Let $\Omega_e^{h_{n+1}} \in \mathcal{T}_{h_{n+1}}$ be a generic element of the triangulation $\mathcal{T}_{h_{n+1}}$ and denote with $\boldsymbol{x}_{e,i}^{h_{n+1}} \in \Omega_e^{h_{n+1}}$ the *i*th Gauss point of the element $\Omega_e^{h_{n+1}}$ and ngp their number. Problem (4.4), formulated next with generic data $\tilde{\epsilon}_n^p(\boldsymbol{x}), \tilde{\alpha}_n(\boldsymbol{x})$, reads as

Given: $\forall \Omega_e^{h_{n+1}} \in \mathcal{T}_{h_{n+1}} \text{ and for } i = 1, \dots, ngp,$ External Loading $\boldsymbol{b}_{n+1}(\boldsymbol{x}_{e,i}^{h_{n+1}})$ State of the system at t_n $\tilde{\boldsymbol{\epsilon}}_n^p(\boldsymbol{x}_{e,i}^{h_{n+1}})$ $\tilde{\boldsymbol{\alpha}}_n(\boldsymbol{x}_{e,i}^{h_{n+1}})$

Find:

$$oldsymbol{u}_{n+1}^{h_{n+1}}(oldsymbol{x}) \,\in\, \mathcal{V}^{h_{n+1}}$$

Such that:

$$\sum_{\substack{\Omega_{e}^{h_{n+1}} \in \mathcal{T}_{h_{n+1}}}} \sum_{i=1}^{ngp} j_{e,i} w_{e,i} \Big\{ {}^{h_{n+1}} \sigma_{n+1} \big(\nabla_s \boldsymbol{u}_{n+1}^{h_{n+1}} (\boldsymbol{x}_{e,i}^{h_{n+1}}) \big) \colon \nabla \boldsymbol{\eta}^{h_{n+1}} (\boldsymbol{x}_{e,i}^{h_{n+1}}) + \\ -\boldsymbol{b}_{n+1} (\boldsymbol{x}_{e,i}^{h_{n+1}}) \colon \boldsymbol{\eta}^{h_{n+1}} (\boldsymbol{x}_{e,i}^{h_{n+1}}) \Big\} = 0, \quad \forall \boldsymbol{\eta}^{h_{n+1}} \in \mathcal{V}_{0}^{h_{n+1}},$$

$$(4.6)$$

where the stress tensor ${}^{h_{n+1}}\sigma_{n+1}(\boldsymbol{\epsilon}_{n+1}^{h_{n+1}}(\boldsymbol{x}_{e,i}^{h_{n+1}}) = \nabla_s \boldsymbol{u}_{n+1}^{h_{n+1}}(\boldsymbol{x}_{e,i}^{h_{n+1}}))$ is obtained by solving at $\boldsymbol{x}_{e,i}^{h_{n+1}}$ the CInNP with data $\tilde{\boldsymbol{\epsilon}}_n^p(\boldsymbol{x}_{e,i}^{h_{n+1}})$, $\tilde{\boldsymbol{\alpha}}_n(\boldsymbol{x}_{e,i}^{h_{n+1}})$ and prescribed strain $\boldsymbol{\epsilon}_{n+1}^{h_{n+1}}(\boldsymbol{x}_{e,i}^{h_{n+1}}) = \nabla_s \boldsymbol{u}_{n+1}^{h_{n+1}}(\boldsymbol{x}_{e,i}^{h_{n+1}})$. The symbols $j_{e,i}$, $w_{e,i}$ retain the usual meaning in Gaussian quadrature scheme for which we refer to Ciarlet (1978); Zienkiewicz & Taylor (2000) and de Souza Neto *et al.* (2002). Also, note that for the sake of notation, we have dropped the work of the traction forces in (4.6).

The choice of the quantities in rate form as secondary variables, for which no a priori interpolation assumptions have been made, and the use of backward Euler as time discrete scheme poses, however, the question on how to define the data $\tilde{\epsilon}_n^p(\boldsymbol{x}_{e,i}^{h_{n+1}})$ and $\tilde{\alpha}_n(\boldsymbol{x}_{e,i}^{h_{n+1}})$ for the InBVP in the case the finite element mesh adopted for its discretization is different from the one used in the previous time interval. In this case, inasmuch as the mesh changes across the time node t_n , the Gauss points change as well and it is no more possible to define the history of the secondary variables at these points if they have not been considered from the initial time t = 0.

The procedures currently in use for the definition of these data, in general, try to compute the value of the unknowns fields $\tilde{\epsilon}_n^p(\boldsymbol{x}_{e,i}^{h_{n+1}})$, $\tilde{\alpha}_n(\boldsymbol{x}_{e,i}^{h_{n+1}})$ at the new integration points, $\boldsymbol{x}_{e,i}^{h_{n+1}} \in \Omega_{e,i}^{h_{n+1}}$ with $\Omega_e^{h_{n+1}} \in \mathcal{T}_{h_{n+1}}$, in terms of the values $h_n \epsilon_n^p(\boldsymbol{x}_{e,i}^{h_n})$, $h_n \alpha(\boldsymbol{x}_{e,i}^{h_n})$, solution relative to the previous time interval $[t_{n-1}, t_n]$ at the old integration points, $\boldsymbol{x}_{e,i}^{h_n} \in \Omega_e^{h_n}$ with $\Omega_e^{h_n} \in \mathcal{T}_{h_n}$.

These procedures are usually known by the name of Transfer of Data and represent a very delicate issue for the global accuracy of a finite element adaptive solution. With this regard, in fact, several vaguely defined properties are usually invoked for a mapping scheme in order to prevent corrupting the quality of the resulting finite element solution. These are, for instance, listed in Perić *et al.* (1996) and Rashid (2002), among others, and are referred to as self-consistency, locality, consistency with the constitutive equations, equilibrium, compatibility of the state transfer with the displacement field on the new mesh, minimisation of the numerical diffusion of these variables, just to mention a few. The remapping schemes proposed in literature attempt somehow to meet these properties and are different from each other according to the main aspect they address. As a result, it appears difficult trying to draw a classification. However, the fundamental approaches and ideas can be referred by some means to the following procedures:

- Variationally Consistent Transfer
- Weak Enforcement Continuity Transfer
- Smoothing Transfer

These transfer processes are in the following described succinctly with reference to the specific problem at hand. This means that only transfer of ϵ^p and α will be analysed. It is also worth noting that all the following operations share the same underlying idea of defining first a field for the state variables which depends on the old mesh with its relative distribution of the elemental Gauss points. This field is then transformed, according to the specific procedure, into a new field on the new mesh which allows the sampling at the new Gauss points. In Chapter 6, in comparing the effects of the above transfer procedures, we will consider the state variables at the new Gauss points after equilibration of the initial state and along with an assumption of prolongation over each element. The resulting function is denoted by $h_{n+1}(\bullet)_n$. The difference between the two fields, $h_n(\bullet)_n - h_{n+1}(\bullet)_n$, defines the discontinuity. Finally, it must be observed that it does not have to be of concern that the incremental form of the constitutive equations may be satisfied only at the Gauss points whereas it may be violated in other points of the domain Ω because of the prolongation operation. The error in the constitutive equations will quantify this discrepancy and in this sense, it must be considered as the error associated with the given assumption for the variables distribution.

4.5.1 Variationally consistent transfer

We define variationally consistent transfers as those remapping procedures where the initial data is obtained from sampling at new Gauss points the solution of the variational formulation of the incremental boundary value problem. For this to happen, the equations that define the secondary variables, and appearing as data of the problem, must be formulated into a variational form and consequently an interpolation for those variables must be explicated. With the aim of shedding light on this class of transfers, which have been analysed by Ortiz and coworkers in Ortiz & Quigley (1991); Camacho & Ortiz (1997) and Radovitzky & Ortiz (1999), it appears appropriate first to consider for motivation the simpler model problem of heat conduction and discuss the several definitions of data in posing the fully discrete scheme obtained by a backward Euler integration in time and finite element interpolation in space. This analysis will serve as a motivation to introduce the so-called variational consistent transfers for the problem at hand.

4.5.1.1 Heat conduction

We consider the classical heat conduction problem in an isotropic body $\Omega \subset \mathbb{R}^d$ with heat capacity $\lambda = 1$ and conductivity $\mu = 1$ described by the following equations

Find
$$u = u(\boldsymbol{x}, t)$$
 with $\boldsymbol{x} \in \Omega, \ t \in \mathcal{I} = [0, T]$
 $\begin{vmatrix} \frac{\partial u}{\partial t} - \Delta u = f & \text{in } \Omega \times \mathcal{I} \\ u(\boldsymbol{x}, t) = 0 & \text{in } \partial\Omega \times \mathcal{I} \\ u(\boldsymbol{x}, t = 0) = u_0 & \text{in } \Omega \text{ at } t = 0 \end{cases}$ (boundary condition) (4.7)

In (4.7), u(x,t) is the temperature at $x \in \Omega$ at time $t \in \mathcal{I}$, u_0 is a given initial temperature and f is a given heat production. The symbol Δ denotes the Laplacian operator.

In order to define the functional setting in which the weak form of (4.7) is posed, we need first to introduce the following notation. Let $H_0^1(\Omega)$ denote the Sobolev space of functions of $L^2(\Omega)$ with the first derivatives in the sense of distributions belonging to $L^2(\Omega)$ and trace vanishing on $\partial\Omega$ (Raviart & Thomas, 1983) and with (\bullet, \bullet) the inner product in $L^2(\Omega)$. Also, denote with $\mathcal{C}^0(\mathcal{I}; \mathcal{V})$ and $L^2(\mathcal{I}; \mathcal{V})$ the space of the vector valued functions defined over the time interval \mathcal{I} and with values in the space of functions \mathcal{V} which are continuous and square integrable over \mathcal{I} , respectively.

By assuming $u_0 \in H^1_0(\Omega)$ and $f \in L^2(\mathcal{I}; L^2(\Omega))$, the weak form of (4.7) reads as follows (Raviart & Thomas, 1983)

Data
$$u_0 \in H_0^1(\Omega), f \in L^2(\mathcal{I}; L^2(\Omega))$$

Find $u \in L^2(\mathcal{I}; H_0^1(\Omega)) \cap \mathcal{C}^0(\mathcal{I}; H_0^1(\Omega))$
 $\left| \frac{\mathrm{d}}{\mathrm{d}t}(u(t), v) + a(u(t), v) = (f(t), v) \quad \forall v \in H_0^1(\Omega), u(0) = u_0 \right|$

$$(4.8)$$

where the time derivative is meant in the sense of distributions over \mathcal{I} and the bilinear form a(u(t), v) is defined as follows

$$a(u(t), v) = \sum_{i=1}^{d} \left(\frac{\partial u}{\partial x_i}, \frac{\partial v}{\partial x_i}\right).$$

In (4.8) we have identified the real function u(x,t) defined over $\Omega \times \mathcal{I}$ with the vector valued function u(t) defined over \mathcal{I} and values in the space of functions of Ω into \mathbb{R} (Raviart & Thomas, 1983).

We consider fully discrete schemes for the problem (4.8) obtained by a backward Euler discretization in time and a Galerkin approximation in space. We refer, therefore, to the same notation as introduced in Section 4.3 and 4.4.1; that is, $\mathcal{I}_{n+1} = [t_n, t_{n+1}]$ denotes the generic time interval of the partition of the time interval \mathcal{I} of interest, and $\mathcal{V}^{h_{n+1}}$ is the conforming finite element space associated with the mesh $\mathcal{T}_{h_{n+1}}$ for the discretization of the one single step problem relative to \mathcal{I}_{n+1} .

The fully discrete scheme analysed in Dupont (1982) is

For
$$n = 1, ..., N$$

Data: $u_n^{h_n} \in \mathcal{V}^{h_n}$
Find: $u_{n+1}^{h_{n+1}} \in \mathcal{V}^{h_{n+1}}$
 $\left| \frac{1}{k_n} (u_{n+1}^{h_{n+1}}, v^{h_{n+1}}) + a(u_{n+1}^{h_{n+1}}, v^{h_{n+1}}) = (f_{n+1}, v^{h_{n+1}}) + \frac{1}{k_n} (u_n^{h_n}, v^{h_{n+1}}) \right|$

$$(4.9)$$

$$\left| \frac{1}{k_n} (u_{n+1}^{h_{n+1}}, v^{h_{n+1}}) + a(u_{n+1}^{h_{n+1}}, v^{h_{n+1}}) = (f_{n+1}, v^{h_{n+1}}) + \frac{1}{k_n} (u_n^{h_n}, v^{h_{n+1}}) \right|$$

where for n = 1, $u_1^{h_1}$ denotes any element of the space $H_0^1(\Omega)$, whose choice is relevant for the error on the initial state u_0 .

Remark 4.9. Problem (4.9) is the finite element formulation of an elliptic equation of the form $(I - k_n \Delta)u = k_n f + u_n$ (Thomée, 1997).

In Dupont (1982) an asymptotic *a priori* error estimate is obtained under fairly general assumptions on the change of mesh. These are: Each mesh \mathcal{T}_{n+1} is a refinement of some given coarse partition \mathcal{T} of Ω . Moreover, \mathcal{T}_{n+1} is obtained by at most one level of refinement or coarsening of the mesh \mathcal{T}_n .

An improvement of the rate of convergence, and with similar constraints on the change of mesh, has been obtained by Eriksson & Johnson (1991) who consider, however, the discontinuous Galerkin discretization in time with order zero, which reduces to the backward Euler.

We note that, in (4.9), we must compute $(u_n^{h_n}, v^{h_{n+1}})$. That is, we have to compute the L^2 projection of the solution $u_n^{h_n} \in \mathcal{V}^{h_n}$ relative to the previous time step, onto $\mathcal{V}^{h_{n+1}}$. If numerical quadrature is used, we get

$$(u_n^{h_n}, v^{h_{n+1}}) = \sum_{\Omega_e^{h_{n+1}} \in \mathcal{T}_{h_{n+1}}} \sum_{i=1}^{ngp} j_{e,i} w_{e,i} u_n^{h_n}(x_{e,i}^{h_{n+1}}) v^{h_{n+1}}(x_{e,i}^{h_{n+1}}).$$

Thus, the data of the one step problem relative to the new mesh $\mathcal{V}^{h_{n+1}}$ are obtained by sampling the field $u_n^{h_n}$, solution of the variational problem relative to the previous time step, at the new Gauss points $x_{e,i}^{h_{n+1}}$. For this reason, the term $(u_n^{h_n}, v^{h_{n+1}})$ is referred to as the variationally consistent data. **Remark 4.10.** The previous sampling does not involve any complication, for the variable u is a primary variable of the formulation, therefore, it is defined over all the domain Ω as finite element interpolation.

Nevertheless, also other fully discrete schemes have been proposed in relation to the way that the term $(u_n^{h_n}, v^{h_{n+1}})$ is treated. For example, in the analysis of the twophase Stefan problem carried out by Nochetto *et al.* (1997) the aforementioned term is replaced in (4.9) with $(\mathcal{I}^{V_{h_{n+1}}}u_n^{h_n}, v^{h_{n+1}})$, where $\mathcal{I}^{V_{h_{n+1}}}$ is the nodal Lagrangian operator with respect to $\mathcal{V}_{h_{n+1}}$ (Ciarlet, 1978). The resulting fully discrete scheme is stable and convergent under similar constraints on the change of the finite element mesh.

Another interesting fully discrete scheme is proposed by Dawson & Kirby (1999) for the analysis of 1D linear parabolic problems discretized in space with mixed finite element method. A piecewise constant approximation is assumed for the variable u appearing in rate form. The term $(u_n^{h_n}, v^{h_{n+1}})$ is replaced with

$$(\bar{u_n}, v^{h_{n+1}})$$

where now $\bar{u_n}$ is a piecewise linear function obtained by *ad hoc* postprocessing of the piecewise constant function $u_n^{h_n}$. In this case, the authors prove convergence of the resulting discrete scheme which preserves the optimal convergence rate under very general changes in the mesh.

In conclusion, the definition of data as given in Nochetto *et al.* (1997) and in Dawson & Kirby (1999) are not variationally consistent, for the data of the one single fully discrete problem is not obtained from sampling at the new Gauss points the solution of the problem relative to the previous time step. Nevertheless, the resulting fully discrete schemes are convergent and stable.

Remark 4.11. For the definition of an *a posteriori* error estimate, it is not important to have a convergence result of the fully discrete scheme. This result is important, however, in the context of effectivity of the estimate and convergence of the adaptive process. This information is, in turn, implied by bounding above the *a posteriori* estimate with an *a priori* error estimate (Eriksson & Johnson, 1991). \Box

4.5.1.2 Weak enforcement of the constitutive equations

The unambiguous definition of the initial state for the fully discrete scheme has been made possible by the assumption of the state variables appearing as data of the problem as primary variables of the formulation. As a result, their values could then be computed at any point of the domain by means of the respective interpolation functions. In a displacement formulation, on contrary, the primary unknown is only the displacement field whereas the secondary variables are obtained from the pointwise solution of the equations used for the reduction of the global problem with respect to only the displacement, that is, the CInNP. If these equations are also imposed in weak form along with an interpolation assumption for the secondary variables involved, it will be the same variational formulation to indicate how to state the data for the fully discrete problem in case of change of mesh. This observation, therefore, hints to enforce the CInNP in a weak form and not in a pointwise manner, as it is implied by a pure use of the displacement formulation. However, inasmuch as we are interested to make transfer of the variables obtained from the displacement formulation, we must make sure that the solution obtained from this more general variational formulation is the same as the one obtained from the displacement formulation.

This can be easily achieved, as it is asserted in Ortiz & Quigley (1991), by an appropriate choice of the interpolation functions for the secondary variables and assuming the same element-base quadrature scheme. Since the fields $\sigma(x)$, $\epsilon^p(x)$, $\alpha(x)$ are not involved in spatial derivative in the general variational formulation, unlike the displacement field u(x), the respective finite element interpolation functions are not required to be continuous over the element and across the element boundaries (Ortiz & Quigley, 1991) but only to meet general regularity properties. For example, the interpolated fields $\sigma(x)$, $\epsilon^p(x)$, $\alpha(x)$ must be bounded (Radovitzky & Ortiz, 1999) or at most square integrable over Ω . This allows the variational equations of the constitutive equations to be imposed element by element.

Furthermore, if we assume the values of the field at the quadrature points as degrees of freedom for the element interpolant of the state variable, then the Galerkin finite element approximation becomes equivalent to the set of equations that enforce the constitutive equations at each Gauss point of the element. This equivalence of the displacement formulation based on element quadrature with underlying more general variational formulations will be shown next for the u - p formulation of linear elasticity and for a mixed formulation of plasticity which enforces in weak form some equations of the incremental form of the evolution laws.

In the u-p formulation of linear elasticity, the displacement u and the pressure p are assumed as independent variables. The additional variational equation other than the weak form of the equilibrium equation, is, therefore, given by (Bathe, 1996)

$$\int_{\Omega} (\frac{p}{\kappa} + \epsilon_v) \delta p \, \mathrm{d}\Omega = 0, \quad \forall \, \delta p \in L^2(\Omega)$$
(4.10)

where δp denotes the weighting function, κ is the bulk modulus,

$$\kappa = \frac{E}{3(1-2\nu)},$$

and ϵ_v is the volumetric strain,

$$\epsilon_v = \epsilon_{xx} + \epsilon_{yy} + \epsilon_{zz}$$

If we replace the space of the test functions with a finite dimensional space $\mathcal{P}^{h_{n+1}}$ built from discontinuous functions, and the degree of freedom of the element interpolant are the values of the function at the Gauss points of the element, equation (4.10) can be read element by element

$$\int_{\Omega_{\epsilon}^{h_{n+1}}} (\frac{p}{\kappa} + \epsilon_v) \delta p^{h_{n+1}} \, \mathrm{d}\Omega = 0, \quad \forall \, \delta p^{h_{n+1}} \in \mathcal{P}^{h_{n+1}}.$$
(4.11)

If element basis quadrature is adopted, one gets

$$\frac{p(\boldsymbol{x}_{e,i}^{h_{n+1}})}{\kappa} + \epsilon_v = 0$$

which is the constitutive equation for the pressure enforced at the Gauss point $x_{e,i}^{h_{n+1}}$. This is an example of limitation principle for mixed formulations introduced by Fraeijs de Veubeke (1965). This principle shows that no particular advantage is gained by the use of the mixed formulation against the displacement one with the above choice of the interpolation functions (Malkus & Hughes, 1978). However, this is of no concern, for here we are interested that such equivalence with an underlying mixed formulation does exist.

In the mixed formulation of plasticity given in Simo *et al.* (1989) $\boldsymbol{u}, \boldsymbol{\sigma}, \boldsymbol{\alpha}$ and the plastic multiplier λ are the independent variables. If we introduce the following set

$$K^{p} = \left\{ \delta \lambda \in L^{2}(\Omega) | \delta \lambda \ge 0 \right\}$$

the additional variational equations, other than the weak form of the equilibrium equation, are obtained by enforcing in a weak sense the following equations of the incremental form of the evolution laws

$$\begin{split} &\int_{\Omega} \left[\nabla_{s} \boldsymbol{u}_{n+1} - \boldsymbol{\epsilon}_{n}^{p} - \frac{\partial \psi_{e}^{*}}{\partial \sigma} (\boldsymbol{\sigma}_{n+1}) - \lambda_{n+1} \frac{\partial f}{\partial \sigma} (\boldsymbol{\sigma}_{n+1} \boldsymbol{A}_{n+1}) \right] : \, \delta \boldsymbol{\sigma} \, \mathrm{d}\Omega = 0, \, \forall \delta \boldsymbol{\sigma} \in (L^{2}(\Omega))^{6} \\ &\int_{\Omega} \left[- \frac{\partial \psi_{p}^{*}}{\partial \boldsymbol{A}} (\boldsymbol{A}_{n+1}) + \frac{\partial \psi_{p}^{*}}{\partial \boldsymbol{A}} (\boldsymbol{A}_{n}) - \lambda_{n+1} \frac{\partial f}{\partial \boldsymbol{A}} (\boldsymbol{\sigma}_{n+1} \boldsymbol{A}_{n+1}) \right] : \, \delta \boldsymbol{\alpha} \, \mathrm{d}\Omega = 0, \, \forall \delta \boldsymbol{\alpha} \in (L^{2}(\Omega))^{n_{dim}} \\ &\int_{\Omega} f(\boldsymbol{\sigma}_{n+1}, \boldsymbol{A}_{n+1}) \delta \lambda \, \mathrm{d}\Omega = 0, \, \forall \delta \lambda \in K^{p} \end{split}$$

where $\delta \sigma$, $\delta \alpha$, $\delta \lambda$ denote the weighting functions and n_{dim} equals the number of components of the tensor α with respect to a given basis. All the other equations of the constitutive model, on the other hand, are enforced pointwise.

For illustrative purposes, likewise the previous example, only the affine space of the test functions is replaced with conforming finite element spaces. If these spaces are built from discontinuous functions and the element shape interpolation functions of $\delta \sigma$, $\delta \alpha$, $\delta \lambda$ have the value at the Gauss points as degrees of freedom, the discrete equations can be posed element by element

$$\forall \Omega_{e}^{h_{n+1}} \in \mathcal{T}_{h_{n+1}}$$

$$\int_{\Omega_{e}^{h_{n+1}}} \left[\nabla_{s} \boldsymbol{u}_{n+1} - \boldsymbol{\epsilon}_{n}^{p} - \frac{\partial \psi_{e}^{*}}{\partial \boldsymbol{\sigma}}(\boldsymbol{\sigma}_{n+1}) - \lambda_{n+1} \frac{\partial f}{\partial \boldsymbol{\sigma}}(\boldsymbol{\sigma}_{n+1}\boldsymbol{A}_{n+1}) \right] : \delta \boldsymbol{\sigma}_{e}^{h_{n+1}} \, \mathrm{d}\Omega = 0$$

$$\int_{\Omega_{e}^{h_{n+1}}} \left[-\frac{\partial \psi_{p}^{*}}{\partial \boldsymbol{A}}(\boldsymbol{A}_{n+1}) + \frac{\partial \psi_{p}^{*}}{\partial \boldsymbol{A}}(\boldsymbol{A}_{n}) - \lambda_{n+1} \frac{\partial f}{\partial \boldsymbol{A}}(\boldsymbol{\sigma}_{n+1}\boldsymbol{A}_{n+1}) \right] : \delta \boldsymbol{\alpha}_{e}^{h_{n+1}} \, \mathrm{d}\Omega = 0$$

$$\int_{\Omega_{e}^{h_{n+1}}} f(\boldsymbol{\sigma}_{n+1}, \boldsymbol{A}_{n+1}) \delta \lambda_{e}^{h_{n+1}} \, \mathrm{d}\Omega = 0$$

and after using element based quadrature, one gets

$$\begin{aligned} \forall \Omega_{e}^{h_{n+1}} \in \mathcal{T}_{h_{n+1}} \\ \text{for } i = 1, \dots, ngp \\ \epsilon_{n+1}^{p}(\boldsymbol{x}_{e,i}^{h_{n+1}}) - \epsilon_{n}^{p}(\boldsymbol{x}_{e,i}^{h_{n+1}}) = \lambda_{n+1}(\boldsymbol{x}_{e,i}^{h_{n+1}}) \frac{\partial f}{\partial \sigma}(\sigma_{n+1}(\boldsymbol{x}_{e,i}^{h_{n+1}})\boldsymbol{A}_{n+1}(\boldsymbol{x}_{e,i}^{h_{n+1}})) \\ - \alpha_{n+1}(\boldsymbol{x}_{e,i}^{h_{n+1}})) + \alpha_{n}(\boldsymbol{x}_{e,i}^{h_{n+1}})) = \lambda_{n+1}(\boldsymbol{x}_{e,i}^{h_{n+1}}) \frac{\partial f}{\partial \boldsymbol{A}}(\sigma_{n+1}(\boldsymbol{x}_{e,i}^{h_{n+1}})\boldsymbol{A}_{n+1})(\boldsymbol{x}_{e,i}^{h_{n+1}}) \\ f(\sigma_{n+1}(\boldsymbol{x}_{e,i}^{h_{n+1}}), \boldsymbol{A}_{n+1}(\boldsymbol{x}_{e,i}^{h_{n+1}})) = 0 \end{aligned}$$

which enforce at the Gauss points $\boldsymbol{x}_{e,i}^{h_{n+1}}$ of each element $\Omega_e^{h_{n+1}}$ the equations of the incremental form of the evolution laws which have been considered in weak form.

The previous examples indicate that, in general, the displacement formulation can be obtained from an underlying more general variational formulation which has also ϵ^p and α as independent variables and the following interpolation assumptions hold over each element

$${}^{e}\epsilon_{n+1}^{p,h_{n+1}} = \sum_{\substack{i=1\\ngp}}^{ngp} {}^{e,i}\mathbb{N}_{\epsilon^{p}}^{h_{n+1}}(x) {}^{e,i}\bar{\epsilon^{p}}_{n+1}^{h_{n+1}}$$

$${}^{e}\alpha_{n+1}^{h_{n+1}} = \sum_{i=1}^{ngp} {}^{e,i}\mathbb{N}_{\alpha}^{h_{n+1}}(x) {}^{e,i}\bar{\alpha}_{n+1}^{h_{n+1}}$$

$$(4.12)$$

with the elemental shape functions piecewise continuous and meeting the following requirements

$$\begin{vmatrix} {}^{e,i} \mathbb{N}_{\epsilon^{p}}^{h_{n+1}}(\boldsymbol{x}_{e,j}^{h_{n+1}}) = \delta_{i,j} \\ {}^{e,i} \mathbb{N}_{\boldsymbol{\alpha}}^{h_{n+1}}(\boldsymbol{x}_{e,j}^{h_{n+1}}) = \delta_{i,j} \end{aligned}$$
(4.13)

where $\delta_{i,j}$ is the Kronecker symbol and the coefficients ${}^{e,i}(\bar{\bullet})_{n+1}^{h_{n+1}}$ identify with the value of the respective field at the Gauss points $x_{e,i}^{h_{n+1}}$. Figure 4.4 depicts some possible choices for ${}^{e,i}\mathbb{N}_{e_{lm}}^{h_{n+1}}$, for example, where ${}^{e}_{lm}$ denotes a component of the second order tensor e .

Remark 4.12. The equivalence of the displacement formulation with mixed variational formulations for models of associative plasticity has been object of analysis also in Comi & Perego (1995) and in Alfano *et al.* (1998). In these studies, the relevant variational equations are obtained as stationary conditions of a suitable functional. Hence, with an appropriate choice of the interpolation functions for the secondary variables, the equations of the displacement formulation using element basis quadrature are retrieved. These interpolations for the state variables may, therefore, be assumed as definition of other variationally consistent transfers in the sense of Ortiz & Quigley (1991).



Figure 4.4: Possible choices for the interpolation functions of the internal variables which comply with the requirements of equation (4.13).

In summary, variationally consistent transfer is obtained by making an interpolation assumption for ϵ^p and α which complies with the requirements of equation (4.13) and by sampling these fields at the Gauss points of the new mesh. Given the generality of the conditions (4.13), the specific choice of the interpolation functions comes from considerations of the accuracy of the solution. For example, Ortiz & Quigley (1991) refer to quadratic triangular elements with three Gauss points, thus assuming a linear interpolation for the secondary state variables. However, it was then noted in Camacho & Ortiz (1997), that the local properties of the solution at the Gauss points, in particular the isochoricity of the plastic strain, was not preserved by the transfer because of the linear interpolation. Thus, a first cure was to apply the transfer procedure to the logarithmic measures of the plastic strain rather than to the plastic strain. However, in areas undergoing rapid transients, the piecewise linear interpolation was observed to contain considerable noise away from the Gauss points. This effect was therefore mitigated by smoothing the state variable fields prior to effecting their transfer. In this sense, however, the technique resembles the smoothing technique which is to be described next. In Radovitzky & Ortiz (1999), on the other hand, the authors consider a piecewise constant distribution of the state variables over the Voronoi cells (Frey & George, 2000), of all the Gauss points with respect to the whole domain. This results in defining the value at the new Gauss point equal to the one relative to the nearest Gauss point of the old mesh. With this choice for the interpolation of the state variables, the local properties of the solution are preserved, as well.

Remark 4.13. The definition of a variationally consistent transfer has the only advantage for providing naturally a variational setting where an error analysis could be carried out more easily. The outcome of this analysis would then be an error estimate which accounts for the weak enforcement, against the pointwise, of the equilibrium and of the incremental constitutive equations. However, this analysis seems to have been invoked but never actually performed by the cited authors. We will discuss in the following Chapters that the error in the constitutive equations

will be able to account for the effects of the above approximations, in addition to the time discretization error in replacing the initial value constitutive problem with the incremental constitutive problem. In conclusion, it is worth noting that the definition of a variationally consistent transfer, by itself, does not guarantee that the most accurate adaptive finite element solution is delivered, as has been pointed out in the previous Section in the analysis of the heat conduction problem. \Box

4.5.2 Weak enforcement of the continuity

This transfer procedure is obtained from an approximation a la Galerkin of the variational equation which imposes in weak form the continuity across the time node t_n of the variables which appear as data of the InBVP.

Let us assume, in the following, as general regularity property that $(\bullet)_n \in (L^2(\Omega))^{n_{dim}}$ and also that $(\bullet)_n \in (L^2(\Omega))^{n_{dim}}$, with n_{dim} equal to the number of components of the respective tensor field with respect to a given basis. The field $(\tilde{\bullet})_n$ is the data for the InBVP whereas $(\bullet)_n$ is relative to the solution at the previous time interval.

The following condition

$$\langle (\tilde{\bullet})_n - (\bullet)_n, \theta \rangle = 0, \quad \forall \theta \in (L^2(\Omega))^{n_{dim}}$$

$$(4.14)$$

enforces, in the weak form, the continuity of the field $(\bullet)_n$ across the time node t_n with $\langle \bullet, \bullet \rangle$ being an inner product for the space $(L^2(\Omega))^{n_{dim}}$.

In Rashid (2002), condition (4.14) is enforced in a Galerkin sense by replacing the infinite dimensional space $(L^2(\Omega))^{n_{dim}}$ with finite dimensional spaces defined by piecewise constant functions representing the distribution assumption for the variables ϵ_n^p and α_n^p . These spaces are defined by constant functions over the Voronoi tessellation of each element with respect to its integration points and are denoted next with \mathcal{C}^{h_n} and $\mathcal{C}^{h_{n+1}}$ relative to \mathcal{T}_{h_n} and $\mathcal{T}_{h_{n+1}}$, respectively. If $\omega_{e,i}^{h_n}$ is the Voronoi cell relative to the point $\boldsymbol{x}_{e,i}^{h_n} \in \Omega_e^{h_n}$, the field ${}^{h_n} \epsilon_n^p(\boldsymbol{x})$, for instance, is defined as follows

$${}^{h_n}\epsilon^p_n(x) = \sum_{\Omega^{h_n}_e \in \mathcal{T}_{h_n}} \sum_{i=1}^{ngp} \mathfrak{I}_{\omega^{h_n}_{e,i}}(x) {}^{h_n}\epsilon^p_n(x^{h_n}_{e,i})$$

where $\mathfrak{I}_{\omega_{a,i}^{h_n}}(\boldsymbol{x})$ is the following function

$$\Im_{\omega_{e,i}^{h_n}}(\boldsymbol{x}) \stackrel{\text{def}}{=} \begin{cases} 1 & \text{if } \boldsymbol{x} \in \omega_{e,i}^{h_n} \\ 0 & \text{if } \boldsymbol{x} \in \Omega - \omega_{e,i}^{h_n} \end{cases}$$

and ${}^{h_n}\epsilon_n^p(\boldsymbol{x}_{e,i}^{h_n})$ are the computed values from the solution at the previous time interval $[t_{n-1}, t_n]$.

Likewise, the data field will be represented by

$$ilde{\epsilon}_n^p(m{x}) = \sum_{\Omega_e^{h_{n+1}}\in\mathcal{T}_{h_{n+1}}}\sum_{i=1}^{ngp}\mathfrak{I}_{\omega_{e,i}^{h_{n+1}}}(m{x})\, ilde{\epsilon}_n^p(m{x}_{e,i}^{h_n})$$

where $\tilde{\epsilon}_n^p(x_{e,i}^{h_n})$ are the unknowns to be determined.

Thus, equation (4.14) can be used to formulate the following problem,

Given:

$$\begin{split} h_n \epsilon_n^p(x) &\in \mathcal{C}^{h_n} \\ \text{Find:} & \tilde{\epsilon}_n^p(x) \in \mathcal{C}^{h_{n+1}} \\ \text{Such that:} & (4.15) \\ & \langle^{h_n} \epsilon_n^p - \tilde{\epsilon}_n^p, \theta \rangle = 0, \quad \forall \, \theta \in \mathcal{C}^{h_{n+1}} \end{split}$$

which finally delivers $\tilde{\epsilon}_n^p(x) \in \mathcal{C}^{h_{n+1}}$ as projection of ${}^{h_n} \epsilon_n^p(x) \in \mathcal{C}^{h_n}$ onto $\mathcal{C}^{h_{n+1}}$ with respect to the inner product $\langle \bullet, \bullet \rangle$ of $(L^2(\Omega))^{n_{dim}}$.

Same procedure can be applied to compute $\tilde{\alpha}_n(x) \in \mathcal{C}^{h_{n+1}}$.

By definition, this transfer is, therefore, self-consistent, inasmuch as it is a projection operator. Furthermore, because of the prolongation assumption into piecewise constant functions, it preserves the local character of the state variables at the relative Gauss point, as well.

4.5.3 Smoothing transfer

This procedure represents perhaps the most widely used remapping algorithm in solid mechanics applications for its relatively simple implementation. Details on the transfer operation can be found, among others, in Lee & Bathe (1994) and Perić *et al.* (1996). Next, we just sketch the main steps which are summarized in Figure 4.5. The values of the state variables (•) at the old Gauss points, (•)($\boldsymbol{x}_{e,i}^{h_n}$), are first

$$\begin{array}{ccc} (\bullet)(\boldsymbol{x}_{e,i}^{h_n}) & \stackrel{(a)}{\longrightarrow} & _{e,i}(\bullet)(\boldsymbol{x}_N^{h_n}) & \stackrel{(b)}{\longrightarrow} & (\bullet)(\boldsymbol{x}_N^{h_n}) & \stackrel{(c)}{\longrightarrow} & (\bullet)(\boldsymbol{x}^{h_n}) & \stackrel{(d)}{\longrightarrow} \\ \\ \xrightarrow{(d)} & (\bullet)(\boldsymbol{x}^{h_{n+1}}) = \mathcal{I}^{\mathcal{V}^{h_{n+1}}} & (\bullet)(\boldsymbol{x}^{h_n}) & \stackrel{(e)}{\longrightarrow} & (\bullet)(\boldsymbol{x}_{l,i}^{h_{n+1}}) \end{array}$$

Figure 4.5: Smoothing transfer. (a) Extrapolation of Gauss points value to nodes of old mesh; (b) Average at nodes of old mesh; (c) Finite element interpolation on old mesh; (d) Nodal interpolation onto new mesh; (e) Sampling at new Gauss points

transferred to the nodes of the old mesh, $_{e,i}(\bullet)(\boldsymbol{x}_N^{h_n})$, possibly also with a weight. A weighted average is then carried out at each node, $(\bullet)(\boldsymbol{x}_N^{h_n})$, and a smooth field, $(\bullet)(\boldsymbol{x}^{h_n})$, is consequently defined by interpolation of the nodal values by means of the basis functions of the finite element space, \mathcal{V}^{h_n} , associated with the old mesh. The nodal interpolant of this field with respect to the new finite element space, $(\bullet)(\boldsymbol{x}^{h_{n+1}}) = \mathcal{I}^{\mathcal{V}^{h_{n+1}}}(\bullet)(\boldsymbol{x}^{h_n})$, is constructed and the resulting field is sampled at the new Gauss points delivering therein the transferred values of the state variables, $(\bullet)(\boldsymbol{x}_{l,j}^{h_{n+1}})$.

Some of the above steps can be by-passed and each of them can be tackled in different ways, delivering a fairly large spectrum of transfer procedures. On one end,

there are those that obtain the values at the new Gauss points directly in terms of the values at the old Gauss points by means of interpolations which can be global or local over predefined neighbourhoods of the new Gauss points, such as in Tabbara et al. (1994); Boroomand & Zienkiewicz (1998) and Villon et al. (2000). On the other end, there are the procedures described in Figure 4.5, such as in Lee & Bathe (1994) and Perić et al. (1996). However, when the extrapolation to the nodes is considered, this represents a delicate point of the whole transfer process. Many studies have been carried out in the field of the so-called recovery procedures and various schemes such as least square fitting (Hinton & Campbell, 1974), superconvergence-patch recovery (SPR) (Zienkiewicz & Zhu, 1992a), moving least square (Tabbara et al., 1994), etc. have been proposed. Despite its simplicity, the procedure however is not self-consistent and if frequent remeshing takes place diffusion of plastic strain can spread all over the domain because of the smoothing operation. Also, special care must be paid in accepting the transferred values. In fact, some properties can not be inherited by the values at the new Gauss points, namely the incompressibility of the plastic strain, for this constraint does not commute in general with the extrapolation operation.

Remark 4.14. The previous classification is by no means exhaustive. A fairly general up-date account of current adaptive strategies in elastoplasticity can be found, however, in Ladevèze & Oden (1998). \Box

4.6 Numerical techniques. Newton-Raphson method

Without loss of generality and only for notational convenience, in the following we refer to the formulations of the discrete problems given in Section 4.4 rather than to problem (4.6), which is the one to be solved in an actual computation. Nevertheless, the considerations are general and can be easily adapted to (4.6).

The discrete problem (4.4), which here we formulate with generic data $\tilde{\epsilon}_n^p(\boldsymbol{x})$, $\tilde{\alpha}_n(\boldsymbol{x})$, represents an algebraic system of nonlinear equations that is convenient to

write in the following form,

where the stress tensor ${}^{h_{n+1}}\sigma_{n+1}(\epsilon_{n+1}^{h_{n+1}}(x) = \nabla_s u_{n+1}^{h_{n+1}}(x))$, we recall, is the function of $\epsilon_{n+1}^{h_{n+1}}$ defined implicitly at any point $x \in \Omega$ by the solution of the following problem with generic data $\tilde{\epsilon}_n^p(x), \tilde{\alpha}_n(x)$,

$$\begin{aligned}
\text{Data:} \quad \tilde{\epsilon}_{n}^{p}, \tilde{\alpha}_{n} \\
\text{Given:} \quad \epsilon_{n+1}^{h_{n+1}} = \nabla_{s} u_{n+1}^{h_{n+1}}(x) \\
\text{Find:} \quad \epsilon_{n+1}^{h_{n+1}} \epsilon_{n+1}^{e}, \\ \frac{h_{n+1} \epsilon_{n+1}^{e}}{n+1}, \\ \frac{h_{n+1} \epsilon_{n+1}^{e}}{n+1} = \frac{h_{n+1} \epsilon_{n+1}^{e}}{n+1} + \\
\frac{h_{n+1} \sigma_{n+1}}{\partial \epsilon^{e}} (\\ \frac{h_{n+1} \epsilon_{n+1}^{e}}{\partial \epsilon^{e}} (\\ \frac{h_{n+1} \epsilon_{n+1}^{e}}{n+1}, \\ \frac{h_{n+1} \epsilon_{n+1}^{p}}{k_{n}} \in \mathfrak{F}(\\ \frac{h_{n+1} \sigma_{n+1}}{h_{n+1}} A_{n+1}) \\
\frac{h_{n+1} \alpha_{n+1} - \tilde{\alpha}_{n}}{k_{n}} \in \mathfrak{G}(\\ \frac{h_{n+1} \sigma_{n+1}, \\ h_{n+1} A_{n+1}) \\
\frac{h_{n+1} \alpha_{n+1} - \tilde{\alpha}_{n}}{k_{n}} \in \mathfrak{G}(\\
\end{aligned}$$
(4.17)

For the considerations of Section 4.4 we assume that problem (4.16) has solution for generic data $\tilde{\epsilon}_n^p(\boldsymbol{x}), \tilde{\alpha}_n(\boldsymbol{x})$ as long as minimum regularity requirements are satisfied. **Remark 4.15.** The functional $\mathcal{L} = \mathcal{L}(\nabla_s \boldsymbol{u}_{n+1}^{h_{n+1}}, \boldsymbol{\eta}^{h_{n+1}})$ is defined over $\mathcal{V}^{h_{n+1}} \times \mathcal{V}_0^{h_{n+1}}$. However, next, whenever necessary, we will refer only to the dependence on $\boldsymbol{u}_{n+1}^{h_{n+1}} \in \mathcal{V}^{h_{n+1}}$.

There exist several techniques to solve the system (4.16) (Crisfield, 1991; Dennis & Schnabel, 1996) which are mostly of iterative type. They, fundamentally, differentiate each other for their rate of convergence and for being locally or globally convergent according to whether or not, they require a starting value which is close to the exact solution.

A technique particularly attractive is the fully consistent Newton's method which replaces iteratively in the equation $\mathcal{L}(\nabla_s \boldsymbol{u}_{n+1}^{h_{n+1}}, \boldsymbol{\eta}^{h_{n+1}}) = 0$ the function $\mathcal{L}(\nabla_s \boldsymbol{u}_{n+1}^{h_{n+1}}, \boldsymbol{\eta}^{h_{n+1}})$ by its first order expansion around the point $\boldsymbol{u}_{n+1}^{h_{n+1},(i)} \in \mathcal{V}^{h_{n+1}}$ (Le Tallec, 1994; Bonet & Wood, 1997). Further to the assumption of small displacements, this is obtained simply by replacing the stress function in the weak form of the equilibrium with its first order Taylor series, consequently the generic iterate $\boldsymbol{u}_{n+1}^{h_{n+1},(i+1)}$ is obtained by solving the following linear system:

Data:
$$u_{n+1}^{h_{n+1},(i)} \in \mathcal{V}^{h_{n+1}}$$

Find: $u_{n+1}^{h_{n+1},(i+1)} \in \mathcal{V}^{h_{n+1}}$
 $\langle D^{h_{n+1}} \sigma_{n+1} \Big(\nabla_s u_{n+1}^{h_{n+1},(i)} \Big) \Big[\nabla_s u_{n+1}^{h_{n+1},(i+1)} - \nabla_s u_{n+1}^{h_{n+1},(i)} \Big], \nabla_s \eta^{h_{n+1}} \rangle =$
 $= \langle b_{n+1}, \eta^{h_{n+1}} \rangle + \langle t_{n+1}, \eta^{h_{n+1}} \rangle_{\partial \Omega_t} - \langle^{h_{n+1}} \sigma_{n+1} \Big(\nabla_s u_{n+1}^{h_{n+1},(i)} \Big), \nabla_s \eta^{h_{n+1}} \rangle$
 $\forall \eta^{h_{n+1}} \in \mathcal{V}^{h_{n+1}}$
(4.18)

where ${}^{h_{n+1}}\sigma_{n+1}\left(\nabla_{s}u_{n+1}^{h_{n+1},(i)}\right)$ is obtained from the solution, at any point $\boldsymbol{x} \in \Omega$, of the problem (4.17) with $\epsilon_{n+1}^{h_{n+1}}(\boldsymbol{x}) = \nabla_{s}u_{n+1}^{h_{n+1},(i)}(\boldsymbol{x})$ and data $\tilde{\epsilon}^{p}(\boldsymbol{x}), \tilde{\alpha}(\boldsymbol{x})$, whereas

$$D^{h_{n+1}} \sigma_{n+1} \Big(\nabla_s u_{n+1}^{h_{n+1},(i)} \Big) \Big[\nabla_s u_{n+1}^{h_{n+1},(i+1)} - \nabla_s u_{n+1}^{h_{n+1},(i)} \Big]$$
(4.19)

is the directional derivative of the function ${}^{h_{n+1}}\sigma_{n+1} = {}^{h_{n+1}}\sigma_{n+1}\left(\epsilon_{n+1}^{h_{n+1}}(x)\right)$ at $\epsilon_{n+1}^{h_{n+1}} = \epsilon_{n+1}^{h_{n+1},(i)}$ along the direction $\Delta \epsilon_{n+1}^{h_{n+1}} = (\epsilon_{n+1}^{h_{n+1},(i+1)} - \epsilon_{n+1}^{h_{n+1},(i)})$, and is referred to as the algorithmic stiffness (Simo & Taylor, 1985).

The iterative process is stopped when a norm of the residual

$$^{h_{n+1}}\mathrm{R}_{n+1}^{(i+1)}(\eta^{h_{n+1}}) = \langle b_{n+1}, \eta^{h_{n+1}} \rangle + \langle t_{n+1}, \eta^{h_{n+1}} \rangle_{\partial \Omega_l} - \langle^{h_{n+1}}\sigma_{n+1} \left(\nabla_s u_{n+1}^{h_{n+1},(i+1)} \right), \nabla_s \eta^{h_{n+1}} \rangle$$

and/or of the displacement increment

$$m{u}_{n+1}^{h_{n+1},(i+1)} - m{u}_{n+1}^{h_{n+1},(i)}$$

is sufficiently small.

The algorithm (4.18) for given starting value $u_{n+1}^{h_{n+1},(0)}$ generates a sequence of iterates which under some conditions is shown to be convergent. More precisely, if $u_{n+1}^{h_{n+1}}$ is the exact solution of (4.16) and $u_{n+1}^{h_{n+1},(0)}$ is sufficiently close to $u_{n+1}^{h_{n+1}}$, the following is valid (Ortega & Rheinboldt, 2000)

$$\|m{u}_{n+1}^{h_{n+1},(i+1)} - m{u}_{n+1}^{h_{n+1}}\| \le eta \|m{u}_{n+1}^{h_{n+1},(i)} - m{u}_{n+1}^{h_{n+1}}\|^2$$

Remark 4.16. In applying Newton's method for the solution (4.16), the starting point $u_{n+1}^{h_{n+1},(0)}$ has no physical meaning. It is only the solution of (4.16) to have physical meaning. The starting point is only required to be close to the exact solution and to belong to the domain of the function $\mathcal{L}(\nabla_s u_{n+1}^{h_{n+1}}, \eta^{h_{n+1}})$.

The initialization strategy represents, therefore, an important issue for the success of the method. If mesh does not change from one time step to the other, the functional $\mathcal{L}(\nabla_s \boldsymbol{u}_{n+1}^h, \boldsymbol{\eta}^h)$ is defined over $\mathcal{V}^h \times \mathcal{V}_0^h$. Therefore, we can assume $\boldsymbol{u}_{n+1}^{h,(0)} = \boldsymbol{u}_n^h$, solution at the previous time step. As a result, the initial residual is given by

$${}^{h}\mathrm{R}_{n+1}^{(0)}(\boldsymbol{\eta}^{h}) = \langle \boldsymbol{b}_{n+1}, \boldsymbol{\eta}^{h} \rangle + \langle \boldsymbol{t}_{n+1}, \boldsymbol{\eta}^{h} \rangle_{\partial \Omega_{t}} - \langle {}^{h}\boldsymbol{\sigma}_{n+1} \Big(\nabla_{s} \boldsymbol{u}_{n+1}^{h,(0)} \Big), \nabla_{s} \boldsymbol{\eta}^{h} \rangle,$$

and for the choice of the starting value, it is also

$$\langle {}^{h} \boldsymbol{\sigma}_{n+1} \Big(
abla_{s} \boldsymbol{u}_{n+1}^{h,(0)} \Big),
abla_{s} \boldsymbol{\eta}^{h}
angle = \langle \boldsymbol{b}_{n}, \boldsymbol{\eta}^{h}
angle + \langle \boldsymbol{t}_{n}, \boldsymbol{\eta}^{h}
angle_{\partial \Omega_{t}}.$$

Thus, in this case, if $u_{n+1}^{h,(0)}$ is not close to u_{n+1}^{h} , that is, the method fails to converge, then by reducing the external load level we reduce the initial residual, which eventually approaches to zero, that is,

$$\lim_{t_{n+1}\to t_n}{}^h \mathbf{R}_{n+1}^{(0)}(\boldsymbol{\eta}^h) = 0.$$

Thus, one may expect that with the above choice of the starting value the algorithm finally will converge.

If the mesh changes from one time step to the other, the functional $\mathcal{L}(\nabla_s u_{n+1}^{h_{n+1}}, \eta^{h_{n+1}})$ is defined over $\mathcal{V}^{h_{n+1}} \times \mathcal{V}_0^{h_{n+1}}$, therefore, $u_n^{h_n} \in \mathcal{V}^{h_n}$ cannot be used as starting value, for $u_{n+1}^{h_{n+1},(0)}$ must belong to $\mathcal{V}^{h_{n+1}}$. In this case, the definition of the starting value becomes an important issue, for the residual does not vanish by reducing the external load level, that is,

$$\lim_{t_{n+1}\to t_n} {}^{h_{n+1}} \mathbf{R}_{n+1}^{(0)}(\boldsymbol{\eta}^{h_{n+1}}) \neq 0.$$

Thus, even though we adopt the incremental load procedure within the step as initialization procedure, the algorithm may still fail to converge. This is to be related, finally, to the value of the residual

$${}^{h_{n+1}}\mathbf{R}_{n}^{(0)}(\boldsymbol{\eta}^{h_{n+1}}) = \langle \boldsymbol{b}_{n}, \boldsymbol{\eta}^{h_{n+1}} \rangle + \langle \boldsymbol{t}_{n}, \boldsymbol{\eta}^{h_{n+1}} \rangle_{\partial \Omega_{t}} - \langle {}^{h_{n+1}}\boldsymbol{\sigma}_{n+1} \Big(\nabla_{s}\boldsymbol{u}_{n+1}^{h_{n+1},(0)} \Big), \nabla_{s}\boldsymbol{\eta}^{h_{n+1}} \rangle.$$

As a result, other solution techniques must be envisaged to solve the system (4.16) or to build a starting value which is good for the success of Newton's method, such as, for instance, the line search method whose general principles are recalled in the next section.

4.6.1 Line search method

Lack of convergence of the Newton–Raphson's method is usually shown either by an increase or by an oscillation of the norm of the residual of the iterates. The line search method avoids these behaviours by imposing a decrease in the residual at each iteration.

The principle of the method is quite simple (Dennis & Schnabel, 1996): If the residual at the proposed new iterate $u_{n+1}^{h_{n+1},(i+1)} = u_{n+1}^{h_{n+1},(i)} + \Delta u_{n+1}^{h_{n+1},(i)}$ is bigger than the residual at the previous iterate $u_{n+1}^{h_{n+1},(i)}$, one tries to find along a given direction d an acceptable point $u_{n+1}^{h_{n+1},(i+1)} = u_{n+1}^{h_{n+1},(i)} + \eta d$ whose residual is lower.

As direction d, one usually takes the solution $\Delta u^{h_{n+1},(i)}$ obtained from the Newton-Raphson of the current step which has resulted unsatisfactory, whereas for the choice of the scalar η , we refer to the criterion given in Bonet & Wood (1997) which approximates the norm of the residual dependent on η with a quadratic function. The resulting algorithm is described in Box 4.5 where the method,

Box 4.5. Newton-Raphson with line search method

1. Given $\boldsymbol{u}_{n+1}^{h_{n+1},(0)}, \qquad \ ^{h_{n+1}} \mathbf{R}_{n+1}^{(0)}\ $
2. Compute $u_{n+1}^{h_{n+1},(1)}$ with Newton-Raphson
3. Set $\boldsymbol{d} = \boldsymbol{u}_{n+1}^{h_{n+1},(1)} - \boldsymbol{u}_{n+1}^{h_{n+1},(0)}$ and compute $\ ^{h_{n+1}} \mathbf{R}_{n+1}^{(1)}\ $
4. IF $\ _{n+1}^{h_{n+1}} \mathbb{R}_{n+1}^{(1)}\ \leq \rho \ _{n+1}^{h_{n+1}} \mathbb{R}_{n+1}^{(0)}\ $ use Newton-Raphson
ELSE $\alpha = \frac{\ h_{n+1} \mathbf{R}_{n+1}^{(0)}\ }{\ h_{n+1} \mathbf{R}_{n+1}^{(1)}\ }$
IF $\alpha > 0$, $\eta = \frac{\alpha}{2}$
ELSE $\eta = \frac{\alpha}{2} + \sqrt{\left(\frac{\alpha}{2}\right)^2 - \alpha}$
END IF
END IF
5. Compute $u_{n+1}^{h_{n+1},(1)} = u_{n+1}^{h_{n+1},(0)} + \eta d$ and GOTO 3
where, typically, a value of $\rho = 0.5$ is used (Bonet & Wood, 1997).

which is a globally convergent strategy, has been combined with a Newton–Raphson method, which is a fast local convergent strategy, obtaining a strategy which inherit the benefits of both.

4.7 Concluding Remarks

In this Chapter we have described the displacement finite element solution of an elastoplastic model with internal variables discretized in time with the backward Euler method. A critical analysis of the nature of the discretization errors introduced in the formulation of the fully discrete scheme has been provided. In particular, the fundamental observation has been made that change of data and/or finite element mesh from one time interval to the other can be both related to a discontinuity jump of the approximate solution across the time instant t_n . Also, a critical review of the current techniques to transfer of the data from one mesh to the other has been given.

However, before proceeding to the study of the transfer procedures in terms of the error produced, in the next Chapter we will first consider how to use the extended dissipation error to assess the quality of finite element solutions with meshes constant throughout the loading process. The objective is to show that this error measure is able to account for effects of time and space discretization error, and also to provide the relative importance of the error components associated with the residual in the state law and in the evolution law.

Chapter 5

The error in the constitutive equations to assess the quality of the finite element solution with mesh *constant* in time

5.1 Introduction

Objective of this chapter is the application of the general theory of the error in the constitutive equations developed in Chapter 3 to the assessment of the quality of finite element solutions of elastoplasticity problems.

The general properties of an admissible solution of the problem under consideration have been given and discussed in Section 3.5. For instance, with respect to the computation of the extended dissipation error, an admissible solution is the set of the time dependent fields of the state variables which satisfy the compatibility and equilibrium equations along with the initial conditions. As a result, a finite element solution obtained for instance by a displacement formulation, is not in general an admissible solution. The values are discrete in time and the finite element stresses do not satisfy the equilibrium equations in a pointwise manner.

In order to apply the theory of the error in the constitutive equations, given the finite element solution, a corresponding admissible solution must be therefore defined. However, if the admissible solution must reflect the approximations associated with the finite element solution, it is understood that some conditions are necessary for its definition (Ladevèze & Pelle, 2001). A rather general one can be expressed in the following terms

> Given a family of finite element solutions which converges to the exact solution as the discretization parameters approach their corresponding limit values, then also the family of corresponding admissible solutions converges to the exact solution and possibly with the same rate of convergence.

In linear elasticity, this condition refers to the definition of the so called prolongation conditions, that is, to the conditions posed on the definition of the statically admissible stress fields in terms of the computed finite element stresses, so that the error in the constitutive equation can be bounded above by the exact error in solution. A thorough analysis of the meaning of this condition can be found in Ladevèze & Leguillon (1983) and Ladevèze & Pelle (2001) for the constant triangular element and is referred to as strong prolongation condition.

In this chapter, the finite element mesh is assumed to be constant throughout the time evolution of the continuum. As a result, the admissible solution corresponding to the finite element solution will be time continuous. Criteria to build a such admissible solution for the computation of the extended dissipation error are first given in general and then detailed for the Prandtl-Reuss model. Numerical analysis of its performance compared to classical measures of the error are illustrated on a 1D model problem. In the second part of the chapter, we recall also the dissipation error and compare its performance to the extended dissipation error.

5.2 Extended Dissipation Error

In the definition of an admissible solution for the computation of the extended dissipation error, the statically admissible variables are not constrained to their conjugate variables by means of the state laws as it happens in the dissipation error. This allows more information from the finite element solution to be included in building the corresponding admissible solution and strengthen the link between the two solutions. For example, in the case of a J_2 -plasticity model and of finite element solution which delivers plastic strains meeting the incompressibility condition, the computed plastic strain field can be assumed, in some circumstances, as part of the admissible solution. In particular, in a displacement finite element formulation the plastic strain, known only at the Gauss points $x_{e,i}^h$ of the element used for the numerical integration of the constitutive equations, can be extended over the element in a field which continues to meet the incompressibility condition. This can be realized, for example, by assuming each element partitioned by the Voronoi cells associated with each Gauss point and assuming a constant distribution of the plastic strain over the cell equal to the value of the strain at the respective Gauss point. In this way we define a plastic strain field whose incompressibility is guaranteed at almost every point of the domain. For elements with only one Gauss point, the field will be clearly constant over the element. Figure 5.1 depicts, for instance, the definition of the Voronoi cells for the triangular element when one or three Gauss points are used for the numerical integration of the constitutive equations, respectively, along with the assumed distribution of internal variables. Clearly, one can envisage also other partitions of the element along with relative definition of the state variables distribution. It will follow that the error in the constitutive equations will have to be considered as the error associated with the given assumption for the state variables distribution.



Figure 5.1: Partition of a triangular element with Voronoi cells relative to its Gauss points along with respective assumed distribution for the internal variables

Likewise, conforming finite element displacements can be used as part of the admissible solution and do not need to be modified, unlike for the definition of the admissible solution to compute the dissipation error, as it will be discussed in the following (Moës, 1996).

A key feature of the analysis and implementation of the error in the constitutive equations, however, is in general the definition of an equilibrated stress field $\sigma_{ad}(x, t_{n+1})$ linked to the finite element solution ${}^{h}\sigma_{n+1}(x_{e,i}^{h})$. A substantial body of research has been devoted to recover more or less equilibrated stress field on the basis of the computed finite element stresses such as in Stein & Ahmand (1977), Zienkiewicz & Zhu (1992c), Ainsworth & Oden (1993), Wiberg et al. (1994) and de Miranda & Umbertini (2002), among others. In these works different conditions are given to relate the recovered stress field to the finite element stresses. In the following we refer to the techniques initiated by Ladevèze (1975) where the so called prolongation condition depending in general on the regularity of the mesh establishes the aforementioned link (see, e.g., Ladèveze, 1994; Ladevèze & Pelle, 2001). In particular, hereafter, we apply the strong prolongation condition as introduced and analysed in Ladevèze (1975), Ladevèze & Leguillon (1983) and Ladevèze & Pelle (2001). This condition distinguishes the statically admissible stress fields $\sigma_{ad}(x, t_{n+1})$ which satisfy the following equation for every shape function $\mathbb{N}_i(x)$ and for all the elements Ω_e ,

$$\int_{\Omega_e} \left(\boldsymbol{\sigma}_{ad}(\boldsymbol{x}, t_{n+1}) - {}^{h} \boldsymbol{\sigma}_{n+1}(\boldsymbol{x}) \right) \colon \nabla \mathbb{N}_i(\boldsymbol{x}) \, \mathrm{d}\Omega = 0.$$
 (5.1)

For anisotropic meshes, the condition (5.1) is required to hold for any element Ω_e and for any shape function of higher order which is associated with a non vertex node. This condition has been termed as weak prolongation condition in Ladèveze (1994) and Ladevèze & Rougeot (1997).

The only unknowns left apart and necessary to determine a complete admis-

sible solution are, therefore,

$$p_{ad}(x, t_{n+1}), \, \alpha_{ad}(x, t_{n+1}); \ R_{ad}(x, t_{n+1}), \, X_{ad}(x, t_{n+1})$$

For their computation, the general method of minimization of the error introduced in Ladevèze *et al.* (1999) can be adopted. However, in general, it may be much more convenient to resort to the simpler criterion given in Ladevèze & Moës (1997) which resembles the integration of the evolution law for the constitutive model which is used. Next, we detail this construction for the Prandtl-Reuss model.

5.2.1 Construction of the admissible solution

Plasticity Model P1: Prandtl Reuss Model with Linear Hardening

The essential equations for this model are hereafter recalled, cf. Section 3.2.4.5,

Yield Condition:

$$\begin{aligned} \|\boldsymbol{\sigma}_{ad}^{D}\| - (R_{ad} + R_{0}) &\leq 0, \quad R_{ad} \geq 0 \\ s^{l}\eta_{\boldsymbol{x},t}^{2}(\boldsymbol{\sigma}_{ad}, R_{ad}; \boldsymbol{\epsilon}_{ad}^{e}, p_{ad}) &= \left(\boldsymbol{\sigma}_{ad} - \mathbf{C}\boldsymbol{\epsilon}_{ad}^{e}\right): \ \mathbf{C}^{-1}\left(\boldsymbol{\sigma}_{ad} - \mathbf{C}\boldsymbol{\epsilon}_{ad}^{e}\right) + \\ &+ \left(R_{ad} - \mathbf{H}p_{ad}\right)\mathbf{H}^{-1}\left(R_{ad} - \mathbf{H}p_{ad}\right); \\ {}^{d}\eta_{\boldsymbol{x},t}^{2}(\boldsymbol{\sigma}_{ad}, R_{ad}; \boldsymbol{\epsilon}_{ad}^{p}, \dot{p}_{ad}) &= R_{0}\|\boldsymbol{\epsilon}_{ad}^{p}\| - \boldsymbol{\sigma}_{ad}: \boldsymbol{\epsilon}_{ad}^{p} + R_{ad}\dot{p}_{ad}, \end{aligned}$$

with $\operatorname{Tr}[\dot{\epsilon}_{ad}^p] = 0$ and $\dot{p}_{ad} \geq \|\dot{\epsilon}_{ad}^p\|$, where $\|\boldsymbol{q}\| = \sqrt{\boldsymbol{q}:\boldsymbol{q}}$ is the norm of the second order tensor \boldsymbol{q} .

In the following, we refer to the solution of a fully implicit conforming finite element displacement formulation of the initial boundary value problem of the plasticity model under consideration. The primary variable is given at the discrete time instants t_{n+1} in terms of the finite element displacement field $u_{n+1}^h(x)$, whereas the secondary variables, such as ${}^{h}\sigma_{n+1}$, ${}^{h}\epsilon_{n+1}^{p}$, ${}^{h}p_{n+1}$, are issued only at the Gauss points x_{ei}^{h} used for the numerical quadrature of the internal virtual work. As a result of the considerations of Section 4.2, we recall that on conceptual level, all the secondary variables can be obtained at any point $\boldsymbol{x} \in \Omega$ provided that at this point and for all the previous discrete time instants t_n , the incremental constitutive value problem has been solved. This would require to store the computed finite element displacement field at each discrete time instant t_n and consequently huge memory capacities would be necessary. By contrast, in an incremental solution procedure, only the accepted solution at the previous discrete time instant is stored at most. The stress field which would be so obtained would not satisfy anyway the equilibrium equations in a pointwise manner, thus the finite element solution cannot be used to compute the extended dissipation error.

Objective of this section is to propose a procedure to build an admissible solution corresponding as close as possible to the computed finite element solution and make use of minimum memory requirements. This is feasible because equation (3.48) shows that the extended dissipation error at the time t_{n+1} can be expressed

in terms of only the error at the time t_n and the admissible solution over $[t_n, t_{n+1}]$. As a result, the admissible solution can be defined in an incremental manner, as well. Consistently with an assumed linear variation of the external load over each time interval and the convexity of the equilibrium and compatibility conditions, the admissible solution is taken to vary linearly over $[t_n, t_{n+1}]$. Therefore for its complete definition we need to solve the following problem

> **Given:** the admissible solution at t_n , the finite element solution at t_{n+1} , **Find:** a corresponding admissible solution at t_{n+1} .

As far as the definition of the admissible displacement field and the equilibrated stress field are concerned, the general considerations given in the previous section apply. In particular, the equilibrated stress field is obtained with a two-stage procedure. The first consists, by means of the prolongation condition, in defining an equilibrated traction forces along the boundary of each element and the second in solving the equilibrium equations over each element, usually, with higher order elements. For technical details we refer to Rougeot (1989), Ladevèze *et al.* (1991), Ladevèze & Rougeot (1997) and Ladevèze & Pelle (2001). The outcome is the definition of a statically admissible stress field $\sigma_{ad}(x, t_{n+1})$, which is known at any point $x \in \Omega$ and is continuous over Ω .

Also, we consider hereafter the computed finite element plastic strain to be extended over the domain with the criterion given in the previous section so that it makes sense to write ${}^{h}\epsilon_{n+1}^{p}(\boldsymbol{x})$ for each $\boldsymbol{x} \in \Omega$. For the definition of other admissible state variables $\epsilon_{ad}^{p}(\boldsymbol{x}, t_{n+1})$, $p_{ad}(\boldsymbol{x}, t_{n+1})$, $R_{ad}(\boldsymbol{x}, t_{n+1})$ we start from the general procedure indicated by Ladevèze *et al.* (1999), which considers the minimization of the extended dissipation error over the set of the admissible values for the remaining state variables. The minimization can be carried out at each point of the domain since there are no spatial derivative involved in the constitutive equations, in particular it will be done at the Gauss points used to compute numerically the space integrals that define the error (cf. Ladevèze & Pelle, 2001). These quadrature points do not have to be confused with those where the constitutive equations are integrated numerically. These points in turn are the one used to compute numerically the integral that appear in the internal virtual power.

The minimization problem in its general terms is described in Box 5.1. An approximate solution of this problem can be obtained as follows. We assume the admissible thermodynamic force $R_{ad}(\boldsymbol{x}, t_{n+1})$ given by

$$R_{ad}(\boldsymbol{x}, t_{n+1}) = \operatorname{Max}\left\{R_1, R_2\right\}$$

where

$$R_1 = \|\boldsymbol{\sigma}_{ad}^D(\boldsymbol{x}, t_{n+1})\| - R_0$$
$$R_2 = R_{ad}(\boldsymbol{x}, t_n).$$
Box 5.1. Definition of the state variables as minimization of the extended dissipation error

 $\pmb{x}\in \Omega$

Given:

For each

 $\begin{array}{ll} \textbf{ren:} & \boldsymbol{\sigma}_{ad}(\boldsymbol{x},t_n), R_{ad}(\boldsymbol{x},t_n); \\ & \boldsymbol{\epsilon}_{ad}(\boldsymbol{x},t_n), \boldsymbol{\epsilon}_{ad}^p(\boldsymbol{x},t_n), p_{ad}(\boldsymbol{x},t_n). \\ & \boldsymbol{\sigma}_{ad}(\boldsymbol{x},t_{n+1}), \boldsymbol{\epsilon}_{ad}(\boldsymbol{x},t_{n+1}) \end{array}$

Find:

$$egin{aligned} R_{ad}(oldsymbol{x},t_{n+1}); \ \epsilon^p_{ad}(oldsymbol{x},t_{n+1}), p_{ad}(oldsymbol{x},t_{n+1}) \end{aligned}$$

such that by assuming a time linear variation over $\left[t_n,\,t_{n+1}\right]$ of the variables

$$egin{aligned} m{\sigma}_{ad}(m{x},t), R_{ad}(m{x},t); \ m{\epsilon}_{ad}(m{x},t), m{\epsilon}_{ad}^p(m{x},t), p_{ad}(m{x},t), \end{aligned}$$

we realize the minimum of the following function

$$\begin{split} F(\boldsymbol{\sigma}_{ad}, R_{ad}; \, \boldsymbol{\epsilon}_{ad}, \boldsymbol{\epsilon}_{ad}^{p}, p_{ad}) &= \\ &= \sup_{t_{n} \leq t \leq t_{n+1}} \left\{ \left[\boldsymbol{\sigma}_{ad}(\boldsymbol{x}, t) - \mathbf{C} \boldsymbol{\epsilon}_{ad}^{e}(\boldsymbol{x}, t) \right] \colon \, \mathbf{C}^{-1} \Big[\boldsymbol{\sigma}_{ad}(\boldsymbol{x}, t) - \mathbf{C} \boldsymbol{\epsilon}_{ad}^{e}(\boldsymbol{x}, t) \Big] + \right. \\ &+ \Big[R_{ad}(\boldsymbol{x}, t) - \mathbf{H} p_{ad}(\boldsymbol{x}, t) \Big] \mathbf{H}^{-1} \Big[R_{ad}(\boldsymbol{x}, t) - \mathbf{H} p_{ad}(\boldsymbol{x}, t) \Big] + \\ &+ \int_{t_{n}}^{t} \Big[R_{0} \| \dot{\boldsymbol{\epsilon}}_{ad}^{p}(\boldsymbol{x}, \tau) \| - \boldsymbol{\sigma}_{ad}(\boldsymbol{x}, \tau) \colon \, \dot{\boldsymbol{\epsilon}}_{ad}^{p}(\boldsymbol{x}, \tau) + R_{ad}(\boldsymbol{x}, \tau) \dot{p}_{ad}(\boldsymbol{x}, \tau) \Big] \, \mathrm{d}\tau \Big\} \end{split}$$

under the following constraints

$$\forall t \in [t_n, t_{n+1}] \left| \begin{array}{l} \|\boldsymbol{\sigma}_{ad}^{D}(\boldsymbol{x}, t)\| - (R_{ad}(\boldsymbol{x}, t) + R_0) \leq 0, \\ \dot{p}_{ad}(\boldsymbol{x}, t) \geq \|\dot{\boldsymbol{\epsilon}}_{ad}^{p}(\boldsymbol{x}, t)\|, \\ \mathrm{Tr}[\dot{\boldsymbol{\epsilon}}_{ad}^{p}(\boldsymbol{x}, t)] = 0 \\ \boldsymbol{\epsilon}_{ad}(\boldsymbol{x}, t) = \boldsymbol{\epsilon}_{ad}^{e}(\boldsymbol{x}, t) + \boldsymbol{\epsilon}_{ad}^{p}(\boldsymbol{x}, t) \\ R_{ad}(\boldsymbol{x}, t) \text{ non-negative and non-decreasing} \end{array} \right|$$

The admissible plastic strain, on the other hand, will be given by

$$\epsilon_{ad}^{p}(\boldsymbol{x},t_{n+1}) = {}^{h} \epsilon_{n+1}^{p}(\boldsymbol{x})$$

if the following condition is satisfied,

$$\boldsymbol{\sigma}_{ad}(\boldsymbol{x}, t_{n+1}) \colon \left[{}^{h} \boldsymbol{\epsilon}_{n+1}^{p}(\boldsymbol{x}) - \boldsymbol{\epsilon}_{ad}^{p}(\boldsymbol{x}, t_{n})\right] \ge 0.$$
(5.2)

Otherwise we choose

$$\epsilon_{ad}^p(\boldsymbol{x}, t_{n+1}) = \epsilon_{ad}^p(\boldsymbol{x}, t_n).$$

Finally, with regard to the admissible accumulated plastic strain $p_{ad}(x, t_{n+1})$ it is

$$p_{ad}(\boldsymbol{x}, t_{n+1}) = p_{ad}(\boldsymbol{x}, t_n) + \|\boldsymbol{\epsilon}_{ad}^p(\boldsymbol{x}, t_{n+1}) - \boldsymbol{\epsilon}_{ad}^p(\boldsymbol{x}, t_n)\|$$

which corresponds to the integration of the equation $\dot{p} = \|\dot{\epsilon}_{ad}^p\|$ that occurs for the model under consideration by assuming linear variation of the variables over $[t_n, t_{n+1}]$.

Condition (5.2), which represents the discrete implicit expression of the plastic power, can be interpreted in the light of the minimization of only the pointwise contribution to the error in the evolution law, θ_d^2 , within the time interval $[t_n, t_{n+1}]$, given by

$$_{[t_n,t_{n+1}]}\zeta_d^2(\boldsymbol{x}) \stackrel{\text{def}}{=} \int_{t_n}^{t_{n+1}} \left[R_0 \| \dot{\boldsymbol{\epsilon}}_{ad}^p(\boldsymbol{x},\tau) \| - \boldsymbol{\sigma}_{ad}(\boldsymbol{x},\tau) \colon \dot{\boldsymbol{\epsilon}}_{ad}^p(\boldsymbol{x},\tau) + R_{ad}(\boldsymbol{x},\tau) \dot{p}_{ad}(\boldsymbol{x},\tau) \right] \mathrm{d}\tau$$

$$(5.3)$$

under the constraints

$$\operatorname{Tr}[\dot{\boldsymbol{\epsilon}}_{ad}^{p}(\boldsymbol{x},\tau)] = 0, \quad \dot{p}_{ad}(\boldsymbol{x},\tau) \ge \|\dot{\boldsymbol{\epsilon}}_{ad}^{p}(\boldsymbol{x},\tau)\|.$$
(5.4)

Further to the assumption of linear variation of the variables which enter into (5.3), and by accounting for (5.4), after some rearrangements, it also follows

$$\begin{split} R_{0} \|\Delta \boldsymbol{\epsilon}_{ad}^{p}\| &- \frac{\boldsymbol{\sigma}_{ad}(\boldsymbol{x}, t_{n+1}) + \boldsymbol{\sigma}_{ad}(\boldsymbol{x}, t_{n})}{2} \colon \Delta \boldsymbol{\epsilon}_{ad}^{p} + \frac{R_{ad}(\boldsymbol{x}, t_{n+1}) + R_{ad}(\boldsymbol{x}, t_{n})}{2} \Delta p_{ad} \geq \\ &\geq \frac{\|\Delta \boldsymbol{\epsilon}_{ad}^{p}\|}{2} \Big[R_{0} + R_{ad}(\boldsymbol{x}, t_{n+1}) - \|\boldsymbol{\sigma}_{ad}^{D}(\boldsymbol{x}, t_{n+1})\| \boldsymbol{n}_{\boldsymbol{\sigma}_{ad}^{D}}(\boldsymbol{x}, t_{n+1}) \colon \boldsymbol{n}_{\Delta \boldsymbol{\epsilon}_{ad}^{p}}(\boldsymbol{x}) \Big] + \\ &+ \frac{\|\Delta \boldsymbol{\epsilon}_{ad}^{p}\|}{2} \Big[R_{0} + R_{ad}(\boldsymbol{x}, t_{n}) - \|\boldsymbol{\sigma}_{ad}^{D}(\boldsymbol{x}, t_{n})\| \boldsymbol{n}_{\boldsymbol{\sigma}_{ad}^{D}}(\boldsymbol{x}, t_{n}) \colon \boldsymbol{n}_{\Delta \boldsymbol{\epsilon}_{ad}^{p}}(\boldsymbol{x}) \Big], \end{split}$$

where we have let

$$\Delta \epsilon^p_{ad} = \epsilon^p_{ad}(\boldsymbol{x}, t_{n+1}) - \epsilon^p_{ad}(\boldsymbol{x}, t_n), \quad \Delta p_{ad} = p_{ad}(\boldsymbol{x}, t_{n+1}) - p_{ad}(\boldsymbol{x}, t_n)$$

 and

$$\boldsymbol{\sigma}_{ad}^{D}(\boldsymbol{x},t) = \|\boldsymbol{\sigma}_{ad}^{D}(\boldsymbol{x},t)\|\boldsymbol{n}_{\boldsymbol{\sigma}_{ad}^{D}}(\boldsymbol{x},t), \quad \Delta\boldsymbol{\epsilon}_{ad}^{p}(\boldsymbol{x}) = \|\Delta\boldsymbol{\epsilon}_{ad}^{p}(\boldsymbol{x})\|\boldsymbol{n}_{\Delta\boldsymbol{\epsilon}_{ad}^{p}}(\boldsymbol{x}).$$

Finite lower bounds for the function (5.3) can therefore be obtained by the minimization of the expression

$$\frac{\|\Delta \boldsymbol{\epsilon}_{ad}^{p}\|}{2} \Big[R_{0} + R_{ad}(\boldsymbol{x}, t_{n+1}) - \|\boldsymbol{\sigma}_{ad}^{D}(\boldsymbol{x}, t_{n+1})\| \boldsymbol{n}_{\boldsymbol{\sigma}_{ad}^{D}}(\boldsymbol{x}, t_{n+1}) \colon \boldsymbol{n}_{\Delta \boldsymbol{\epsilon}_{ad}^{p}}(\boldsymbol{x}) \Big].$$
(5.5)

However, by inspection, we note that lower values of this function correspond to $\Delta \epsilon^p_{ad}$ such that

$$\boldsymbol{n}_{\boldsymbol{\sigma}_{ad}^{D}}(\boldsymbol{x}, t_{n+1}) \colon \boldsymbol{n}_{\Delta \boldsymbol{\epsilon}_{ad}^{p}} \ge 0.$$
(5.6)

By choosing the admissible plastic strain $\epsilon_{ad}^p(x, t_{n+1})$ as described above we respect both condition $\text{Tr}[\Delta \epsilon_{ad}^p] = 0$ and (5.6), and also we include information pertinent to the finite element plastic strain.

Once all the admissible state variables have been computed at t_{n+1} , the use of a time linear interpolation over $[t_n, t_{n+1}]$ guarantees the admissibility of the solution for the convexity of the equilibrium and compatibility conditions. Also, the above procedure delivers an admissible solution which produces a finite value of the error for the convexity of the domains \mathbb{E} and \mathbb{C} introduced in Section 3.2.4.5.

Remark 5.1. The use of a time linear interpolation of the computed nodal values has the implied assumption that the accuracy of this time dependent function is the same as the computed nodal values. This is not the case in general but it holds for small values of the time step Δt .

External Loads at t_n	External Loads at t_{n+1}
t _n	t_{n+1}
FE Solution at t _n	FE Solution at <i>t</i> _{n+1}
u ^h	$\mathbf{u}_{n+1}^{\mathbf{h}}$
h _o ,	h _{On+1}
$h \epsilon_n^p$	hep-1
^h p _n	${}^{h}\mathbf{p}_{n+1}$
\mathcal{T}_{h} t_{n} \mathcal{T}_{h}	\tilde{h} t_{n+1} T_{h}
$u_{ad}(t_n)$	$u_{ad}(t_{n+1})$
$\sigma_{ad}(t_n)$	$\sigma_{ad}(t_{n+1})$
$\varepsilon_{\mathrm{ad}}^{\mathrm{p}}(t_{\mathrm{n}})$	$\epsilon_{ad}^{P}(t_{n+i})$
$p_{ad}(t_n)$	$p_{ad}(t_{n+1})$
Adm Solution at t.	Adm Solution at t_{n+1}

Figure 5.2: Finite element solution and admissible solution for finite element mesh \mathcal{T}_h constant in time.

Figure 5.2 reports schematically the notation relative to the finite element solutions and corresponding admissible solutions in the case of finite element mesh constant in time.

5.2.2 Error Expressions

In the section Error Analysis which follows, comparisons of the extended dissipation error with classical measures of the exact error in solution will be reported. The aim is to illustrate that the extended dissipation error reflects quite well the evolution of the admissible solution with respect to the exact one as described by more classical measures of the error. This section therefore presents the expressions of the classical measures of the error used for the subsequent numerical comparison and recalls, for the reader's convenience, the expressions of the extended dissipation error and the error in solution.

Classical measures of the exact error in solution

As shown in the previous section, the extended dissipation error applied to the finite element solution measures actually the accuracy of an admissible solution corresponding to the finite element solution. The admissible solution is a time dependent function which is obtained as time linear interpolation of the discrete values at the time instants used for the time discretization of the initial boundary value problem under consideration. Thus, it is this time dependent solution that will be assumed in the definition of the exact error. Furthermore, as generalization of the error in elasticity, it has been observed in Section 3.5.3 that the extended dissipation error can be interpreted as a global measure of the error of the kinematically admissible solution $s_{ad}^{kin} = (\epsilon_{ad}, \epsilon_{ad}^p, p_{ad})$. Thus, it appears quite natural to consider the $L^{\infty}L^2$ norm of the exact error of the admissible total strain,

$$\begin{split} \|\boldsymbol{e}_{\boldsymbol{\epsilon}_{ad}}\|_{L^{\infty}([0,T];(L^{2}(\Omega))^{d\times d})} &= \underset{t\leq T}{\mathrm{SUP}} \|\boldsymbol{e}_{\boldsymbol{\epsilon}_{ad}}(\boldsymbol{x},t)\|_{(L^{2}(\Omega))^{d\times d}} = \\ &= \underset{1\leq n\leq N}{\mathrm{MAX}} \underset{t\in[t_{n},t_{n+1}]}{\mathrm{SUP}} \left\{ \int_{\Omega} \left(\boldsymbol{\epsilon}_{ex}(\boldsymbol{x},t) - \boldsymbol{\epsilon}_{ad}(\boldsymbol{x},t) \right) \colon \left(\boldsymbol{\epsilon}_{ex}(\boldsymbol{x},t) - \boldsymbol{\epsilon}_{ad}(\boldsymbol{x},t) \right) \mathrm{d}\boldsymbol{x} \right\}^{\frac{1}{2}}, \end{split}$$

along with the L^{∞} control in time of the free complementary energy norm of the exact error of the generalised stress field conjugate of the kinematically admissible solution. That is, if we let

$$\begin{split} \|\boldsymbol{e}_{\bar{\boldsymbol{\sigma}}}(\boldsymbol{x},t)\|_{\mathcal{V}}^{2} &= \int_{\Omega} \left(\boldsymbol{\sigma}_{ex}(\boldsymbol{x},t) - \mathbf{C}\boldsymbol{\epsilon}_{ad}^{e}(\boldsymbol{x},t)\right) : \ \mathbf{C}^{-1}\left(\boldsymbol{\sigma}_{ex}(\boldsymbol{x},t) - \mathbf{C}\boldsymbol{\epsilon}_{ad}^{e}(\boldsymbol{x},t)\right) \mathrm{d}\boldsymbol{x} \\ \|\boldsymbol{e}_{\bar{R}}(\boldsymbol{x},t)\|_{\mathcal{M}}^{2} &= \int_{\Omega} \left[R_{ex}(\boldsymbol{x},t) - \mathbf{H}p_{ad}(\boldsymbol{x},t)\right] \mathbf{H}^{-1} \Big[R_{ex}(\boldsymbol{x},t) - \mathbf{H}p_{ad}(\boldsymbol{x},t)\Big] \mathrm{d}\boldsymbol{x} \end{split}$$

then, it follows

$$\|\boldsymbol{e}_{GSF}\|_{L^{\infty}([0,T];\mathcal{V}\times\mathcal{M})} = \sup_{t\leq T} \left\{ \|\boldsymbol{e}_{\bar{\boldsymbol{\sigma}}}(\boldsymbol{x},t)\|_{\mathcal{V}}^{2} + \|\boldsymbol{e}_{\bar{R}}(\boldsymbol{x},t)\|_{\mathcal{M}}^{2} \right\}^{\frac{1}{2}}$$

This choice is motivated by the following result **Theorem 5.1.**

Given a kinematically admissible solution $s_{ad}^{kin} = (\epsilon_{ad}, \epsilon_{ad}^p, p_{ad}),$

$$\begin{aligned} \|\boldsymbol{e}_{\boldsymbol{\epsilon}}\|_{L^{\infty}([0,T];(L^{2}(\Omega))^{d\times d})} &= 0 \\ \|\boldsymbol{e}_{GSF}\|_{L^{\infty}([0,T];\mathcal{V}\times\mathcal{M})} &= 0 \end{aligned} \Rightarrow \begin{aligned} \boldsymbol{\epsilon}_{ad}(\boldsymbol{x},t) &= \boldsymbol{\epsilon}_{ex}(\boldsymbol{x},t) \\ \boldsymbol{\epsilon}_{ad}^{e}(\boldsymbol{x},t) &= \boldsymbol{\epsilon}_{ex}^{e}(\boldsymbol{x},t) \\ \boldsymbol{\epsilon}_{ad}^{p}(\boldsymbol{x},t) &= \boldsymbol{\epsilon}_{ex}^{p}(\boldsymbol{x},t) \\ p_{ad}(\boldsymbol{x},t) &= p_{ex}(\boldsymbol{x},t) \end{aligned}$$

The proof follows by the same arguments as in the proof of Theorem 3.9, and will not be repeated here.

Error in the constitutive equations

This error measure is given by equation (3.47) which is here rewritten in a form that fits better the incremental origin of the admissible solution $s_{ad} = (\sigma_{ad}, R_{ad}; \epsilon_{ad}, \epsilon_{ad}^p, p_{ad})$.

$$e_{ext}^{2}(T) = \frac{\theta_{st}^{2}(t)}{\prod_{1 \le n \le N} \sup_{t \in [t_{n}, t_{n+1}]} \left\{ \int_{\Omega} \left(\sigma_{ad}(x, t) - \mathbf{C} \epsilon_{ad}^{e}(x, t) \right) : \mathbf{C}^{-1} \left(\sigma_{ad}(x, t) - \mathbf{C} \epsilon_{ad}^{e}(x, t) \right) dx + \frac{\theta_{st}^{p^{2}}(t)}{\prod_{n \ge 1} \left[R_{ad}(x, t) - \mathbf{H} p_{ad}(x, t) \right] \mathbf{H}^{-1} \left[R_{ad}(x, t) - \mathbf{H} p_{ad}(x, t) \right] dx} + \frac{\theta_{d}^{2}(t)}{\theta_{d}^{2}(t)} + \theta_{d}^{2}(t_{n}) + 2 \int_{\Omega} \int_{t_{n}}^{t} \left[R_{0} \| \dot{\epsilon}_{ad}^{p}(x, \tau) \| - \sigma_{ad}(x, \tau) : \dot{\epsilon}_{ad}^{p}(x, \tau) + R_{ad}(x, \tau) \dot{p}_{ad}(x, \tau) \right] d\tau dx}{(5.7)}$$

where

$$\theta_d^2(t_n) = 2 \int_0^{t_n} \int_\Omega \left[R_0 \| \dot{\boldsymbol{\epsilon}}_{ad}^p(\boldsymbol{x}, t) \| - \boldsymbol{\sigma}_{ad}(\boldsymbol{x}, t) \colon \dot{\boldsymbol{\epsilon}}_{ad}^p(\boldsymbol{x}, t) + R_{ad}(\boldsymbol{x}, t) \dot{p}_{ad}(\boldsymbol{x}, t) \right] \mathrm{d}\boldsymbol{x} \, \mathrm{d}t.$$

Equation (5.7) basically expresses the global error at the time T in terms of the error at time t_n and of the admissible solution over $[t_n, T]$.

Since the admissible solution $s_{ad}(x, t)$ is continuous piecewise linear over each time step, the time integral in the error expression can be computed easily. Hence, we have for $t \in [t_n, t_{n+1}]$,

$$\begin{split} &\int_{\Omega} \int_{t_n}^t \Big\{ R_0 \| \dot{\boldsymbol{\epsilon}}_{ad}^p(\boldsymbol{x},\tau) \| - \boldsymbol{\sigma}_{ad}(\boldsymbol{x},\tau) \colon \dot{\boldsymbol{\epsilon}}_{ad}^p(\boldsymbol{x},\tau) + R_{ad}(\boldsymbol{x},\tau) \dot{p}_{ad}(\boldsymbol{x},\tau) \Big\} \mathrm{d}\tau \, \mathrm{d}\boldsymbol{x} = \\ &= \frac{t - t_n}{t_{n+1} - t_n} \int_{\Omega} \Big\{ R_0 \| \Delta \boldsymbol{\epsilon}_{ad}^p(\boldsymbol{x}) \| - \frac{\boldsymbol{\sigma}_{ad}(\boldsymbol{x},t) + \boldsymbol{\sigma}_{ad}(\boldsymbol{x},t_n)}{2} \colon \Delta \boldsymbol{\epsilon}_{ad}^p(\boldsymbol{x}) + \\ &+ \frac{R_{ad}(\boldsymbol{x},t) + R_{ad}(\boldsymbol{x},t_n)}{2} \Delta p_{ad}(\boldsymbol{x}) \Big\} \mathrm{d}\boldsymbol{x} \end{split}$$

where

$$\Delta \epsilon^p_{ad}(\boldsymbol{x}) = \epsilon^p_{ad}(\boldsymbol{x}, t_{n+1}) - \epsilon^p_{ad}(\boldsymbol{x}, t_n)$$
$$\Delta p_{ad}(\boldsymbol{x}) = p_{ad}(\boldsymbol{x}, t_{n+1}) - p_{ad}(\boldsymbol{x}, t_n),$$

whereas $\sigma_{ad}(x, t)$ and $R_{ad}(x, t)$ are the time linear interpolation over $[t_n, t_{n+1}]$ of the respective values at t_n and t_{n+1} . Also, it follows

$$\begin{split} \theta_d^2(t_n) &= \sum_{i=1}^{n-1} 2 \int_{\Omega} \Big\{ R_0 \| \Delta \epsilon_{ad,i}^p(\boldsymbol{x}) \| - \frac{\boldsymbol{\sigma}_{ad}(\boldsymbol{x}, t_{i+1}) + \boldsymbol{\sigma}_{ad}(\boldsymbol{x}, t_i)}{2} \colon \Delta \epsilon_{ad,i}^p(\boldsymbol{x}) + \\ &+ \frac{R_{ad}(\boldsymbol{x}, t_{i+1}) + R_{ad}(\boldsymbol{x}, t_i)}{2} \Delta p_{ad,i}(\boldsymbol{x}) \Big\} \mathrm{d}\boldsymbol{x} \end{split}$$

with

$$\Delta \epsilon^p_{ad,i}(\boldsymbol{x}) = \epsilon^p_{ad}(\boldsymbol{x}, t_{i+1}) - \epsilon^p_{ad}(\boldsymbol{x}, t_i)$$
$$\Delta p_{ad,i}(\boldsymbol{x}) = p_{ad}(\boldsymbol{x}, t_{i+1}) - p_{ad}(\boldsymbol{x}, t_i).$$

The value of the dissipation error at $t_1 = 0$, $\theta_d^2(t_1)$, is assumed equal to zero.

Error in solution

This error measure is given by equation (3.74) which is the extended dissipation error associated with the admissible solution $s_{ex,ad} = (\sigma_{ex}, R_{ex}; \epsilon_{ad}, \epsilon_{ad}^{p}, p_{ad})$.

$$\begin{aligned} e_{ex}^{2}(T) &= \\ &= \max_{1 \le n \le N} \sup_{t \in [t_{n}, t_{n+1}]} \left\{ \int_{\Omega} \left(\boldsymbol{\sigma}_{ex}(\boldsymbol{x}, t) - \mathbf{C} \boldsymbol{\epsilon}_{ad}^{e}(\boldsymbol{x}, t) \right) : \ \mathbf{C}^{-1} \left(\boldsymbol{\sigma}_{ex}(\boldsymbol{x}, t) - \mathbf{C} \boldsymbol{\epsilon}_{ad}^{e}(\boldsymbol{x}, t) \right) \, \mathrm{d}\boldsymbol{x} + \\ &+ \int_{\Omega} \left[R_{ex}(\boldsymbol{x}, t) - \mathsf{H} p_{ad}(\boldsymbol{x}, t) \right] \mathsf{H}^{-1} \left[R_{ex}(\boldsymbol{x}, t) - \mathsf{H} p_{ad}(\boldsymbol{x}, t) \right] \, \mathrm{d}\boldsymbol{x} + \theta_{d, ex}^{2}(t_{n}) + \\ &+ 2 \int_{\Omega} \int_{t_{n}}^{t} \left[R_{0} \| \dot{\boldsymbol{\epsilon}}_{ad}^{p}(\boldsymbol{x}, \tau) \| - \boldsymbol{\sigma}_{ex}(\boldsymbol{x}, \tau) : \ \dot{\boldsymbol{\epsilon}}_{ad}^{p}(\boldsymbol{x}, \tau) + R_{ex}(\boldsymbol{x}, \tau) \dot{p}_{ad}(\boldsymbol{x}, \tau) \right] \, \mathrm{d}\tau \, \mathrm{d}\boldsymbol{x} \right\}, \end{aligned}$$

where, as before, it is

$$\theta_{d,ex}^2(t_n) = 2 \int_0^{t_n} \int_\Omega \left[R_0 \| \dot{\boldsymbol{\epsilon}}_{ad}^p(\boldsymbol{x},t) \| - \boldsymbol{\sigma}_{ex}(\boldsymbol{x},t) \colon \dot{\boldsymbol{\epsilon}}_{ad}^p(\boldsymbol{x},t) + R_{ex}(\boldsymbol{x},t) \dot{p}_{ad}(\boldsymbol{x},t) \right] \mathrm{d}\boldsymbol{x} \,\mathrm{d}t.$$

Unlike the analogous expression (5.7), $\sigma_{ex}(x,t)$ and $R_{ex}(x,t)$, as functions of time, are not in general linear over each time step, therefore the time integral must be computed by accounting for the actual time variation of the functions.

For the ease of implementation, it may be useful rewriting the error in the evolution law as follows:

$$\begin{split} &\int_{t_n}^t \left[R_0 \| \dot{\boldsymbol{\epsilon}}_{ad}^p(\boldsymbol{x},\tau) \| - \boldsymbol{\sigma}_{ex}(\boldsymbol{x},\tau) \colon \dot{\boldsymbol{\epsilon}}_{ad}^p(\boldsymbol{x},\tau) + R_{ex}(\boldsymbol{x},\tau) \dot{p}_{ad}(\boldsymbol{x},\tau) \right] \mathrm{d}\tau = \\ &= \frac{t - t_n}{t_{n+1} - t_n} R_0 \| \Delta \boldsymbol{\epsilon}_{ad}^p \| - \frac{1}{t_{n+1} - t_n} \left(\int_{t_n}^t \boldsymbol{\sigma}_{ex}(\boldsymbol{x},\tau) \, \mathrm{d}\tau \right) \colon \Delta \boldsymbol{\epsilon}_{ad}^p + \\ &+ \frac{1}{t_{n+1} - t_n} \left(\int_{t_n}^t R_{ex}(\boldsymbol{x},\tau) \, \mathrm{d}\tau \right) \Delta p_{ad}. \end{split}$$

5.2.3 Numerical example

The general theory developed in the previous Sections of this Chapter is here applied to assess the quality of the finite element solution of a one dimensional elastoplastic bar under distributed axial loads. Despite the simplicity of the model, it allows one to emphasize the physical concepts of the theory and illustrate all the ingredients which characterize an error estimator of a finite element solution of elastoplastic problems (Orlando & Peric, 2000).

The model problem is shown in Figure 5.3 along with the variation of the external load multiplier. The bar is assumed to be composed of an elastoplastic material which obeys the Prandtl-Reuss plasticity law with linear hardening. The hardening law and elastic domain in the space of the generalized stresses are also depicted in Figure 5.3. For the problem at hand a closed form solution was not available, hence an "overkill" procedure has been adopted, that is, the backward Euler finite element solution of a uniform mesh of 2000 elements with linear interpolation and time step dt = 0.025 has been used as an "exact" solution.

The evolution of the state variables is reported in Figure 5.4 and Figure 5.5. Here, a steep gradient of the total strain is observed over the subdomain $\Omega' = [0.725, 0.7275]$ starting at t = 16.375 which spreads over [0.628, 0.73] as the load is increased (in absolute value). This is due to the fact that plastic strains of opposite sign are therein produced further to the sign reversing of the load. For the associative model under consideration, the plastic strain rate is, indeed, given by $\dot{\epsilon}^p = \lambda \operatorname{sign}(\sigma)$ where the plastic multiplier is nonnegative, i.e. $\lambda \geq 0$, thus if plastic loading occurs, the sign of $\dot{\epsilon}^p$ is the same as σ .



Figure 5.3: 1D Model Problem

t_n	t_{n+1}	Load Mult.	Domain Ω'
		μ_{n+1}	
1.500	1.525	1.525	[0.995, 1.0]
2.525	2.550	2.550	$[0., 0.031] \cup [0.8865, 1.0]$
5.975	6.000	6.000	$[0., 0.439] \cup [0.726, 1.0]$
16.375	16.400	-4.400	[0.725, 0.7275]
17.500	17.525	-5.525	[0.7075, 0.9115]
17.525	17.550	-5.550	[0.709, 0.9215]
17.675	17.700	-5.700	[0.756, 1.0]
17.700	17.725	-5.725	[0.76, 1.0]
18.175	18.200	-6.200	[0.76, 1.0]
18.200	18.225	-6.225	$[0, 0.0705] \cup [0.7595, 1.0]$
29.975	30.000	-18.000	$[0, 0.5315] \cup [0.628, 1.0]$

Table 5.1: Parts of the domain which experience plastic loading for the given loading history. "Exact" solution is obtained as finite element solution with 2000 elements and dt = 0.025



Figure 5.4: Exact Stress and Total Strain Distribution at each dt = 0.75 for $t \in [0.0, 30.0]$



Figure 5.5: Exact Plastic Strain and Accumulated Plastic Strain Distribution at each dt = 0.75 for $t \in [0.0, 30.0]$

The Table 5.1 reports the parts Ω' of the domain Ω which experience plastic loading for the given loading history.

Before proceeding to the error analysis of several fully discrete schemes of the problem given in Figure 5.3, some considerations are due on the particularity of time and finite element discretization of 1D associative plasticity problems which will help to gain critical insight into the error behaviour.

As a result of finite element interpolation for the displacement field, the initial boundary value problem which governs the evolution of the elastoplastic continuum is transformed into a system of ordinary differential equations given in implicit form and algebraic constraints on the variables, which are usually called differential– algebraic equations (Brenan *et al.*, 1996). The differential problem is stated over the time interval of interest and its unknowns are given by the nodal displacements.

The exact solution of this system of equations would allow one to build a displacement field which is affected by only space discretization error. Unfortunately, the resulting system is complex, for the main difficulty comes from the exact integration of the initial value constitutive problem which has been obtained only for very special conditions, such as in Krieg & Krieg (1977) and Ristinmaa & Tryding (1993). Consequently, as we have discussed in Section 4.3, one must generally resort to numerical integration algorithms, such as, the implicit backward Euler difference scheme. For an autonomous ordinary differential equation, for example,

$$\begin{vmatrix} \dot{y} = F(y) \\ y(t_0) = y_0 \end{vmatrix}$$

the backward Euler would be displayed as in Figure 5.6(a) where the slope $\frac{y_{n+1} - y_n}{\Delta t}$



Figure 5.6: Backward Euler method graphically displayed: (a) For a general ordinary differential equation (b) For the initial value constitutive problem of 1D associative plasticity. The solution is exact at the time instants of the discretization

is given by the value of the function F(y) at y_{n+1} . Also, note that the approximate solution does not lie in general on the exact curve.

In 1D associative plasticity, update stresses equations obtained from backward Euler integration scheme applied to the evolution equations are solved by a predictorcorrector algorithm which reads as in Box 5.2 (see, e.g., de Souza Neto *et al.*, 2002), and has the geometrical interpretation given in Figure 5.7.

By inspection of the same Figure, it follows at once that the use of backward Euler will deliver the exact solution at the load levels that are considered. Under the assumption that the projection direction on the yield surface does not depend on the imposed total strain, which occurs surely in 1D plasticity and isotropic hardening, the application of the total strain increment $\Delta \epsilon$ or, for instance, of two increments $\Delta \epsilon^{(i)}$ such that $\Delta \epsilon = \Delta \epsilon^{(1)} + \Delta \epsilon^{(2)}$ delivers the same final state. This can also



Figure 5.7: Return mapping in 1D associative plasticity for $\Delta \epsilon$ and $\Delta \epsilon = \Delta \epsilon^{(1)} + \Delta \epsilon^{(2)}$

be read on the isoerror maps given in Krieg & Krieg (1977) and Schreyer *et al.* (1979). Finally, this means that time discretization effects for different values of the time step depend upon the difference in the time linear interpolations of the corresponding discrete values, Figure 5.6(b).



Figure 5.8: Exact and finite element solutions for an elastic bar with constant properties. The finite element solutions are exact at the nodes of the mesh.

The one dimensionality of the problem has also particularities in terms of space discretization. For elastic behaviour, the orthogonality of the residual with respect to the finite element space and the local properties of the shape functions allows one to show quite easily that the finite element solution is exact at the nodes if, for example, the material properties of the bar are uniform and the elements are linear. This result holds, however, also for more general conditions, for which we refer to Babuska & Strouboulis (2001). The approximation, therefore, arises in the difference between the displacement finite element interpolation within each element and the exact displacement field, Figure 5.8.

For an elastoplastic material, this is no more true. Nevertheless one can expect that the finite element solution at the nodes is close to the exact one.

Box 5.2. Fully implicit Elastic predictor/Return mapping algorithm for numerical integration of 1D associative plasticity constitutive equations. Isotropic hardening

Data:	ϵ_n^p, p_n
Given:	ϵ_{n+1}
Evaluate:	Elastic Trial State $\begin{vmatrix} \epsilon_{n+1}^{p,tr} = \epsilon_{n}^{p}, \ \epsilon_{n+1}^{e,tr} = \epsilon_{n+1} - \epsilon_{n+1}^{p,tr}, \ p_{n+1}^{tr} = p_{n} \\ \sigma_{n+1}^{tr} = C\epsilon_{n+1}^{e,tr}, \ R_{n+1}^{tr} = g(p_{n+1}^{tr}) \end{vmatrix}$
Check:	IF $f_{n+1}^{tr} \stackrel{\text{def}}{=} f(\sigma_{n+1}^{tr}, R_{n+1}^{tr}) \le 0$ THEN set $(\bullet)_{n+1} = (\bullet)_{n+1}^{tr}$ ELSE \Rightarrow Return mapping
Return mapping:	END IF Solve the system $\left \begin{array}{c} \epsilon_{n+1}^{e} - \epsilon_{n+1}^{e,tr} + \lambda_{n+1} \frac{\partial f}{\partial \sigma}(\sigma_{n+1}, R_{n+1}) = 0 \\ tr \end{array} \right _{\sigma} \left \begin{array}{c} \delta f \\ \delta f \end{array} \right _{\sigma} \left \begin{array}{c} \delta f \\ \delta f \end{array} \right _{\sigma} \left \begin{array}{c} \delta f \\ \delta f \end{array} \right _{\sigma} \left \begin{array}{c} \delta f \\ \delta f \end{array} \right _{\sigma} \left \begin{array}{c} \delta f \\ \delta f \end{array} \right _{\sigma} \left \begin{array}{c} \delta f \\ \delta f \end{array} \right _{\sigma} \left \begin{array}{c} \delta f \\ \delta f \end{array} \right _{\sigma} \left \begin{array}{c} \delta f \\ \delta f \end{array} \right _{\sigma} \left \begin{array}{c} \delta f \\ \delta f \end{array} \right _{\sigma} \left \begin{array}{c} \delta f \\ \delta f \end{array} \right _{\sigma} \left \begin{array}{c} \delta f \\ \delta f \end{array} \right _{\sigma} \left \begin{array}{c} \delta f \\ \delta f \end{array} \right _{\sigma} \left \begin{array}{c} \delta f \\ \delta f \end{array} \right _{\sigma} \left \begin{array}{c} \delta f \\ \delta f \end{array} \right _{\sigma} \left \begin{array}{c} \delta f \\ \delta f \end{array} \right _{\sigma} \left \begin{array}{c} \delta f \\ \delta f \end{array} \right _{\sigma} \left \begin{array}{c} \delta f \\ \delta f \end{array} \right _{\sigma} \left \begin{array}{c} \delta f \\ \delta f \end{array} \right _{\sigma} \left \begin{array}{c} \delta f \\ \delta f \end{array} \right _{\sigma} \left \begin{array}{c} \delta f \\ \delta f \end{array} \right _{\sigma} \left \begin{array}{c} \delta f \\ \delta f \end{array} \right _{\sigma} \left \begin{array}{c} \delta f \\ \delta f \end{array} \right _{\sigma} \left \begin{array}{c} \delta f \\ \delta f \end{array} \right _{\sigma} \left \begin{array}{c} \delta f \\ \delta f \end{array} \right _{\sigma} \left \begin{array}{c} \delta f \\ \delta f \end{array} \right _{\sigma} \left \begin{array}{c} \delta f \\ \delta f \end{array} \right _{\sigma} \left \begin{array}{c} \delta f \\ \delta f \end{array} \right _{\sigma} \left \begin{array}{c} \delta f \\ \delta f \end{array} \right _{\sigma} \left \begin{array}{c} \delta f \\ \delta f \end{array} \right _{\sigma} \left \begin{array}{c} \delta f \\ \delta f \end{array} \right _{\sigma} \left \begin{array}{c} \delta f \\ \delta f \end{array} \right _{\sigma} \left \begin{array}{c} \delta f \\ \delta f \end{array} \right _{\sigma} \left \left \begin{array}{c} \delta f \\ \delta f \end{array} \right _{\sigma} \left \left \begin{array}{c} \delta f \\ \delta f \end{array} \right _{\sigma} \left \left \left \begin{array}{c} \delta f \\ \delta f \end{array} \right _{\sigma} \left $
	$\begin{vmatrix} p_{n+1} - p_{n+1}^{e_{n+1}} - \lambda_{n+1} \frac{1}{\partial R} (\sigma_{n+1}, R_{n+1}) = 0 \\ \lambda_{n+1} > 0, \ f_{n+1} = 0 \\ \text{for } \lambda_{n+1}, \epsilon_{n+1}^{e}, p_{n+1}, \text{ with} \\ \sigma_{n+1} = C\epsilon_{n+1}^{e}, \ R_{n+1} = g(p_{n+1}) \end{vmatrix}$

In the light of the aforementioned observations, we will consider fully discrete schemes where time steps and meshes are tailored *ad hoc* so that effects of space and time discretization can be relevant. This has been realized by requiring that there are different finite elements to yield in the model once the load level varies from 3.0 to 6.0 with step equal to 1.5. In this way, the response of the finite element model departs from the time linear interpolation of the solutions at t = 3.0 and t = 6.0. Therefore, an underlying non uniform mesh *me*0 has been first constructed with this criterion whereas the other meshes have been obtained by halving each element



Figure 5.9: Fully discrete schemes analysed

of the corresponding parent mesh. Also, uniform partition of the time interval of interest with time steps $k = \Delta t = 6.0, 3.0, 1.5, 0.75$ have been assumed. The fully discrete schemes are shown in Figure 5.9. These have been analysed using an incremental solution based on the backward Euler stepping scheme and the Newton-Raphson procedure. Only one Gauss point has been used for the integration of the constitutive equations, because linear finite elements are used. Plastic strain ${}^{h}\epsilon_{n+1}^{p}$ and accumulated plastic strain ${}^{h}p_{n+1}$, obtained at the single quadrature point of each element, are prolongated into uniform field over the respective element.

For the definition of the corresponding admissible solution necessary to compute the extended dissipation error the criteria described in the previous Section apply. In particular, if the 1D system is statically indeterminated, the statically admissible stress field $\sigma_{ad}(x, t_{n+1})$ is obtained by solving over each element $\Omega_e^h \in \mathcal{T}_h$ the following equilibrium problem,

$$\forall \Omega_{e}^{h} =]x_{e}^{h}, x_{e+1}^{h} [\in \mathcal{T}_{h}$$

$$\left| \begin{array}{c} -\frac{\mathrm{d}}{\mathrm{d}x} \sigma_{ad}^{e}(x, t_{n+1}) + f(x, t_{n+1}) = 0 \\ \sigma_{ad}^{e}(x_{e}^{h}, t_{n+1}) = -\int_{x_{e}^{h}}^{x_{e+1}^{h}} h \sigma(x) \frac{\mathrm{d}\mathbb{N}_{e}^{h}}{\mathrm{d}x}(x) \,\mathrm{d}x + \int_{x_{e}^{h}}^{x_{e+1}^{h}} f(x, t_{n+1}) \mathbb{N}_{e}^{h}(x) \,\mathrm{d}x \end{array} \right|$$

$$(5.9)$$

where the boundary condition is obtained from the prolongation condition (cf. Ladevèze & Leguillon, 1983)

$$\int_{x_e^h}^{x_{e+1}^n} \left(\sigma_{ad}^e(x, t_{n+1}) - {}^h \sigma(x) \right) v(x) \mathrm{d}x \quad \forall v \in \mathcal{V}^h(\Omega_e^h)$$
(5.10)

with $\mathcal{V}^h(\Omega_e^h)$ being the finite dimensional space generated by the element Lagrangian shape functions $\mathbb{N}_e^h(x)$ and $\mathbb{N}_{e+1}^h(x)$ depicted in Figure 5.10.



Figure 5.10: Element Lagrangian shape functions of a linear finite element

Remark 5.2. It is an easy matter to check that condition (5.10), albeit local, is consistent with the continuity of $\sigma_{ad}^e(x, t_{n+1})$ across the nodes, i.e.,

$$\sigma_{ad}^{e-1}(x_e, t_{n+1}) = \sigma_{ad}^e(x_e, t_{n+1}),$$

and also, that the following equilibration condition

$$\sigma_{ad}^{e}(x_{e}, t_{n+1}) - \sigma_{ad}^{e}(x_{e+1}, t_{n+1}) + \int_{x_{e}^{h}}^{x_{e+1}^{h}} f(x, t_{n+1}) dx = 0$$

is satisfied. Furthermore, it is possible to show that condition (5.10) corresponds to a splitting assumption for the singular component of the residual in the equilibrium equation associated with the finite element stress, ${}^{h}\sigma$, and given by ${}^{h}\sigma^{e}(x_{e}^{+}, t_{n+1}) {}^{h}\sigma^{e-1}(x_{e}^{-}, t_{n+1})$, if a distribution assumption, over the element, of the finite element stresses computed at the quadrature points has also been respected.

As for the admissible plastic strain, in 1D associative plasticity it may be written

$$\boldsymbol{n}_{\sigma_{ad}}(x, t_{n+1})\boldsymbol{n}_{\Delta\epsilon_{\sigma_{ad}}^p} = \pm 1.$$
(5.11)

Thus, we define $\epsilon_{ad}^p(x, t_{n+1})$ such that +1 holds in equation (5.11). This, finally, means that we assume $\epsilon_{ad}^p(x, t_{n+1}) = {}^h \epsilon_{n+1}^p(x)$.

For the computation of the error expressions given in Section 5.2.2, the supremum over the generic time interval $[t_n, t_{n+1}]$ is computed as maximum of a discrete set given by the value of the functions sampled at the time $t_i \in [t_n, t_{n+1}]$ where the "exact" solution is known, whereas the space integrals have been computed with Gauss quadrature by using 12 quadrature points. This does not incur a supplementary computational effort, for only sampling of functions is involved.

The extended dissipation error computed at each time t_i , as given by equation (5.7), and the error in solution, as given by equation (5.8), have been used to define the effectivity index

$$\xi(t) = \frac{e_{ext}(t)}{e_{ex}(t)},\tag{5.12}$$

likewise for elliptic problems (Babuska & Rheinboldt, 1978b). The time evolution of $\xi(t)$ is shown in Figure 5.11 for the fully discrete schemes analysed. For all the computations, the effectivity index was ranging between 1.00 to 1.45. As a result,



Figure 5.11: Evolution in time of the effectivity index for different fully discrete schemes

the extended dissipation error $e_{ext}(t)$ can be used as a reliable estimate of the error in solution $e_{ex}(t)$. Furthermore, it is noted that $\xi(t)$ does not approach to 1 as the discretization becomes finer. The condition $\xi \to 1$ defines the asymptotic exactness of the error estimator. This is a desirable property, though not strictly necessary for the reliability and efficiency of the estimator which results from the boundness of the effectivity index from above and below (≥ 1) , respectively. However, the asymptotic properties of the extended dissipation error will be analysed numerically in the next section. There, it will be shown that $e_{ext}(t) \to 0$ as $h, \Delta t \to 0$ and, consequently, the approximate solution will converge to the exact solution. Therefore, at this stage, the default of the extended dissipation error of being an asymptotically exact estimator of the error in solution is not of concern and deserves certainly major investigation in the future. Also, one must be aware that, for instance, for elliptic problems, the asymptotic exactness of an error estimator is a very fragile property (Verfurth, 1996) that depends on or requires, amongst other things, very regular meshes and a fairly smooth solution. In fact, proofs of asymptotic exactness of some estimators given in literature are all based on the validity of superconvergence results (Babuska & Rheinboldt, 1979a, 1981; Duran & Rodriguez, 1992; Babuska & Rodriguez, 1993; Verfurth, 1996). Therefore, the importance of this property does not have to be over-emphasized (Ainsworth & Oden, 1997) and certainly it is not expected to hold for the general types of meshes used in the practical engineering computations (Babuska et al., 1994a) and a fortiori for non linear problems.



Figure 5.12: Time variation of the extended dissipation error and its components

A typical time evolution of the extended dissipation error $e_{ext}(t) = \sup_{\tau \leq t} \theta(\tau)$ for a fully discrete finite element model is given in Figure 5.12. In this picture, also the current value of the components, $\theta(t) = \sqrt{\theta_{sl}^2(t) + \theta_d^2}$ with $\theta_{sl}(t) = \sqrt{\theta_{sl}^{e^2}(t) + \theta_{sl}^{p^2}(t)}$, defined in equation (5.7), are reported. By definition, $e_{ext}(t)$ is a nondecreasing function of time, which allows one to assess the quality of the approximate solution over the whole interval of interest. The same monotone character is presented also by the current value of the error component in the evolution law, $\theta_d(t)$, which reflects the irreversible phenomena associated with the admissible solution up to the current time t. On the contrary, the error in the state law, $\theta_{sl}(t)$, which is defined in terms of the free Helmholtz energy and its conjugate, will depend on the energy associated with the current approximate solution. Thus its time variation will depend on the behaviour of the error in the elastic law at t = 28.5.

In particular, because of the linearity of the elastic law and definition of the statically admissible stress field as in (5.9), for the problem at hand, it follows that

$$\theta_{sl}^{e\,2}(t) = \sum_{e} \frac{1}{\mathsf{C}} \int_{x_{e}^{h}}^{x_{e+1}^{h}} |\sigma_{ad}^{e}(x,t) - \mathsf{C}\epsilon_{ad}^{e,e}(x,t)|^{2} \,\mathrm{d}x = \\ = \mu^{2}(t) \sum_{e} \frac{1}{\mathsf{C}} \int_{x_{e}^{h}}^{x_{e+1}^{h}} \left| \int_{x_{e}^{h}}^{x_{e+1}^{h}} f(\xi) \mathbb{N}_{e}^{h}(\xi) \mathrm{d}\xi - \int_{x_{e}^{h}}^{x} f(\xi) \mathrm{d}\xi \right|^{2} \mathrm{d}x,$$
(5.13)

which shows that the time variation of $\theta_{sl}^e(t)$ is the same as the variation of the external load, even in presence of plastic loading. The previous result is obtained further



Figure 5.13: The error in the state law at t=28.5 (a) Admissible stress σ_{ad} versus stress conjugate to the admissible elastic strain $C\epsilon^{e}_{ad}$. (b) Pointwise contribution to the error in the elastic law, $\frac{1}{C}(\sigma_{ad} - C\epsilon^{e}_{ad})^{2}$, at t=28.5 (c) Admissible thermodynamic force R_{ad} versus force conjugate of the admissible accumulated plastic strain Hp_{ad} . (d) Pointwise contribution to the error in the hardening law, $\frac{1}{H}(R_{ad} - Hp_{ad})^{2}$, at t=28.5

to the assumption that $\epsilon_{ad}^{p,e}(x,t) = {}^{h}\epsilon^{p}(x_{e,GP},t)$, thus $C\epsilon_{ad}^{e,e}(x,t) = {}^{h}\sigma(x_{e,GP},t)$, where only one Gauss point per element is used for the numerical integration of the constitutive equations.

With regard, then, to the time variation of $\theta_{sl}^p(t)$, changes of θ_{sl}^p are noted if plastic loading occurs otherwise θ_{sl}^p remains constant if the admissible thermodynamic force R_{ad} , and the thermodynamic force conjugate to the admissible accumulated plastic strain, Hp_{ad} , do not vary, such as during elastic unloading. Also, when plastic deformations should occur in the model, as identified by value different from zero of R_{ad} , but they are not detected by the admissible kinematic solution, p_{ad} , a contribution different from zero to θ_{sl}^p comes from these parts of the domain Ω . This circumstance is shown in Figure 5.13(c) and Figure 5.13(d) which report, for instance, the pointwise contribution to the error in the hardening law at t = 28.5. Even though the admissible accumulated plastic strain is zero all over the element el = 4, a contribution different from zero to the error associated with the plastic energy is obtained from this element for being, therein, the thermodynamically admissible force different from zero. This may be interpreted as the plastic energy that we have somehow to supply to this part of the domain, since the kinematically admissible solution was not able to describe it.

Major critical insight in the causes that produce variation of the error in the evolution law, θ_d , is obtained by analysing the pointwise contribution to θ_d within a given time interval $[t_n, t_{n+1}]$,

$$[t_n, t_{n+1}]\zeta_d^2(\boldsymbol{x}) \stackrel{\text{def}}{=} \int_{t_n}^{t_{n+1}} \left[R_0 \| \dot{\boldsymbol{\epsilon}}_{ad}^p(\boldsymbol{x}, t) \| - \boldsymbol{\sigma}_{ad}(\boldsymbol{x}, t) \colon \dot{\boldsymbol{\epsilon}}_{ad}^p(\boldsymbol{x}, t) + R_{ad}(\boldsymbol{x}, t) \dot{p}_{ad}(\boldsymbol{x}, t) \right] \mathrm{d}t$$

which is next considered in its general tensorial notation.

By accounting for the definition of the admissible plastic strain and accumulated plastic strain, the assumption of time linear variation for the admissible solution over the time interval $[t_n, t_{n+1}]$ and the fact that $n_{\sigma_{ad}^D}(x, t) : n_{\Delta \epsilon_{ad}^p}(x) = +1$, it follows that

$${}_{[t_n,t_{n+1}]}\zeta_d^2(\boldsymbol{x}) = \frac{\|\Delta \boldsymbol{\epsilon}_{ad}^p\|}{2} \Big[R_0 + R_{ad}(\boldsymbol{x},t_{n+1}) - \|\boldsymbol{\sigma}_{ad}^D(\boldsymbol{x},t_{n+1})\| \Big] + \\ + \frac{\|\Delta \boldsymbol{\epsilon}_{ad}^p\|}{2} \Big[R_0 + R_{ad}(\boldsymbol{x},t_n) - \|\boldsymbol{\sigma}_{ad}^D(\boldsymbol{x},t_n)\| \Big],$$

$$(5.14)$$

where, in general, $R_0 + R_{ad}(\boldsymbol{x}, t) - \|\boldsymbol{\sigma}_{ad}^D(\boldsymbol{x}, t)\| \ge 0$. As a result, if at the point $\boldsymbol{x} \in \Omega$ the following expressions are valid

$$\begin{aligned} R_0 + R_{ad}(\boldsymbol{x}, t_n) - \|\boldsymbol{\sigma}_{ad}^D(\boldsymbol{x}, t_n)\| &= 0, \\ R_0 + R_{ad}(\boldsymbol{x}, t_{n+1}) - \|\boldsymbol{\sigma}_{ad}^D(\boldsymbol{x}, t_{n+1})\| &= 0, \end{aligned}$$

then there is no local contribution to the error in the evolution law from the current time interval. This, in turn, means that given

$$\sigma_{ad}^{D}(\mathbf{x}, t) = \frac{t - t_{n}}{t_{n+1} - t_{n}} \sigma_{ad}^{D}(\mathbf{x}, t_{n+1}) + \frac{t_{n-1} - t}{t_{n+1} - t_{n}} \sigma_{ad}^{D}(\mathbf{x}, t_{n})$$

$$R_{ad}(\mathbf{x}, t) = \frac{t - t_{n}}{t_{n+1} - t_{n}} R_{ad}(\mathbf{x}, t_{n+1}) + \frac{t_{n-1} - t}{t_{n+1} - t_{n}} R_{ad}(\mathbf{x}, t_{n})$$

$$\epsilon_{ad}^{p}(\mathbf{x}, t) = \frac{t - t_{n}}{t_{n+1} - t_{n}} \epsilon_{ad}^{p}(\mathbf{x}, t_{n+1}) + \frac{t_{n-1} - t}{t_{n+1} - t_{n}} \epsilon_{ad}^{p}(\mathbf{x}, t_{n})$$

$$p_{ad}(\mathbf{x}, t) = \frac{t - t_{n}}{t_{n+1} - t_{n}} p_{ad}(\mathbf{x}, t_{n+1}) + \frac{t_{n-1} - t}{t_{n+1} - t_{n}} p_{ad}(\mathbf{x}, t_{n})$$
(5.15)

with

$$\begin{aligned} \|\boldsymbol{\sigma}_{ad}^{D}(\boldsymbol{x}, t_{n})\| - (R_{0} + R_{ad}(\boldsymbol{x}, t_{n})) &\leq 0 \\ \|\boldsymbol{\sigma}_{ad}^{D}(\boldsymbol{x}, t_{n+1})\| - (R_{0} + R_{ad}(\boldsymbol{x}, t_{n+1})) &\leq 0 \\ \operatorname{Tr}[\Delta \boldsymbol{\epsilon}_{ad}^{p}] &= 0 \\ \Delta p_{ad} &= \|\Delta \boldsymbol{\epsilon}_{ad}^{p}\| \\ \boldsymbol{n}_{\boldsymbol{\sigma}_{ad}^{D}}(\boldsymbol{x}, t) \colon \boldsymbol{n}_{\Delta \boldsymbol{\epsilon}_{ad}^{p}}(\boldsymbol{x}) &= +1, \quad \forall t \in [t_{n}, t_{n+1}] \end{aligned}$$
(5.16)



Figure 5.14: (a)Variation of admissible plastic strain between t = 18.0 and t = 19.5 (b)Space distribution of $f(\sigma_{ad}, R_{ad})$ at t = 18.0 and t = 19.5 (c)Pointwise contribution to the increment of the error in the evolution law.

then, it follows that

$$\forall t \in [t_n, t_{n+1}], \begin{cases} \|\boldsymbol{\sigma}_{ad}^D(\boldsymbol{x}, t)\| - \left[R_0 + R_{ad}(\boldsymbol{x}, t)\right] \le 0 \\ \|\dot{\boldsymbol{\epsilon}}_{ad}^p(\boldsymbol{x}, t)\| - \dot{p}_{ad}(\boldsymbol{x}, t) \le 0 \\ \operatorname{Tr}[\dot{\boldsymbol{\epsilon}}_{ad}^p(\boldsymbol{x}, t)] = 0 \\ R_0 \|\dot{\boldsymbol{\epsilon}}_{ad}^p(\boldsymbol{x}, t)\| - \boldsymbol{\sigma}_{ad}(\boldsymbol{x}, t) \colon \dot{\boldsymbol{\epsilon}}_{ad}^p(\boldsymbol{x}, t) + R_{ad}(\boldsymbol{x}, t)\dot{p}_{ad}(\boldsymbol{x}, t) = 0. \end{cases}$$

That is, the function $(\sigma_{ad}(\boldsymbol{x}, t), R_{ad}(\boldsymbol{x}, t); \epsilon^p_{ad}(\boldsymbol{x}, t), p_{ad}(\boldsymbol{x}, t))$ defined as in (5.15) upon the conditions (5.16) is an integral of the evolution law over the time interval $[t_n, t_{n+1}]$, (see Section 3.2.4.5). The difference from the exact solution, however, lies in the diversity of initial condition met at t_n . These observations are finally visualized in Figure 5.14 which show the pointwise contribution to θ_d within the time interval [18.0, 19.5]. In conclusion, $[t_n, t_{n+1}]\zeta_d(\boldsymbol{x})$ can be equal to zero without implying that the state variables do not change between t_n and t_{n+1} . Their variation is, in fact, detected by the variation of the current error in the state law.



Figure 5.15: Extended dissipation error along with its components for different discretizations. Evolution of the error in the state law and in the evolution law with respect to the global error as the time step is reduced.

5.2.4 Analysis of the error

In this Section the behaviour of the extended dissipation error with respect to the parameters that control the approximation, namely time step size and mesh size, is investigated numerically. We start by analysing the relative importance of the error components θ_{sl} and θ_d with respect to the global error e_{ext} .

Figure 5.15 describes the influence of the time discretization for a given finite element mesh, which is the mesh me1 defined in Figure 5.9, whereas Figure 5.16 shows the influence of the space discretization for given time discretization realized by uniform time step $\Delta t = 1.5$. In general, one notes a reduction of the absolute error e_{ext} along with the variation of the relative importance of its components.

155



Figure 5.16: Extended dissipation error along with its components for different discretizations. Evolution of the error in the state law and in the evolution law with respect to the global error as the finite element mesh is refined.

In particular, Figure 5.15 shows that by reducing the time step, the variation of $\theta_{sl}(t)$ remains almost unchanged whereas $\theta_d(t)$ reduces, so that θ_{sl} represents the main error component. This behaviour is easier to comprehend if we consider the time step contribution to the error in the evolution law given by $\int_{\Omega} [t_n, t_{n+1}] \zeta_d(x) dx$. This is depicted in Figure 5.17 which compares two fully discrete schemes having the same finite element mesh me1 and different time step, $\Delta t = 1.5$ and $\Delta t = 0.75$, respectively. For the problem at hand, at a given point $x \in \Omega$, the error in the evolution law results from assuming a linear variation for the variables ϵ_{ad}^p , p_{ad} over the whole time interval $[t_n, t_{n+1}]$, whereas the time t_c , which is the time when the exact plastic loading occurs, belongs to $]t_n, t_{n+1}[$. As a result, the exact linear



Figure 5.17: Contribution to the error in the evolution law for given finite element mesh, me1, and different time discretizations, time step $\Delta t = 1.5$ and $\Delta t = 0.75$ (a) Time step contribution to the error in the evolution law (b) Time variation for the extended dissipation error along with its components





157

variation of ϵ^p and p occurs only over $[t_c, t_{n+1}]$.

The reduction of the time step reduces the time interval during which the plastic flow contributes to the error because of the linear interpolation. This situation is sketched in Figure 5.18.

Figure 5.16, as well, shows that by refining the finite element mesh size the error component $\theta_{sl}(t)$ reduces drastically so that θ_d constitutes the main error component. The dependence of θ_{sl} on the mesh size h, and more specifically of θ_{sl}^e on h, can also be revealed by a direct analysis of the equation (5.13). Figure 5.16 also shows that the dependence of θ_d on the mesh size is not well defined as it can be inferred from comparing the variation of θ_d for the meshes me0 and me1. Here, it is noted that θ_d increases by refining the mesh size. This behaviour is visualized in Figure 5.19 which compares the time step contribution to the global error in the evolution law for the previous schemes. This circumstance occurs as a result of a larger area experiencing plastic loading detected by the mesh me1 with the elements el = 7 and el = 8, as shown in Figure 5.20.

Another aspect taken into account in this section is to motivate the use of the extended dissipation error by showing that this measure of the error does indeed reflect the global quality of an admissible solution. This is proved numerically on the model problem under consideration by investigating the behaviour of the family of finite element approximations corresponding to values of the discretization parameters approaching their limit values, that is,

$$\begin{vmatrix} e_{ext} \to 0 & \text{as } h, \Delta t \to 0 \\ e_{ext} \to e_{ext,\Delta t} & \text{as } h \to 0 \\ e_{ext} \to e_{ext,h} & \text{as } \Delta t \to 0 \end{cases}$$
(5.17)

where $e_{ext,\Delta t}$ and $e_{ext,h}$ denote the error due to only time and space discretization, respectively. That is, $e_{ext,\Delta t}$ is defined as the error associated with the exact solution of the nonlinear incremental boundary value problem obtained by performing only the time discretization of the initial boundary value problem; whereas, $e_{ext,h}$ is the error associated with the exact solution of the system of differential algebraic equations obtained by performing only the space finite element discretization of the initial boundary value problem.

We also compare the time evolution of the extended dissipation error (which can be thought of as an estimate of the error of the kinematically admissible variables) with the classical measures of the exact error defined as difference between exact and approximate solution and introduced in Section 5.2.2. The aim of this comparison is to show that the extended dissipation error describes quite well the evolution of the approximate solution compared to the exact one. That is, the occurrence of (5.17) corresponds effectively to have the approximate solution approaching to the exact one, to the solution of the time discrete scheme and of the space discrete scheme, respectively.

The results of these studies are delivered in Figures 5.21 and 5.22. In particular, Figure 5.21 shows the effects of the space discretization error by comparing fully



Figure 5.19: Contribution to the error in the evolution law for given time discretization, time step $\Delta t = 1.5$, and different finite element mesh, me0 and me1 (a) Time step contribution to the error in the evolution law (b) Time variation for the extended dissipation error along with its components



Figure 5.20: (a) Admissible plastic strain distribution for mesh me0 and time step $\Delta t = 1.5$ (b) Space distribution of $f(\sigma_{ad}, R_{ad})$ for mesh me0 and time step $\Delta t = 1.5$ (c) Admissible plastic strain distribution for mesh me1 and time step $\Delta t = 1.5$ (d) Space distribution of $f(\sigma_{ad}, R_{ad})$ for mesh me1 and time step $\Delta t = 1.5$

discrete schemes which present the same uniform time discretization defined by $\Delta t = 1.5$ and the finite element meshes depicted in Figure 5.9. The asymptotic behaviour of the error measures, however, is better appreciated by considering the variation for fixed values of the time. These diagrams are reported in the same figure and describe the variation of the error at t = 1.5, t = 6, t = 24.0 and t = 30.0.



Figure 5.21: Extended dissipation error and classical measures of the exact error. Effect of space discretization for given time discretization. Time step $\Delta t = 0.75$.

At t = 1.5, the behaviour of the bar is elastic, consequently the extended dissipation error along with the other measures of the error, which involve gradient of the displacement, exhibit a linear rate of convergence which is the type of convergence of the error in energy norm for linear elements, (see, e.g., Babuska & Rheinboldt, 1978b) or (Ciarlet, 1978). In fact, for elastic behaviour it is

$$\sigma_{ad}(\boldsymbol{x},t) = \mu(t)\sigma_{ad}(\boldsymbol{x}); \quad \boldsymbol{\epsilon}_{ad}^{e}(\boldsymbol{x},t) = \boldsymbol{\epsilon}_{ad}(\boldsymbol{x},t) = \mu(t)\boldsymbol{\epsilon}_{ad}(\boldsymbol{x})$$

whereas the other state variables vanish. Thus equation (5.7) reduces to the same expression obtained in linear elasticity

$$e_{ext}^{2}(T) = \sup_{t \leq T} \mu^{2}(t) \int_{\Omega} \left(\boldsymbol{\sigma}_{ad}(\boldsymbol{x}) - \mathbf{C}\boldsymbol{\epsilon}_{ad}^{e}(\boldsymbol{x}) \right) : \ \mathbf{C}^{-1} \left(\boldsymbol{\sigma}_{ad}(\boldsymbol{x}) - \mathbf{C}\boldsymbol{\epsilon}_{ad}^{e}(\boldsymbol{x}) \right) \mathrm{d}\boldsymbol{x}$$
(5.18)

which is the norm of the error induced by the elastic energy once one normalizes the error with respect to the load multiplier.

At the other time instants t_n plastic deformations occur in the bar. Hence, in this case, it is interesting to note that the functions $e_{ext}(t_n, h)$, $\|e_{\epsilon_{ad}}\|_{L^{\infty}([0,t_n];L^2(\Omega)}(h)$ and $\|e_{GSF}\|_{L^{\infty}([0,t_n];\mathcal{V}\times\mathcal{M})}(h)$ take values different from zero for $h \to 0$. This is due to the presence of the time discretization error. Hence, only enrichment of the finite dimensional space, without also refining the discretization in time, may not improve the accuracy of the numerical solution.

Finally, the diagrams given in Figure 5.22 aim to highlight the effect of the space discretization error. To this end, numerical simulations have been carried out on given finite element mesh whereas the time step size Δt was changed. In general a reduction of the extended dissipation error is observed as the time step size Δt is reduced, though the reduction is not as pronounced as the one obtained by the enrichment of the mesh. Also here, the error presents a value different from zero as $\Delta t \rightarrow 0$, due to the influence of the space discretization error.

As for the time variation of the classical measures of the exact error, as a result of the little influence of the time discretization, it is interesting to note that, for example, the diagram of the L^2L^{∞} of the exact error of the total strain shows that starting from $t \geq 21$ we have the same time evolution of the error using different time discretizations. This behaviour is to be related to the one dimensionality of the model problem under consideration. Even considering other time discretizations, the elements of the discrete model experiencing plastic loading are the same for $t \geq 21$. Hence, since at the same load level, we have the same system state, it follows that the variation of state from one time instant to the other is the same.

Figure 5.22 reports as well the variation for the different time discretizations of the error computed at the time instants t = 1.5, t = 6.0 and t = 21.0. These diagrams, likewise the previous one, allow one to appreciate better the asymptotic behaviour of the error with respect to Δt . At t = 1.5 a constant value is obtained for all the error measures. This is due to the fact that the behaviour of the discrete model is elastic, hence no time discretization error is introduced since the elastic constitutive equation is integrated exactly. The plots relative to t = 6.0 and t = 21.0, on the other hand, show that reducing the time step size beyond a certain limit value has no effect on the error, hence no benefit can be expected on the improvement of the accuracy of the solution, which relates to the error due to the discretization in space.

Remark 5.3. In Orlando & Peric (2000) by generalizing the error in the constitutive equations for linear elasticity, the extended dissipation error has been interpreted as an estimate of the error of the kinematically admissible variables. For linear elastic behaviour of the discrete model, we have seen that the extended dissipation



Figure 5.22: Extended dissipation error and classical measures of the exact error. Effect of time discretization for given finite element mesh. Mesh with 56 elements.

error reduces to the energy norm of the error in the displacement which shows a linear convergence rate with the mesh size. When plastic loading occurs, several rate of convergence for the state variables are noted, (see, e.g., Johnson, 1977; Han & Reddy, 1999). Since the extended dissipation error account for the error in all the variables, its rate of convergence should be influenced by the lower one.

5.3 Dissipation Error

The dissipation error has been defined in Section 3.5.1.1 as the error in the constitutive equations produced by a field $s_{ad}(\boldsymbol{x},t) = (\boldsymbol{\sigma}_{ad}(\boldsymbol{x},t), \boldsymbol{X}_{ad}(\boldsymbol{x},t), R_{ad}(\boldsymbol{x},t);$

163

 $u_{ad}(x,t), \epsilon_{ad}(x,t), \epsilon_{ad}^p(x,t), \alpha_{ad}(x,t), p_{ad}(x,t))$ which meets the compatibility equations, the equilibrium equations, the initial conditions and the state law. In order to apply this theory for the assessment of the quality of a finite element solution, the general considerations given in Section 5.2 for the use of the extended dissipation error apply. The solution of a fully implicit conforming finite element displacement formulation of the elasto-plastic problem, to which we refer in the following, is in general not admissible, for the finite element stresses do not satisfy the equilibrium equations in a pointwise manner. As a result, to evaluate the dissipation error as indication of the error of the finite element solution, one needs first to build an admissible solution which is as close as possible to the computed finite element solution. This will be the object of the first part of this section where such criteria are given for the Prandtl-Reuss model with linear elasticity and linear hardening following the works of Moës (1996) and Ladevèze & Pelle (2001). In the second part, after recalling the expressions of the dissipation error and of the error in solution, numerical comparisons of this error with the extended dissipation error will be illustrated on a 1D model problem.

5.3.1 Construction of the admissible solution

Plasticity Model P1: Prandtl Reuss Model with Linear Hardening

The equations for this model have been recalled in Section 5.2.1. Likewise the extended dissipation error, equation (3.46) expresses the dissipation error at the time t_{n+1} in terms of its value at t_n and of the admissible solution over $[t_n, t_{n+1}]$. As a result, the admissible solution can be obtained in an incremental manner.

Consistently with an assumed linear variation of the external load over each time interval and a normal formulation of the model as introduced in Ladevèze (1989), the admissible solution can be taken to vary linearly over $[t_n, t_{n+1}]$ so that for its complete definition one needs to compute only the value at t_{n+1} .

The criteria to build a statically admissible stress field in terms of the computed finite element stresses are the same as the one presented for the extended dissipation error. These criteria, in fact, are not dependent on the constitutive model.

The definition of an admissible displacement field, on the other hand, requires some further consideration because of the constraint imposed by the Hooke's law,

$$\boldsymbol{\sigma}_{ad}(\boldsymbol{x}, t_{n+1}) = \mathbf{C}\boldsymbol{\epsilon}_{ad}^{e}(\boldsymbol{x}, t_{n+1}).$$

In fact, for a plasticity model which does not conserve volume, one can simply assume $u_{ad}(x, t_{n+1}) = u_h(x, t_{n+1})$. The associated admissible plastic strain field would then be given by

$$\boldsymbol{\epsilon}_{ad}^{p}(\boldsymbol{x}, t_{n+1}) = \nabla_{\boldsymbol{s}} \boldsymbol{u}_{n+1}^{h}(\boldsymbol{x}) - \mathbf{C}^{-1} \boldsymbol{\sigma}_{ad}(\boldsymbol{x}, t_{n+1}).$$
(5.19)

In the case of a J_2 -flow theory, such as for the Prandtl-Reuss model under consideration, on the contrary, the choice $u_{ad}(x, t_{n+1}) = u_{n+1}^h(x)$ is not always

possible. The plastic strain given by (5.19) may not meet the incompressibility condition

$$\operatorname{Tr}[\nabla_{s}\boldsymbol{u}_{n+1}^{h}(\boldsymbol{x}) - \mathbf{C}^{-1}\boldsymbol{\sigma}_{ad}(\boldsymbol{x}, t_{n+1}) - \boldsymbol{\epsilon}_{ad}^{p}(\boldsymbol{x}, t_{n})] = 0$$
(5.20)

and, consequently, the dissipation error would not assume a finite value. In such a case, then, one needs to define $u_{ad}(x, t_{n+1})$ so that condition (5.20) is met.

A procedure to build such an admissible displacement field is given in Moës (1996) who adapts the method proposed in Gastine *et al.* (1992) for incompressible elasticity. However, for 1D problems and plane stress state problems, one can assume without restriction $\boldsymbol{u}_{ad}(\boldsymbol{x}, t_{n+1}) = \boldsymbol{u}_{n+1}^{h}(\boldsymbol{x})$ and adjust the transversal component of the plastic strain to realize the incompressibility condition.

As for the definition of the admissible accumulated plastic strain $p_{ad}(\boldsymbol{x}, t_{n+1})$, we can follow the general procedure indicated in Section 5.2.1. This determines $p_{ad}(\boldsymbol{x}, t_{n+1})$ as minimizer of the pointwise contribution to the dissipation error within the time step $[t_n, t_{n+1}]$ and under the further constraint $R_{ad}(\boldsymbol{x}, t_{n+1}) = Hp_{ad}(\boldsymbol{x}, t_{n+1})$, which imposes the respect of the hardening law.

The general problem given in Box 5.1 would then specialize as in Box 5.3.

It is trivial to check that the exact solution of this problem is given by

$$p_{ad}(t_{n+1}) = \max\left\{ \|\boldsymbol{\sigma}_{ad}^{D}(t_{n+1})\| - R_0; \ p_{ad}(t_n) + \|\boldsymbol{\epsilon}_{ad}^{p}(t_{n+1}) - \boldsymbol{\epsilon}_{ad}^{p}(t_n)\| \right\}.$$

Once all the admissible state variables have been computed at t_{n+1} , the use of a time linear interpolation over $[t_n, t_{n+1}]$ guarantees the admissibility of the solution for the convexity of the equilibrium and compatibility conditions and for the linearity of the state law as a result of having expressed the model into normal form. Yet, the above procedure delivers an admissible solution which produces a finite value of the error for the convexity of the domains \mathbb{E} and \mathbb{C} introduced in Section 3.2.4.5.

5.3.2 Error Expressions

Error in the constitutive equations

In the previous section we have seen that the admissible solution for the computation of the dissipation error, unlike the one used for the definition of the extended dissipation error, is required to meet the state laws. As a result of this constraint between the admissible static variables and their conjugate kinematic one, the residual in the state law vanishes and the accuracy of the admissible solution is defined only in terms of the residual produced in the evolution equations, which are the only equations of the model not to be satisfied. The error in the constitutive equations is, therefore, given by

$$e_{dis}^{2}(t_{n+1}) = e_{dis}^{2}(t_{n}) +$$

$$+ 2 \int_{\Omega} \int_{t_{n}}^{t_{n+1}} \left[R_{0} \| \dot{\boldsymbol{\epsilon}}_{ad}^{p}(\boldsymbol{x}, t) \| - \boldsymbol{\sigma}_{ad}(\boldsymbol{x}, t) : \dot{\boldsymbol{\epsilon}}_{ad}^{p}(\boldsymbol{x}, t) + R_{ad}(\boldsymbol{x}, t) \dot{\boldsymbol{p}}_{ad}(\boldsymbol{x}, t) \right] dt d\Omega$$
(5.21)

Box 5.3. Definition of the admissible accumulated plastic strain as minimization of the dissipation error

For each	$oldsymbol{r}\in\Omega$			
Given:	$\boldsymbol{\sigma}_{ad}(\boldsymbol{x},t_n),~ \boldsymbol{\sigma}_{ad}(\boldsymbol{x},t_{n+1}),$			
	$\boldsymbol{\epsilon}_{ad}^{p}(\boldsymbol{x},t_{n}), \boldsymbol{\epsilon}_{ad}^{p}(\boldsymbol{x},t_{n+1}),$			
	$p_{ad}(\boldsymbol{x},t_n),\ R_{ad}(\boldsymbol{x},t_n) = Hp_{ad}(\boldsymbol{x},t_n)$			
Find:	$p_{ad}(oldsymbol{x},t_{n+1})$			
such that by assuming a time linear variation over $[t_n, t_{n+1}]$ of the variables				
$oldsymbol{\sigma}_{ad}(oldsymbol{x},t), R_{ad}(oldsymbol{x},t);$				
$oldsymbol{\epsilon}^p_{ad}(oldsymbol{x},t), p_{ad}(oldsymbol{x},t),$				
we realize the minimum of the following function				
$F(p_{ad}(\boldsymbol{x}, t_{n+1})) =$				
$= \int_{t_n}^{t_{n+1}} \left[R_0 \ \dot{\boldsymbol{\epsilon}}_{ad}^p(\boldsymbol{x},t) \ - \boldsymbol{\sigma}_{ad}(\boldsymbol{x},t) \colon \dot{\boldsymbol{\epsilon}}_{ad}^p(\boldsymbol{x},t) + R_{ad}(\boldsymbol{x},t) \dot{p}_{ad}(\boldsymbol{x},t) \right] \mathrm{d}t$				
under the following constraints				
	$\ \boldsymbol{\sigma}_{ad}^{D}(\boldsymbol{x},t)\ - (R_{ad}(\boldsymbol{x},t) + R_{0}) \leq 0,$			
	$\dot{p}_{ad}(\boldsymbol{x},t) \geq \ \dot{\boldsymbol{\epsilon}}_{ad}^{p}(\boldsymbol{x},t)\ ,$			
$\lor \iota \in [\iota_n, \iota_{n+1}]$	$\mathrm{Tr}[\dot{\epsilon}^p_{ad}(m{x},t)]=0$			

where

$$e_{dis}^{2}(t_{n}) =$$

$$= \sum_{i=1}^{n-1} 2 \int_{\Omega} \int_{t_{i}}^{t_{i+1}} \left[R_{0} \| \dot{\boldsymbol{\epsilon}}_{ad}^{p}(\boldsymbol{x},t) \| - \boldsymbol{\sigma}_{ad}(\boldsymbol{x},t) \colon \dot{\boldsymbol{\epsilon}}_{ad}^{p}(\boldsymbol{x},t) + R_{ad}(\boldsymbol{x},t) \dot{p}_{ad}(\boldsymbol{x},t) \right] \mathrm{d}t \,\mathrm{d}\Omega.$$

 $R_{ad}(\boldsymbol{x},t) = \mathsf{H}p_{ad}(\boldsymbol{x},t)$

The expression (5.21) fits well the incremental origin of the admissible solution as described in the previous section, and because $s_{ad}(\boldsymbol{x},t)$ is continuous piecewise

linear over each time interval $[t_i, t_{i+1}]$, it follows

$$\begin{split} &\int_{\Omega} \int_{t_i}^{t_{i+1}} \left[R_0 \| \dot{\boldsymbol{\epsilon}}_{ad}^p(\boldsymbol{x}, t) \| - \boldsymbol{\sigma}_{ad}(\boldsymbol{x}, t) \colon \dot{\boldsymbol{\epsilon}}_{ad}^p(\boldsymbol{x}, t) + R_{ad}(\boldsymbol{x}, t) \dot{p}_{ad}(\boldsymbol{x}, t) \right] \mathrm{d}t \,\mathrm{d}\Omega = \\ &= \int_{\Omega} \left\{ R_0 \| \Delta \boldsymbol{\epsilon}_{ad,i}^p(\boldsymbol{x}) \| - \frac{\boldsymbol{\sigma}_{ad}(\boldsymbol{x}, t_{i+1}) + \boldsymbol{\sigma}_{ad}(\boldsymbol{x}, t_i)}{2} \colon \Delta \boldsymbol{\epsilon}_{ad,i}^p(\boldsymbol{x}) + \right. \\ & \left. + \mathsf{H} \big(p_{ad}^2(\boldsymbol{x}, t_{i+1}) - p_{ad}^2(\boldsymbol{x}, t_i) \big) \Big\} \mathrm{d}\boldsymbol{x} \end{split}$$

where we have let

$$\Delta \boldsymbol{\epsilon}_{ad,i}^{p}(\boldsymbol{x}) = \boldsymbol{\epsilon}_{ad}^{p}(\boldsymbol{x}, t_{i+1}) - \boldsymbol{\epsilon}_{ad}^{p}(\boldsymbol{x}, t_{i})$$

The value of the dissipation error at $t_1 = 0$, $e_{dis}^2(t_1)$, is assumed equal to zero.

Error in solution

In Section 3.5.3.1, as a result of the inequality (3.78) which represents an extension of the Prager-Synge theorem for the dissipation error, it was shown that $e_{dis}^2(T)$ provides an upper bound for the error in solution, $e_{ex}^2(T)$, defined as the error in the constitutive equations produced by $s_{ex,ad} = (\sigma_{ex}, R_{ex}; u_{ad}, \epsilon_{ad}^p, p_{ad})$ and given by equation (3.74).

5.3.3 Comparison between the two errors

The numerical performance of the dissipation error has been object of study in Moës (1996) and Ladevèze & Moës (1997). Therein, numerical applications are given for 2D models with constitutive equations obeying the Prandtl-Reuss law and the corresponding viscoplastic law. In the following, we are mainly interested in looking at how the dissipation error compares with the extended dissipation error in the assessment of the quality of the same finite element solution. With this regard, the same 1D model problem described in Figure 5.3 and discretized in Figure 5.9 has been considered.

The plots in Figure 5.23 and 5.24 recall the asymptotic behaviour of the dissipation error with respect to the discretization parameters h and Δt , respectively. These diagrams show the ability of the dissipation error to detect effects of time and space discretization, respectively.

Furthermore, it is worth noting that the time evolution of the dissipation error of an elastic finite element solution is not linear, thus the dissipation error does not reduce to the error in the constitutive equations obtained in the case of linear elasticity, unlike the extended dissipation error. The admissible solution corresponding to an elastic finite element solution must be necessarily a plastic solution: if this was not the case, the admissible solution would be the exact solution, for its associated dissipation would be zero.

Nevertheless, the variation of the error e_{dis} at t = 1.5 with the time step size and for the discrete models which present t = 1.5 in the definition of the load levels,



Figure 5.23: Time Variation of the Dissipation error. Effect of space discretization for given time discretization. Time step $\Delta t = 0.75$.



Figure 5.24: Time Variation of the Dissipation error. Effect of time step refinement for given finite element mesh. Mesh with 56 elements.

show that the time step has no influence on the value of the error. The finite element solution is, in fact, elastic, thus it does not depend on the time discretization and so also the corresponding admissible solution.

Figure 5.25 compares more specifically the time evolution of the dissipation error and of the extended dissipation error for two different discrete schemes. These schemes have been chosen as examples of two extreme situations. In fact, according to the analysis with the extended dissipation error, the error in the evolution law, θ_d , for one scheme and the error in the state law, θ_{sl} , for the other, represent the main error components, respectively. We recall that both the dissipation error, e_{dis} , and the extended dissipation error, e_{ext} , are measures of the error in the constitutive equations, but they assess the quality of different admissible solutions corresponding to the same finite element solution.

In both the discrete models, the dissipation error presents values of the error higher than the extended dissipation error and it is closer to the latter for the discrete model which has θ_d as main error component of e_{ext} . This behaviour is to be related finally to the major dissipation associated with the admissible solution defined for the computation of the dissipation error. Thus, the previous plots point

168







Figure 5.26: Contribution to the error in the evolution law for the two fully discrete schemes analysed (a) Dissipation error (b) Extended dissipation error

to the higher effectiveness of the extended dissipation error for the assessment of the quality of the finite element solution.

In conclusion, it is interesting to note the strictly increasing character of the dissipation error during the whole time evolution due to the L^1 accumulation in time of the residual of the admissible solution in the evolution law, which is always positive, as it is shown in Figure 5.26. In fact, by adapting the observations of Section 3.5.2, one concludes that there is variation of the state variables if and only if the dissipation error is different from zero.

5.4 Concluding Remarks

In this Chapter we have shown how to use the extended dissipation error introduced by Ladevèze *et al.* (1999) to assess the quality of finite element solutions of elastoplastic problems with the mesh constant throughout the loading process. The main problem was, therefore, the definition of a corresponding admissible solution, which reflects the approximations associated with the finite element solution. After giving

169

general guidelines, actual criteria to construct an admissible solution in the case of the Prandtl–Reuss model have been given. The general theory has then been applied to assess the quality of the finite element solution of a one dimensional elastoplastic bar under axial load. The extended dissipation error allows one to appreciate the importance of the effects of time discretization and also to show that the regions at the initial stage of plastic deformation contribute significantly to the error component associated with the evolution law, whereas already plastified sub–domains add only low values to the error. This circumstance has been justified on the basis of the specific material model under consideration.

Analysis of the relative importance of the error components associated with the residual in the state law, which depends on the current approximation of the finite element mesh, and the error component associated with the residual in the evolution law, which has nondecreasing character and accounts for the error in the history of the variables, has been presented.

Notable has also been the comparison with classical measures of the exact error in solution. This has showed that the extended dissipation error reflects quite well the evolution of the admissible solution with respect to the exact one as described by more classical measures of the error.

Comparison with the dissipation error defined by Ladevèze (1989) has also been given. Extended dissipation error and dissipation error assess the accuracy of different admissible solutions associated with the same finite element solution. The dissipation error delivered values of the error higher than the extended dissipation error and it was closer to the latter for the discrete model which had θ_d as the main error component of e_{ext} .

In this Chapter, the finite element mesh was constant in time. As a result, the time linear interpolation of the discrete values was a continuous function over the time interval of interest. Consequently, also the corresponding admissible solutions were time continuous. Objective of the next Chapter is to prescribe a change of finite element mesh at a given time instant t_n . In this case, the time linear interpolation of the computed finite element solutions and the associated admissible solution will have a discontinuity jump at the time instant t_n . The global accuracy in time of this solution will be assessed by means of the augmented extended dissipation error developed in Section 3.5.2.2.
Chapter 6

Numerical studies of transfer operations for adaptive finite element solutions

6.1 Introduction

Use of adaptive strategies in the finite element solution of history-dependent elastoplastic problems with incremental procedures is of paramount importance. An adaptive strategy can be defined as a computational procedure which delivers the finite element solution for the problem at hand to the prescribed accuracy. Key ingredients are, therefore, among others, the availability of an error estimator, which accounts for the sources of error associated with the approximation, and of a transfer procedure, which defines the data of the one step fully discrete problem in the case the current finite element mesh is different from the one of the previous time step.

In the previous chapter, it has been shown that the extended dissipation error applied to the assessment of the accuracy of the finite element solution obtained by a fully implicit displacement formulation of the elastoplastic problem is able to account for the effects of time and space discretization. Therein, the analysis has been carried out by assuming finite element mesh constant throughout the whole evolution. A property of this error is its non-decreasing character in time due to the accumulation of the discretization errors. As a result, during the computation with incremental procedures, one may need to modify the parameters which define the fully discrete scheme, namely time step size and finite element mesh, in order to obtain the corresponding solution to the prescribed global accuracy.

When only variation of the time step is sufficient to improve the accuracy of the solution, the extended dissipation error presented in the previous chapter can be used to assess the global quality of the finite element approximation because of the time continuity of the associated admissible solution. On contrary, when the finite element mesh is changed at time t_n , two finite element solutions are considered for the same load level: the one at t_n^- , which is associated with the mesh \mathcal{T}_{h_n} , (henceforth, called old mesh), and the other at t_n^+ , which is associated with the mesh $\mathcal{T}_{h_{n+1}}$, (henceforth, referred to as a new mesh). The solution at t_n^+ is employed to define a time linear interpolation function which we require to satisfy the following property

$$\lim_{\Delta t\downarrow 0} f_d(t_n + \Delta t) = \lim_{\Delta t\downarrow 0} f_i(t_n + \Delta t)$$

where $f_i = f_i(t_n + \Delta t)$ denotes the time linear interpolation over the time interval $[t_n^+, t_{n+1}^-]$ of the discrete values f_n^+ and f_{n+1}^- whereas $f_d = f_d(t_n + \Delta t)$ is the function which associates with any given Δt the solution of the discrete scheme corresponding to the given Δt and data f_n^+ . Consequently, a discontinuity jump appears in the time linear interpolation of the discrete values across the time node t_n as a result of the change of mesh and transfer procedure. The global accuracy in time of the solution, therefore, will have to depend not only on the time step and finite element mesh size but also on the value of the jump.

The extended dissipation error, augmented in Section 3.5.2.2 by the term which accounts for time discontinuity in the admissible solution, lends itself to be used for this objective. Its applicability will be illustrated on a 1D model problem where several type of change of meshes and transfer procedures (Ortiz & Quigley, 1991; Perić *et al.*, 1996; Rashid, 2002) have been analysed.

6.2 Numerical studies of transfer operations for adaptive finite element solutions

6.2.1 Augmented Extended Dissipation Error

For the implementation of the augmented extended dissipation error, the general concepts given in Section 5.1 remain still valid. An admissible solution as close as possible to the given finite element solution needs first to be defined so that it can mirror all the approximations affecting the finite element solution.

In the following, for the Prandtl Reuss model, we first present, how, given the admissible solution at t_n^- and the finite solution at t_n^+ , we build the corresponding admissible solution at t_n^+ . Successively, numerical applications of the augmented extended dissipation error aimed to compare the quality of different transfer procedures will be illustrated. Analysis of the reliability of the new error estimator in reflecting the quality of the finite element solution in the presence of change of mesh will be also performed.

6.2.1.1 Construction of the admissible solution

Plasticity Model P1: Prandtl Reuss Model with Linear Hardening The essential equations of this model are given in Section 3.2.4.5 and are next recalled.

Yield Condition:
$$\|\boldsymbol{\sigma}_{ad}^D\| - (R_{ad} + R_0) \leq 0, \quad R_{ad} \geq 0$$

$$s^{l}\eta_{\boldsymbol{x},t}^{2}(\boldsymbol{\sigma}_{ad}, R_{ad}; \boldsymbol{\epsilon}_{ad}^{e}, p_{ad}) = \left(\boldsymbol{\sigma}_{ad} - \mathbf{C}\boldsymbol{\epsilon}_{ad}^{e}\right): \mathbf{C}^{-1}\left(\boldsymbol{\sigma}_{ad} - \mathbf{C}\boldsymbol{\epsilon}_{ad}^{e}\right) + \left(R_{ad} - \mathbf{H}p_{ad}\right)\mathbf{H}^{-1}\left(R_{ad} - \mathbf{H}p_{ad}\right);$$
$${}^{d}\eta_{\boldsymbol{x},t}^{2}(\boldsymbol{\sigma}_{ad}, R_{ad}; \boldsymbol{\epsilon}_{ad}^{p}, \dot{p}_{ad}) = R_{0}\|\boldsymbol{\epsilon}_{ad}^{p}\| - \boldsymbol{\sigma}_{ad}: \boldsymbol{\epsilon}_{ad}^{p} + R_{ad}\dot{p}_{ad},$$

with $\operatorname{Tr}[\dot{\epsilon}_{ad}^p] = 0$ and $\dot{p}_{ad} \geq ||\dot{\epsilon}_{ad}^p||$, where $||q|| = \sqrt{q \cdot q}$ is the norm of the second order tensor q.

In Section 5.2.1, we have seen in the case of finite element mesh constant in time that the criteria for definition of the admissible solution corresponding to the finite element solution at t_{n+1} were expressed only in terms of the admissible solution at t_n and of the finite element solution at t_{n+1} . Therein, the cause of the variation of the state of the system was a change of load. The latter is important only as far as the definition of a statically admissible stress field was concerned. Hence, the same criteria can be used to define the admissible solution corresponding to the finite element solution at t_n^+ , provided that one replaces t_{n+1} with t_n^+ . The general procedure is recalled in Box 6.1.

DATA:	
Admissible solution at t_n^-	$ \begin{array}{c c} \boldsymbol{\sigma}_{ad}(\boldsymbol{x},t_n^-), R_{ad}(\boldsymbol{x},t_n^-), \\ \boldsymbol{\epsilon}_{ad}(\boldsymbol{x},t_n^-), \boldsymbol{\epsilon}_{ad}^p(\boldsymbol{x},t_n^-), p_{ad}(\boldsymbol{x},t_n^-). \end{array} $
Finite element solution at t_n^+	$ \begin{array}{c} \boldsymbol{u}_n^{h_n+1}(\boldsymbol{x}), \boldsymbol{\epsilon}_n^{h_n+1}(\boldsymbol{x}) = \nabla_{\boldsymbol{s}} \boldsymbol{u}_n^{h_n+1}(\boldsymbol{x}) \\ & \overset{h_{n+1}}{\to} \boldsymbol{\epsilon}_n^p(\boldsymbol{x}), \overset{h_{n+1}}{\to} p_n(\boldsymbol{x}), \overset{h_{n+1}}{\to} \boldsymbol{\sigma}_n(\boldsymbol{x}) \end{array} $
Admissible solution at t_n^+	$ \begin{array}{l} \boldsymbol{\sigma}_{ad}(\boldsymbol{x},t_n^+), R_{ad}(\boldsymbol{x},t_n^+) \\ \boldsymbol{\epsilon}_{ad}(\boldsymbol{x},t_n^+), \boldsymbol{\epsilon}_{ad}^p(\boldsymbol{x},t_n^+), p_{ad}(\boldsymbol{x},t_n^+) \end{array} $
Admissible generalised stress field at t_n^+ : $\sigma_{ad}(\boldsymbol{x}, t_n^+), R_{ad}(\boldsymbol{x}, t_n^+)$	$\begin{split} \int_{\Omega} \boldsymbol{\sigma}_{ad}(\boldsymbol{x}, t_n^+) \colon \nabla \boldsymbol{\eta}(\boldsymbol{x}) \mathrm{d}\Omega &= \int_{\Omega} \boldsymbol{b}_n(\boldsymbol{x}) \boldsymbol{\eta}(\boldsymbol{x}) \mathrm{d}\Omega + \\ &+ \int_{\partial \Omega_t} \boldsymbol{t}_n(\boldsymbol{x}) \boldsymbol{\eta}(\boldsymbol{x}) \mathrm{d}s, \ \forall \boldsymbol{\eta} \in \mathcal{V}_0, \\ \forall \Omega_e^{h_n + 1} \in \mathcal{T}_{h_{n+1}}, & \int_{\Omega_e^{h_n + 1}} [\boldsymbol{\sigma}_{ad}(\boldsymbol{x}, t_n^+) - {}^{h_n + 1} \boldsymbol{\sigma}_n(\boldsymbol{x})] \colon \nabla \mathbb{N}_i \mathrm{d}\Omega = 0 \\ \forall \mathbb{N}_i, \ \forall \text{ vertex nodes } i & \\ R_{ad} &= \max \{R_1, R_2\} \\ \text{where } R_1 &= \ \boldsymbol{\sigma}_{ad}^D(\boldsymbol{x}, t_n^+) \ - R_0 \\ R_2 &= R_{ad}(\boldsymbol{x}, t_n^-) & \end{split}$
Admissible kinematic solution at t_n^+ : $\epsilon_{ad}(\boldsymbol{x}, t_n^+),$ $\epsilon_{ad}^p(\boldsymbol{x}, t_n^+), p_{ad}(\boldsymbol{x}, t_n^+)$	$\begin{aligned} \boldsymbol{u}_{ad}(\boldsymbol{x}, t_n^+) &= \boldsymbol{u}_n^{h_n+1}(\boldsymbol{x}), \boldsymbol{\epsilon}_{ad}(\boldsymbol{x}, t_n^+) = \nabla_s \boldsymbol{u}_{ad}(\boldsymbol{x}, t_n^+) \\ \text{IF} \ \boldsymbol{\sigma}_{ad}(\boldsymbol{x}, t_n^+) &: \begin{bmatrix} h_{n+1}\boldsymbol{\epsilon}_n^p(\boldsymbol{x}) - \boldsymbol{\epsilon}_{ad}^p(\boldsymbol{x}, t_n^-) \end{bmatrix} \ge 0 \\ \boldsymbol{\epsilon}_{ad}^p(\boldsymbol{x}, t_n^+) &= h_{n+1}\boldsymbol{\epsilon}_n^p(\boldsymbol{x}) \\ \text{ELSE} \\ \boldsymbol{\epsilon}_{ad}^p(\boldsymbol{x}, t_n^+) &= \boldsymbol{\epsilon}_{ad}^p(\boldsymbol{x}, t_n^-) \\ \text{END IF} \\ p_{ad}(\boldsymbol{x}, t_n^+) &= p_{ad}(\boldsymbol{x}, t_n^-) + \ \boldsymbol{\epsilon}_{ad}^p(\boldsymbol{x}, t_n^+) - \boldsymbol{\epsilon}_{ad}^p(\boldsymbol{x}, t_n^-) \ \end{aligned}$

Box 6.1. Procedure to build an admissible solution at t_n^+ in presence of change of mesh

Remarks

1. As already mentioned in Section 4.5, in order to define an admissible solution, a hypothesis on the distribution over each element of the state variables, which are obtained from the finite element solution at t_n^+ at the Gauss points of the new mesh, must be made. This assumption is implicitly required by the transfer procedures

described in Ortiz & Quigley (1991) and Rashid (2002), for example, and we require it to apply also for the variables defined with the smoothing transfer by Perić *et al.* (1996). Hereafter, we refer to the distributions depicted in Figure 5.1. As a result, the error in the constitutive equations must be considered as the error associated with this given postulation for the variables distribution which will also allow one to quantify the discontinuity of the fields across the time node t_n .

2. The admissible solution at t_n^- is known at the Gauss points of the old mesh, which are employed to compute numerically the space integrals that define the error at t_n^- . The element based quadrature of the space integrals that define the error at t_n^+ , on the other hand, requires the knowledge of the admissible solution at t_n^+ at the quadrature points of the new mesh. In order to implement the procedure shown in Box 6.1, the values of the fields $\epsilon_{ad}^p(\boldsymbol{x}, t_n^-)$ and $p_{ad}(\boldsymbol{x}, t_n^-)$ also at the quadrature points of the new mesh are necessary. These are obtained simply by suitable interpolation of their values at the Gauss points of the old mesh.

3. Finally, a special remark deserve the statically admissible stress fields $\sigma_{ad}(x, t_n^-)$ and $\sigma_{ad}(x, t_n^+)$ which correspond to the same load level but they are defined as prolongation of different finite element stresses.



Figure 6.1: Finite element solutions and admissible solutions for change of finite element mesh $\mathcal{T}_{h_n} \to \mathcal{T}_{h_{n+1}}$ at the time instant t_n .

Figure 6.1 reports schematically the notation relative to the finite element solutions and corresponding admissible solutions in the case of change of finite element mesh at the time instant t_n .

6.2.1.2 Error Expressions

We recall hereafter the error expressions for the Prandtl Reuss model under consideration by highlighting the terms due to the change of mesh both in the classical measures of the exact error and in the augmented extended dissipation error.

Classical measures of the exact error in solution

For the same reasons expressed in Section 5.2.2, here we also assume a global control of the exact error, which is of L^{∞} type in time and L^2 in space of the exact error of the total admissible strain,

$$\begin{aligned} \|\boldsymbol{e}_{\boldsymbol{\epsilon}_{ad}}\|_{L^{\infty}([0,T];(L^{2}(\Omega))^{d\times d})} &= \underset{t\leq T}{\text{SUP}} \|\boldsymbol{e}_{\boldsymbol{\epsilon}_{ad}}(\boldsymbol{x},t)\|_{(L^{2}(\Omega))^{d\times d}} = \\ &= \text{MAX}\left\{\underset{t\leq t_{n}^{-}}{\text{SUP}} \|\boldsymbol{e}_{\boldsymbol{\epsilon}_{ad}}(\boldsymbol{x},t)\|_{(L^{2}(\Omega))^{d\times d}}, \underset{t_{n}^{+}\leq t\leq T}{\text{SUP}} \|\boldsymbol{e}_{\boldsymbol{\epsilon}_{ad}}(\boldsymbol{x},t)\|_{(L^{2}(\Omega))^{d\times d}}\right\} \end{aligned}$$

along with the L^{∞} norm in time of the free complementary energy norm of the exact error of the generalised stress field conjugate of the kinematically admissible solution, that is,

$$\|\boldsymbol{e}_{GSF}\|_{L^{\infty}([0,T];\mathcal{V}\times\mathcal{M})} = \sup_{t\leq T} \left\{ \|\boldsymbol{e}_{\tilde{\boldsymbol{\sigma}}}(\boldsymbol{x},t)\|_{\mathcal{V}}^{2} + \|\boldsymbol{e}_{\tilde{R}}(\boldsymbol{x},t)\|_{\mathcal{M}}^{2} \right\}^{\frac{1}{2}} =$$
$$= \operatorname{MAX} \left\{ \|\boldsymbol{e}_{GSF}\|_{L^{\infty}([0,t_{n}^{-}];\mathcal{V}\times\mathcal{M})}, \|\boldsymbol{e}_{GSF}\|_{L^{\infty}([t_{n}^{+},T];\mathcal{V}\times\mathcal{M})} \right\}.$$

In both the above expressions, the exact error has been split into two terms corresponding to the two different meshes at the time t_n . The term relative to the time interval $[0, t_n^-]$ refers to the old mesh, that is, to the initial mesh, whereas the term relative to the time interval $[t_n^+, T]$ refers to the new mesh, that is, to the mesh with improved approximation. The splitting of the error shows that as a result of the L^{∞} control in time of the error, which is expressed as a suitable norm in space of the current exact error, there is reduction in the value of the error, that is, the error will not increase, if the current error at t_n^+ is not greater than the global error at t_n^- . These observations will appear clearer in the following.

Error in the constitutive equations

This error measure is given by equation (3.70) which we recall for reader's convenience,

$$e_{ext}^{n,c^{2}}(T) = \mathrm{MAX}\left\{\underbrace{\sup_{t \le t_{n}^{-}} \left[\underbrace{2\int_{\Omega} s^{l} \eta_{x,t}^{2} \,\mathrm{d}\Omega}_{t \le t_{n}^{-}} + \underbrace{2\int_{\Omega} \int_{0}^{t} d\eta_{x,\tau}^{2} \,\mathrm{d}\tau \,\mathrm{d}\Omega}_{\theta_{a}^{0}(t)}\right]}_{\theta^{o^{2}}(t)}, \underbrace{\sup_{\theta^{o^{2}}(t)}}_{\theta^{o^{2}}(t)} \left[\underbrace{2\int_{\Omega} s^{l} \eta_{x,t}^{2} \,\mathrm{d}\Omega}_{\theta_{a}^{0}(t)} + \underbrace{2\int_{\Omega} \Delta\zeta_{d}^{2}(x,t_{n}) \,\mathrm{d}\Omega}_{\theta^{n}(c^{2}}(t)} + \underbrace{2\int_{\Omega} \int_{t_{n}^{t}}^{t} d\eta_{x,\tau}^{2} \,\mathrm{d}\tau \,\mathrm{d}\Omega}_{\theta^{n}(c^{2}}(t)}\right]\right\}.$$
(6.1)

In equation (6.1), the notation $e_{ext}^{n,c^2}(T)$ has been adopted in place of ${}^{\Delta}e_{ext}^2(T)$ used in equation (3.70). Here, the superscripts "o" and "n, c" stand for old and new mesh (after change), respectively. The expressions of ${}^{sl}\eta_{x,t}^2$ and ${}^{d}\eta_{x,t}^2$ are reported at the beginning of section 6.2.1.1, whereas for the Prandtl–Reuss model, the term ${}^{\Delta}\zeta_d^2(x,t_n)$ is given by equation (3.52), i.e.,

$$\begin{aligned} \forall \boldsymbol{x} \in \Omega \\ & ^{\Delta}\zeta_{d}^{2}(\boldsymbol{x}, t_{n}) = \lim_{\Delta t \to 0+} \int_{t_{n}}^{t_{n}+\Delta t} \left\{ R_{0} \| \dot{\boldsymbol{\epsilon}}_{ad,\Delta t}^{p}(\boldsymbol{x}, \tau) \| + \right. \\ & - \boldsymbol{\sigma}_{ad,\Delta t}(\boldsymbol{x}, \tau) \colon \dot{\boldsymbol{\epsilon}}_{ad,\Delta t}^{p}(\boldsymbol{x}, \tau) + R_{ad,\Delta t}(\boldsymbol{x}, \tau) \dot{p}_{ad,\Delta t}(\boldsymbol{x}, \tau) \right\} \mathrm{d}\tau = \\ & = R_{0} \| \boldsymbol{\epsilon}_{ad}^{p}(\boldsymbol{x}, t_{n}^{+}) - \boldsymbol{\epsilon}_{ad}^{p}(\boldsymbol{x}, t_{n}^{-}) \| + \\ & - \frac{\boldsymbol{\sigma}_{ad}(\boldsymbol{x}, t_{n}^{+}) + \boldsymbol{\sigma}_{ad}(\boldsymbol{x}, t_{n}^{-})}{2} \colon (\boldsymbol{\epsilon}_{ad}^{p}(\boldsymbol{x}, t_{n}^{+}) - \boldsymbol{\epsilon}_{ad}^{p}(\boldsymbol{x}, t_{n}^{-})) + \\ & + \frac{R_{ad}(\boldsymbol{x}, t_{n}^{+}) + R_{ad}(\boldsymbol{x}, t_{n}^{-})}{2} (p_{ad}(\boldsymbol{x}, t_{n}^{+}) - p_{ad}(\boldsymbol{x}, t_{n}^{-})) \end{aligned}$$

At this point, some comments are deemed useful on the structure of equation (6.1) which will help to gain some insight on the error evolution in presence of change of mesh. These observations recall the one exposed previously for the exact error. Similar remarks are also reported in Ladéveze *et al.* (1986) and Coffignal (1987) with regard to the error in the constitutive equations according to Drucker inequality. However, in this case, no additional term has been assumed to account for the discontinuity of the admissible solution.

In (6.1) we can distinguish primarily two terms. One, θ_d^2 , is related to the history of the variables by means of an L^1 accumulation in time of the error in the evolution law, whereas the other term, θ_{sl}^2 , depends on the current value of the error in the state law. As a result, further to change of mesh, only the term θ_{sl}^2 can be reduced whereas the term θ_d^2 increases by the quantity $\Delta \theta_d^2$. Therefore, there is an advantage to change mesh for given definition of the initial data if at least the following inequality is satisfied

$$\theta_{sl}^{o^2}(t_n^-) \ge \theta_{sl}^{n,c^2}(t_n^+) + \Delta \theta_d^2(t_n).$$
(6.2)

The occurrence of (6.2) guarantees that $\theta^{n,c}(t_n^+) \leq e_{ext}^o(t_n^-)$. Figure 6.2 visualizes the meaning of the several terms which appear in (6.1). In the picture, the jump Δ is given by $\sqrt{\theta_d^{o^2}(t_n^-) + \Delta \theta_d^2(t_n)} - \theta_d^o(t_n^-)$. The augmented extended dissipation error gives the same qualitative informa-

The augmented extended dissipation error gives the same qualitative information of the exact error in solution which, finally, is to be related to the type of error control in time. However, it also shows that there will be no convenience to change mesh if the error associated with the evolution law, which is the error component



Figure 6.2: Components of the augmented extended dissipation error for change of mesh at t_n . Terms relative to the old mesh (o) and to the new mesh (n, c) after change of mesh.

that cannot be reduced for being associated with the quality of the solution up to the current time t_n , assumes values close to the prescribed global tolerance, that is, if the error associated with the past history of the solution has been relevant. This circumstance would indicate that if a global control of the solution is sought for, the incremental finite element analysis should be repeated from the beginning by starting with a finer initial mesh (see Ladéveze *et al.*, 1986).

Error in solution.

The expression of the error in solution is given by equation (5.8). The jump term is not included because of the time continuity of the exact static solution $(\sigma_{ex}(\boldsymbol{x},t), R_{ex}(\boldsymbol{x},t))$ as discussed in Section (3.5.3).

6.2.1.3 Numerical examples

The performance of the error (6.1) to assess the quality of the finite element solution obtained with an incremental procedure and in presence of change of the finite element mesh at the time instant t_n is here illustrated on the same 1D model problem as introduced in Section 5.2.3.

The initial fully discretization of the model problem is realized with uniform time step $\Delta t = 1.5$ and the non uniform mesh me1 of 14 linear elements depicted in Figure 6.3. A prescribed type of change of mesh along with a certain definition of the initial data is then assumed to occur at the time $t_n = 25.5$. At this time instant plastic loading starts to localize once the load has been reversed in sign. This is shown in Figure 5.5 which illustrates the evolution of the exact solution.

In the following, we will consider first the case of change between embedded meshes, and then the case of not embedded meshes. In both cases, the augmented extended dissipation error (6.1) will then be used to assess the quality of the resulting



Figure 6.3: Change between embedded meshes

finite element solution.

Analysis of the error: Change between embedded meshes

The type of change of finite element meshes which is considered in this section is given in the same Figure 6.3. For this example of change of meshes the condition $\mathcal{V}^{h_n} \subset \mathcal{V}^{h_{n+1}}$ is realized between the interpolating spaces of the displacement field by means of a refinement of the old mesh *me*1. In particular, the new mesh *me*2 has been obtained by halving the corresponding elements of the mesh *me*1.

Three types of definition of initial state $\tilde{\epsilon}_n^p(x)$, $\tilde{p}_n(x)$ on the new mesh me2 to restart the finite element analysis at the time $t_n = 25.5$ have been taken into account. These definitions exemplify the three groups of transfer procedures introduced in Section 4.5: variationally consistent transfer, weak enforcement of continuity and smoothing transfer.

The variationally consistent transfer is obtained by sampling at the Gauss points of the new mesh the fields ${}^{h_n} \epsilon_n^p(x)$ and ${}^{h_n} p_n(x)$, whose distribution assumption must comply with the requirements dictated by equation (4.13). Here, these fields have been obtained over each element as prolongation into a constant function of the value at the respective unique Gauss point used for the quadrature of the elemental contribution to the internal virtual power. Since the elements of the new mesh are obtained by refinement of the corresponding old element, it follows that the mapping of ${}^{h_n}(\bullet)_n(x)$ into $(\tilde{\bullet})_n(x)$ reduces to the identity operator. This transfer particularizes to linear elements the transfer adopted by Ortiz & Quigley (1991) for quadratic triangular elements and it is consistent with the constant total strain formulation of the element. This definition is, however, different from the transfer proposed by Radovitzky & Ortiz (1999). These authors, indeed, consider the Voronoi tessellation of the whole domain Ω defined by all the Gauss points, whereas here we have assumed the Voronoi tessellation of the element to which the Gauss points belong. Consequently, in the former case, the resulting partition of



Figure 6.4: Different plastic strain distribution assumption referred to the finite element mesh me1 and to the Voronoi tessellation of Ω with respect to Gauss points of me1.

 Ω will not coincide, in general, with the finite element triangulation and over each linear element a piecewise constant distribution of the variables could be defined. This circumstance is shown, for example, for the element 7 in Figure 6.4 in the case we assume the plastic strain distribution referred to the Voronoi tessellation of Ω .

In the transfer procedure obtained by imposing the weak enforcement of the continuity as in Rashid (2002), the field $(\tilde{\bullet})_n(x) \in C^{h_{n+1}}$ is obtained as L^2 projection onto $C^{h_{n+1}}$ of the respective field ${}^{h_n}(\bullet)_n(x) \in C^{h_n}$. The sets C^{h_n} and $C^{h_{n+1}}$ denote the spaces of the piecewise constant functions over each element of the old mesh \mathcal{T}_{h_n} and of the new mesh $\mathcal{T}_{h_{n+1}}$, respectively. Since $\mathcal{V}^{h_n} \subset \mathcal{V}^{h_{n+1}}$, it follows $C^{h_n} \subset C^{h_{n+1}}$, therefore, this transfer, which in the following we refer to as L^2 transfer, coincides with the variational consistent transfer defined beforehand, that is, with the identity operator.

Finally, the transfer introduced in Perić *et al.* (1996) has been used as an example of smoothing transfer. The value of the state variable ${}^{h_n}(\bullet)_n$ at the Gauss point of each element of the old mesh is first transferred unaltered to the two nodes of the element. An averaging is then carried out at each node and a continuous piecewise linear field is successively built by interpolation of the nodal values by means of the basis functions of the finite element space, \mathcal{V}^{h_n} , associated with the old mesh. The sampling of this field at the Gauss points of the new mesh provides therein the value of the initial state $(\bullet)_n$. Note that step (d) in Figure 4.5 is not required, since \mathcal{V}^{h_n+1} so that the nodal interpolant of a function $v_{h_n} \in \mathcal{V}^{h_n}$ with respect to $\mathcal{V}^{h_{n+1}}$ reduces to the identity operator. The transfer of the state variable ${}^{h_n} \epsilon_n^p$ is shown, for example, in Figure 6.5.

Once the data \tilde{e}_n^p , \tilde{p}_n at the Gauss points of the new mesh have been assigned, we consider the finite element solution at t_n^+ corresponding to load increment equal to zero, i.e., load level equal to $q(x, t_n) = \mu(t_n)x$. In Section 4.4.1 this solution was said to be obtained from the equilibration of the initial state and the time instant



Figure 6.5: Smoothing transfer of the variable ${}^{h_n}\epsilon_n^p$

 t_n^+ was denoting the time instant $t_n + \Delta t$ with $\Delta t \to 0$. Consequently, the resulting finite element solution will deliver a system state which is in equilibrium with respect to the new mesh.

Likewise for the case of finite element mesh constant in time, plastic strain ${}^{h_{n+1}}\epsilon_n^p$ and accumulated plastic strain ${}^{h_{n+1}}p_n$ obtained at the single Gauss points of each element of the new mesh are prolongated into a uniform field over the respective element. The effects of the data equilibration for each transfer are visualized in Figure 6.6. Here, a variation of the initial state defined by the given transfer procedure is noted. In particular, then, a saw-teeth distribution has been obtained in the case of L^2 transfer. In this same picture we have also plotted the distribution ${}^{h_n}\epsilon_n^p(x)$, ${}^{h_n}p_n(x)$ so that one can appreciate the discontinuity of these fields as a result of the change of mesh.

For the definition of the corresponding admissible solution necessary to compute the augmented extended dissipation error we implement the criteria given in Box 6.1. In particular, for the 1D model problem under consideration, the equilibrated stress field is given by equation (5.9) which can be used also in this context, for its definition depends only upon the current finite element stresses. The admissible plastic strain, on the other hand, is obtained by letting

$$\epsilon_{ad}^p(x,t_n^+) = {}^{h_{n+1}}\epsilon_n^p(x) \tag{6.3}$$

if

$$\sigma_{ad}(x,t_n^+) \left[{}^{h_{n+1}} \epsilon_n^p(x) - \epsilon_{ad}^p(x,t_n^-) \right] \ge 0, \tag{6.4}$$

otherwise we assume

$$\epsilon^p_{ad}(x,t_n^+) = \epsilon^p_{ad}(x,t_n^-),\tag{6.5}$$

where for the definition of $\epsilon_{ad}^p(x, t_n^-)$ the remarks expressed in Section 6.2.1 have been taken into account.



Figure 6.6: Plastic strain and accumulated plastic strain distributions at $t_n = 25.5$ resulting from different transfer assumptions. Finite element solution at t_n^- , ${}^{h_n}(\bullet)_n(x)$; Definition of the data following the transfer operation, $(\bullet)_n(x)$; Equilibration of the data with respect to the new mesh, ${}^{h_{n+1}}(\bullet)_n(x)$.

Figure 6.7 shows the admissible plastic strain and the admissible accumulated plastic strain distribution at t_n^- and t_n^+ . These pictures allow one to appreciate the time discontinuity in these fields as a result of the time discontinuity of the corresponding finite element solutions. Furthermore, in the case of L^2 transfer the admissible plastic strain is equal to the corresponding finite element solution almost everywhere, apart from a neighbourhood of the node 15 of the mesh *me2*. When we use the smoothing transfer, the admissible plastic strain is different also in the elements 17 and 18 of the mesh *me2*. Therein, in order to guarantee (6.4), definition (6.5) has been used. The accumulated plastic strain $p_{ad}(x, t_n^+)$, conversely, differs from the corresponding finite element solution in almost all elements for both transfers. This happens because $p_{ad}(x, t_n^+)$ is defined in terms of $\Delta \epsilon_{ad}^p(x)$ and $p_{ad}(x, t_n^-)$ with the latter accounting for the history of the solution up to the current time t_n .



Figure 6.7: Admissible plastic strain and admissible accumulated plastic strain distributions at $t_n = 25.5^-$, $t_n = 25.5^+$ and plots of $h_{n+1}(\bullet)_n(x)$ obtained from equilibration of the data with respect to the new mesh.



Figure 6.8: Evolution in time of the effectivity index for different transfers



Figure 6.9: Time Evolution of the Augmented Extended Dissipation Error with its components for different type of transfer at $t_n = 25.5$. L^2 and Smoothing transfer. Change between embedded meshes. For the meaning of the symbols we refer to Figure 6.2



Figure 6.10: The error in the elastic law at $t = 25.5^+$ after change of mesh with different transfer assumptions (a) L^2 transfer: Admissible stress σ_{ad} versus stress conjugate of the admissible elastic strain $C\epsilon^e_{ad}$ (b) L^2 transfer: Pointwise contribution to the error in the elastic law, $\frac{1}{C}(\sigma_{ad} - C\epsilon^e_{ad})^2$ (c) Smoothing transfer: Admissible stress σ_{ad} versus stress conjugate of the admissible elastic strain $C\epsilon^e_{ad}$ (d) Smoothing transfer: Pointwise contribution to the error in the elastic law, $\frac{1}{C}(\sigma_{ad} - C\epsilon^e_{ad})^2$



Figure 6.11: Total Strain distributions at the time $t_n = 25.5$ after change of mesh

The finite element solution ${}^{h_{n+1}}p_n(x)$, on contrary, is computed in terms of $\tilde{p}_n(x)$ which is given by the specific transfer procedure. Consequently, the definition of p_{ad} allows one to account for the approximations associated with the variable up to the current time t_n . This is a piece of information essential for the assessment of the global quality in time of the solution.

Figure 6.8 shows that the time variation of the effectivity index for the two schemes resulting from the different transfer assumptions is identical. Both the augmented extended dissipation error and the exact error which enter equation (5.12) involve L^{∞} control in time. Consequently, following change of mesh, reduction of the error with value equal to the one related to the same initial mesh *mel* is obtained.

In Figure 6.9 we plot the time evolution of the augmented extended dissipation error along with its components. We observe that for both the finite element solutions resulting from the two different transfers, we have similar qualitative behaviour: reduction in the value of the error due to the reduction of the error in the state law and a slight increase of the error in the evolution law. This behaviour of the error shows an improvement of the quality of the solution by considering both types

	$t_n = 25.5$					$t_{n+1} = 27.0$			
	е	θ_{sl}	$\frac{\theta_{sl}^e}{\theta_{sl}^p}$	θ_d	$\Delta \theta_d$	е	θ_{sl}	$\frac{\theta^e_{sl}}{\theta^p_{sl}}$	θ_d
me1	0.355	0.282	$\begin{array}{r} 0.141 \\ 0.244 \end{array}$	0.215	-	0.396	0.319	$\begin{array}{r} 0.160 \\ 0.276 \end{array}$	0.235
me2	0.245	0.140	$0.069 \\ 0.122$	0.201	-	0.254	0.156	$\begin{array}{c} 0.076 \\ 0.136 \end{array}$	0.201
$\frac{me1 \to me2}{L^2 \text{ transfer}}$	$0.355 \\ t_n^+$	$ \begin{array}{c} 0.170 \\ t_n^+ \end{array} $	$\begin{array}{r} 0.073 \\ 0.154 \end{array}$	$0.219 \\ t_n^+$	0.040	0.355	0.159	0.078 0.139	0.227
$me1 \rightarrow me2$ Sm. transfer	$0.355 \\ t_n^+$	$\begin{bmatrix} 0.218 \\ t_n^+ \end{bmatrix}$	$\begin{array}{r} 0.114 \\ \hline 0.186 \end{array}$	$\begin{array}{c} 0.230 \\ t_n^+ \end{array}$	0.080	0.355	0.208	0.080 0.192	0.237

Table 6.1: Comparison of the error components at time $t_n = 25.5$ and $t_{n+1} = 27.0$



Figure 6.12: The error in the hardening law at $t = 25.5^+$ after change of mesh with different transfer assumptions (a) L^2 transfer: Admissible thermodynamic force R_{ad} versus force conjugate of the admissible accumulated plastic strain Hp_{ad} (b) L^2 transfer: Pointwise contribution to the error in the hardening law, $\frac{1}{H}(R_{ad} - Hp_{ad})^2$ (c) Smoothing transfer: Admissible thermodynamic force R_{ad} versus force conjugate of the admissible accumulated plastic strain Hp_{ad} (b) Smoothing transfer: Pointwise contribution to the error in the hardening law, $\frac{1}{H}(R_{ad} - Hp_{ad})^2$ (c) Smoothing transfer: Hp_{ad} (b) Smoothing transfer: Pointwise contribution to the error in the hardening law, $\frac{1}{H}(R_{ad} - Hp_{ad})^2$

of transfers. In particular, the finite element solution resulting from L^2 transfer appears to behave slightly better. This is shown by the time variation of the current error $\theta^{n,c}$ which is closer to the time variation of the error e^n which is obtained with the finite element mesh me2 constant in time. This can be better appreciated also in Table 6.1 which reports the values at t_n^- , t_n^+ and t_{n+1} of the several components of the augmented extended dissipation error defined by equation (6.1). In the same table, for completeness, we have also given the values which are obtained by assuming the finite element meshes me1 and me2 constant in time throughout the whole evolution. The values at the time t_{n+1} , conversely, are reported to illustrate the influence of the transfer procedure also at later time. In the case at hand, however,



Figure 6.13: L^2 transfer (a) Variation of admissible plastic strain at $t_n = 25.5$ (b) Space distribution of $f(\sigma_{ad}, R_{ad})$ at t_n^- and t_n^+ (c) Pointwise contribution to the jump term ${}^{\Delta}\theta_d(t_n)$

the error behaviour at t_n^+ is similar to the one at t_{n+1} .

The Figures 6.10–6.14 permit to appreciate the source of the difference of values of the error in the case of the two transfers. This is accomplished by showing the pointwise contribution to the error components at the time t_n^+ . In particular, Figure 6.10 displays the pointwise contribution to the error in the elastic law which shows a major contribution coming from the elements 12, 15, 17 and 18 as a result of the different distribution of the admissible elastic strain therein. This, in turn, reflects for the problem at hand the difference of admissible plastic strain, given that the distribution of the total strain is similar, as shown in Figure 6.11.

The pointwise contribution to the error in the hardening law is given in Figure 6.12 which shows a slightly higher contribution in the case of smoothing transfer whereas the Figures 6.13 and 6.14 allow comparison of different contributions to the jump term ${}^{\Delta}\theta_d(t_n)$ for the L^2 transfer and Smoothing transfer, respectively.



Figure 6.14: Smoothing transfer (a) Variation of admissible plastic strain at $t_n = 25.5$ (b) Space distribution of $f(\sigma_{ad}, R_{ad})$ at t_n^- and t_n^+ (c) Pointwise contribution to the jump term ${}^{\Delta}\theta_d(t_n)$

For the latter case, the major contribution comes from the elements 14 and 15. In these elements, the smoothing transfer assumption produces a variation of admissible plastic strain, as shown in Figure 6.14(a), whereas the variation of $f(\sigma_{ad}, R_{ad})$, given in Figure 6.14(b) indicates that the behaviour associated with (σ_{ad}, R_{ad}) should be elastic, since $f \leq 0$ therein.

Finally, Figure 6.15 reports the classical measures of the error introduced in Section 6.2.1.2. These error measures exhibit the same qualitative behaviour as the augmented extended dissipation error with the reduction in the value of the error. They also show an improved behaviour of the finite element solution corresponding to the L^2 transfer. This is made clear from the time evolution of the current error of the generalised stress field $\|^c e_{GSF}^{n,c}\|$, which is closer to the evolution of the global error $\|e_{GSF}^n\|$. The latter is obtained by assuming the constant finite element mesh me_2 throughout the whole loading process. Therefore, it can be said that the augmented extended dissipation error is capable to mirror the approximation associated with finite element solutions in the presence of change of mesh.



Figure 6.15: Time Evolution of the Exact Error for different type of transfers at $t_n = 25.5$. The meaning of the symbols used is the following: $\forall t \in \mathcal{I}, (\bullet)(t) = \sup_{\tau \leq t} ||(\bullet)(x, \tau)||_{L^2(\Omega)}$ and ${}^c(\bullet)(t) = ||(\bullet)(x, t)||_{L^2(\Omega)}$, whereas the superscripts "o", "n, c" and "n" retain the usual meaning

Analysis of the error: Change between non–embedded meshes

In this second example, a change between non-embedded meshes is assumed to occur at the time $t_n = 25.5$. As a result, the condition $\mathcal{V}^{h_n} \subset \mathcal{V}^{h_{n+1}}$ is no more realized. Nevertheless, the mesh associated with $\mathcal{V}^{h_{n+1}}$ is chosen to contain a bigger number of elements. In particular, the new mesh *me2es* has been obtained by considering 28 linear equally spaced finite elements. Figure 6.16 shows the time discretization and the time instant when the change from the old mesh *me1* to the new mesh *me2es* occurs.



Figure 6.16: Change between non-embedded meshes

Likewise the previous example, the three types of transfers introduced in Section 4.5 are next particularized for the change of mesh considered here.

When one adopts the variationally consistent transfer, the initial state $\tilde{\epsilon}_n^p(x)$, $\tilde{p}_n(x)$ is obtained by sampling the fields ${}^{h_n}\epsilon_n^p(x)$ and ${}^{h_n}p_n(x)$ at the Gauss points of the new mesh, respectively. Consistently with the choice of one Gauss point per element for the integration of the constitutive equations, the fields ${}^{h_n}\epsilon_n^p(x)$ and ${}^{h_n}p_n(x)$ are in turn assumed constant over each element of the old mesh. The piecewise value is equal to the computed finite element solution at the respective Gauss point of the element. Therefore, in order to determine the value of the data $\tilde{\epsilon}_n^p(x)$, $\tilde{p}_n(x)$ at the element Gauss point of the new mesh, $x_{e,GP}^{h_{n+1}} \in \Omega_e^{h_{n+1}}$, one needs to identify first the element $\Omega_l^{h_n} \in \mathcal{T}_{h_n}^{\iota}$ of the old mesh such that $x_{e,GP}^{h_{n+1}} \in \Omega_l^{h_n}$. Then, by denoting with $x_{l,GP}^{h_n}$ the Gauss point of the element $\Omega_l^{h_n}$ contained the element $\Omega_l^{h_n}$.

$$(\tilde{\bullet})_n(x_{e,GP}^{h_{n+1}}) = {}^{h_n}(\bullet)_n(x_{l,GP}^{h_n})$$
(6.6)

If $x_{e,GP}^{h_{n+1}}$ lies on the boundary of the element $\Omega_l^{h_n}$, the average value of the variables between the two neighbouring elements is assumed. The definition of the variationally consistent transfer of the variable ${}^{h_n}\epsilon_p^n(x)$ is visualized in Figure 6.17.

The initial state $\tilde{\epsilon}_n^p(x)$, $\tilde{p}_n(x)$ obtained from the weak enforcement of the con-



Figure 6.17: Variationally consistent transfer of ${}^{h_n} \epsilon_n^p(x)$ for the definition of $\tilde{\epsilon}_n^p(x)$



Figure 6.18: Weak enforcement of the continuity between the fields ${}^{h_n}\epsilon_n^p(x)$ and $\tilde{\epsilon}_n^p(x)$

tinuity with the fields ${}^{h_n}\epsilon_n^p(x)$, ${}^{h_n}p_n(x)$ as introduced in Rashid (2002) is given by

$$\forall e = 1, 2, \dots, N_{h_{n+1}}$$

$$(\tilde{\bullet})_n^e(x) = \frac{\int_{x_e^{h_{n+1}}}^{x_{e+1}^{h_{n+1}}} h_n(\bullet)_n(x) \, \mathrm{d}x}{x_e^{h_{n+1}} - x_e^{h_n}} \quad \forall x \in]x_e^{h_{n+1}}, x_{e+1}^{h_{n+1}}[$$
(6.7)

where $N_{h_{n+1}}$ is the number of elements in the triangulation $\mathcal{T}_{h_{n+1}}$ and the superscript "e" stands for element. The transfer defined by equation (6.7) assumes a constant value for $(\tilde{\bullet})_n^e(x)$ over each element $\Omega_e^{h_{n+1}}$. This value is equal to the weighted average of the field ${}^{h_n}(\bullet)_n(x)$, with the weight given by the area of the so called tributary regions. For the problem at hand, these regions are defined as the parts of the element, $\Omega_e^{h_{n+1}} = [x_e^{h_{n+1}}, x_{e+1}^{h_{n+1}}]$, of the new mesh where the field ${}^{h_n}(\bullet)_n(x)$ is constant. The definition of $\tilde{\epsilon}_n^p$ is visualized in Figure 6.18.

Remark 6.1. Unlike the case $\mathcal{V}^{h_n} \subset \mathcal{V}^{h_{n+1}}$, since now $\mathcal{V}^{h_n} \not\subset \mathcal{V}^{h_{n+1}}$, the L^2 transfer and the variationally consistent transfer describe two different procedures which are not coincident with the identity operator.

Finally, the smoothing transfer has been described in general in Figure 4.5. For the problem at hand, the transfer of the variable ${}^{h_n}\epsilon_n^p(x)$ is given in Figure 6.19.



Figure 6.19: Smoothing transfer of the variable ${}^{h_n} \epsilon_n^p(x)$

The stresses obtained by solving the incremental constitutive equations with data $\tilde{\epsilon}_n^p$, \tilde{p}_n and the displacement field $u_{h_n} = u_{h_n}(x, t_n^-)$ are not in equilibrium with respect to the test functions associated with the new mesh. Therefore, the resulting residual must be equilibrated with consequent redistribution of the stresses and of the state variables. To achieve this, a finite element analysis with the new mesh and load increment equal to zero, i.e., load level equal to $q(x, t_n) = \mu(t_n)x$, is performed. In this manner, the ensuing finite element solution will deliver at t_n^+ a system state which is in equilibrium with respect to the new mesh.

The computed plastic strain ${}^{h_{n+1}}\epsilon_n^p$ and accumulated plastic strain ${}^{h_{n+1}}p_n$ obtained at the single Gauss points of each element of the new mesh are prolongated into a uniform field over the respective element. The effects of the data equilibration for each transfer are visualized in Figure 6.20. In general, a variation of the initial state defined by the given transfer procedure is obtained. In particular, when the variationally consistent transfer is used, the initial state $\tilde{\epsilon}_n^p(x)$, $\tilde{p}_n(x)$, by definition, presents very little variation with respect to the finite element solution ${}^{h_n}\epsilon_n^p(x)$, ${}^{h_n}p_n(x)$ obtained on the old mesh me1. Consequently, the equilibration of the data with respect to the new mesh me2es produces a non uniform redistribution of plastic strain and accumulated plastic strain with concentration of plastic strain distribution at t_n^+ which appears to be substantially different from ${}^{h}\epsilon_n^p(x)$, where ${}^{h}\epsilon_n^p(x)$ denotes the plastic strain distribution at t_n obtained from the finite element solution with mesh me2es constant throughout the whole evolution.

Also with the L^2 and smoothing transfer we obtain non uniform redistribution of the variables following the equilibration of the data. However, this does not produce plastic distributions at t_n^+ substantially different from ${}^h \epsilon_n^p(x)$.



Figure 6.20: Plastic strain and accumulated plastic strain distribution at $t_n = 25.5$ resulting from different transfer operations.



Figure 6.21: Admissible plastic strain and admissible accumulated plastic strain distribution at $t_n = 25.5^-$, $t_n = 25.5^+$ and plots of $h_{n+1}(\bullet)_n(x)$ obtained from equilibration of the data with respect to the new mesh



Figure 6.22: Evolution in time of the effectivity index.

Once the finite element solution has been computed at t_n^+ , the criteria given in Box 6.1 are implemented to build the corresponding admissible solution necessary to evaluate the augmented extended dissipation error. The distribution of the admissible plastic strain and admissible accumulated plastic strain at t_n^- and t_n^+ are given in Figure 6.21. A time discontinuity for these fields is introduced at t_n as a result of the time discontinuity of the corresponding finite element solutions.

The admissible plastic strain at time t_n^+ is equal to the corresponding finite element solution wherever condition (6.4) is satisfied, otherwise it is assumed equal to the value at t_n^- . The accumulated plastic strain $p_{ad}(x, t_n^+)$, conversely, differs from the corresponding finite element solution in all the elements of the fully discrete schemes resulting from the three different transfer procedures. This difference is due to the definition of $p_{ad}(x, t_n^+)$ in terms of $p_{ad}(x, t_n^-)$ and represents an essential feature in the assessment of the global quality in time of the solution. In fact, $p_{ad}(x, t_n^-)$ accounts for the history of the solution up to the current time t_n whereas the finite element solution $h_{n+1}p_n(x)$ is computed in terms of $\tilde{p}_n(x)$. The latter field is given by the specific transfer procedure, thus, information on the accuracy associated with the past values of the solution could be lost. The definition of p_{ad} , on contrary, allows one to account for the approximations associated with the variable up to the current time t_n .

Figure 6.22 shows the time variation of the effectivity index which is identical for the three schemes. Both the augmented extended dissipation error and the error in solution, which enter equation (5.12), involve L^{∞} control in time. Consequently, following the change of mesh, reduction in the value of the error with value equal to the one relative to the same initial mesh me1 is obtained for both the augmented extended dissipation error and the error in solution.

The time evolutions of the augmented extended dissipation error of the admissible solutions corresponding to the finite element solutions resulting from the three different transfers are given in Figure 6.23. All the diagrams present similar qualitative behaviour: reduction in the value of the error due to the reduction of

the error in the state law and a slight increase of the error in the evolution law is noted. Therefore, there is globally an improvement of the quality of the solution by considering the proposed transfers for the given change of mesh. In particular, the solution resulting from L^2 transfer appears to behave better out of the proposed transfer procedures. This is shown by the time variation of the current error $\theta^{n,c}(t)$. In the case of the L^2 transfer, the evolution of $\theta^{n,c}(t)$ is the closest to $e_{ext}^n(t)$, where $e_{ext}^n(t)$ is the extended dissipation error which is obtained with the constant finite element mesh *me2es* throughout the loading process.

Table 6.2 contains the values at t_n^- , t_n^+ and t_{n+1} of the several components of the augmented extended dissipation error defined by equation (6.1). This table allows one to appreciate the differences between different transfer procedures. For completeness, we have also given the values which are obtained by assuming the finite element meshes *mel* and *me2es* constant in time throughout the whole loading process, whereas the values at the time t_{n+1} are reported to illustrate the influence of the transfer procedure at a later time.

	$t_n = 25.5$				$t_{n+1} = 27.0$				
	e _{ext}	θ_{sl}	$ extstyle heta_{sl}^{e} heta_{sl}^{p}$	θ_d	Δ_{θ_d}	e	θ_{sl}	$ extstyle rac{ heta_{sl}^e}{ heta_{sl}^p}$	θ_d
me1	0.355	0.282	$\begin{array}{r} 0.141 \\ 0.244 \end{array}$	0.215	-	0.396	0.319	0.160 0.276	0.235
me2es	0.236	0.131	0.057 0.118	0.196	-	0.245	0.146	0.063 0.132	0.197
$me1 \rightarrow me2es$ Var. transfer	$\begin{array}{c} 0.355 \\ t_n^+ \end{array}$	$0.173 \\ t_n^+$	0.076 0.156	$0.218 \\ t_n^+$	0.033	0.355	0.190	0.075 0.174	0.224
$\begin{array}{c} me1 \rightarrow me2es \\ L^2 \text{ transfer} \end{array}$	$0.355 \\ t_n^+$	$0.184 \\ t_n^+$	$\begin{array}{c} 0.110\\ 0.148\end{array}$	$0.216 \\ t_n^+$	0.014	0.355	0.159	0.074 0.141	0.218
$me1 \rightarrow me2es$ Sm. transfer	$\begin{array}{c} 0.355 \\ t_n^+ \end{array}$	$\begin{array}{c} 0.222 \\ t_n^+ \end{array}$	$\begin{array}{r} 0.144 \\ 0.168 \end{array}$	$\begin{array}{c} 0.225 \\ t_n^+ \end{array}$	0.064	0.355	0.179	0.081 0.160	0.229

Table 6.2: Comparison of the error components at time $t_n = 25.5$ and $t_{n+1} = 27.0$

The examination of this table shows that when the variationally consistent transfer is used, the least free energy norm of the error, $\theta_{sl}^{n,c}$, at t_n^+ is attained. This is essentially the result of the best fit between the admissible stress and the stress conjugate to the admissible elastic strain as shown in Figure 6.24. The fit between the admissible thermodynamic forces and the forces conjugate to the admissible accumulated plastic strain, depicted in Figure 6.25, conversely, appears to be best in the case of the L^2 transfer. As for the effects of the transfer, we have already mentioned that with the adoption of the variationally consistent transfer, the non uniform redistribution of the initial state \tilde{e}_n^p , \tilde{p}_n following the equilibration of the data produces concentration of plastic strain in the elements 23 and 27. This, in turn, gives rise to an admissible accumulated plastic distribution $p_{ad}(x, t_n^+)$ which is substantially different from the distribution of the admissible hardening forces at t_n^+ . This difference is kept also at t_{n+1} and is the cause of the increase of the error associated with the residual in the hardening law at the time t_{n+1} .

With the L^2 and smoothing transfer, on contrary, the error in the state law at t_{n+1} decreases with respect to t_n^+ . This decrease can be considered due mainly to the enhanced approximation properties of the new interpolation space whose effects are soon evident on the variation of admissible plastic strain. Therefore, for these two transfers, unlike the variationally consistent transfer, the values of the error at t_n^+ can be assumed to reflect the effects more pertinent to the transfer procedure. For the variationally consistent transfer, on contrary, also the value of the error at t_{n+1} must be considered.

Likewise the case of change between embedded meshes, Figures 6.26–6.28 allow one to compare the different contributions to the jump term ${}^{\Delta}\theta_d(t_n)$ for the transfer procedures under consideration. This appears to be highest in the case of the smoothing transfer assumption. The major contribution arises also here from the elements 15, 16 and 18 and must be related to the plastic strain which is therein introduced with the transfer and to the evolution of the admissible generalised stress field.

Finally, Figure 6.29 reports the classical measures of the error introduced in Section 6.2.1.2. These diagrams present the same qualitative behaviour as the augmented extended dissipation error. Therefore, it is possible to assess the advantage of change of mesh which is to be related to the reduction in the value of the error. They also show a better behaviour of the admissible solution corresponding to the L^2 transfer. This does appear from the time evolution of the current error of the generalised stress field $\|^c e_{GSF}^{n,c}(t)\|$. In fact, the variation of $\|^c e_{GSF}^{n,c}(t)\|$ is closer to the evolution of the global error $\|e_{GSF}^n(t)\|$ that is obtained by assuming the constant finite element mesh *me2es* throughout the loading process. Therefore, likewise the previous example, it can be said that the augmented extended dissipation error is capable to mirror the approximations associated with finite element solutions also in presence of change between non-embedded meshes.



Figure 6.23: Time Evolution of the Augmented Extended Dissipation Error with its components for different type of transfer at $t_n = 25.5$. Variationally consistent, L^2 and Smoothing transfer.



Figure 6.24: The error in the elastic law at $t = 25.5^+$ after change of mesh with different transfer operations (a) Variationally consistent transfer: Admissible stress σ_{ad} versus stress conjugate of the admissible elastic strain $C\epsilon^e_{ad}$ (b) Variationally consistent transfer: Pointwise contribution to the error in the elastic law, $\frac{1}{C}(\sigma_{ad} - C\epsilon^e_{ad})^2$ (c) L^2 transfer: Admissible stress σ_{ad} versus stress conjugate of the admissible elastic strain $C\epsilon^e_{ad}$ (d) L^2 transfer: Pointwise contribution to the error in the elastic law, $\frac{1}{C}(\sigma_{ad} - C\epsilon^e_{ad})^2$ (e) Smoothing transfer: Admissible stress σ_{ad} versus stress conjugate of the admissible elastic strain $C\epsilon^e_{ad}$ (f) Smoothing transfer: Pointwise contribution to the error in the elastic law, $\frac{1}{C}(\sigma_{ad} - C\epsilon^e_{ad})^2$



Figure 6.25: The error in the hardening law at $t = 25.5^+$ after change of mesh with different transfer operations (a) Variationally consistent transfer: Admissible thermodynamic force R_{ad} versus force conjugate of the admissible accumulated plastic strain Hp_{ad} (b) Variationally consistent transfer: Pointwise contribution to the error in the hardening law, $\frac{1}{H}(R_{ad} - Hp_{ad})^2$ (c) L^2 transfer: Admissible thermodynamic force R_{ad} versus force conjugate of the admissible accumulated plastic strain Hp_{ad} (d) L^2 transfer: Pointwise contribution to the error in the hardening law, $\frac{1}{H}(R_{ad} - Hp_{ad})^2$ (e) Smoothing transfer: Admissible thermodynamic force R_{ad} versus force conjugate of the admissible accumulated plastic strain Hp_{ad})² (e) Smoothing transfer: Admissible thermodynamic force R_{ad} versus force conjugate of the admissible accumulated plastic strain Hp_{ad} (f) Smoothing transfer: Pointwise contribution to the error in the hardening law, $\frac{1}{H}(R_{ad} - Hp_{ad})^2$ (e) Smoothing law, $\frac{1}{H}(R_{ad} - Hp_{ad})^2$



Figure 6.26: Variationally consistent transfer (a) Variation of admissible plastic strain at $t_n = 25.5$ (b) Space distribution of $f(\sigma_{ad}, R_{ad})$ at t_n^- and t_n^+ (c) Pointwise contribution to the jump term $\Delta \theta_d(t_n)$



Figure 6.27: L^2 transfer (a) Variation of admissible plastic strain at $t_n = 25.5$ (b) Space distribution of $f(\sigma_{ad}, R_{ad})$ at t_n^- and t_n^+ (c) Pointwise contribution to the jump term ${}^{\Delta}\theta_d(t_n)$



Figure 6.28: Smoothing transfer (a) Variation of admissible plastic strain at $t_n = 25.5$ (b) Space distribution of $f(\sigma_{ad}, R_{ad})$ at t_n^- and t_n^+ (c) Pointwise contribution to the jump term ${}^{\Delta}\theta_d(t_n)$



Figure 6.29: Time Evolution of the Exact Error for different type of transfers at $t_n = 25.5$. The meaning of the symbols used is the following: $\forall t \in \mathcal{I}, (\bullet)(t) = \sup_{\tau \leq t} \|(\bullet)(x,\tau)\|_{L^2(\Omega)}$ and $^c(\bullet)(t) = \|(\bullet)(x,t)\|_{L^2(\Omega)}$, whereas the superscripts "o", "n, c" and "n" retain the usual meaning

6.3 Concluding remarks

When the finite element mesh is changed at time instant t_n , a discontinuity jump is introduced in the solution at the time instant t_n . The jump produces a deterioration of the global accuracy of the approximation which is due both to the low order regularity of the approximation across the time node t_n and to the eventual diffusion of inelastic deformations as a result of the transfer procedure. However, change of mesh along with suitable definition of the data can enhance the quality of the solution because of the improved approximation properties of the new finite element subspace. Therefore, the advantage of changing mesh with a transfer procedure depends on the interplay between the above opposing features.

In this chapter we have presented a general methodology for the assessment of the global quality of displacement finite element solutions of elastoplastic problems discretized in time with the backward Euler method on dynamically changing mesh.

This methodology employs the extended dissipation error, augmented by the term which accounts for the time discontinuity in the admissible solutions. The applicability of this new error estimator has been illustrated on a one dimensional model problem. Here, the behaviour of the finite element solution with respect to several definitions of transfer procedures and type of change of meshes has been considered. We have shown that the proposed error estimator reflects quite well the several sources of approximations, which result from the change of finite element mesh with the time step. Furthermore, the new error estimator accounts also for the improved approximation property, which arises from the enrichment of the finite element space and from suitable definition of the data on the new mesh. Therefore, the augmented extended dissipation error represents an effective tool for the critical assessment of the effects of transfer procedures for evolving meshes in small strain elastoplasticity.
Chapter 7

Conclusions

A general methodology for the assessment of the global quality of displacement finite element solutions of elastoplastic problems discretized in time with the backward Euler method on dynamically changing mesh has been presented.

The motivating idea has been the observation that change of data and/or finite element mesh from one time interval to the other can be both related to a discontinuity jump of the approximate solution across the time instant t_n . Consequently, in the development of reliable *a posteriori* error estimates one needs to account not only for the time step and finite element mesh size but also for the value of the jump. Two simple error analysis of a first order ordinary differential equation, chosen as elementary prototype of the evolution law of the internal variables, show the influence on the error of the discontinuity jump, thus, the need for including such term in an *a posteriori* error estimate.

As a result, only measures of error that account for time discretization effects can reflect the low order regularity of the approximation across the time instant t_n when the change of mesh occurs. Thus, the extended dissipation error introduced by Ladevèze *et al.* (1999) can be used for this aim. This is a measure of the error in the constitutive equations, that is, the state law and the evolution law, produced by an admissible solution which is time continuous, and satisfies the equilibrium and the kinematic conditions at any time instant.

The extended dissipation error capability to capture the effects of time and space discretization has been shown in the assessment of the quality of finite element solutions of elastoplastic problems with the mesh constant throughout the loading process. Criteria to construct an admissible solution, which mirrors the approximations associated with the finite element solution, have been given for the Prandtl–Reuss plasticity model and illustrated with a numerical example. This has shown that all trends on the state law and dissipation contribution to the error were meaningful. Notable was also the comparison with classical measures of the exact error in solution showing that the extended dissipation error reflects quite well the evolution of the admissible solution with respect to the exact one as described by more classical measures of the error.

However, the extended dissipation error given in Ladevèze et al. (1999) assumes

time continuity of the admissible solution. Thus, in the second part of Chapter 3, we have defined a new measure of the error in the constitutive equations which accounts for the discontinuity jump in the admissible solution. The theory has been developed for rate-independent plasticity material models and exploits the observation that the solution of the initial boundary value problem, which governs the evolution of this class of material models, depends only on the sequence of load levels while time has just the function of ordering this sequence. Thus, fictitious continuous admissible processes have been defined over the time interval $[t_n, t_n + \Delta t]$, with Δt not influential in the solution, along which the discontinuity was assumed to be taking place.

The analysis of the error in the constitutive equations along the aforementioned fictitious processes has lead to the definition of an additional nonnegative term which was depending on the jump, in agreement with the simple a posteriori error analysis of the first order differential equation. This term and the behaviour of the error component in the state law characterize completely the discontinuity jump. This is the content of Theorem 3.8 of Chapter 3 which motivates the use of the augmented extended dissipation error as basis of a methodology for the assessment of the global accuracy in time of finite element solutions on evolving meshes.

Applications of the theory have been presented for the displacement finite element solution of the Prandtl-Reuss plasticity model solved with incremental procedure. For this material model we have given the criteria to build admissible solutions as close as possible to the computed finite element solution.

The applicability of the methodology has been finally illustrated on a one dimensional model problem where a detailed study of transfer operators (Ortiz & Quigley, 1991; Perić *et al.*, 1996; Rashid, 2002) has been carried out, with the numerical experiments providing confirmation of the theoretical developments. The augmented extended dissipation error was able to mirror the several sources of approximations which are incurred by the change of finite element mesh with the time step. Also, it was able to account for the improved approximation property, which were arising from the enrichment of the finite element space and from suitable definition of the data on the new mesh. Therefore, the augmented extended dissipation error has proven to be an effective tool for the critical assessment of the effects of transfer procedures for evolving meshes in small strain elastoplasticity.

7.1 Suggestions for further research

Although notable are the advances in the numerical simulation of more and more complex physical models, very little has been done on the corresponding side of assessment of the accuracy of the produced approximation.

This work has an aim to indicate a new way of designing adaptive strategies for problems solved with incremental procedure. In this sense, still further research is necessary. The following points bring up some issues that deserve special attention.

Definition of a new transfer.

It has since been recognized that the issues of how to change mesh and the definition of data is a very delicate matter in the adaptive solution of history dependent material models solved with incremental procedures.

To date, the available transfer operations have been introduced upon the request to meet several vaguely defined properties which are invoked to prevent corrupting the quality of the resulting finite element solution (Ortiz & Quigley, 1991; Perić *et al.*, 1996; Rashid, 2002). With the methodology set in this work, a more rational treatment of the transfer operation seems possible to be devised in the context of the ensuing error. This should therefore lead to the definition of a transfer operation such as the one that minimizes the error produced. The definition of time step size, mesh size and indication on how to change mesh and to give data are not separate steps arising from heuristic arguments but should result from a unified analysis of the error contribution of each component.

Extension to higher dimensional problems.

The theory presented in this work has been formulated in tensorial notations. Thus, its application to higher dimensional problems should not give further complications apart from the implementation aspects. Criteria to build statically admissible stress fields are already in place (Ladevèze & Leguillon, 1983; Ladevèze *et al.*, 1991; Ladevèze & Rougeot, 1997) whereas the definition of the other internal variables to build an admissible solution has been indicated in this work.

Extension to viscoplasticity.

The considerations of Section 3.5.2 cannot be carried over as they stand to a viscoplastic model. For example, in the viscoplastic model corresponding to the Prandtl–Reuss plasticity model (Ladevèze & Pelle, 2001), the dissipation potential $\varphi(\dot{\epsilon}^p, -\dot{p})$ is given by

$$\varphi(\dot{\epsilon}^p, -\dot{p}) = R_0 \|\dot{\epsilon}^p\| + k \frac{n}{n+1} \left(\frac{\dot{p}}{k}\right)^{\frac{n+1}{n}} + I_{\mathbb{C}}$$

where \mathbb{C} is the same domain defined as in the Prandtl-Reuss plasticity model and k, n are material constants, with n > 0. Thus, it follows that $\gamma \stackrel{\text{def}}{=} \frac{n+1}{n} > 1$. As a result, in presence of a discontinuity jump in the internal variable p, \dot{p} is a δ -Dirac type distribution. Hence, $(\delta)^{\gamma}$ with $\gamma > 1$ is no more a distribution (Schwartz, 1966) and the limit (3.49), which had been assumed as error associated with the discontinuity jump, is not finite.

In this case, it can be worth exploiting the following idea: Starting from the expression of the dissipation error for a time continuous fictitious admissible solution defined over $[t_n, t_n + T_c]$, where T_c stands for a critical time to be computed, we assume that, for instance, the continuous admissible plastic strain has the following expression

$$\boldsymbol{\epsilon}^{p}(\boldsymbol{x},t) = \boldsymbol{\epsilon}^{p}(\boldsymbol{x},t_{n}^{-}) + [\boldsymbol{\epsilon}^{p}(\boldsymbol{x},t_{n}^{+}) - \boldsymbol{\epsilon}^{p}(\boldsymbol{x},t_{n}^{-})]\varphi(t)$$

with the jump $[\epsilon^p]_{t_n}$ very small so that $[\epsilon^p]_{t_n}\varphi(t)$ may be assumed as a variation. Applying the methods of the calculus of the variations, we can state the following problem: For given T_c find the function $\varphi(t)$ which minimizes the error.

The function $\varphi(t)$ is required to meet the boundary conditions $\varphi(t=0) = 0$, $\varphi(t=T_c) = 1$ and the minimization should be carried out in a-dimensional format so that a certain value for T_c can be computed.

The extension to viscoplasticity should also be such that when the viscoplastic model reduces to the corresponding rate independent model, the expression of the error reduces to the one obtained for the corresponding rate independent model.

,

Bibliography

- ABBO, A. J., & SLOAN, S. W. 1996. An automatic load stepping algorithm with error control. International Journal for Numerical Methods in Engineering, 39, 1737-1759.
- AINSWORTH, M. 1996. The influence and selection of subspaces for a posteriori error estimators. *Numerische Mathematik*, **73**, 399–418.
- AINSWORTH, M., & ODEN, J. T. 1992. A procedure for a posteriori error estimation for h-p finite element methods. Computer Methods in Applied Mechanics and Engineering, 101, 73–96.
- AINSWORTH, M., & ODEN, J. T. 1993. A unified approach to a posteriori error estimation using element residual methods. *Numerische Mathematik*, 65, 23–50.
- AINSWORTH, M., & ODEN, J. T. 1997. A posteriori error estimation in finite element analysis. Computer Methods in Applied Mechanics and Engineering, 142, 1-88.
- AINSWORTH, M., & ODEN, J. T. 2000. A Posteriori Error Estimation in Finite Element Analysis. John Wiley & Sons and B. G. Teubner.
- AINSWORTH, M., ZHU, J. Z., CRAIG, A., & ZIENKIEWICZ, O. C. 1989. Analysis of the Zienkiewicz–Zhu a posteriori error estimator in the finite element method. International Journal for Numerical Methods in Engineering, 28, 2161–2174.
- ALBERTY, J., & CARSTENSEN, C. 2000. Numerical analysis of time-depending primal elastoplasticity with hardening. SIAM Journal on Numerical Analysis, 37, 1271-1294.
- ALBERTY, J., CARSTENSEN, C., & ZARRABI, D. 1999. Adaptive numerical analysis in primal elastoplasticity with hardening. Computer Methods in Applied Mechanics and Engineering, 171, 175–204.
- ALFANO, G., ROSATI, L., & VALOROSO, N. 1998. A displacement-like finite element model for J_2 elastoplasticity: variational formulation and finite-step solution. Computer Methods in Applied Mechanics and Engineering, 155, 325–358.

- ARMERO, F., & PÉREZ-FOGUET, A. 2002. On the formulation of closest-point projection algorithms in elastoplasticity-part I: The variational structure. International Journal for Numerical Methods in Engineering, 53, 297-329.
- BABUSKA, I., & MILLER, A. 1987. A feedback finite element method with a posteriori error estimation: Part I. The finite element method and some basic properties of the a posteriori error estimator. *Computer Methods in Applied Mechanics and Engineering*, **61**, 1–40.
- BABUSKA, I., & RHEINBOLDT, W. C. 1978a. Error estimates for adaptive finite element computations. SIAM Journal on Numerical Analysis, 15, 736-754.
- BABUSKA, I., & RHEINBOLDT, W. C. 1978b. A posteriori error estimates for the finite element method. International Journal for Numerical Methods in Engineering, 12, 1597-1615.
- BABUSKA, I., & RHEINBOLDT, W. C. 1979a. Analysis of optimal finite-element meshes in \mathbb{R}^1 . Mathematics of Computation, 33, 435-463.
- BABUSKA, I., & RHEINBOLDT, W. C. 1979b. On the reliability and optimality of the finite element method. Computers & Structures, 10, 87–94.
- BABUSKA, I., & RHEINBOLDT, W. C. 1981. A posteriori error analysis of finite element solutions for one-dimensional problems. SIAM Journal on Numerical Analysis, 18, 565-589.
- BABUSKA, I., & RODRIGUEZ, R. 1993. The problem of the selection of an a posteriori error indicator based on smoothening techniques. *International Journal for Numerical Methods in Engineering*, **36**, 539–567.
- BABUSKA, I., & STROUBOULIS, T. 2001. The Finite Element Method and its Reliability. Oxford University Press.
- BABUSKA, I., & YU, D. 1987. Asymptotically exact a posteriori error estimator for biquadratic elements. *Finite Elements in Analysis and Design*, **3**, 341–354.
- BABUSKA, I., STROUBOULIS, T., & UPADHYAY, C. S. 1994a. A model study of the quality of a posteriori error estimators for linear elliptic problems. Error estimation in the interior of patchwise uniform grids of triangles. Computer Methods in Applied Mechanics and Engineering, 114, 307–378.
- BABUSKA, I., STROUBOULIS, T., UPADHYAY, C. S., GANGARAJ, S. K., & COPPS, K. 1994b. Validation of a posteriori error estimators by numerical approach. International Journal for Numerical Methods in Engineering, **37**, 1073–1123.
- BANK, R. E., & SMITH, R. K. 1993. A posteriori error estimates based on hierarchical bases. SIAM Journal on Numerical Analysis, 30, 921–935.

- BANK, R. E., & WEISER, A. 1985. Some a posteriori error estimators for elliptic partial differential equations. *Mathematics of Computation*, 44, 283–301.
- BARLOW, J. 1976. Optimal stress locations in finite element models. International Journal for Numerical Methods in Engineering, 10, 243–251.
- BARTHOLD, F.-J., SCHMIDT, M., & STEIN, E. 1996. Error estimation and mesh adaptivity for elasto-plastic deformations. Z. Angew. Math. Mech.: Proceedings ICIAM/GAMM, 76, 159-162.
- BARTHOLD, F. J., SCHMIDT, M., & STEIN, E. 1997. Error estimation mesh adaptivity for elasto-plastic deformations. *Pages 597-602 of:* OWEN, D. R. J., NATE, E. O & HINTON, E. (eds), *Computational Plasticity: Fundamentals and Applications.* CIMNE.
- BARTHOLD, F.-J., SCHMIDT, M., & STEIN, E. 1998. Error indicators and mesh refinements for finite-element-computations of elastoplastic deformations. *Computational Mechanics*, **22**, 225–238.
- BATAILLE, J., & KESTIN, J. 1979. Irreversible processes and physical interpretation of rational thermodynamics. *Journal of Non-equilibrium Thermodynamics*, 4, 229–258.
- BATHE, K. J. 1996. Finite Element Procedures. Prentice-Hall.
- BERNARDI, C., & GIRAULT, Y. 1998. A local regularization operator for triangular and quadrilateral finite elements. SIAM Journal on Numerical Analysis, 35, 1893– 1916.
- BESSON, J., CAILLETAUD, G., CHABOCHE, J. L., & FOREST, S. 2001. Mécanique Non Linéaire des Matériaux. Hermes Science Publications.
- BONET, J., & WOOD, R. 1997. Nonlinear Continuum Mechanics for Finite Element Analysis. Cambridge University Press.
- BONET, J., HUERTA, A., & PERAIRE, J. 2002. The efficient computation of bounds for functionals of finite element solutions in large elasticity. *Computer Methods in Applied Mechanics and Engineering (in press)*.
- BOROOMAND, B., & ZIENKIEWICZ, O. C. 1998. Recovery procedures in error estimation and adaptivity. Adaptivity in non-linear problems of elasto-plasticity behaviour. Pages 383-410 of: LADEVÈZE, P., & ODEN, J. T. (eds), Advances in Adaptive Computational Methods in Mechanics. Elsevier Science Ltd.
- BRENAN, K.E., CAMPBELL, S.L., & PETZOLD, L.R. 1996. Numerical Solution of Initial-Value Problems in Differential-Algebraic Equations. SIAM, Society for Industrial and Applied Mathematics.

- BREZIS, H. 1986. Analisi Funzionale: teoria e applicazioni. Liguori Editore. (In Italian).
- BUSSY, P., & REMOND, Y. 1985. Dualité en mécanique des matériaux et des strucutures. Internal Report 52. LMT-E.N.S. de Cachan, France. Lecture's notes from the course by P. Ladèveze.
- CAMACHO, G. T., & ORTIZ, M. 1997. Adaptive Lagrangian modelling of ballistic penetration of metallic targets. *Computer Methods in Applied Mechanics and Engineering*, 142, 269–301.
- CARSTENSEN, C., & FUNKEN, S. A. 2000. Averaging technique for FE-a posteriori error control in elasticity. Part I: Conforming FEM. Tech. rept. Mathematisches Seminar, Christian-Albrechts-Universität zu Kiel, Germany.
- CARSTENSEN, C., & VERFURTH, R. 1999. Edge residuals dominate a-posteriori error estimates for low order finite element methods. *SIAM Journal on Numerical Analysis*, **36**, 1571–1587.
- CHABOCHE, J.L. 1996. Unified cyclic viscoplastic constitutive equations: Development, capabilities and thermodynamical framework. *Pages 1-68 of:* KRAUSZ, A.S., & KRAUSZ, K. (eds), *Unified Constitutive Laws of Plastic Deformation*. Academic Press.
- CHEN, Z., NOCHETTO, R. H., & SCHMIDT, A. 2000a. A characteristic Galerkin method with adaptive error control for the continuous casting problem. *Computer Methods in Applied Mechanics and Engineering*, 189, 249–276.
- CHEN, Z., NOCHETTO, R. H., & SCHMIDT, A. 2000b. Error control and adaptivity for a phase relaxation model. *ESAIM-Mathematical Modelling and Numerical Analysis-Modelisation Mathematique et Analyse Numerique*, **34**, 775–797.
- CIARLET, P. G. 1978. The Finite Element Method for Elliptic Problems. North Holland Publisher.
- CIRAK, F., & RAMM, E. 2000. A posteriori error estimation and adaptivity for elastoplasticity using the reciprocal theorem. International Journal for Numerical Methods in Engineering, 47, 379–393.
- CLÉMENT, P. 1975. Approximation by finite element functions using local regularization. Revue française d'automatique et de recherche opérationnelle, 2, 77-84.
- COFFIGNAL, G. 1987. Optimisation et Fiabilité des Calculs Eléments Finis en Elastoplasticité. Ph.D. thesis, Université Pierre et Marie Curie - Paris 6, France.
- COLEMAN, B. D., & GURTIN, M. E. 1967. Thermodynamics with internal state variables. The Journal of Chemical Physics, 47, 597-613.

- COMI, C., & PEREGO, U. 1995. A unified approach for variationally consistent finite elements in elastoplasticity. *Computer Methods in Applied Mechanics and Engineering*, **121**, 323–344.
- COOREVITS, P., DUMEAU, J.P., & PELLE, J. P. 1998. Error estimator and adaptivity for three-dimensional analysis. *Pages 443-458 of:* LADEVÈZE, P., & ODEN, J. T. (eds), *Advances in Adaptive Computational Methods in Mechanics*. Elsevier Science Ltd.
- CRISFIELD, M. 1991. Nonlinear Finite Element Analysis of Solids and Structures. Vol. 1. John Wiley & Sons, Inc.
- DAWSON, C., & KIRBY, R. 1999. Solution of parabolic equations by backward Euler-mixed finite element methods on a dynamically changing mesh. SIAM Journal on Numerical Analysis, 37, 423-442.
- DE MIRANDA, S., & UMBERTINI, F. 2002. Recovery of consistent stresses for compatible finite elements. Computer Methods in Applied Mechanics and Engineering, 191, 1595–1609.
- DE SAXCÈ, G. 1992. Une généralisation de l'inégalité de Fenchel et ses applications aux lois constitutives. Comptes Rendus Académie des Sciences, II, **314**, 125–129.
- DE SOUZA NETO, E. A., PERIC, D., & OWEN, D. R. J. 2002. Computational Plasticity: Small and Large Strain Finite Element Analysis of Elastic and Inelastic Solids. In Preparation.
- DEMKOWICZ, L., DEVLOO, PH., & ODEN, J. T. 1985. On *h*-type meshrefinement strategy based on minimization of interpolation errors. *Computer Methods in Applied Mechanics and Engineering*, **53**, 67-89.
- DENNIS, J. E., & SCHNABEL, R. B. 1996. Numerical Methods for Unconstrained Optimization and Nonlinear Equations. SIAM, Society for Industrial and Applied Mathematics.
- DIAZ, A.R., KIKUCHI, N., & TAYLOR, J.E. 1983. A method of grid optimization for finite element methods. Computer Methods in Applied Mechanics and Engineering, 41, 29–45.
- DORFLER, W., & WILDEROTTER, O. 2000. An adaptive finite element method for a linear elliptic equation with variable coefficients. Z. Angew. Math. Mech., 80, 481-491.
- DRUCKER, D. C. 1964. On the postulate of stability of material in the mechanics of continua. *Journal de Mécanique*, **3**, 235–249.
- DUPONT, T. 1982. Mesh modification for evolution equations. Mathematics of Computation, 39, 85-107.

- DURAN, R., & RODRIGUEZ, R. 1992. On the asymptotic exacteness of Bank-Weiser's estimator. Numerische Mathematik, 62, 297–303.
- DUVAUT, G., & LIONS, J. L. 1976. Inequalities in Mechanics and Physics. Springer Verlag.
- EKELAND, I., & TEMAM, R. 1976. Convex Analysis and Variational Problems. North Holland Publisher.
- ERIKSSON, K., & JOHNSON, C. 1991. Adaptive finite element methods for parabolic problems I: A linear model problem. SIAM Journal on Numerical Analysis, 28, 43–77.
- ERIKSSON, K., ESTEP, D., HANSBO, P., & JOHNSON, C. 1995. Introduction to adaptive methods for differential equations. Acta Numerica, 105–158.
- ERIKSSON, K., ESTEP, D., HANSBO, P., & JOHNSON, C. 1996. Computational Differential Equation. Cambridge University Press.
- ESTEP, D. J., LARSON, M. G., & WILLIAMS, R. D. 2000. Estimating the Error of Numerical Solutions of Systems of Reaction-Diffusion Equations. American Mathematical Society. Memoirs of the American Mathematical Society.
- EVANS, L. C. 1999. Partial Differential Equations. American Mathematical Society.
- FIORENZA, R. 1988. Elementi di Analisi Funzionale ed Applicazioni: Appunti delle Lezioni. Universita' di Napoli-Italy. (In Italian).
- FRAEIJS DE VEUBEKE, B. 1965. Displacement and equilibrium models in the finite element method. Chap. 9, pages 145–197 of: ZIENKIEWICZ, O.C., & HOLISTER, G.S. (eds), Stress Analysis. John Wiley & Sons.
- FREY, P. J., & GEORGE, P. L. 2000. Mesh Generation. Hermes Science Publications.
- FUCHS, M., & SEREGIN, G. 2000. Variational Methods for Problems from Plasticity Theory and for Generalized Newtonian Fluids. Springer Verlag.
- GALLIMARD, L. 1994. Contrôle adaptatif des calculs en élastoplasticité et en viscoplasticité. Ph.D. thesis, LMT - E.N.S. de Cachan, France.
- GALLIMARD, L., LADEVÈZE, P., & PELLE, J.P. 1996. Error estimation and adaptivity in elastoplasticity. International Journal for Numerical Methods in Engineering, 39, 189–217.
- GASTINE, J. L., LADEVÈZE, P., MARIN, P., & PELLE, J.P. 1992. Accuracy and optimal meshes in finite element computation for nearly incompressible materials. *Computer Methods in Applied Mechanics and Engineering*, **94**, 303–314.

- GEAR, C.W. 1971. Numerical Initial Value Problems in Ordinary Differential Equations. Prentice Hall.
- GERMAIN, P, NGUYEN, Q. S., & SUQUET, P. 1983. Continuum thermodynamics. Journal of Applied Mechanics - Transactions of the ASME, 50, 1010–1020.
- GLOWINSKI, R., LIONS, J. L., & TREMOLIERES, R. 1981. Numerical Analysis of Variational Inequalities. North Holland Publisher.
- GURTIN, M. E. 1972. The Linear Theory of Elasticity. *Pages 1–295 of:* TRUESDELL, C., & FLUGGE (eds), *Handbuch der Physik*, vol. VIa/2. Springer Verlag.
- GURTIN, M.E. 1981. An Introduction to Continuum Mechanics. Academic Press.
- HALPHEN, B., & NGUYEN, Q. S. 1975. Sur les materiaux standards generalises. Journal de Mécanique, 14, 39–63.
- HAN, W., & REDDY, B. D. 1999. Plasticity: Mathematical Theory and Numerical Analysis. Springer Verlag.
- HILL, R. 1950. The Mathematical Theory of Plasticity. Oxford University Press.
- HINTON, E., & CAMPBELL, J. S. 1974. Local and global smoothing of discontinuous finite element functions using a least square method. *International Journal* for Numerical Methods in Engineering, 8, 461–480.
- HIRIART-URRUTY, J. B., & LEMARÉCHAL, C. 2001. Fundamentals of Convex Analysis. Springer Verlag.
- HJIAJ, M. 1994. Sur la classe des matériaux standard implicites: Concept, Aspects discrétisés et Estimation de l'erreur a posteriori. Ph.D. thesis, LMT E.N.S. de Cachan, France.
- HLAVÁČEK, I. 1980. A finite element solution for plasticity with strain hardening. Revue française d'automatique et de recherche opérationnelle, 14, 347-368.
- JOHNSON, C. 1976a. Existence theorems for plasticity problems. Journal de Mathematiques Pures et Appliquees, 55, 431-444.
- JOHNSON, C. 1976b. On finite element methods for plasticity problems. Numerische Mathematik, 26, 79–84.
- JOHNSON, C. 1977. A mixed finite element method for plasticity problems with hardening. SIAM Journal on Numerical Analysis, 14, 575–583.
- JOHNSON, C. 1978. On plasticity with hardening. Journal of Mathematical Analysis and Applications, **62**, 325–336.

- JOHNSON, C. 1994. A new paradigma for adaptive finite element method. Chap. 6 of: WHITEMAN, J. R. (ed), The Mathematics of Finite Elements and Applications. John Wiley & Sons.
- JOHNSON, C., & HANSBO, P. 1992. Adaptive finite element methods in computational mechanics. Computer Methods in Applied Mechanics and Engineering, 101, 143-181.
- KACUR, J. 1985. Method of Rothe in Evolution Equations. John Wiley & Sons and B. G. Teubner.
- KELLY, D.W. 1984. The self-equilibration of residuals and complementary a posteriori error estimates in the finite element method. International Journal for Numerical Methods in Engineering, 20, 1491–1506.
- KRIEG, R.D., & KRIEG, D.B. 1977. Accuracies of numerical solution methods for the elastic-perfectly plastic model. Journal of Pressure Vessel Technology -Transactions of the ASME, 99, 510-515.
- LABORDE, P., & NGUYEN, Q. S. 1990. Étude de l'équation d'évolution des systèmes dissipatifs standards. Modélisation Mathématique et Analyse Numérique, 24, 67-84.
- LADEVÈZE, P. 1975. Comparaison de modèles de milieux continus. Ph.D. thesis, Université Pierre et Marie Curie - Paris 6, France.
- LADEVÈZE, P. 1985. Sur une famille d'algorithmes en mécanique des structures. Comptes Rendus Académie des Sciences, II, 300, 41-44.
- LADEVÈZE, P. 1989. La méthode à grand incrément de temps pour l'analyse des structures à comportement non linéaire décrit par variables internes. *Comptes Rendus Académie des Sciences, II*, **309**, 1095–1099.
- LADÈVEZE, P. 1994. La méthode d'évaluation d'erreur en relation de comportement appliquée à la MEF: qualité et amélioration. Internal Report 153. LMT-E.N.S. de Cachan, France.
- LADÈVEZE, P. 1995. Les bases de la méthode des erreurs en relation de comportement pour le contrôle adaptatif des calculs éléments finis: les travaux des années 1977. Internal Report 163. LMT-E.N.S. de Cachan, France.
- LADEVÈZE, P. 1999. Nonlinear Computational Structural Mechanics. Springer Verlag.
- LADEVÈZE, P. 2001. Constitutive relation errors for F. E. analysis considering (visco-)plasticity and damage. International Journal for Numerical Methods in Engineering, 52, 527–542.

- LADEVÈZE, P., & LEGUILLON, D. 1983. Error estimate procedure in the finite element method and applications. SIAM Journal on Numerical Analysis, 20, 485–509.
- LADEVÈZE, P., & MAUNDER, E. A. W. 1996. A general method for recovering equilibrating element tractions. Computer Methods in Applied Mechanics and Engineering, 137, 111–151.
- LADEVÈZE, P., & MOËS, N. 1997. A new a posteriori error estimation for nonlinear time dependent finite element analysis. Computer Methods in Applied Mechanics and Engineering, 157, 45–68.
- LADEVÈZE, P., & MOËS, N. 1999. Adaptive control for finite element analysis in plasticity. Computers and Structures, 73, 45–60.
- LADEVÈZE, P., & ODEN, J. T. (eds). 1998. Advances in Adaptive Computational Methods in Mechanics. Elsevier Science Ltd.
- LADEVÈZE, P., & PELLE, J.-P. 2001. La Maîtrise du Calcul en Mécanique Linéaire et Non Linéaire. Hermes Science Publications.
- LADÈVEZE, P., & ROUGEE, P. 1984. Plasticité et viscoplasticité sous chargement cyclique. Internal Report 44. LMT-E.N.S. de Cachan, France.
- LADEVÈZE, P., & ROUGEOT, PH. 1997. New advances on a posteriori error on constitutive relation in f.e. analysis. Computer Methods in Applied Mechanics and Engineering, 150, 239-249.
- LADÉVEZE, P., COFFIGNAL, G., & PELLE, J. P. 1986. Accuracy of elastoplastic and dynamic analysis. Chap. 11, pages 181-203 of: BABUSKA, I., ZIENKIEWICZ, O. C., GAGO, J., & DE A. OLIVEIRA, E. R. (eds), Accuracy Estimates and Adaptive Refinements in Finite Element Computations. John Wiley & Sons.
- LADEVÈZE, P., PELLE, J. P., & ROUGEOT, PH. 1991. Error estimation and mesh optimization for classical finite elements. *Engineering Computation*, 8, 69–80.
- LADEVÈZE, P., MOËS, N., & DOUCHIN, B. 1999. Constitutive relation error estimators for (visco)plastic finite element analysis with softening. Computer Methods in Applied Mechanics and Engineering, 176, 247-264.
- LARSSON, F., RUNESSON, K., & HANSBO, P. 2001. Computation of goal-oriented a posteriori error measures in space-time finite elements for viscoplasticity. *Pages* 499-510 of: WALL, W. A., BLETZINGER, K. U., & SCHWEIZERHOF, K. (eds), *Trends in Computational Structural Mechanics.* CIMNE.
- LE TALLEC, P. 1994. Numerical methods for nonlinear three-dimensional elasticity. Pages 465-622 of: CIARLET, P.G., & LIONS, J.L (eds), Handbook of Numerical Analysis, vol. 3. Elsevier Science Ltd.

- LEE, N., & BATHE, K. J. 1994. Error indicators and adaptive remeshing in large deformation finite element analysis. *Finite Elements in Analysis and Design*, 16, 99–139.
- LEMAITRE, J., & CHABOCHE, J. L. 1990. *Mechanics of Solid Materials*. Cambridge University Press.
- LUENBERGER, D. G. 1984. Linear and Nonlinear Programming. Adison-Wesley Publishing Company, Reading, Mass.
- MALKUS, D. S., & HUGHES, T. J. R. 1978. Mixed finite element methods Reduced and selective integration techniques: a unification of concepts. Computer Methods in Applied Mechanics and Engineering, 15, 63-81.
- MAUGIN, G. A. 1992. The Thermomechanics of Plasticity and Fracture. Cambridge University Press.
- MIKHLIN, S. G. 1964. Variational Methods in Mathematical Physics. Pergamon Press.
- MOËS, N. 1996. Une méthode de mesure d'erreur a posteriori pour les modèles de matériaux déscrits par variables internes. Ph.D. thesis, LMT E.N.S. de Cachan, France.
- MOREAU, J. J. 1974. On unilateral constraints, friction and plasticity. Pages 175– 322 of: CAPRIZ, G., & STAMPACCHIA, G. (eds), New Variational Techniques in Mathematical Physics. Edizioni Cremonese.
- NADAI, A. 1937. Plastic behaviour of metals in the strain hardening range. Journal of Applied Physics, 8, 205–213.
- NGUYEN, Q. S. 1994. Bifurcation and stability in dissipative media (plasticity, friction, fracture. Applied Mechanics Review, 47, 1-31.
- NGUYEN, Q. S. 2000. Stability and Nonlinear Solid Mechanics. John Wiley & Sons, LTD.
- NOCHETTO, R. H., SCHMIDT, A., & VERDI, C. 1997. Adapting meshes and time steps for phase change problems. *Atti Accademia Nazionale dei Lincei*, 8, 273–292.
- NOCHETTO, R. H., SAVARÉ, G., & VERDI, C. 2000. A posteriori error estimates for variable time step discretizations of nonlinear evolution equations. *Communications on Pure and Applied Mathematics*, **53**, 525–589.
- ODEN, J. T., DEMKOWICZ, L., RACHOWICZ, W., & WESTERMANN, T. A. 1989. Toward a universal *h-p* adaptive finite element strategy. Part 2: A posteriori error estimation. Computer Methods in Applied Mechanics and Engineering, 77, 113– 180.

- ORLANDO, A., & PERIC, D. 2000. A study on a posteriori error estimation for elastoplastic solids. In: Proceedings of the 6th International Conference on Computational Plasticity COMPLAS VI.
- ORTEGA, J. M., & RHEINBOLDT, W. C. 2000. Iterative Solution of Nonlinear Equations in Several Variables. SIAM, Society for Industrial and Applied Mathematics.
- ORTIZ, M., & POPOV, E. P. 1985. Accuracy and stability of integration algorithms for elastoplastic constitutive relations. *International Journal for Numerical Meth*ods in Engineering, 21, 1561–1576.
- ORTIZ, M., & QUIGLEY, J. J. 1991. Adaptive mesh refinement in strain localization problems. *Computer Methods in Applied Mechanics and Engineering*, 90, 781-804.
- ORTIZ, M., & STAINIER, L. 1999. The variational formulation of viscoplastic constitutive updates. Computer Methods in Applied Mechanics and Engineering, 171, 419-444.
- PANAGIOTOPOULOS, P. D. 1985. Inequality Problems in Mechanics and Applications. Birkhauser Boston Inc.
- PARASCHIVOIU, M., PERAIRE, J., & PATERA, A. T. 1997. A posteriori finite element bounds for linear functional outputs of elliptic partial differential equations. *Computer Methods in Applied Mechanics and Engineering*, 150, 289–312.
- PATERA, A., & PERAIRE, J. 2001. Implicit a posteriori computation of bounds. In: DECONINCK, H., & BARTH, T. J. (eds), VKI Lecture Series: Error Estimation and Solution Adaptive Discretization in CFD. Von Karman Institute, Belgium & NASA Ames Research Center, USA.
- PERIĆ, D., YU, J., & OWEN, D. R. J. 1994. On error estimates and adaptivity in elastoplastic solids: Applications to the numerical simulation of strain localization in classical and Cosserat continua. *International Journal for Numerical Methods* in Engineering, 37, 1351–1379.
- PERIĆ, D., HOCHARD, CH., DUTKO, M., & OWEN, D. R. J. 1996. Transfer operators for evolving meshes in small strain elasto-plasticity. *Computer Methods* in Applied Mechanics and Engineering, 137, 331-344.
- PRAGER, W., & SYNGE, J. L. 1947. Approximations in elasticity based on the concept of function space. *Quarterly of Applied Mathematics*, 5, 241–269.
- RADOVITZKY, R., & ORTIZ, M. 1999. Error estimation and adaptive meshing in strongly nonlinear dynamic problems. *Computer Methods in Applied Mechanics and Engineering*, **172**, 203–240.

- RANNACHER, R., & SUTTMEIER, F. T. 1997. A feed-back approach to error control in finite element methods: application to linear elasticity. *Computational Mechanics*, 19, 434-446.
- RANNACHER, R., & SUTTMEIER, F. T. 1998. A posteriori error control in finite element methods via duality techniques: Application to perfect plasticity. *Computational Mechanics*, 21, 123–133.
- RANNACHER, R., & SUTTMEIER, F. T. 1999. A posteriori error estimation and mesh adaptation for finite element models in elasto-plasticity. *Computer Methods* in Applied Mechanics and Engineering, **176**, 333–361.
- RASHID, M. M. 2002. Material state remapping in computational solid mechanics. International Journal for Numerical Methods in Engineering, 55, 431–450.
- RAVIART, P. A., & THOMAS, J. M. 1977. Primal hybrid finite element methods for 2nd order elliptic equations. *Mathematics of Computation*, **31**, 391–413.
- RAVIART, P. A., & THOMAS, J. M. 1983. Introduction à l'Analyse Numérique des Équations aux Dérivées Partielles. Masson, Paris.
- REDDY, B. D., & MARTIN, J. B. 1994. Internal variable formulations of problems in elastoplasticity: Constitutive and algorithmic aspects. *Applied Mechanics Review*, 47, 429–456.
- REIHER, TH. 1987. An adaptive method for linear parabolic differential equations. Z. Angew. Math. Mech., 67, 557-565.
- REPIN, S. I., & XANTHIS, L. 1996. A posteriori error estimation for elasto-plastic problems based on duality theory. Computer Methods in Applied Mechanics and Engineering, 138, 317-339.
- RISTINMAA, M., & TRYDING, J. 1993. Exact integration of constitutive equations in elasto-plasticity. International Journal for Numerical Methods in Engineering, 36, 2525-2544.
- ROCKAFELLAR, R. T. 1970. Convex Analysis. Princeton University Press.
- RODRIGUEZ, R. 1994. Some remarks on Zienkiewicz-Zhu estimator. International Journal on Numerical Methods in Partial Differential Equations, 10, 625-635.
- ROUGEOT, P. 1989. Sur le contrôle de la qualité des maillages eléments finis. Ph.D. thesis, Université Pierre et Marie Curie Paris 6, France.
- SCHREYER, H.L., KULAK, R.F., & KRAMER, J.M. 1979. Accurate numerical solutions for elastic- plastic models. *Journal of Pressure Vessel Technology Transactions of the ASME*, **101**, 226–234.

SCHWARTZ, L. 1966. Mathemathics for the Physical Sciences. Addison–Wesley.

- SIMO, J. C., & TAYLOR, R. 1985. Consistent tangent operators for rate independnet elasto-plasticity. Computer Methods in Applied Mechanics and Engineering, 48, 101-118.
- SIMO, J. C., KENNEDY, J. G., & TAYLOR, R. 1989. Complementary mixed finite element formulations for elastoplasticity. Computer Methods in Applied Mechanics and Engineering, 74, 177–206.
- SIMO, J.C. 1998. Numerical analysis and simulation of plasticity. Pages 183-499 of: CIARLET, P.G., & LIONS, J.L (eds), Handbook of Numerical Analysis, vol. 6. Elsevier Science Ltd.
- SIMO, J.C., & HUGHES, T.J.R. 1998. Computational Inelasticity. Springer Verlag.
- STEIN, E., & AHMAND, R. 1977. An equilibrium method for stress calculation using finite element displacements models. Computer Methods in Applied Mechanics and Engineering, 10, 175–198.
- STEWART, G. W. 1973. Introduction to Matrix Computations. Academic Press.
- STROUBOULIS, T., & HAQUE, K. A. 1992. Recent experiences with error estimation and adaptivity. Part I: Review of error estimators for scalar elliptic problems. Computer Methods in Applied Mechanics and Engineering, 97, 399-436.
- SULI, E., & HOUSTON, P. 2001. Adaptive finite element approximation of hyperbolic problems. In: DECONINCK, H., & BARTH, T. J. (eds), VKI Lecture Series: Error Estimation and Solution Adaptive Discretization in CFD. Von Karman Institute, Belgium & NASA Ames Research Center, USA.
- SUQUET, P. 1981. Evolution problems for a class of dissipative materials. *Quarterly* of Applied Mathematics, **38**, 391–414.
- SYNGE, J.L. 1957. The Hypercircle in Mathematical Physics. Cambridge University Press.
- TABBARA, M., BLACKER, T., & BELYTSCHKO, T. 1994. Finite element derivative recovery by moving least square interpolants. Computer Methods in Applied Mechanics and Engineering, 117, 211-223.
- TEMAM, R. 1985. Mathematical Problems in Plasticity. Gauthier-Villars.
- TEMAM, R. 1986. A generalized Norton-Hoff model and the Prandtl-Reuss law of plasticity. Archive for Rational Mechanics and Analysis, 95, 137–183.
- TETAMBE, R. P., YUNUS, S. M., RAJAKUMAR, C., & SAIGAL, S. 1995. Examination of flux projection-type error estimators in nonlinear finite element analysis. *Computers & Structures*, 54, 641–653.

- THOMÉE, V. 1997. Galerkin-Finite Element Methods for Parabolic Problems. Springer Verlag.
- VERFURTH, R. 1989. A posteriori error estimators for Stokes equations. Numerische Mathematik, 55, 309–325.
- VERFURTH, R. 1996. A Review of A Posteriori Error Estimation and Adaptive Mesh-Refinement Techniques. Wiley-Teubner.
- VERFURTH, R. 1999. A review of a posteriori error estimation techniques for elasticity problems. Computer Methods in Applied Mechanics and Engineering, 176, 419-440.
- VILLON, P., BOROUCHAKI, H., & SAANOUNI, K. 2000. Constrained interpolation based on plasticity criterion. In: Proceedings of the 6th International Conference on Computational Plasticity COMPLAS VI.
- WIBERG, N. E., & ABDULWAHAB, F. 1993. Patch recovery based on superconvergent derivatives and equilibrium. International Journal for Numerical Methods in Engineering, 36, 2703–2724.
- WIBERG, N. E., ABDULWAHAB, F., & ZIUKAS, S. 1994. Enhanced superconvergent patch recovery incorporating equilibrium and boundary conditions. *Interna*tional Journal for Numerical Methods in Engineering, 37, 3417–3440.
- ZHU, J. Z., & ZIENKIEWICZ, O. C. 1990. Superconvergent recovery technique and a posteriori error estimators. International Journal for Numerical Methods in Engineering, 30, 1321–1339.
- ZIENKIEWICZ, O. C., & TAYLOR, R. L. 2000. The Finite Element Method. Fifth edn. Vol. 1–3. Butterworth Heinemann.
- ZIENKIEWICZ, O. C., & ZHU, J. Z. 1987. A simple error estimator and adaptive procedure for practical engineering analysis. International Journal for Numerical Methods in Engineering, 24, 337–357.
- ZIENKIEWICZ, O. C., & ZHU, J. Z. 1992a. The superconvergent patch recovery and a posteriori error estimates. Part I: The recovery technique. *International Journal for Numerical Methods in Engineering*, **33**, 1331–1364.
- ZIENKIEWICZ, O. C., & ZHU, J. Z. 1992b. The superconvergent patch recovery and a posteriori error estimates. Part II: Error estimates and adaptivity. International Journal for Numerical Methods in Engineering, 33, 1365–1382.
- ZIENKIEWICZ, O. C., & ZHU, J. Z. 1992c. The superconvergent patch recovery (SPR) and adaptive finite element refinement. Computer Methods in Applied Mechanics and Engineering, 101, 207–224.

ZIENKIEWICZ, O. C., LIU, Y. C., & HUANG, G. C. 1988. Error estimation and adaptivity in flow formulation for forming problems. *International Journal for Numerical Methods in Engineering*, **25**, 23-42.